FRANCHISE DISCLOSURE DOCUMENT



Ace Handyman Franchising, Inc. a Colorado corporation 390 Union Boulevard, Suite 520 Lakewood, Colorado 80228 303-984-0177 or 1-866-808-8401 www.AceHandymanServices.com www.AceHandymanFranchising.com

Ace Handyman Franchising, Inc. ("we") offers the opportunity to serve your local community by operating a business providing much-needed, high-quality repair, and maintenance services for residential and commercial properties (an "AHS Business"). By leveraging technology, an AHS Business is simplified such that prior experience or knowledge of the construction trade is not needed. You will be able to offer a fostering culture where craftsmen are treated with dignity and respect toward the greatest good. This business operates during normal business hours using the trade name ACE HANDYMAN SERVICES.

The total investment necessary to begin operation of an AHS Business is estimated at \$\frac{131,077 \text{ to}}{131,997 \text{ to}}\$\$\\$\frac{5220,297223,797}{223,797}.-\] This includes \$\frac{71,497 \text{ to}}{71,797 \text{ to}}\$\$\\$\frac{1101,497 \text{ 101},797}{101,797}\$\$ that must be paid to the franchisor or an affiliate. If you acquire a Mini AHS Business, the total investment is estimated at \$\frac{96,977 \text{ to}}{96,997 \text{ to}}\$\$\\$\frac{96,997 \text{ to}}{46,497 \text{ 46},797}\$\$ that must be paid to the franchisor or an affiliate.

This Disclosure Document summarizes certain provisions of your Franchise Agreement and other information in plain English. Read this Disclosure Document and all accompanying agreements carefully. You must receive this Disclosure Document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your Disclosure Document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact our corporate headquarters at 390 Union Boulevard, Suite 520, Lakewood, Colorado 80228 and (303) 984-0177.

The terms of your contract will govern your franchise relationship. Don't rely on the Disclosure Document alone to understand your contract. Read all of your contract carefully. Show your contract and this Disclosure Document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this Disclosure Document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this Disclosure Document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: March 2727, 2024 2025

For use in: AL, AK, AZ, AR, CA, CO, CT, DE, DC, GA, FL, HI, ID, IL, IA, IN, KS, KY, LA, ME, MD, MA, MI, MN, MS, MO, MT, NE, NV, NH, NJ, NM, NY, NC, ND, OH, OK, OR, PA, RI, SC, SD, TN, TX, UT, VT, VA, WV, WI, WY, and U.S. TERRITORIES (see State Effective Dates page for effective dates in certain states.)

Not for use in: WA

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Attachments H and I.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Attachment J includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Ace Handyman Services business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be an Ace Handyman Services franchisee?	Item 20 or Attachments H and I lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this Disclosure Document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

<u>Business model can change</u>. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Attachment K.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The franchise agreement requires you to resolve disputes with the franchisor by arbitration and/or litigation only in Colorado. Out-of-state arbitration or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to arbitrate or litigate with the franchisor in Colorado than in your own state.
- **Mandatory Minimum Payments.** You must make minimum royalty or advertising fund payments, regardless of your sales levels. Your inability to make the payments may result in termination of your franchise and loss of your investment.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

INFORMATION FOR PROSPECTIVE FRANCHISEES IN MICHIGAN

The State of Michigan prohibits certain unfair provisions that are sometimes in franchise documents. If any of the following provisions are in these franchise documents, the provisions are void and cannot be enforced against you.

Each of the following provisions is void and unenforceable if contained in any documents relating to a franchise:

- (A) A prohibition on the right of a franchisee to join an association of franchisees.
- (B) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (C) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provisions of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (D) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration, of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than five years; and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least six months advance notice of franchisor's intent not to renew the franchise.
- (E) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (F) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state. (The above language has been included in this Disclosure Document as a condition for registration. We and you do not agree that the parties are restricted from choosing to conduct arbitration outside of Michigan and believe that each of the provisions of the Franchise Agreement, including each of the arbitration provisions, is fully enforceable. We and you intend to rely on the federal pre-emption under the Federal Arbitration Act.)
- (G) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
- (i) The failure of the proposed franchisee to meet the franchisor's then current reasonable qualifications or standards.

- (ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (H) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (C).
- (I) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000, the franchisor shall, at the request of a franchisee, arrange for the escrow of initial investment and other funds paid by the franchisee or subfranchisor until the obligations to provide real estate, improvements, equipment, inventory, training, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

The fact that there is a notice of this offering on file with the Attorney General does not constitute approval, recommendation, or endorsement by the Attorney General.

Any questions regarding the notice should be directed to:

State of Michigan
Department of Attorney General
Franchise Section - Consumer Protection Division
G. Mennen Williams Building, 1st Floor
525 W. Ottawa Street
Lansing, Michigan 48933
Telephone Number: (517) 373-7117

TABLE OF CONTENTS

ITEM		PAGE
1.	THE FRANCHISOR AND ANY PARENTS, PREDECESSORS AND AFFILIATES	1
2.	BUSINESS EXPERIENCE	
3.	LITIGATION	
4.	BANKRUPTCY	
5.	INITIAL FEES	
ó.	OTHER FEES	
' .	ESTIMATED INITIAL INVESTMENT	
3.	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	15 16
).	FRANCHISEE'S OBLIGATIONS	
0.	FINANCING	
1.	FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND	-
	TRAINING	19
2.	TERRITORY	
3.	TRADEMARKS	
4.	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	
5.	OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCE	
	BUSINESS	
6.	RESTRICTIONS ON WHAT YOU MAY SELL	
7.	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	40
8.	PUBLIC FIGURES	42
19.	FINANCIAL PERFORMANCE REPRESENTATIONS	
20.	OUTLETS AND FRANCHISEE INFORMATION	
21.	FINANCIAL STATEMENTS	
22.	CONTRACTS	
23.	RECEIPTS	
<u>ATTA</u>	<u>CHMENTS</u>	
A	Franchise Agreement with Exhibits	
3	Nondisclosure and Noncompetition Agreement	
\mathbf{C}	Statement of Prospective Franchisee	
)	Multi-Territory Addendum to Franchise Agreements	
3	Form of Successor Franchise Rider to Franchise Agreement	
7	Current Form of General Release	
G	Operations Manual Table of Contents	
Η	List of Franchisees	
	List of Franchisees Who Have Left the System	
ſ	Financial Statements	
K	List of State Franchise Administrators and Agents for Service of Process	
-	State Specific Addendum to Disclosure Document	
M	AHF Data Collection Agreement	
N	IGX Participation Agreement	

ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language in this Franchise Disclosure Document, "we," "us," or "our" means Ace Handyman Franchising, Inc., the Franchisor. "You" means the person, corporation, partnership or other entity that buys the franchise. If a corporation, partnership or other entity is the franchisee, "you" also includes the franchisee's owners in certain situations.

The Franchisor and Our Parents, Predecessors and Affiliates

We are a Colorado corporation that was incorporated on August 22, 2000 under the name Handyman Matters Franchise Corporation. On September 5, 2019, Handyman Matters, Inc. ("Ace Services"), which was our former direct parent company, was acquired by Ace Services Holdings LLC ("Ace Services"), a wholly-owned subsidiary of Ace Hardware Corporation, our ultimate parent company. We changed our name to Ace Handyman Franchising, Inc. on September 18, 2019. We do business under our corporate name and under the name and service mark: ACE HANDYMAN SERVICES. We are a franchising company that sells and grants franchises for the operation of businesses (an "AHS Business") using the name "ACE HANDYMAN SERVICES." We have been offering franchised businesses similar to an AHS Businesses since February 2001. We have not and do not operate any businesses similar to the franchised business being offered. Our principal business address is 390 Union Boulevard, Suite 520, Lakewood, Colorado 80228. Our agents for service of process are disclosed in Attachment K.

Parents

We have the following parent company structure:

Name	Principal Business Address	Parent Status
Ace Services Holdings LLC ("Ace Services")	2915 Jorie Boulevard Oak Brook, Illinois 60523-2100	Ace Services is a wholly owned subsidiary of Ace Hardware and our direct parent company.
Ace Hardware Corporation ("Ace Hardware")	2915 Jorie Boulevard Oak Brook, Illinois 60523-2100	Ace Hardware is our ultimate parent company. It is a retailer-owned cooperative. As of December 31, 20232024, there were 4,994 5,144 Ace Hardware cooperative locations (4,411-4,539 Ace Hardware branded stores, which includes 235-259 company owned Ace Hardware stores, and 348-346 non-Ace Hardware branded stores) operating in the United States. Ace Hardware branded cooperative locations owned by members are deemed franchises in some states.

Affiliates

We are affiliated with the following companies that (1) conduct business of the type being offered under this Disclosure Document, (2) offer franchises in any line of business (in addition to our ultimate parent, Ace Hardware, as shown above), or (3) provide products or services to our franchisees:

Name	Principal Business Address	Business Operations
Ace Hardware Home Services, Inc. ("AHHS")	2915 Jorie Boulevard, Oak Brook, Illinois 60523-2100	AHHS operates our company owned home services businesses, including 12 18 Ace Handyman Services Businesses similar to the franchise business offered under this Disclosure Document in Colorado (10), Kansas (1) and Missouri (1)2), Missouri (3), and Texas (3). AHHS also operates businesses offering residential and commercial plumbing, electrical and HVAC services.
Ace Painting Franchising, LLC ("APF")	390 Union Boulevard, Suite 520 Lakewood, Colorado 80228	APF franchises businesses offering indoor and outdoor painting, staining and other painting services to residential and commercial customers under the name and service mark ACE HARDWARE PAINTING SERVICES SM through a separate franchise disclosure document. APF commenced selling franchises in late 2022. As of December 31, 2023 2024, it had no there are three franchised and three company-owned Ace Hardware Painting Services businesses operating in the United States and there is one Ace Hardware Painting Services business operated in conjunction with an AHS Business.
Ace Plumbing Franchising, LLC ("APLF")	2915 Jorie Blvd., Oak Brook, Illinois 60523-2100	APLF is currently not active. It intends to franchise businesses offering plumbing services for residential and commercial properties under the name and service mark ACE HARDWARE PLUMBING SERVICES SM .
Westlake Hardware, Inc. ("Westlake")	14000 Marshall Drive Lenexa, Kansas 66215	Westlake is owned by Ace Retail Holdings LLC ("Ace Retail"), a subsidiary of Ace Hardware. Westlake currently operates 169–180 Ace Hardware retail stores and may provide products and services to our franchisees.
Great Lake Ace Hardware, Inc. ("GLA")	27555 Farmington Road, Suite 110 Farmington Hills, Michigan 48334	GLA is also owned by Ace Retail. It currently operates 66-79 Ace Hardware retail stores and may provide products and services to our franchisees.

Except as described above, neither we nor any of our parents, predecessors or affiliates have (1) conducted a business of the type being franchised under this Disclosure Document, (2) offered franchises in any line of business or (3) provides products or services to our franchisees.

The Franchise We Offer

We franchise to you or authorize you to operate an AHS Business under the mark ACE HANDYMAN SERVICES and related service marks, trademarks and trade names ("Marks"). In addition, in certain markets we may require our franchisees to offer their handyman services to the public under the service mark, ACE HARDWARE HANDYMAN SERVICES. In the event we require you to use the mark ACE HARDWARE HANDYMAN SERVICES, you will be required to adopt and use that mark.

An AHS Business provides handyman, and repair and maintenance services, and sales to residential and commercial customers by operating under a uniform system consisting of high standards of service, use of quality products, proprietary and exclusive technology, and the business format created and developed by us and our predecessors (the "Ace Handyman Services System"). These handyman services include basic services in the categories of carpentry, plumbing and electrical, drywall, painting, flooring, landscaping and many other miscellaneous repairs, and services to residential and commercial clients. The services provided by an AHS Business do not include advanced services in the categories of plumbing and electrical, painting, heating, cooling, air conditioning, and other categories. Those more advanced services are better suited for professionals that specialize in these services.

When you obtain a franchise from us, we license to you the right to use our Marks and the Ace Handyman Services System within a protected territory that will be assigned to you ("Territory"). An AHS Business franchise grants you a Territory of between 70,000 and 100,000 households generally delineated by zip codes. All Territories include the businesses in the Territories. If your Territory size exceeds 100,000 households, you must purchase an additional franchise.

In our discretion, we also offer franchises to operate an AHS Business in a Territory that is smaller than 70,000 households in size. We call this franchise our "Mini AHS Business." Mini AHS Business franchises are only granted by us when the Territory has 45,000 or fewer households. All references in this Disclosure Document to an AHS Business include our Mini AHS Business unless we specifically note otherwise.

You will operate from a small office outside of your home ("Business Location"). You pay us the initial fees discussed in ITEM 5 of this Disclosure Document and sign our standard franchise agreement ("Franchise Agreement") prior to operating your AHS Business. A copy of our Franchise Agreement is attached to this Disclosure Document as Attachment A.

In certain cases where a franchisee is entering into two or more Franchise Agreements to operate multiple AHS Businesses with adjacent Territories, or where an existing franchisee is entering into another Franchise Agreement to acquire an additional AHS Business with a territory adjacent to the territory of its existing AHS Business, we may agree to enter into a Multi-Territory Addendum to Franchise Agreements in the form attached as Attachment D hereto (a "Multi-Territory Addendum") with that franchisee. The Multi-Territory Addendum allows for the adjacent AHS Businesses to be operated together with certain modifications to allow for more efficient operations, such as use of a single franchisee business entity, a single Business Location, and the appointment of a single Franchise Manager (as defined in ITEM 11).

Market and Competition

An AHS Business currently focuses on serving residential and commercial customers in urban and suburban areas. The market for handyman services is well developed. You may have to compete with other businesses, including franchised operations, national chains and independently owned companies or individuals offering handyman services to residential and commercial customers. The sales of the services are not seasonal.

Industry Specific Laws and Regulations

As a franchisee, you may be required to obtain a specific construction license, registration or permits in certain states to operate an AHS Business. There are also federal, state, and in some cases, local laws and regulations pertaining specifically to the building and construction industry and your franchise, which vary and can change over time. These include laws and regulations pertaining to lead paint removal and disposal, asbestos handling and removal, and hazardous waste handling and disposal. You will need to become knowledgeable of, and comply with, all applicable local, state, and national uniform building codes, including plumbing, structural and electric codes, and licensing regulations and requirements. The National Association of Remodeling Industry publishes a "Summary of State Contractor Licensing Laws," which we strongly suggest you review.

Among the federal regulations that are applicable to your franchise are regulations promulgated by the Environmental Protection Agency ("EPA"). On April 22, 2010, the EPA's Renovation, Repair and Maintenance Final Rule ("RRP") went into effect. This RRP details both information that must be given to certain homeowners as well as processes and procedures that must be followed in homes and other structures built before 1978. See https://www.epa.gov/lead/lead-renovation-repair-and-painting-program. You will also need to comply with the "Lead Renovation, Repair, and Painting" rule (the "LRRP Rule") of the EPA. The LRRP Rule generally applies to any renovation, repair, and painting activities that may disturb paint in homes and certain other facilities constructed prior to 1978, on the basis that lead paint may be present in those facilities. Your AHS Business will likely be involved in these activities. To comply with the LRRP Rule, both your AHS Business and the individual representative or representatives of your AHS Business carrying out these activities must be certified by the EPA (the "LRRP Certification"). To obtain an LRRP Certification, the applicable representative or representatives must complete an initial training course provided or accredited by the EPA. We do not provide this training course. You must obtain the LRRP Certification for your AHS Business and at least one individual representative prior to participating in our initial training program. Recertification and refresher training courses are required under the LRRP Rule every five years to maintain the LRRP Certification. You must maintain the LRRP Certification for your AHS Business and at least one individual representative at all times. You must provide us with evidence of the LRRP Certification. The LRRP Rule additionally imposes certain requirements related to how the relevant renovation, repair, and painting activities are carried out, which include the distribution of a mandatory informational pamphlet to the customer, paint testing, cleaning verification, waste disposal, recordkeeping, and other requirements. There may be other EPA regulations that will apply to your AHS Business.

The Occupational Safety and Health Administration ("OSHA") has issued regulations pertaining to worker safety and health in the workplace. OSHA has also issued regulations pertaining to exposure to airborne concentrations of lead above certain levels. Many states and municipalities also have laws regulating the handling and disposal of hazardous materials, including paint and solvents, as well as laws and regulations pertaining to worker safety and health in the workplace.

There are federal, state and possibly local laws covering how to classify workers, for example, whether as independent contractors or employees, or as exempt or non-exempt, for different purposes, such as tax, wage and hour laws, unemployment compensation and workers' compensation. These laws and regulations can vary from state to state, city to city and at the federal level, and could affect, in some instances materially, the operation of your AHS Business.

In addition to the laws referenced above, you will be required to comply with all federal, state and local laws, rules and ordinances that apply generally to all businesses, including, but not limited to, taxes, data protection and privacy laws. It is your responsibility to investigate the laws, rules and regulations that apply

within your Territory (as defined in ITEM 12 below) before purchasing your AHS Business. You should consider both the effect of any such laws, rules and regulations on your AHS Business and the cost of compliance. You are responsible for obtaining any licenses, registrations or permits required to operate an AHS Business in your Territory.

ITEM 2 BUSINESS EXPERIENCE

William Guzik - Director

William Guzik has been our Director since September 2019. He is the Executive Vice President, Chief Financial Officer and Chief Risk Officer for Ace Hardware, positions he has held since January 2014. Between November 2021 and June 2023, he served as the Vice President, Finance, of APF. He served in a similar capacity for other entities owned by Ace Hardware that have since been merged into AHHS. Mr. Guzik is the sole-a Director of AHHS, a position he has held since June 7, 2023. Mr. Guzik is located at the headquarters of Ace Hardware in Oak Brook, Illinois.

Andy Bell – Chief Executive Officer and Founder

Andy Bell is the founder of the Ace Handyman Services System. He currently serves as our Chief Executive Officer. He also serves as Vice President, Handyman Services of AHHS. Mr. Bell has previously served as one of our Directors since our inception in 2000 to September 2019, and has held the positions of Chief Executive Officer-or. President-, or Director at various times since our inception in 2000 to September 2019. He also serves as Vice President, Ace Hardware Home Services – Ace Handyman Services Division of AHHS, a position he has held since June 2023. He served in similar positions for our former parent and operating entities that have merged into AHHS. Since December 2021, Mr. Bell has been the Chief Executive Officer of APF. Since July 2023, Mr. Bell he also serves as a Manager of Bell Franchising, LLC, located in Lakewood, Colorado. From September 2015 to September 2019, he served as CEO of Stair Fitter Franchising, LLC located in Lakewood, Colorado. Mr. Bell also served on the Board of Directors of Tetras Properties, LLC located in Lakewood, Colorado from March 2005 until it was dissolved in September 2023. He previously served on the Management Committee of WebScheduler Technologies, LLC located in Lakewood, Colorado, from its inception until it was dissolved in March 2020. Mr. Bell retains management responsibility relating to the sale and operation of franchises offered under this Disclosure Document. He is located in our offices in Lakewood, Colorado.

Chris Bue - President

Chris Bue has been employed by us since September 2014, first as our Chief Operating Officer, and from September 2018 to September 2019 as our Chief Executive Officer, and currently as our President. From September 2019 to June 2023, he has also served as President, Franchising of Ace Handyman Services, Inc. ("AHS"), a former affiliate of ourserved in similar positions for our former operating entities that have merged into AHHS. Mr. Bue is located in our offices in Lakewood, Colorado.

Colette Bell – Vice President of Franchise Development

Colette Bell has been employed as Vice President of Franchise Development for us since January 2019. From September 2019 to June 2023, she served in the same position for AHS similar positions for our former operating entities that have merged into AHHS. Since July 2023, Ms. Bell also serves as a Manager of Bell Franchising, LLC. From 2016 to September 2019, she was a consultant to SMB Franchise Advisors located in Philadelphia, Pennsylvania. Ms. Bell is located in our offices in Lakewood, Colorado. Since December 2021, she has also been Vice President, Franchise Development for APF.

LaShelle L. Taylor – Vice President of Franchise Operations

LaShelle Taylor has been our Vice President of Franchise Operations since August 2021. She served as our Franchise Consultant from March 2021 to August 2021. From June 2006 to August 2019, Ms. Taylor served as Vice President of Operations and Support for Homewatch Caregivers International, Inc., located in Greenwood Village, Colorado.—Ms. Taylor is located in our offices in Lakewood, Colorado.

Karen T. Bloomfield – Vice President, Marketing

Karen Bloomfield has been our Vice President, Marketing since September 2023. She has also served in the same position for AHHS and APF since September 2023. Ms. Bloomfield previously was engaged by Ace Hardware as Director of Merchandising Management - Ecommerce from January 2022 to August 2023 and Director of Merchandising Management - Plumbing Division from January 2021 to December 2021. Prior to that, from August 2016 to January 2021, she served as Vice President of Merchandising for Power Equipment Direct, a wholly owned subsidiary of Ferguson Enterprises, in Bolingbrook, Illinois. Ms. Bloomfield is located in Oak Brook, Illinois.

<u>Jeff A. Klos – Vice President, Finance</u>

Jeff Klos has been our Vice President, Finance since June 2023. He has served in the same position for APF since June 2023. Mr. Klos has been employed by Ace Hardware since April 2000 and currently serves as Vice President Finance, Ace Hardware Home Services (since June 2023). Mr. Klos has previously held the positions of Corporate Vice President, Financial Planning and Analysis and Strategy, from July 2020 to May 2023; and Director, Financial Planning and Analysis from March 2017 to June 2020. Mr. Klos is located in Oak Brook, Illinois.

ITEM 3 LITIGATION

LITIGATION INVOLVING US:

Concluded Actions

Ace Handyman Services, Inc. v. Ace Hardware Corporation et al. (Cause No. 2020-CI-03722, 468th Judicial District Bexar County, Texas). On February 25, 2020, Ace Handyman Services, Inc., an unaffiliated third party ("Ace Handyman Services"), filed a complaint and obtained a temporary restraining order against us, Ace Hardware, Ace Handyman Operating, Inc. (a former affiliated company of ours), and two of our franchisees (collectively, the "Ace Defendants") alleging that it has superior rights in the mark "ACE HANDYMAN SERVICES" in eight counties in the San Antonio, Texas region in connection with a handyman services business. The plaintiff alleged common law trademark infringement and common law unfair competition and sought to prevent the Ace Defendants from using the name "ACE HANDYMAN SERVICES" in the eight counties. The plaintiff sought a permanent injunction, unspecified actual and exemplary damages, pre-judgment and post-judgment interest, attorneys' fees, and costs. On April 16, 2020, the parties settled this matter, whereby the Plaintiff was paid \$150,000 in consideration for our franchisees and us to immediately be permitted to use the ACE HANDYMAN SERVICES mark and the plaintiff's agreement to discontinue its use of ACE and ACE HANDYMAN SERVICES.

LITIGATION OF OUR PARENT, ACE HARDWARE:

Concluded Actions

Cathay Bank v. Ace Hardware Corporation, et al. (Case No. KC069647, Superior Court of the State California, County of Los Angeles, Eastern District.) In February 2017, Cathay Bank ("Bank") obtained several money judgments against Sentron International, Inc. ("Sentron"), a former Ace Hardware vendor, in the aggregate amount of \$10,642,452, and had a receiver, Robb Evans & Associates LLC ("Receiver") appointed to liquidate Sentron's assets. In May 2017, the Bank obtained a money judgment against Smith Barnett LLC ("Smith Barnett"), also a former Ace Hardware vendor, in the amount of \$2,332,080, for successor liability, among other claims. In September 2017, the Bank and the Receiver filed a civil action against Ace Hardware and other defendants alleging that Ace Hardware made wrongful payments to Smith Barnett to purchase products belonging to Sentron in the amount of \$2,020,084, and possible additional wrongful payments totaling \$4,352,224, which may include the \$2,020,084. The eight-count third amended complaint alleges violations of various California commercial code sections, conversion, breach of fiduciary duty, constructive fraud, involuntary and constructive trust, and tort in essence, and seeks the return of Sentron's proceeds, unspecified monetary damages, punitive damages, the imposition of a constructive trust, disgorgement and attorneys' fees. The breach of fiduciary duty count was dismissed in December 2020. On March 15, 2021, Ace Hardware filed an Amended Cross-Complaint against the Bank alleging that, in connection with its loans to Sentron, the Bank violated California's Unfair Competition Law by: (i) aiding and abetting Sentron's fraudulent business practices; and (ii) violating its internal commercial lending policies and federal banking guidelines and regulations. On May 13, 2022, the parties settled this matter in consideration for the payment of \$1.1 million to the plaintiffs, \$450,000 of which was paid by co-defendant JMI, Inc. and \$650,000 of which was paid by Ace Hardware. The plaintiffs' complaint and Ace Hardware's cross-complaint were dismissed on July 14, 2022.

Advanced Caregivers, LLC d/b/a Hialeah Ace Hardware and William Bloodworth v. Ace Hardware Corporation (Case No. 12-20082-CIV-Williams, U.S. District Court, Southern District of Florida). In January 2012, a former member of Ace Hardware and its owner brought a civil action against Ace Hardware on behalf of themselves and a putative nationwide class alleging that Ace Hardware defrauded them in connection with their decision to become an Ace Hardware member and operate a Vision 21 store in Hialeah, Florida. The three-count complaint, which was amended on December 5, 2012 asserting the same complaints, alleged violations of the Illinois Franchise Disclosure Act, fraudulent inducement and fraud, and sought class action certification and unspecified monetary damages and attorneys' fees. Counsel for the putative class subsequently filed numerous motions for leave to file several amended complaints which named new and additional putative class representatives, and Ace Hardware filed motions to dismiss and oppositions to the various motions to compel. Finally, the then named plaintiffs (fifteen corporate entities and seven bankruptcy trustees) and Ace Hardware attended a mediation on July 26, 2016. The plaintiffs' sought \$2.5 million for a global settlement. Ace Hardware made an offer to settle with the plaintiffs for \$25,000 each, which was subsequently accepted by all of the plaintiffs. The case was subsequently dismissed with prejudice on July 27, 2017.

<u>Ace Handyman Services, Inc. v. Ace Hardware Corporation et al.</u> (Cause No. 2020-CI-03722, 468th Judicial District Bexar County, Texas). SEE DISCLOSURE UNDER LITIGATION INVOLVING US ABOVE.

Other than the cases listed above, no litigation is required to be disclosed in this ITEM.

ITEM 4 BANKRUPTCY

No bankruptcy information is required to be disclosed in this ITEM.

ITEM 5 INITIAL FEES

Initial Fees

The Initial Franchise Fee for a single AHS Business franchise is \$70,000 for a Territory size of up to 70,000 households. If your actual Territory size exceeds 70,000 households, the Initial Franchise Fee is increased \$1.00 for each additional household located in the Territory, up to a maximum Territory size of 100,000 households. On occasion, we have, and may continue to, negotiate a reduced Initial Franchise Fee for people acquiring more than five Territories at the same time.

If, with our consent, you acquire a Mini AHS Business, the Initial Franchise Fee is \$35,000 for a Territory size of up to 35,000 households. If your actual Territory size exceeds 35,000 households, the Initial Franchise Fee is increased \$1.00 for each additional household located in the Territory, up to a maximum Territory size of 45,000 households.

All Territories include the businesses in the Territory.

If you acquire multiple Territories, you will sign a separate Franchise Agreement and pay the Initial Franchise Fee for each Territory you acquire.

The amount of the Initial Franchise Fee payable by you will be noted in <u>Exhibit I</u> to the Franchise Agreement (the "Addendum") and is due in full upon the signing of the Franchise Agreement.

Discounts

Additional Territory Discount. If you are an existing franchisee of ours that: (a) has been operating an AHS Business for at least six months, (b) meets our qualifications for a new franchisee of ours, and (c) are purchasing an additional Territory from us (for which you are signing a separate franchise agreement), you may receive a discount of \$14,000 off of the Initial Franchise Fee for the additional Territory you may acquire.

Additional Concept Discount. If you own and operate an existing franchisee that you acquired from an affiliate of ours, are in full compliance under your franchise agreement with our affiliate for at least one year, and you meet our qualifications for a new franchisee of ours, you may receive a discount of \$14,000 off of the Initial Franchise Fee for a new franchise from us.

VetFran Discount. We are a member of the International Franchise Association's VetFran initiative. If you have served as a veteran of any branch of the United States Armed Forces and received an honorable discharge, you may receive a discount of \$7,000 off of the Initial Franchise Fee for the first Franchise Agreement you sign.

Discounts may not be combined. If you qualify for multiple discounts, the largest discount available to you will be granted.

<u>General</u>

All of the initial fees described in this ITEM 5 are fully earned by us on signing of the Franchise Agreement and are entirely non-refundable in all events.

ITEM 6 OTHER FEES

Column 1	Column 2	Column 3	Column 4
Type of Fee	Amount	Due Date	Remarks
Royalty Fee (1)(2)(3)(4)(8)	6% of your actual Gross Revenues of the previous month, subject to annual reconciliation based on your Minimum Annual Gross Revenues requirements	Payable monthly on the 23 rd -18 th day of the month (or shortly thereafter if paid by EFT) based on the prior month's Gross Revenues. See Notes 2 and 8 below.	You must meet certain Minimum Annual Gross Revenues requirements. See Note 4 below and ITEM 12.
National Brand Fee (1)(2)(3)(4)(8)	2% of your actual Gross Revenues, subject to annual reconciliation based on your Minimum Annual Gross Revenues requirements	Payable monthly with the Royalty Fee based on the prior month's Gross Revenues. <u>See</u> Note 2 below.	You must meet certain Minimum Annual Gross Revenues requirements. See Note 4 below and ITEM 12.
Software and Internet Fees (1)(8)	Currently \$499-599 per month, but may be changed by us upon 30 days' notice	Payable on the 1 st of each month for that month, beginning in the first full calendar month after you commence operations.	For use of our scheduling software, e-mail, location website, online marketing services, for our expenses associated with maintenance of our e-mail and Intranet sites, and for the website development, search engine optimization, and related online marketing services provided by our designated suppliers.
Minimum Individual Marketing Expenditure (5)(6)	\$30,000 from your third full month through the end of first year of operations, \$40,000 in your second year of operations, and \$50,000 for each subsequent year For a Mini AHS Business, the minimum	As incurred.	You must spend a minimum amount on local marketing on an annual basis. See also, Note 5 below.
	expenditures are \$15,000 from your third full month through the end of first year of operations, \$20,000 in your second year of operations, and \$25,000 for each subsequent year		

Column 1 Type of Fee	Column 2 Amount	Column 3 Due Date	Column 4 Remarks
Meeting and Convention Fees	-0- If you do not attend: \$1,500 per meeting If you attend but do not stay at the designated hotel or do not stay for the entire meeting or convention: \$750 per meeting	As incurred.	The additional meeting and convention requirements are discussed in ITEM 11.
Transfer Fee (1)(8)	\$10,000. There is no Transfer Fee for a transfer to a business entity wholly owned by franchisee, and the Transfer Fee is reduced to \$3,500 for transfers to a family member	Before acceptance of transfer.	Payable before you transfer your franchise. You will also be required to pay additional fees to outside franchise brokers or to us if we, or our resources, find the buyer for you.
Audit Fee (1)(7)(8)	Cost of audit	Upon completion of audit and determination of amounts due.	You also pay the cost of the audit if audit shows an understatement of at least 2% of Gross Revenues during the audit period. Any amounts unpaid, unreported, or underreported must be paid in full, with a late fee of 18% interest per annum.
Returned Electronic Funds Transfer Fee (1)(8)	\$100	Upon notice.	This fee is imposed when an Electronic Funds Transfer transaction is not honored, in addition to any late fees and interest that may apply.
Successor Franchise (Renewal) Fee (1)(8)	\$5,500	Upon execution of successor Franchise Agreement.	If you are in good standing, you may acquire a successor franchise after your initial term.
Supplier Approval (1)	Currently none. However, we reserve the right to charge a fee in the future	As incurred.	If you request approval of a supplier, we may charge this fee. This only applies for those items for which we have previously designated an approved supplier.
Additional training (1)(8)	Our then current rate. The current rate as of the date of this Disclosure Document is \$55 per hour per person plus out-of-pocket expenses	Payable within 30 days of date of invoice.	We provide opening training for free.

Column 1 Type of Fee	Column 2 Amount	Column 3 Due Date	Column 4 Remarks
Interest and late fees on late payments and/or reports (1)(8)	Not to exceed 18% per annum on all amounts owed. Also, \$75 per occurrence per late report or fee	Due upon payment of late fee or submittal of late report.	No amounts charged will exceed any legal limits or restrictions.
Costs and Attorneys' Fees (1)	Will vary under circumstances	As incurred.	May be payable if you do not prevail in an action based on the Franchise Agreement.
Indemnification (1)	Will vary under circumstances	As incurred.	You will have to reimburse us if we are held liable for claims arising from your AHS Business.

Notes:

- (1) Fees that are payable to us. All fees are non-refundable. You must maintain in a bank account from which funds can be transferred to us a minimum balance of \$4,000 against which we will withdraw funds due to us. All fees are generally imposed uniformly on all franchisees who sign our current Franchise Agreement, but we may in unique situations modify certain fees. Certain fees set forth in the current Franchise Agreement have changed from the amounts charged in the past and may change in the future. Therefore, existing and future franchisees may have fees imposed on them that are different from those represented in this table.
- (2) The Royalty Fee and National Brand Fees are payable monthly. You will also report Gross Revenues through monthly, quarterly, and annual reports, and in the event of any discrepancy in the Gross Revenues previously estimated or reported and the most recent applicable report you will either be required to pay the amount of any underpayment or be granted a credit for the amount of any overpayment, which will apply to future fees due.
- (3) Gross Revenues include all revenues you receive from your customers, including amounts collected for both labor and materials. Gross Revenues do not include sales or use tax. Gross Revenues are net of any discounts you provide to your customers, to the extent the customers pay less to you. Cash refunded and credit given to customers, and any receivable uncollected from customers more than 120 days after billing, may be deducted in computing Gross Revenues to the extent that the receivable was previously included in Gross Revenues on which a Royalty Fee was paid.

If you render services to a customer or otherwise recognize a sale (whether or not you have received payment from the customer), the Royalty Fee is due for the period the service was performed and completed.

If a job is partially complete at the end of a month, and it represents less than 16 hours of labor, it can be included in the next month's calculations. However, a larger job that will continue beyond this time period will need to be settled and the Gross Revenues will be recognized in each month.

You will participate in our Electronic Royalty Payment program, which authorizes us to use a preauthorized bank transfer system. You will also participate in an electronic reporting system covering sales, operating reports and other items. We may designate and require another manner of payment from time to time in our discretion.

- (4) You will be required to achieve the Minimum Annual Gross Revenues described in ITEM 12. Following the first anniversary of your commencement of operations and on each subsequent anniversary, if your actual Gross Revenues for the then completed operations year do not equal or exceed the Minimum Annual Gross Revenues, you will be required to pay us a reconciliation amount equal to amount of Royalty Fees and National Brand Fees that are due to us based on the applicable Minimum Annual Gross Revenues minus the actual Royalty Fees and National Brand Fees paid to us during the applicable operations year.
- (5) You must spend between \$9,000 to \$12,000 as a required initial marketing expense in the period starting 30 days before, and continuing through the end of the second full month after, the opening of your franchise. You will pay this amount directly to our supplier conducting your initial marketing campaign. You must also spend between \$2,000 1,000 and \$4,000 for a required Grand Opening event.
- (6) We may require you to join a Local Advertising Group that includes franchisees in your market area. If so, you may be directed to pay all or a portion of your Minimum Individual Marketing Expenditures to the Local Advertising Group.
- (7) We estimate that the cost of an audit will range from \$2,500 to \$5,000, but may be more.
- (8) These payments will be made to us electronically. We currently require you to pay fees and other amounts due to us or our affiliates via electronic funds transfer ("EFT") or other similar means. You are required to complete an EFT authorization in the form attached to the Franchise Agreement as <u>Exhibit V</u>.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT

Column 1	Column 2	Column 3	Column 4	Column 5
Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment is to be Made
Initial Franchise Fee (1)	\$70,000 - \$100,000	Lump sum	At signing of Franchise Agreement	Us
Travel and living expense while training (2)	\$3,500 - \$ 6,0004 ,500	As incurred	As incurred during training	Airlines, hotels, restaurants, auto rental
Lease Deposit (3)	\$1,000 - \$ 3,000 <u>3,200</u>	Lump sum	At signing of lease and for term	Lessor
Rent for first 3 months (3)	\$ 3,000 - <u>3,500</u> - \$ 4,500 <u>4,800</u>	As incurred	Monthly	Lessor
Leasehold Improvements and Signage (3)	\$ 800 - <u>600</u> - \$ 5,000 4,000	Will vary	As incurred	Contractor or Suppliers
Furnishings (3)	\$ 800 - <u>600</u> - \$ 5,000 4,000	Will vary	As incurred	Suppliers
Tools and equipment (4)	\$0 - \$ 3,500 2,500	Lump sum	At delivery	Suppliers
Truck or Van (5)	\$0 - \$ 6,000 <u>9,000</u>	As incurred	Monthly	Suppliers

Column 1	Column 2	Column 3	Column 4	Column 5
Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment is to be Made
Computer Hardware and Office Equipment ⁽⁶⁾	\$ 2,250 - <u>2,500</u> - \$ 5,000 <u>5,500</u>	As needed	Before training and as incurred	Third Party
Software and Internet Fees (3 months) (7)	\$ 1,497 <u>1,797</u>	As incurred, monthly	Monthly	Us
Initial Marketing Spend (8)	\$ 11,000 <u>9,000</u> - \$16,000	As incurred	Varied times from pre- opening to post-opening	Suppliers
Utility deposits and fees	\$ 100 - <u>500</u> - \$ 700 1,000	As incurred	Monthly	Utility companies
Licenses and permits (9)	\$ 630 -1,000 - \$ 3,500 5,000	Lump sum	At time of application	Government agencies or third parties
Insurance (10)	\$4 ,000 \$5,100 5,500 - \$7,000	As incurred	Monthly	Insurance companies
Professional fees	\$500 - \$6,000	As incurred	As agreed	Attorneys, accountants and other consultants
Miscellaneous opening costs	\$2,000 - \$4,500	As incurred	Varied times	Suppliers, vendors
Additional Funds (3 months) (11)	\$30,000 - \$45,000	As incurred	Varied times	Suppliers, utilities, vendors
TOTAL (12)(13)(14)(15)	\$131,077 - \$220,297 ⁽¹⁾ \$131,997 - \$223,797 ⁽¹⁾			

Unless otherwise indicated in this document, all items payable to us are non-refundable.

Notes:

(1) See ITEM 5 for discussion on the Initial Franchise Fee. The Initial Franchise Fee increases by \$1.00 for each additional household in your Territory over 70,000 households. The high end represents the Initial Franchise Fee for the maximum territory size of 100,000. There may be a discount available to you.

If you acquire a Mini AHS Business, the Initial Franchise Fee is between \$35,000 and \$45,000. All other expenses remain the same.

Column 1 Type of Expenditure	Column 2 Amount	Column 3 Method of Payment	Column 4 When Due	Column 5 To Whom Payment is to
		,		be Made
Initial Franchise Fee (1)	\$35,000 - \$45,000	Lump sum	At signing of Franchise Agreement	Us

Column 1	Column 2	Column 3	Column 4	Column 5
Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment is to be Made
Estimated Initial Investment Expenses Noted Above Minus the Initial Franchise Fee	\$61,077 \$120,297 <u>\$61,997</u> - \$123,797	See Chart Above	See Chart Above	See Chart Above
TOTAL (12)(13)(14)(15)	\$ 96,077 <u>96,997</u> - \$ 165,297 ⁽¹ <u>168,797⁽¹⁾</u>			

- (2) The initial training program is tuition-free for up to four people. However, you will pay for the expenses of attendance, including lodging, meals, transportation, and wages of trainees.
- (3) You must operate from a small office outside of your home. A commercial lease normally requires payment of the first month's rent, a security deposit and a rent deposit equal to one month's rent. A lease deposit may or may not be refundable and lease deposits vary widely from location to location. You will also incur costs to furnish your office, which may include purchasing a desk, office chair, file cabinets, decorations, and other office items.
- (4) You can rent tools or you may find it cost effective to purchase various tools with a high use rate. The tools you are required to acquire include ladders of varying lengths, a tub saw for tile, a chop box with miter saw, a hammer drill, various pneumatic tools with an air compressor, and various hand tools. The low end of this estimate assumes you already own the necessary tools.
- (5) You are not required to purchase or lease a vehicle until you have three fulltime craftsmen (working an average of 30 hours per week or more). You may purchase or lease a vehicle before you reach this requirement. Any vehicle you acquire must be either a van or truck and we recommend that it be white, gray, red, or black.
- (6) Includes a computer and software meeting our minimum specifications that you purchase from suppliers selected by you. You will need a workstation, laser printer, fax software, hub-router, network, and wiring. A high-speed Internet connection is also required. If you do not already own one, you will need a smart phone or other device we approve meeting our standards and specifications. The standards and specifications we are currently utilizing are set forth in the Operations Manual.
- (7) In exchange for the Software and Internet Fee, we give you access to our scheduling software, e-mail, location website, and online marketing services, which includes website development services and search engine optimization. We also maintain an Internet site for prospective customers and an Intranet site for you and other franchisees. We reserve the right to discontinue providing the e-mail account and Intranet sites, if technology advancements make these obsolete or impractical to provide. The amounts shown in the chart are for the fees for the first three months following the commencement of operations of your AHS Business. We pay a portion of the Software and Internet Fee to our designated suppliers for website development services, search engine optimization, and other online marketing services. These services do not include paid, placed media advertising, which is an additional expense paid to the vendor directly by you.
- (8) You must spend between \$9,000 and \$12,000 as a required initial marketing expense in the period starting 30 days before opening, and continuing through the end of the second full month after opening of your franchise. This amount includes collateral marketing materials that will last for a period

of time beyond the first three months of operation and are purchased in quantities based on price considerations. You must also spend between \$2,000 and \$4,000 for a required Grand Opening event.

- (9) Licensing laws and permit requirements, including fees, may vary from state to state and from city to city, depending on state and local laws. You must comply with these laws. This estimate also includes the costs associated with obtaining the required LRRP Certification for your AHS Business and one individual representative, including the estimated costs for the required training course provided by the EPA or a third party.
- You must procure, maintain and provide evidence of (i) workers' compensation insurance with (10)employers liability coverage, with a broad form all-states endorsement coverage sufficient to meet the requirements of law (or, if such coverage is not required by applicable law, then with limits of at least \$1 million and otherwise meeting the requirements we designate); (ii) comprehensive general liability insurance including product liability, property damage, personal injury liability, bodily injury coverage, and premises/operation liability for the Business Location and its operations, with a limit of not less than \$1 million per occurrence and \$2 million in the aggregate, or such greater limit as may be required as part of any lease agreement for the Business Location; with a crime policy endorsement with a limit of not less than \$25,000; (iii) comprehensive automobile liability insurance including hired and non-owned coverage with a limit of not less than \$1 million per occurrence and \$2 million in the aggregate; (iv) umbrella commercial excess coverage with a limit of not less than \$1 million; and (v) all-risk business property insurance covering all equipment, tenant improvements, inventory and other personal property owned or leased by you in an amount equal to at least 100 percent of the replacement costs of the contents and tenant improvements located at the Business Location, less a reasonable deductible which cannot exceed \$5,000. Cyber liability insurance in an amount of not less than \$250,000 is optional but highly recommended. We reserve the right to require that you obtain and maintain, in addition to all other policies of insurance, employment practices insurance in such amount as we specify. All of the required policies of insurance must name us as an additional insured with waiver of subrogation and must provide for a 10 day advance written notice to us of cancellation. Prior to opening your AHS Business, and within 15 days after each successor franchise term begins, you must provide a certificate of insurance as to the insurance coverage maintained to a third-party administrator we use to monitor insurance compliance.
- (11) The disclosure laws require us to include this estimate of all expenses to operate your AHS Business during the "initial period" of the business, which is defined as three months or a longer period if "reasonable for the industry." We are not aware of any established longer "reasonable period," so our disclosures cover an estimate of start-up expenses for an initial three-month period. These expenses do not include your living expenses or expenses if you decide to employ assistants. You are required to appoint an office manager ("Office Manager") who will answer the phone, schedule your craftsmen, and assist with running your office for your AHS Business within 75 days of signing your Franchise Agreement. We have included an estimate for the costs you will incur during the first three months to employ the required Office Manager. Based on records we received from our franchisees, we estimate that you will pay the Office Manager an hourly rate of \$24 to \$30 per hour, for 40 hours per week, for each of the first 12 weeks of operation of your AHS Business.
- (12) The estimated initial investment set forth in this table assumes that you will acquire the franchise rights for one AHS Business with a Territory of up to 100,000 households. If you acquire more than one AHS Business, your estimated initial investment will increase.
- (13) The chart contains estimates only and we cannot guarantee that you will not have additional expenses starting your business. Your actual expenses may exceed our estimates. The information in the chart is not intended to imply that your AHS Business will reach the breakeven point or any other financial

goal by that time. We recommend that you have (or have access to) additional startup capital beyond the amounts shown in the table above.

- (14) We relied on over 75 years of combined experience of certain officers of the company in the handyman service business to compile these estimates. You should review these figures carefully with a business advisor before making any decision to purchase the franchise.
- (15) No payments to us are refundable. Payments to third parties may or may not be refundable depending on your agreement with such third parties; however, usually such payments are nonrefundable.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Operations

You will establish and operate your AHS Business in compliance with your Franchise Agreement, and with the standards and specifications contained in an Operations Manual that we make available to you (the "Operations Manual"). The Operations Manual is designed to protect our reputation and the goodwill of the Marks. It is not designed to control the day-to-day operations of your AHS Business.

You will use our scheduling software, which we grant you access to in exchange for your payment of the Software and Internet Fee. Our third party designated vendor is the supplier of this software. You will also obtain website development services, search engine optimization, and other online marketing services from our designated suppliers, which are independent third parties. We currently bill you for these required services by charging the Software and Internet Fee and pay the suppliers on your behalf. We currently use Ingeniux Corporation ("IGX") to service and maintain our website. You will enter into a Participation Agreement with IGX to permit you to receive access to IGX's Vendor SaaS Services. A copy of the IGX Participation Agreement is attached to this Disclosure Document as Attachment N.

If you do not already have it, you will need to obtain computer hardware and software that meets our minimum specifications. You will also need a smart phone or other device approved by us that meets our minimum specifications for receiving and sending e-mails remotely, unless you already own one you will use in the AHS Business. You are required to answer your AHS Business' phone during regular business hours by a live person, which may be your Office Manager or an answering service.

Approved Suppliers

We currently have approved suppliers from whom you will purchase or otherwise acquire certain materials, supplies, services and items of equipment that you will use in or sell and install in customer locations through your AHS Business. We also currently have approved suppliers from whom you will purchase digital advertising services and merchant processing services. We will provide you with information on these suppliers after you sign the Franchise Agreement. Ace Hardware and its network of retailers (collectively, the "Ace Suppliers"), some of whom are owned by Ace Hardware, are approved suppliers of certain goods, materials, supplies, fixtures, equipment, and inventory used in your AHS Business. You are required to purchase all materials, supplies and goods needed to run your AHS Business from an Ace Supplier, but only when an item is in stock or readily available, and competitively priced.

In addition to the Ace Suppliers, we may in the future require you to purchase certain goods, services, supplies, fixtures, equipment, and inventory used in your AHS Business from other approved suppliers. If we do require you to purchase any other goods and services from only approved or designated suppliers, we will advise you of those suppliers.

Except as noted above and in the Operations Manual, you may purchase any products, services, supplies, fixtures, equipment, and inventory from any supplier.

Once we designate a particular supplier for a particular product or service, if you want to utilize similar materials or services in your AHS Business from a supplier that has not previously been approved by us, you will need to first obtain our approval.

We do not have written criteria for supplier approval, and therefore written criteria are not provided to our franchisees. However, we may require that you submit specifications, information or samples of the items for our review to determine if they meet our specifications and standards set forth in the Operations Manual as to quality, content, composition and service. We will advise you within 30 days after we receive the required information whether the items or proposed suppliers meet our specifications. We do not currently require that a fee be paid to us to secure supplier approval; however, we may require a fee in the future, and in our discretion, we may require the supplier to reimburse us for any expenses we incur in determining if the supplier meets our specifications and standards. We may revoke our approval of any supplier previously approved by written notice to the supplier and each licensee using that supplier.

Ace Hardware is one of our approved suppliers and you are required to purchase certain goods and services from the Ace Suppliers for your AHS Business. Otherwise, neither we nor any affiliate of ours is currently an approved supplier for any goods or services for your AHS Business, although we reserve the right to designate ourselves or other affiliates as approved suppliers for certain goods or services in the future.

Vans and Trucks

You are required to purchase or lease a vehicle for every three craftsmen you employ working an average of 30 hours per week or more. For example, you must purchase one vehicle when you have three craftsmen working an average of 30 hours per week or more, two vehicles when you have six such craftsmen, three vehicles when you have nine such craftsmen, etc.

Currently, our recommendation is that the vehicle be a white, gray, red, or black van or truck and capable of being monikered with our required vehicle designs and logo. We may change our requirements in the future.

You may acquire additional vans or trucks provided all vans and trucks used in your AHS Business are approved by us and must display our approved vehicle design advertising and logo. In the event that you use a personal vehicle in your AHS Business that is not one of the required vehicles, we may require you use a vehicle magnet with our logo or name on such vehicle.

Additional Information

We may derive revenue from purchases or leases of inventory and equipment from any approved supplier of ours on account of their dealings with you and other franchisees. During our last fiscal year (ending December 3128, 2023 2024), we did not derive any revenues from materials sold to our franchisees. We do not derive revenues from real estate leases. No officer of ours owns any interest in any of our approved suppliers.

In the fiscal year ending December 3128, 2023 2024, Ace Hardware's revenues, which includes the revenues from its subsidiaries, Westlake and GLA, from the sale of merchandise and services to its network of Ace Hardware retail stores was approximately \$7.5 8.3 billion out of its consolidated total revenue of approximately \$9.1 9.5 billion. In 2023 2024, based on tracking mechanisms implemented in Ace Hardware

retail stores, approximately \$\frac{826,000}{277,000}\$ of its revenues was derived from sales of products to our franchisees.

We estimate that the cost of your purchases of materials meeting our specifications will represent approximately 8 to 15 percent of your total purchases in establishing your franchise. We estimate that purchases in accordance with our standards and specifications will comprise 80 to 90 percent of the product purchase requirements for the ongoing operations of an AHS Business.

We currently do not privately label any products, but we reserve the right to do so in the future.

Other than the purchasing arrangement with the Ace Suppliers, we do not currently, but reserve the right in the future to, negotiate purchase arrangements with suppliers, including preferred pricing arrangements, for the benefit of our franchisees. Further, we do not currently, but reserve the right in the future to, receive payments from approved suppliers with respect to your purchases.

Except as is described in this ITEM 8, you do not receive a material benefit from us based on your use of any particular designated or approved sources or your purchase of particular products or services.

REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the Franchise Agreement and other agreements. It will help you to find more detailed information about your obligations in these agreements and in the other items of this Disclosure Document.

	Obligation	Section in Agreement	Disclosure Document Item
a.	Site selection and acquisition/lease	Section 4.1	ITEM 7 and 11
b.	Pre-opening purchases/leases	Sections 7.1 and 11.1	ITEMS 7, 8 and 11
c.	Site development and other pre- opening requirements	Article 7	ITEMS 6, 7 and 11
d.	Initial and ongoing training	Article 6	ITEM 11
e.	Opening	Section 7.3	ITEM 11
f.	Fees	Articles 5 and 12, Section 13.5	ITEMS 5 and 6
g.	Compliance with standards and policies/Operations Manual	Article 9 and Section 11.1	ITEM 1 and 11
h.	Trademarks and proprietary information	Article 15 and Sections 18.6 and 20.4	ITEMS 13 and 14
i.	Restrictions on products/services offered	Sections 11.1, 14.2, 18.2 and 20.1	ITEMS 8 and 16
j.	Warranty and customer service requirements	Section 11.1	ITEM 11
k.	Territorial development and sales quotas	Article 4 and Section 12.4	ITEMS 11 and 12

	Obligation	Section in Agreement	Disclosure Document Item
1.	Ongoing product/service purchases	Sections 11.1 and 14.2	ITEMS 8 and 16
m.	Maintenance, appearance and remodeling requirements	Section 11.1	ITEM 7
n.	Insurance	Article 21	ITEM 7
0.	Advertising	Article 13 and Section 12.2	ITEM 11
p.	Indemnification	Section 19.3	ITEMS 6 and 14
q.	Owners participation/ management/staffing	Section 11.1	ITEM 15
r.	Records and reports	Article 16 and Section 18.1	ITEMS 6 and 17
s.	Inspection and audits	Section 16.3	ITEM 6
t.	Transfer	Article 17	ITEM 17
u.	Renewal	Sections 3.3 and 3.4	ITEM 17
v.	Post-termination obligations	Sections 18.6, 18.7, 18.8, 18.9, 20.2, 20.3, 20.4, and 20.5	ITEM 17
w.	Non-competition covenants	Article 20	ITEM 17
x.	Dispute resolution	Article 22	ITEM 17
y.	Guaranty by Owners and Other Individuals	Sections 11.1 and 11.2, and Exhibit III	ITEM 15

ITEM 10 FINANCING

We do not offer direct or indirect financing. We do not guarantee your note, lease or obligation.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

Pre-Opening Assistance

Before you open your business, we or our designee will: (Section 8.1 of Franchise Agreement ("FA") unless otherwise noted)

- 1. Designate your protected Territory.
- 2. Provide you access to our scheduling software, an Internet site for prospective customers, an e-mail account and an Intranet site for you and other franchisees.
- 3. Provide you with the specifications of all initial and replacement equipment, inventory and supplies required for the operation of your AHS Business, all as specified in the Operations Manual, and a list of all approved suppliers of any services, equipment, inventory or supplies that we require you use.

- 4. Provide you with pre-opening and opening activities, including during the first several weeks of operation of your AHS Business. These activities will include providing you with a list of information about tasks you need to complete to establish your business, assisting you in completing those tasks and assigning a field support TEAM member to begin assisting you with completion of the startup list.
- 5. Provide you access to our proprietary project management start-up onboarding program, which includes proprietary and confidential guidance and resources for setting up an AHS Business, insurance resources, and approved supplier information.
- 6. Make available to you during the term of your Franchise Agreement one copy of our confidential Operations Manual, which may be provided electronically to you and which contains mandatory and suggested specifications, standards, operating procedures and rules, arts and graphics for advertising, and required product and service warranties. (Section 9.3 of FA.)
- 7. Within 90 days of you signing the Franchise Agreement, train you and up to three other people during a required 20-29 hour initial training program, which may be conducted in-person or virtually via video conference. We may waive or modify the required initial business training program in our sole discretion.

We do not provide the above services to franchisees acquiring a successor franchise.

Continuing Assistance

During the operation of your AHS Business, we or our designee will: (Section 10.1 of FA unless otherwise noted)

- 1. Within the first 90 days of operation of your AHS Business, provide to you approximately 18 to 24 hours (over two to three business days) of on-site training at your Business Location.
- 2. Research new products, services and methods of doing business and provide you with information concerning developments of this research.
- 3. Make ourselves available to you on a reasonable basis via telephone, text, and/or e-mail during our regular business hours to discuss your operational questions and experiences.
- 4. Provide regular communications on current best practices through telephone calls, webinars, video conferences, e-mail and other electronic communications, internal blogs, and other means.
- 5. Hold annual conferences to discuss sales techniques, new product developments, new service suggestions, accounting, inventory control, performance standards, advertising programs, merchandising procedures and/or other topics deemed relevant by us. We may require you to attend these conferences. (Section 6.4 of FA.)
- 6. Send to you and to customers periodic online newsletters that we produce from time to time.
- 7. At our discretion, conduct additional training programs at a location agreed upon by you and us, or virtually via video conference. (Section 6.4 of FA.)

- 8. Make available to you marketing programs, advertising designs, advertising campaigns, and camera-ready artwork. We may charge you for these materials. (Sections 10.1 and 13.1 of FA.)
- 9. Assist in local market research and provide guidance and assistance to you in recommending the prices to be charged by you for the services and products provided by you. Unless prohibited by applicable law, we may periodically set a maximum, minimum, or set price that you may advertise and charge for services and products offered by your AHS Business. If we establish a maximum price for any services or products, you shall not offer or sell those services or products at any greater price. If we establish a minimum price for any services or products, you shall not offer or sell those services or products at any lesser price. If we establish a set price for any services or products, you shall offer or sell those services or products only at that price. If we do not establish pricing limits, we may establish suggested prices. In that case, any prices that we recommend to you are merely recommendations and you may establish your own prices, which may be higher or lower than our recommended prices. You must abide by our advertising policies related to advertising prices. We may also establish subscription package programs for residential or commercial customers periodically and may require you to offer those subscription packages and otherwise comply with the programs in your AHS Business.
- 10. Provide you access to our scheduling software and all upgrades, an e-mail account, an Internet site for prospective customers, and an Intranet site for you and other franchisees, and to social media accounts for your AHS Business that we deem relevant for the brand. We reserve the right to discontinue providing the e-mail account and Intranet site if technology advancements or solutions make these obsolete or impractical to provide or take the place of these tools. You may not create, have or use any other social media accounts related to your AHS Business other than the ones we provide.
 - 11. Assist with your initial marketing campaign. (Section 13.2 of FA.)
- 12. Permit you to use our Marks in accordance with our specifications. (Sections 2.1 and 15.1 of FA.)
- 13. Permit you to use the Ace Handyman Services System, as it may be modified. (Sections 2.1 and 15.3 of FA.)

Marketing

Local Marketing

At your initial training, we will provide you with a suggested marketing plan, marketing materials, advertising designs, advertising campaigns, camera-ready artwork, and budgeting templates and discuss the budgeting, execution, and maintenance of your marketing program.

For your initial marketing activities, you must spend between \$9,000 and \$12,000 in the period starting 30 days before opening, and continuing through the end of the second full month after opening of your franchise. You will pay this amount to the vendors who will assist you in developing and running your marketing activities. You must also hold a Grand Opening event. For your Grand Opening activity, you must spend between \$2,000 and \$4,000 on that event. You will pay this amount to vendors who will assist you in creating marketing materials and give aways for this activity.

Beginning with the third full month of operation of your AHS Business through the end of your first Contract Year, you must spend at least \$30,000, increasing to \$40,000 for your second Contract Year, and \$50,000 for each Contract Year thereafter as your Minimum Individual Marketing Expenditure. If you acquire a Mini AHS Businesses, these amounts are respectively reduced to \$15,000, \$20,000, and \$25,000

for these periods. These amounts are in addition to the National Brand Fee that you will pay to us. We will review your spend amounts each six months and if you are not on pace to meet or exceed the required Minimum Individual Marketing Expenditure, you must increase your Minimum Individual Marketing Expenditure for the second six months of the applicable 12-month period to meet the required annual Minimum Individual Marketing Expenditure.

We may designate as approved suppliers required or recommended media or advertising agencies or other vendors for certain types of marketing for your AHS Business at our discretion. Expenditures on online marketing services and products acquired through a third-party vendor shall count toward this Minimum Individual Marketing Expenditure.

If there is an Ace Hardware store located in or near your Territory, we encourage, but do not require, you to do promotions of your AHS Business through those stores. Independent Ace Hardware dealers are not required to participate in any such promotions and there is no assurance they will do so.

If you execute a Multi-Territory Addendum for the operation of multiple AHS Businesses in adjacent territories, that Multi-Territory Addendum will specify a combined Minimum Individual Marketing Expenditure amount that will apply to those AHS Businesses and territories as a whole.

To protect the standards of the Ace Handyman Services System and brand, before using any promotional and advertising materials, you are required to submit to us or to our designated agency, for our prior written approval, all information pertaining to promotional materials and advertising initiated by you, including written advertisements, radio and television scripts, Internet sites, social media advertising, vehicle wraps or vehicle magnets, or any promotional creative materials. If we do not give you our written approval or disapproval of any advertising and promotional material submitted to us by you within seven business days from the date the information is received by us, the materials will be deemed approved as submitted. Notwithstanding our approval of any advertising created or used by you, you are responsible for the accuracy of all advertising content used by you, and your obligation to comply with all of our advertising standards and specifications and all applicable legal requirements relating to advertising and promotional material. You may only use gift certificates that have been approved by us.

You will submit to us a local advertising and marketing expenditure report accurately reflecting expenditures, in a form and by methods as required by us in the Operations Manual or otherwise.

In all advertising and promotional materials, you must display your business name only in obvious conjunction with the phrase, "Locally Owned and Independently Operated Franchise of Ace Handyman Franchising, Inc." or similar statement approved by us.

Local Advertising Groups

We may require you to join a Local Advertising Group that includes other franchisees in your market area. If so, we may direct you to pay all or part of your Minimum Individual Marketing Expenditure to the Local Advertising Group. The membership, rules and regulations of a Local Advertising Group, including how advertising fees are to be spent, will be determined by the Local Advertising Group's member franchisees, but must be approved in advance by us. The Local Advertising Group must operate based on written governing documents, and those documents will be available for review by franchisees. A Local Advertising Group must provide quarterly financial reports to us, and those reports will be available for review by franchisees at our headquarters. We can require Local Advertising Groups to be changed, dissolved or merged. If a Local Advertising Group is established in a market area, all company-owned AHS Businesses operating in that market area, if any, will contribute to the Local Advertising Group on the same basis as franchisees.

National Fund

We have established the Ace Handyman Services National Brand Fund (the "National Fund"). You must pay 2 percent of your Gross Revenues, (subject to any adjustment based on you not meeting your required Minimum Annual Gross Revenues as discussed in ITEM 12) to the National Fund (the "National Brand Fee"). Unless we direct otherwise, the National Brand Fee is payable each month with the Royalty Fee based on the prior month's Gross Revenues. Payments into the National Fund are in addition to your Minimum Individual Marketing Expenditure requirements.

The creation of advertising materials and provision of other advertising and marketing services will be provided to you through the National Fund. We may occasionally place advertising for the entire Ace Handyman Services System through the National Fund. We reserve the right to use brand fees from the National Fund to place advertising in regional or national media (including broadcast, print, Internet or other media). The National Fund will be used to promote the services and products sold by franchisees and company-owned operations, to provide media placement for national recruitment of craftsmen, and for such other purposes as determined in our sole discretion to promote the Ace Handyman Services System and brand, including participation in charitable events and programs. It will not be used to sell additional franchises, except as stated below. (See Section 13.5 of FA.)

We may use outside advertising and marketing agencies to create advertising material.

We may charge you the costs of any marketing materials we produce or print for our franchisees. In that event, you must pay us upon receipt for all marketing materials purchased from us unless we agree to different terms. We have never charged for marketing materials we produced or printed for our franchisees in the past.

We may require that you include a brief statement in your advertising that AHS Business franchise locations are available and the contact information for acquiring AHS Businesses, and we may include this type of statement in any advertising developed by the National Fund.

All company-owned Ace Handyman Services locations, if any, will contribute to the National Fund on the same basis as franchisees. All payments to the National Fund will be spent on advertising, promotion and marketing of goods and services provided by us. We cannot ensure that you or any franchisee will benefit directly or on a prorata basis from placement of advertising through the National Fund.

The National Fund will not be required to operate under written governing documents. Our accounting and marketing personnel will administer the National Fund, and a portion of the salaries of those employees may be paid by the National Fund.

The National Fund will be accounted for separately from our other funds and will not be used to defray any of our general operating expenses, except for any accounting and marketing personnel salaries, administrative costs, overhead and other expenses we may reasonably incur in activities related to the National Fund and its programs (including conducting marketing research, preparing advertising and marketing materials, insurance, legal costs and collecting and accounting for the National Fund, as well as protecting the brand in any closed markets). In any event, we may charge the National Fund for attorney's fees and other costs related in any way to our defense of any claims against us regarding the National Fund (including our management of the National Fund) or regarding collecting amounts due and/or expenditures by or from the National Fund. We may, in our sole discretion, spend in any fiscal year an amount greater or less than the aggregate contributions to the National Fund in that year and the National Fund may borrow from us or other lenders to cover deficits of the National Fund or cause the National Fund to invest any

surplus for future use by the National Fund. We will have the right, but not the obligation, to cause the National Fund to be incorporated or operated through an entity separate from us as we deem appropriate, and any successor entity will have all rights and duties of ours relating to the National Fund.

We may remit a portion of National Fund contributions back to one or more franchisees on any terms and conditions we determine in our sole discretion, including reimbursement of local advertising expenditures made by a franchisee. We may waive and/or compromise claims for contributions to, and/or claims against or with respect to, the National Fund in our sole discretion, using the National Fund to pay any of these claims. We will have sole discretion as to whether or not we take legal or other action against any franchisee who is in default of his, her or its obligations concerning the National Fund (including obligations to make contributions) and whether a franchisee may be allowed to make direct advertising expenditures in place of contributions to the National Fund.

During the 2024 fiscal year of the National Fund, 63.8 58.9% percent of the National Fund was spent on media placement, 10.6-14.6% percent was spent on first party marketing, and 7.0% percent was spent on pay per click marketing. The remaining 19.5% percent was spent on website development and hosting, and 8.5 percent on pay per click marketing. The remaining 17.1 percent was spent on point of sale advertising, recruitment, professional fees, photo and video shoots and other administration expenses. The National Fund ended the 2023 fiscal year with a cumulative deficit (spending in excess of collections) of \$103,000. The 2024 spending from the National Fund will be reduced by this amount. No portion of the National Fund was spent to principally solicit new franchisees.

We do not currently intend to have the National Fund audited. An accounting of advertising expenditures will be available for review by you at our headquarters, and an annual unaudited financial statement of the National Fund will be available to you upon request.

Franchisee Advisory Council

We have established a Franchisee Advisory Council, which serves to assist us in determining how to operate the National Fund and other advertising matters. The Franchisee Advisory Council consists of elected and selected franchisees of the Ace Handyman Services System. We reserve the right to terminate the Franchisee Advisory Council or to modify its purposes, in our sole discretion.

Computer Requirements

We provide you access to our scheduling software upon payment of your Initial Franchise Fee. Our scheduling software will track your customers, schedule service calls, dispatch craftsmen, provide information to the customer about the craftsmen, provide mobile access to job information to the craftsmen (such as photos and notes), solicit and syndicate customer reviews, track customer service, and generate invoices and financial, marketing and sales reports and collect payment in the field, all that conform to our standards to maintain consistent business applications in accordance with the Ace Handyman Services System.

We maintain an Internet site for prospective customers, which will also include information for prospective franchisees regarding purchasing an AHS Business franchise. You may not use or own any other Internet sites related to the home improvement or a handyman business. We also maintain an Intranet site for you and other franchisees. We will grant you access to that Intranet site, which you must use in accordance with our specifications and the Ace Handyman Services System. We reserve the right to discontinue providing the Intranet site at any time in our sole discretion.

We will provide an e-mail account to you. You are required to use only the e-mail account provided by us in the operation of your AHS Business. You may not use the e-mail account for any purpose not related to the operation of your AHS Business. You are required to check this e-mail account throughout the business day and to respond to any e-mail messages promptly. We reserve the right to discontinue providing this e-mail account in the future, in which event you will be required to maintain an e-mail account to be used in the operation of your AHS Business. We also reserve the right to discontinue providing the e-mail account and Intranet sites if technology advancements make these obsolete or impractical to provide. You must pay us the Software and Internet Fee of \$499-599 per month for the scheduling software, e-mail, location website, and online marketing services, which includes an e-mail account for national and local use. If you are operating multiple AHS Businesses in adjacent territories, we may agree to allow you to split the Software and Internet Fee and pay a single software fee ("Software Fee") for all of your AHS Businesses and an internet fee ("Internet Fee") based on the number of location websites you have, pursuant to the terms of the Multi-Territory Addendum, in which case you may receive a single e-mail account for the covered AHS Businesses.

You must obtain website development, search engine optimization, and related online marketing services from our designated suppliers. We currently use Ingeniux Corporation (defined above as "IGX") to service and maintain our website. You will enter into a Participation Agreement with IGX to permit you to receive access to IGX's Vendor SaaS Services. A copy of the IGX Participation Agreement is attached to this Disclosure Document as Attachment N. We bill you for these services as part of the Software and Internet Fee, and we pay IGX and any other designated suppliers of these services for you, although in the future we may have our vendors bill you directly for the services they provide. These services do not include paid, placed media advertising, which is an additional expense paid to the vendor directly by you.

If you do not already have it, you will need to acquire computer hardware and software that meets our minimum specifications, as stated in the Operations Manual. As of the date of this Disclosure Document, our minimum specifications are as follows: a PC computer running Windows 11 or higher as its operating system (although the operating system must be supported by Microsoft and you will need to upgrade the operating system to a higher system once your system is no longer supported), and a high-speed Internet connection, with the Microsoft Office suite standard and QuickBooks online version software. We have the right to periodically change our standards. The computer will be used to operate the required software programs and store the data used in that software.

You will need a smart phone or another device approved by us that meets our minimum specifications, for receiving and sending e-mails remotely. As of the date of this Disclosure Document, our minimum specifications are that the mobile device be (i) able to remain on throughout the working day, (ii) portable so that it may be taken to job sites and throughout the Territory, and (iii) capable of accessing, sending and receiving e-mail. We may change those specifications on notice to you.

The estimated cost of purchasing the required computer hardware, software and office equipment including the mobile device for receiving and sending e-mails, but not including the Software and Internet Fee described above, ranges from \$2,250 to 2,500 to \$5,000 5,500.

You must keep your computer hardware and software up to date and purchase platform and/or operating software upgrades as we direct and which meet our specifications as they evolve over time. In some cases, these systems and software may only be available through us, our affiliates and/or designated suppliers. We reserve the right to charge service and development fees to cover the costs of meeting the evolving technology needs of the Ace Handyman Services System. We also may require you to purchase new or additional computer hardware or software from us or from third parties as may be necessary to support the technology utilized now or in the future by the Ace Handyman Services System. The annual cost of any optional or required maintenance, updating, upgrading or support for the computer hardware and software

will vary. We, our affiliates and our designated suppliers have no obligation under the Franchise Agreement to provide ongoing maintenance, repairs, upgrades or updates to your computer hardware and software. As of the date of this Disclosure Document, we do not have independent access to your computer system; however, no contractual restrictions exist concerning our ability to require you to give us independent access in the future.

You must also have a laser printer. All equipment and software must comply with our standards and specifications.

You must accept credit and debit cards from customers of your AHS Business. You may not charge your customers any additional fees or service charges if they elect to pay by credit or debit card. The Payment Card Industry ("PCI") requires all companies that process, store, or transmit credit or debit card information to protect the cardholders' information by complying with the PCI Data Security Standard ("PCI DSS"). Therefore, you must be PCI compliant by following and adhering to the then-current PCI DSS, currently found at www.pcisecuritystandards.org, or any similar or subsequent standard for the protection of cardholder data throughout the term of your Franchise Agreement. PCI mandates the PCI DSS compliance. You will share certain personal information of your customers with AHF and in connection therewith, will enter into the AHF Data Collection Agreement attached to this Disclosure Document as <u>Attachment M</u>.

Site Selection

You must operate from a small office outside of your home. You are responsible for selecting and acquiring the premises for your Business Location, and we do not locate or negotiate the lease or purchase of a site for you. You must obtain our prior written consent for any Business Location. We follow Google Best Practices for office locations for service-based businesses. This currently includes an office in a centralized location within your Territory, although Google changes its requirements from time to time, and we adjust our standards to conform to their requirements.

A Business Location office is typically between 150 to 600 square feet. It must be centrally located in your Territory. We reserve the right to establish additional standards for your Business Location, and once those standards are established, you must comply with our standards no later than 30 days after written notice to you. We do not generally own the premises used by our franchisees. You may not open your AHS Business until such time as you have obtained an approved location.

The typical time to find an acceptable site for your Business Location and enter into a lease is 30 days, however, you must find a site within 90 days of signing the Franchise Agreement unless we agree to grant you additional time, which we will consider granting if you provide evidence that you have made a diligent effort in obtaining a site. Additionally, if we disapprove of any proposed site, we will grant you an additional reasonable time to obtain approval for a different site. If you fail to locate and obtain our approval of a site within the applicable time period, we may terminate the Franchise Agreement. Our acceptance of a location does not infer or guarantee the success or profitability of an approved location.

Time for Commencement of Operations

The time that will elapse between the signing of a Franchise Agreement and the opening of your franchise business is typically 120 days. The actual time that elapses will vary depending upon individual circumstances and local conditions, such as your ability to obtain (1) any permits or licenses required by your state and local governments; (2) a lease for your office and financing or building permits for leasehold improvements (if applicable); and (3) equipment, fixtures and signs; and when you complete your preopening study and initial training. In the event that you cannot keep pace with the timeline of tasks required for pre-opening study, your opening may be delayed. Consideration will be given to the other franchise

owners that are in the process of opening their AHS Businesses in determining the rescheduling of your opening. The Franchise Agreement requires that you begin operations of your AHS Business within 60 days after you, or a fully trained and qualified operations manager designated by you in writing to manage the AHS Business on your behalf (the "Franchise Manager"), completes the initial training program. If extenuating circumstances exist, such as a delay in your ability to obtain a lease for your business or you have incurred a serious illness, we may (in our sole discretion) agree to extend that time period for an additional amount of time we determine. Otherwise, we may terminate your Franchise Agreement for your failure to begin operations of your AHS Business within that time period.

Training

Prior to attending our initial business training program, you will undergo approximately 10 to 12 weeks of onboarding work and study. The onboarding work and study may be adjusted depending on circumstances.

Once you have completed your onboarding work and study, we will provide an initial business training program to be conducted at our office in the Denver, Colorado metropolitan area, or another location designated by us, or via video conference. The initial business training program lasts approximately three business days. The actual length of your training program and your training schedule may be adjusted by us based on your prior experience or training. We conduct our initial business training program on an asneeded basis, but typically not less than seven times per year.

Training is supervised by our President, Chris Bue. Mr. Bue has over 10-11 years of experience with us, and more than 24-25 years of experience in the field across multiple industries. Each of our instructors has demonstrated to us satisfactory knowledge of the topics they instruct and are overseen and reviewed by Mr. Bue.

The Operations Manual is the main source of instructional material. The initial business training program will be conducted within 90 days after execution of the Franchise Agreement, unless otherwise agreed by you and us, and prior to the commencement of operation of your AHS Business. You or your Franchise Manager must obtain the LRRP Certification for your AHS Business prior to participating in our initial business training program, regardless of your intentions to complete projects where you may encounter lead paint.

Yourself, or, if applicable, your Franchise Manager, and your Office Manager, plus up to two other individuals designated by you (for a total of four individuals), are eligible to participate in our initial business training program without a charge. You or your Franchise Manager, and your Office Manager, must successfully complete our initial business training program prior to the commencement of operations of your AHS Business.

You will be responsible for the transportation and living expenses of each person who you designate to attend the initial business training program.

TRAINING PROGRAM

Onboarding Study Program

The subjects covered in the pre-opening onboarding study program and the general times devoted to each subject are described below. However, the program may be modified, at our discretion, to meet the needs of any individual trainee, and as internal processes, best practices, and software may update and change.

ONBOARDING STUDY PROGRAM AGENDA

Subject Type	Hours of Guided Training	Hours of Self-Guided Training	Hours of Research, Preplanning & Setup
Business Establishment, Licensing, & Professional Advisors	0	0	35-40
Establish Key Vendors & Office Space	0	0	25-30
Finance, Banking, & Credit	1	0	25-30
Employees & HR	7-10	4-6	3-4
Accounting & Insurance setup	1	1-3	10-12
Marketing & Networking	4-6	0	25-30
Software & Technology Tools	2	0	2
Market Research	1	0	3-4
TOTALS	16-21	5-9	128-152

Initial Business Training

The subjects covered in the initial business training program and the general times devoted to each subject are described below. However, the program may be modified, at our discretion, to meet the needs of any individual trainee, and as internal processes, best practices, and software may update and change.

INITIAL BUSINESS TRAINING PROGRAM AGENDA

Subject	Hours of Classroom Training	Hours of On-The-Job Training	Location*
Introduction and Orientation	1-2	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
History, Mission, and Brand	1-2	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Franchisor/Franchisee Relations	1-2	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Start-up and Daily Operations	5-6	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Brand, Culture & Service Path (Customer Service)	3-4	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Employee/Craftsman Hiring & Retention	1-2	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Technology Tools and Scheduling Software	1-2	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us

Subject	Hours of Classroom Training	Hours of On-The-Job Training	Location*
Accounting & Software	2-3	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Advertising and Marketing	4	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Corporate Communications	1	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
Week-in-Review, Testing	2	0	Denver, Colorado Metropolitan Area or Another Location Designated by Us
TOTALS	22-30	0	•

^{*} The initial training program is typically conducted at or near our Support Center headquarters in the Denver, Colorado metropolitan area. We may conduct any of the initial business training virtually via video conference.

If, whether as a result of observations, test results or otherwise during or upon completion of onboarding training (including during operation of your AHS Business) we determine that it is appropriate or necessary, we can require that you (or a managing partner or shareholder consented to by us) or your Franchise Manager re-attend and successfully complete the initial business training program, at your sole cost and expense.

On-Site Training

Within the first 90 days of operations of your AHS Business, we will conduct on-site training at your Business Location. At our discretion, we may divide the on-site training program into two separate training sessions to be held at different times at your Business Location. We require you or your Franchise Manager, as applicable, and your Office Manager, to attend and successfully complete this on-site training program. Other employees of yours may attend this on-site training as well. The on-site training program lasts approximately two to three business days. There will be no fee for this on-site training program.

The subjects covered in the on-site training program and the general time devoted to each subject is described below. However, the program may be modified, in our discretion, to meet the needs of any individual trainee, and as internal processes, best practices, and software may update and change.

ON-SITE TRAINING PROGRAM AGENDA

Subject	Hours of On-The-Job Training	Location
Office TEAM Roles & Responsibilities Division	0.5-1	Your Business Location
Office Organization, Setup, & Flow	0.5-1	Your Business Location

Subject	Hours of On-The-Job Training	Location
Technology Tools: Scheduling, Estimating, & Accounting Software	2-3	Your Business Location
Website, Social Media, & Reviews	1-2	Your Business Location
Role Play & Practice of Critical Tasks	3-4	Your Business Location
Support Center Communications	0.5-1	Your Business Location
Daily Operations Best Practices	9-10	Your Business Location
Local Ace Integration Training	0.5-1	Your Business Location
Onboarding Wrap Up	1	
Totals	18-24	

Additional Training

If we determine that you, your Franchise Manager, your Office Manager, or other representatives require training in addition to the initial business training program and on-site training, or if you request additional training, then we will provide notice to you of the additional training and conduct the additional training program at a location we designate or virtually via video conference. You will be responsible for paying the travel, lodging and other costs for you, your Franchise Manager, your Office Manager, or your other representatives, and you will be required to pay us our standard fees for conducting additional training in accordance with our then current fee schedule.

The marketplace is ever-changing and continuous education is crucial. Therefore, we may present seminars, national or regional conventions, continuing development programs, other additional or refresher training programs, or other meetings, which may be conducted in-person or virtually via video conference. Some of these are voluntary and your attendance is not required. However, you (or a managing partner, member or shareholder consented to by us) or your Franchise Manager, and your Office Manager, as applicable, must attend any mandatory seminars, programs, or meetings we conduct. The mandatory additional training will not last more than three days at a time. You or your Franchise Manager, and your Office Manager, must attend one regional meeting for your region and our annual national convention each year if we hold these meetings. You or your Franchise Manager, and your Office Manager, may also be required to attend up to one additional mandatory convention or meeting every 24 months, in addition to any mandatory regional meetings for your region and the annual national convention. We will give you at least 30 days prior written notice of any seminar, convention, program or meeting that is deemed mandatory, unless your attendance is waived by us. (See Section 6.4 of Franchise Agreement.)

We do not charge a fee or costs for attendance at any mandatory seminar, national or regional convention or other mandatory meetings. However, unless we agree otherwise, if you, your Franchise Manager or your Office Manager does not attend the applicable seminar, convention or other mandatory meeting, you will be required to pay us \$1,500 per missed event, and if you attend the applicable seminar, convention or other mandatory meeting but do not stay at the designated hotel or do not stay for the entire meeting or convention, you will pay us \$750 per applicable event. You will be responsible for the cost of attending, including travel, meals and lodging expenses, any seminars, national or regional conventions, continuing development programs, other additional or refresher training programs, or other meetings. If you, your

Franchise Manager, or your Office Manager fails to attend a program at which attendance is deemed mandatory, we may, without waiving any other rights, require you, your Franchise Manager, or your Office Manager to attend and complete a make-up or alternative program at a location determined by us and you will be responsible the costs of the program. (See Section 6.4 of Franchise Agreement.)

All training programs are provided to protect our brand and the Marks and not to control the day-to-day operation of your AHS Business.

Operations Manual

Attached to this Disclosure Document as <u>Attachment G</u> is the table of contents of our Operations Manual. Our Operations Manual is maintained electronically and made available to you online. If it were printed, it would be approximately 70 total pages.

ITEM 12 TERRITORY

Territory and Business Location

You will receive a designated protected territory (the "Territory") for the AHS Business that will be delineated by zip codes as determined by the business map software that we utilize. Your Territory will include the businesses in the Territory. We currently use Fract operated by Fract, Inc. located in San Francisco, California, to determine the number of households in a particular Territory. As long as you are not in default of your Franchise Agreement, we will not operate or permit another AHS Business to operate within your Territory. Your Territory will be determined by you and us before you sign your Franchise Agreement and will be set forth in the Addendum to your Franchise Agreement. The size and configuration of your Territory is based on the number of households.

You will operate from one location that you select as your Business Location. You are solely responsible for selecting and acquiring the premises for your Business Location. You must obtain our prior written approval of any Business Location, including any relocation of your Business Location. We will generally approve your Business Location and any relocation of it, if the Business Location is within your Territory and meets our then-current standards for your Business Location. If we do not approve your Business Location, you must select another location for our approval. Unless extended by us, you must select and acquire a Business Location within 90 days following the effective date of the Franchise Agreement. You may not commence operations of your AHS Business until such time as you have obtained an approved Business Location. If you are operating multiple AHS Businesses in adjacent territories, we may agree to allow you to operate from a single Business Location centrally located in the combined territories pursuant to the terms of the Multi-Territory Addendum. We will not allow the use of a single Business Location under the Multi-Territory Addendum if having a single Business Location will prevent your AHS Businesses from appearing with reasonable priority in a Google search throughout the combined territories, or if we otherwise determine that the use of a single Business Location decreases the business visibility for purposes of online advertising.

Minimum Annual Gross Revenues

You will be required to achieve minimum gross revenues on an annual basis (the "Minimum Annual Gross Revenues") as set forth below:

Operating Year of Franchise Agreement	Minimum Annual Gross Revenues
Year 1	\$200,000
Year 2	\$400,000
Year 3 and thereafter	\$500,000

If you acquire a Mini AHS Business, you will be required to achieve Minimum Annual Gross Revenues as set forth below:

Operating Year of Franchise Agreement	Minimum Annual Gross Revenues
Year 1	\$100,000
Year 2	\$200,000
Year 3 and thereafter	\$250,000

The Minimum Annual Gross Revenues for your AHS Business will not change if there is an increase or decrease in the number of households during the term of your Franchise Agreement. If you have entered into multiple Franchise Agreements with us, you must meet the Minimum Annual Gross Revenues for each Franchise Agreement unless we agree otherwise. However, if you execute a Multi-Territory Addendum for the operation of multiple AHS Businesses in adjacent territories, that Multi-Territory Addendum will specify a combined Minimum Annual Gross Revenues amount that you must meet from all of those AHS Businesses and territories combined. The amount of the Minimum Annual Gross Revenues designated under a Multi-Territory Addendum will be negotiated between the franchisee and us based on the characteristics of the territories, and may vary from the standard amounts described above that apply for each AHS Business.

Following the end of each year after commencement of operations by you (an "operating year"), we will compute your total Gross Revenues for the just completed operating year. If you fail to achieve the minimum gross revenue in any operating year, we reserve the right to charge you for the royalty fee and/or National Brand Fund fee based on the Minimum Annual Gross Revenues instead of your actual Gross Revenues. We will invoice you for the difference between the amount of Royalty Fee and/or National Brand Fee due based on the Minimum Annual Gross Revenues, less the actual Royalty Fee and/or National Brand Fee paid to us during the applicable operating year.

If your actual Gross Revenues are less than the applicable Minimum Annual Gross Revenue for two consecutive operating years, or in any three operating years during the term of your Franchise Agreement, then we have the right to (i) terminate your Franchise Agreement, (ii) award additional franchises within the Territory or otherwise permit another franchisee to operate within the Territory, (iii) reduce the geographic size of your Territory, or (iv) any combination of the above options.

We have the right, in our sole discretion and on a case-by-case basis, to waive the obligation of you or any other franchisee to meet the Minimum Annual Gross Revenues requirement or to pay any fees or make expenditures calculated based on the applicable Minimum Annual Gross Revenues.

Commercial Quality Service Program

If you perform work or provide services to commercial customers in your Territory, you must participate in our Commercial Quality Service Program. Under this program, you agree to a review by us, our representative or the commercial client of the level of services you provided to a commercial customer. If we determine that you have not provided a satisfactory level and quality of service, then we may, in our sole discretion, appoint another franchisee of ours to perform any future services for that commercial customer, regardless of whether the commercial customer is located or operates in your Territory.

Referral Customer Contact

If we or another franchisee of ours operating outside of your Territory refer a prospective customer to you, you are required to contact that prospective customer within 24 hours of receiving the prospective customer's contact information from us or the other franchisee. If you fail to contact the prospective customer within that time period, we, in our sole discretion, shall have the right to assign another franchisee the right to contact the prospective customer and perform the services initially requested and all services later requested by the same prospective customer in the future, regardless of the fact that the prospective customer is located within your Territory.

General

Subject to your requirement to meet Minimum Annual Gross Revenues, comply with our Commercial Quality Service Program and respond to referral customers, all as discussed above, we will not operate, or permit another franchisee to operate, within your Territory. However, because of these requirements, your Territory is not deemed to be an exclusive Territory.

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

You will maintain rights to your Territory (or Territories), even though the number of households may increase.

You do not receive the automatic right to acquire additional franchises adjacent to your Territory. You may be granted, in our sole discretion, permission to sell or service customers in an unsold Territory adjacent to your Territory (an "Adjacent Territory"). However, when an Adjacent Territory is granted to another franchisee, you must, upon receipt of written notice from us, cease all sales and service efforts within the Adjacent Territory, and deliver to us, within 10 days of notice, all customer and prospect information related to customers and prospects located in the Adjacent Territory.

The Franchise Agreement gives you no options, rights of first refusal, or similar options to acquire any additional franchises from us, although we may in our discretion offer you the right to acquire additional franchises in the future.

Except as expressly stated above, you may not operate your AHS Business outside of your Territory without our prior written consent. If you market, provide or sell any of the products or services of the AHS Business within the territory of any AHS Business owned by us, any affiliate of ours or any other franchisee of ours, without our prior written consent, we will have the right to terminate your Franchise Agreement effective

immediately upon notice. You are prohibited from using other channels of distribution, such as the Internet, catalog sales, telemarketing, or other direct marketing, to make sales outside of your Territory, without our prior written consent.

We have and retain the right under the Franchise Agreement to develop and establish other franchise systems for similar, the same, or different products or services utilizing proprietary marks not now or in the future designated as part of the Ace Handyman Services System licensed under the Franchise Agreement, and to grant licenses for these other franchise or license systems without providing you any right in those systems. We retain the right to open, operate, sell, manage and/or franchise AHS Businesses outside your Territory. We are not restricted from the pursuit of any other business concept, or the distribution of proprietary equipment and supplies, if any, to wholesalers or other distribution outlets (other than AHS Businesses) or by Internet, catalog, mail order, telemarketing, other direct marketing sales or otherwise, whether inside or outside the Territory. We may use our Marks and Ace Handyman Services System in connection with the provision of other services and products or in alternative channels of distribution (including by Internet, catalog, mail order, telemarketing or other direct marketing sales), inside or outside of the Territory. We are not required to compensate you for soliciting or accepting orders inside your Territory. Our ultimate parent, Ace Hardware, or any other company affiliated with Ace Hardware other than us, may use trademarks or service marks similar to our Marks and methods of doing business that are similar to the System, in connection with providing or franchising others to provide services and products that may be similar to or include some of the services you may offer in your AHS Business, including, by way of example but not by limitation, stand alone "Ace Hardware Painting" or "Ace Hardware Plumbing" businesses, inside or outside of the Territory. (See Section 4.4 of FA.)

Our affiliate APF began to offer franchises for Ace Hardware Painting Services businesses in 2022 under the service mark ACE HARDWARE PAINTING SERVICESSM, which businesses provide painting services for residential and commercial properties that may be deemed similar to the services offered by an AHS Business. Our affiliate AHHS currently operates Ace Hardware Painting Services businesses in Colorado, Kansas Missouri and Virginia. Since an AHS Business and an Ace Hardware Painting Services business have separate focuses, we do not anticipate conflicts between AHS Businesses and Ace Hardware Painting Services businesses regarding territory, customers, or franchisor support, but in the event of any conflicts we and APF shall resolve them as we deem appropriate in our discretion. These Ace Hardware Painting Services businesses may solicit and accept orders in your Territory. APF shares our principal office address. We also share training facilities with APF.

Our affiliate AHHS also operates an Ace Hardware Plumbing Services business under the name LEGACY PLUMBING® in the Dallas/Fort Worth, Texas metropolitan area. AHHS operates an Ace Hardware Plumbing Services business under the service mark ACE HARDWARE PLUMBING SERVICES® in Oklahoma City, Oklahoma. AHHS may expand into other areas in the future. Our affiliate APLF intends to offer franchises in the future for Ace Hardware Plumbing Services businesses under the service mark ACE HARDWARE PLUMBING SERVICES®, which businesses will provide plumbing services for residential and commercial properties that may be deemed similar to the services offered by an AHS business. Since an AHS Business and an Ace Hardware Plumbing Services business have separate focuses, we do not anticipate conflicts between AHS Businesses and Ace Hardware Plumbing Services businesses regarding territory, customers, or franchisor support, but in the event of any conflicts we and APLF shall resolve them as we deem appropriate in our discretion. These Ace Hardware Plumbing Services businesses may solicit and accept orders in your Territory.

Our parent, Ace Services owns, through its subsidiaries, affiliate AHHS also operates a number of HVAC companies businesses that offer in-home and commercial HVAC services. Some of these businesses, may also provide plumbing and electrical services. While it is not expected that the services offered by any of the HVAC business will compete with the types of services offered through an AHS Business, it is possible

that some overlap of services may exist. In the event of any conflicts we and Ace Services AHHS shall resolve them as we deem appropriate in our discretion. Ace Services AHHS may solicit and accept orders in your Territory.

We can develop, acquire or be acquired by, or engage in any other transaction with other businesses, companies and/or units (competitive or not), that are or will be located anywhere, including arrangements where other units are (or are not) converted to the Ace Handyman Services System or other format, or in which company-owned, franchised or other businesses (including your AHS Business) are (or are not) converted to another format (whether competitive or not), or both, and is maintained as the same concept, as a new concept, or as a separate concept in your protected Territory. You must fully cooperate with any of these conversions, at your sole expense.

ITEM 13 TRADEMARKS

We grant you the right to operate a business under the Marks. Our Marks are owned by our ultimate parent, Ace Hardware. As we grow the equity behind the Ace Handyman Services brand, a key factor is strict compliance to its use. As such, you must explicitly adhere to its proper use.

Ace Hardware has filed for registration on the Principal Register of the United States Patent and Trademark Office the following Marks:

Mark	Registration (Serial) Number	Registration (Application) Date
ACE Handyman Services	6,071,158	June 2, 2020
HANDYMAN SERVICES	6,241,209	January 5, 2021
ACE HANDYMAN SERVICES Bringing Helpful To Your Home	6,241,296	January 5, 2021
HANDYMAN SERVICES An ACE Hardware Company	<u>7,711,957</u>	March 4, 2025

Ace Hardware has granted us, in a License Agreement dated September 18, 2019, a nonexclusive, royalty-free license to use and to permit our franchisees to use its Marks in the United States of America. The license is for 20 years commencing September 18, 2019, but it will automatically renew for successive 20-year periods if we do not materially breach the License Agreement by engaging in any activity that damages those Marks or the goodwill of those Marks. If the license is terminated, Ace Hardware has agreed it will continue to permit any franchisees of ours that have effective Franchise Agreements to continue to use the Marks as permitted by their Franchise Agreements; provided we agree to assign the effective Franchise Agreements to Ace Hardware or its successor or assigns.

In addition to those Marks owned by Ace Hardware listed above, we also use a number of other unregistered trademarks and service marks, in which we claim common law trademark rights. The following statements apply solely to any unregistered trademarks and service marks: We do not have a federal registration for these principal trademarks. As such, these trademarks do not have as many legal benefits and rights as federally registered trademarks. If our right to use these trademarks is challenged, you may have to change to an alternative trademark, which may increase your expenses.

The Marks owned by Ace Hardware and our other trademarks and service marks are collectively referred to as the "Marks" or individually as a "Mark."

You must follow our rules when you use any of our Marks. You cannot use our Marks as part of a business entity name or with modifying words, designs or symbols except as we license to you. You may not use our Marks or any unregistered mark or any variation of them, with the sale of an unauthorized product or service or in a manner not authorized in writing by us.

We reserve the rights to (i) any Internet sites utilizing a domain name incorporating one or more of the words "Ace Handyman," "Ace Handyman Services," "Ace Hardware Home Services," "Ace Home Services," "AHS," "Handyman Matters," or any other derivative using "Ace" or "Handyman" and (ii) advertise the Ace Handyman Services System on the Internet. You cannot establish an Internet site without our prior written consent, including any that uses any domain name containing one or more of the words "Ace," in conjunction with "Handyman," or "Matters," or any variation of these. We may withhold our consent in our sole discretion. We are the owner of all right, title and interest in and to any domain names designated in the Operations Manual.

There are no currently effective material determinations of the United States Patent and Trademark Office, trademark trial and appeal board, the trademark administrator of any state or court pending infringement, opposition or cancellation, or pending material litigation involving our principal Marks.

No currently effective agreements limit our right to use or license the use of the Marks, except that we may not violate the terms of the existing License Agreements between Ace Hardware and us.

You must notify us immediately when you learn about an infringement of or challenge to your use of the Marks. We will have the right to take any action we deem appropriate and have the exclusive right to control any litigation, arbitration or administrative proceeding arising out of any infringement of, or challenge or claim to, any of our Marks or in any way related to our Marks. Any damages or benefits arising out of any infringement, challenge or claim shall accrue exclusively to us. You must notify us immediately when you learn about the infringement or challenge, and you must use our Marks only in accordance with the Franchise Agreement.

You must modify or discontinue the use of a Mark if we modify or discontinue it. If this happens, we will reimburse you for your reasonable out-of-pocket costs (if any) of changing your main identifying signage and other significant signage, but we will not be liable for any other costs, expenses or damages you incur

as a result of our decision to modify or discontinue use of a Mark. You must not directly or indirectly contest our right to our Marks.

There may be one or more businesses similar to an AHS Business, which are operating in or near the area(s) where you may do business or otherwise, which have a name and/or marks similar to ours and with superior rights to these names and/or Marks. We strongly urge you to research this possibility, using telephone directories, internet searches, local filings and other means, before you sign the Franchise Agreement or any other documents related to your business, or expending or paying any sums or making any commitments concerning your business. If you fail to do so, you may be at risk.

ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

No patents or pending patent applications are material to the franchise.

No copyrights, other than common law copyrights in the Operations Manual, are material to the franchise. The Operations Manual contains our confidential and proprietary information and trade secrets regarding the operations of an AHS Business. We may modify the Operations Manual in our sole discretion, but the modifications will not alter your status and rights under your Franchise Agreement. We have included a copy of the Table of Contents for our Operations Manual as <u>Attachment G</u> to this Disclosure Document.

You can use the proprietary information in the Operations Manual only as permitted by your Franchise Agreement. You must maintain the confidentiality of our Operations Manual and adopt reasonable procedures to prevent unauthorized disclosure of it. You are required to promptly tell us if you learn about unauthorized use of this proprietary information. We are not obligated to take any action, but we will respond to this information as we think appropriate. We will indemnify you against losses claimed by a third party concerning your use of this information. Our right to use or license this information is not materially limited by any agreement or known infringing use. There are no determinations of any administrative office or any court regarding these copyrighted materials or the proprietary information.

We own all records concerning the customers, suppliers, and independent contractors and other service professionals of, and/or related in any way to, your AHS Business, including all databases (whether in print, electronic or other form), with customer and potential customer names, addresses, phone numbers and email addresses, and customer purchase records. You own all records concerning your employees. We may use, permit any third party to use, or transfer our records in any way we wish, both before and after any termination, expiration, repurchase, transfer or otherwise. (See Section 15.9 of FA.) We may contact any and/or all of your customers, suppliers, and independent contractors and other service professionals for quality control, market research and any other purposes we deem appropriate, in our sole and absolute discretion. We have the right to inspect and audit your records, the records of certain individuals and companies affiliated with your AHS Business, and the records of the immediate family members, generally meaning spouses, parents, siblings, and adult children, of you and these affiliated individuals.

We may require you to cause each of your officers, directors, partners or shareholders, and any member of your or their immediate families, as applicable, to execute our standard Nondisclosure and Noncompetition Agreement, a copy of which is attached to this Disclosure Document as <u>Attachment B</u>. You must provide us a copy of each Nondisclosure and Noncompetition Agreement at the time it is signed and thereafter upon our request. You also will not disclose the contents of any materials we deem confidential to anyone other than those persons who have an obligation of confidentiality to you.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

If you are an individual, either you individually or an individual who you designate and is approved by us to be an operating manager for your franchise, who we refer to as the Franchise Manager, must directly supervise the daily operation of your franchise at the Business Location. If we believe you lack sufficient business experience, you will be required to designate and hire a Franchise Manager whom we approve. If you are a corporation or other business entity, a Franchise Manager must be approved by us and must perform the direct, on-site supervision of your franchise. A Franchise Manager is not required to have any ownership interest in a business entity franchisee.

The Franchise Manager will need to attend and successfully complete the initial business training program and abide by the obligations in the Franchise Agreement and the Operations Manual.

You are responsible for recruiting, appointing, hiring, firing, and supervising your employees, independent agents, Office Managers, Franchise Managers, and other representatives. You will have sole authority and control over the day-to-day operations of the AHS Business and its employees and representatives. You are solely responsible for implementing training and other programs for your employees and representatives related to the legal, safe, and proper performance of their work, regardless of the fact that we may provide advice, suggestions, and certain training programs.

You have sole authority and control over the decision of whether or not to use independent contractors, although we do not recommend independent contractors exceed 40 percent of your work force. You will be solely responsible and liable for your decision to use independent contractors, any liability arising from the acts or omissions of your independent contractors, and for complying with all laws, rules and regulations related to the engagement and use of independent contractors in your AHS Business.

You must conduct a background check of all employees, contractors, and subcontractors hired or engaged by you and utilized in your AHS Business and System including, but not limited to, authorized representatives. In conducting these background checks, you must abide by all applicable laws, rules and regulations, including, but not limited to, the Fair Credit Reporting Act (and the state equivalents, if any) and/or any employment laws, rules, regulations or ordinances. It is your responsibility to make sure that no employee, contractor, or subcontractor enters a customer's home or business if such person has not successfully completed the background check process, as determined in your sole discretion.

You will have sole authority and control over the day-to-day operations of the AHS Business and its employees, contractors and representatives. You are solely responsible for implementing training and other programs for your employees, contractors and representatives related to the legal, safe, and proper performance of their work, regardless of the fact that we may provide advice, suggestions, and certain training programs. The advice, suggestions, and training we provide are to protect our brand and the Marks and not to control the day-to-day operation of your AHS Business. We are not a joint employer with respect to the operation of your AHS Business, including, without limitation, your decision whether or not to use independent contractors, as determined in your sole discretion.

You must hire an experienced Office Manager to assist in the answering of your phones, scheduling of jobs with customers, assisting with employment applications, paperwork check-in, and other similar customer service and office duties. The Office Manager must complete our initial business training program and onsite training program.

You must ensure that at least one representative of your AHS Business maintains LRRP Certification at all times during the term of your Franchise Agreement. We recommend that you or the Franchise Manager, as applicable, obtain the LRRP Certification. If the representative with the LRRP Certification ceases to be associated with your AHS Business, or if the representative's LRRP Certification is terminated for any reason, another representative of your AHS Business must immediately obtain the LRRP Certification.

You or the Franchise Manager are required to devote substantial full time and best efforts on a daily basis, in person, to the supervision and conduct of the AHS Business, which must not be less than 40 hours per week. However, if you are operating multiple AHS Businesses in adjacent territories, we may agree to allow you to have one Franchise Manager dedicating full time and best efforts to all of the AHS Businesses as a whole pursuant to the terms of the Multi-Territory Addendum.

If you wish to operate another business, including any e-commerce business which operates over the Internet, our prior approval, which we may grant or deny in our sole discretion, will be required.

Unless waived by us, each individual who owns, directly or indirectly, an interest in the franchisee entity will sign the Guaranty and Assumption of Franchisee's Obligations assuming and agreeing to discharge all of your obligations and comply with all restrictions under the Franchise Agreement. (See Exhibit III to FA.)

ITEM 16 RESTRICTIONS ON WHAT YOU MAY SELL

In order to maintain uniformity throughout the Ace Handyman Services system, franchisees must operate their businesses in conformity with our standards and specifications.

You may offer and sell only those products and services that we have approved or that meet the standards and specifications set by us, which may be set forth in the Operations Manual or in other writings.

You are required to offer all handyman, repair, and maintenance services that we designate as required for all franchisees within your market area. We may change the services required in your area, with prior notice to you. (See Section 11.3 of FA.) You must also offer, honor, and otherwise comply with the terms of any subscription package programs for residential or commercial customers that we may establish. It is your responsibility to promptly obtain all licenses and permits necessary to operate the AHS Business in your jurisdiction. You cannot engage in any activities not covered by your liability insurance or workers compensation insurance.

You cannot engage in any trade, practice or other activity that is harmful to our goodwill or reflects unfavorably on our reputation, that constitutes deceptive or unfair competition, or that is in violation of any applicable law.

Certain inventory, supplies, and equipment used in your AHS Business must be purchased from us or a vendor approved by us, to the extent we have approved vendors for such supplies and equipment, or that meet the standards and specifications set by us. We have a purchasing arrangement with Ace Hardware whereby you must purchase all materials, supplies and goods needed to run your AHS Business from an Ace Supplier, but only when the items are in stock or readily available, and competitively priced. Many other items used in your AHS Business can be purchased from local hardware stores, big box outlets, and other sources.

We have established a Franchisee Advisory Council, which serves as an advisory council to assist us in collecting and prioritizing project requests and business process improvement suggestions from our

franchisees and monitoring the status of existing project requests and improvement suggestions. We reserve the right to terminate the Franchisee Advisory Council, or to modify its purposes, in our sole discretion.

We may impose restrictions limiting your access to customers outside of your Territory and to commercial customers within your Territory, both as discussed in ITEM 12 above, but apart from these we do not impose any restrictions on your access to customers.

REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK

ITEM 17 RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the Franchise Agreement and related agreements. You should read these provisions in the agreements attached to this Disclosure Document.

	Provision	Section in Franchise or Other Agreement	Summary
a.	Length of the franchise term	Section 3.1	10 years.
b.	Renewal or extension of the term	Sections 3.3 and 3.4	If you are in good standing, you can acquire additional successor franchise terms of 10 years each.
c.	Requirements for you to renew or extend	Sections 3.3 and 3.4	No less than 120 days' notice, nor more than one year notice; sign the current form of the Franchise Agreement (which may contain terms and conditions materially different from your original Franchise Agreement); be in compliance with current Franchise Agreement; be current in payments; satisfy current standards for new franchisees; modify your AHS Business and operations to conform with current Operations Manual; pay successor franchise fee; execute a Successor Franchise Rider (in its then current form, which will include, except where prohibited by law, a release. The current form of Successor Franchise Rider is attached to this Disclosure Document as Attachment E).
d.	Termination by you	Section 18.5	You can terminate the Franchise Agreement only if we materially breach the Franchise Agreement, if you provide us with written notice within 30 days of the breach and a reasonable opportunity to cure of not less than 90 days (subject to state law).
e.	Termination by us without cause	Section 18.3	We can terminate at our option if there is a force majeure event that prevents us from performing our obligations under the Franchise Agreement and continues for longer than six months
f.	Termination by us with cause	Sections 18.1 and 18.2	We can terminate based on your inability or unwillingness to operate your AHS Business, or upon certain violations of the Franchise Agreement or engaging in illegal or unethical practices.

	Provision	Section in Franchise or Other Agreement	Summary
g.	"Cause" defined – curable defaults	Section 18.2	You have 7 days to cure the filing of a legal action in violation of the dispute resolution terms in the Franchise Agreement; 10 days to cure misuse of Marks, non-payment of fees, non-submission of reports, failure to meet minimums; and 30 days to cure any other default not listed in Section 18.2 (subject to modification by state law). If we provide you with a notice of default, we and our affiliates may suspend services to you until each default is cured.
h.	"Cause" defined – non-curable defaults	Section 18.1	Non-curable defaults: unauthorized disclosures; repeated defaults even if cured; abandonment; unapproved transfers; sales activity in the territory of another AHS Business; sexual harassment or discrimination; violation of noncompete or other in-term restrictive covenant; breach of related agreement; a guaranty becomes unenforceable or inadequate; bankruptcy (which may not be enforceable under Federal Bankruptcy Law (11 U.S.C. § 101 et seq.)); others.
i.	Your obligations on termination/non-renewal	Section 18.6	Obligations include escrow of an amount equal to 1% of the greater of the Gross Revenues or your Minimum Annual Gross Revenues for the prior 12 months to cover warranty and customer dispute issues; complete deidentification; return of proprietary materials; payment of amounts due; payment of an amount equal to the Royalty Fees based on the Minimum Annual Gross Revenues due over the remaining term under the Franchise Agreement, if terminated due to your default; inspection of books; assign all pending and contract business to us; others. See also item "r" below.
j.	Assignment of contract by us	Section 17.1	No restriction on our right to assign.
k.	"Transfer" by you - defined	Section 17.2	Includes transfer of contract or assets or ownership change.
1.	Our approval of transfer by you	Sections 17.2 and 17.3	We have the right to approve all transfers.
m.	Conditions for our approval of transfer	Section 17.3	New franchisee qualified; transfer fee paid, if applicable; purchase agreement approved; training completed by transferee; current agreement signed by transferee; escrow of an amount equal to 1% of the greater of the Gross Revenues or your Minimum Annual Gross Revenues for the prior 12 months to cover warranty and customer dispute issues, and release signed by you (except where prohibited by law). Our current form of general release is attached to this Disclosure Document as <u>Attachment F</u> .
n.	Our right of first refusal to acquire your business	Section 17.5	We can match any offer for your business within 30 days.

	Provision	Section in Franchise or Other Agreement	Summary
0.	Our option to purchase your business	Not applicable	Not applicable.
p.	Your death or disability	Section 17.4	Your estate or legal representative must apply to us for the right to transfer to the next of kin within 120 days.
q.	Non-competition covenants during the term of the franchise	Sections 20.1 and 20.3	Prohibited from owning, operating or performing service for a competing business.
r.	Non-competition covenants after the franchise is terminated or expires	Sections 20.2 and 20.3	No competing business for 24 months within 25 miles from the closest point to the former Territory or the Territory of any other AHS Business (including after assignment).
s.	Modification of the agreement	Section 23.1	The Franchise Agreement may be modified by a writing signed by both parties or, at our option, upon approval of 75% of our franchisees affected by the modification. The Operations Manual is subject to change.
t.	Integration/merger clause	Section 23.2	Only the terms of the Franchise Agreement are binding (subject to state law). Any representations or promises outside of this Disclosure Document and the Franchise Agreement may not be enforceable. Nothing in the Franchise Agreement is intended to disclaim any representations made by us in this Disclosure Document.
u.	Dispute resolution by arbitration or mediation	Section 22	Except for certain claims, all disputes must be arbitrated in Denver, Colorado (subject to state law). If a claim can be brought in court, both you and we agree to waive our rights to a jury trial.
v.	Choice of forum	Section 22.5	Colorado (subject to state law).
W.	Choice of law	Section 22.5	Federal and Colorado law applies (subject to state law). The Colorado Consumer Protection Act does not apply.

ITEM 18 PUBLIC FIGURES

We do not currently use any public figure to promote the Ace Handyman Services System.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the Disclosure Document. Financial performance information that differs from that included in ITEM 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this ITEM 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

In this Item 19, we refer to the AHS Businesses as "Territories" and the franchise owners as "Operators." As of December 31, 2023, there were 367 Territories operated by a total of 205 different Operators. There are 200 Territories operated by 100 Operators shown in the tables below. Of the Territories not shown in the tables below, 84 commenced operations during calendar year 2023 and 15 other Territories were closed by their Operators for periods of time during calendar year 2023 and therefore were not opened for at least 12 months as of December 31, 2023. Another 63 Territories are owned by Operators who did not provide us with complete financial reports showing the full year of 2023 operations. In addition, 5 Territories that were owned by Operators were acquired by our affiliate, AHHS, during calendar year 2023. None of those Territories are shown in the charts below.

There were also 12 affiliated AHS Businesses in operation (including the 5 AHS Businesses acquired by our affiliate as noted above) that are not shown in this Item 19.

As of December 31, 2024, there were 369 franchised Territories and 18 affiliated Territories in operation, with the franchised outlets operated by a total of 214 different franchise owners. In addition, five Operators representing five Territories have purchased AHS Businesses that are expected to be opened in 2025.

A combined total of 35 Territories had not been open for at least 12 months as of December 31, 2024, and are not reported in the tables below. Out of the remaining Territories, 48 Territories are owned by Operators who did not provide us with complete financial reports during the full year of 2024, and we have therefore also excluded those Territories from these tables. There were 33 total Territories, operated by 16 Operators, that closed at some point during the year 2024. None of these closed Territories or Operators are included in this ITEM 19. (Of these closed outlets, three were open less than 12 months as of the date of closing.) The Territories included in the tables include two Mini AHS Businesses.

The results of the remaining 100 Operators, all of whom operated for the full calendar year 2023 The remaining 286 Territories which were open for the full year 2024, operated by 148 Operators, are shown in the tables below. These Territories are referred to as the "Reporting Territories" and these Operators are referred to as the "Reporting Operators." There were 47 Reporting Operators that owned and operated in a single Territory, and 53 Reporting Operators who owned multiple Territories. The 18 outlets operated by our affiliate are not included in this ITEM 19.

Of these 286 Reporting Territories, 72 were owned and operated as a single territory (each, a "Single Reporting Territory"), while the remaining 214 were operated in multiple Territory groups by a single Reporting Operator ("Multi-Territory Reporting Groups"), of which there were 76. The 72 Single Reporting Territories and the 76 Multi-Territory Reporting Groups are referred to herein collectively as the "Reporting Units," for a total of 148 Reporting Units in 2024.

Because Multi-Territory Reporting Operators Groups' results may vary significantly from the results those of Single-Territory Reporting Operators Territories, we include below separate tables for the results of the Single Reporting Operators Territories and Multi-Territory Operators Reporting Groups.

Table 1 shows the results of Single Territory Reporting Operators who have been operating between one and three years. Table 2 shows the results of the Single Territory Reporting Operators who have been operating for more than three years. Table 3 shows the results of the Multi Territory Reporting Operators who have been operating between one and three years. Table 4 shows the results of the Multi Territory Reporting Operators who have been operating for more than three years. Tables 1 through 4 below show the results of the 72 Single Reporting Territories, grouped into four quartiles based on the Total Revenue achieved. Each of the four tables shows the results of one quartile.

Tables 5 through 8 below show the results of the 76 Multi-Territory Reporting Groups, grouped into four quartiles based on the Total Revenue achieved. Each of the four tables shows the results of one quartile.

TABLES BEGIN ON NEXT PAGE

Table 1

Single Reporting Territories

That Have Operated between 1 and 3 Years

Top Quartile

Average, High, Low and Median and Low Performance Chart (43 Reporting Operators with 43 Territories)

Based on Total Revenue 18 Reporting Territories

Single Quartile 1

		LOW									
				AVERAGE# / % ABOVE A	VERAGE 18 Rep	oorting Territor	<u>ries</u>				
HIGH		MEDIAN	Revenue:		<u>Average</u>	%/# Above			<u>High</u>	Low	M
oor					\$818,638	\$332,779					
nue		\$364,115 <u>\$560,845</u>					\$ \$149,452 <u>766,905</u>	\$437,813	<u>\$52</u>	24,800	
erial,					\$188,546	\$61,170					
<u>rials</u> Tra	sh ,										
ment											
nue		\$103,894 \$145,483					\$ \$52,313 210,653	\$43,700	\$11	15,906	
rvice					\$61,985	\$31,335					
ge Reve	ıue	\$37,927 \$53,779					\$ \$12,450 48,533	<u>\$67,040</u>	\$5	58,197	
L REVEN	1UE	\$506,026\$760,107		21/49 39% <u>/ 7</u>	\$1,069,169	\$495,638	\$ \$214,214 1,026,090	<u>\$548,553</u>	<u>\$69</u>	98,903	
of Good	S				-						
aftsmen					\$466,475	\$135,257					
nnel											
nses		\$159,563 <u>\$239,445</u>					\$ \$42,845 <u>436,776</u>	<u>\$167,269</u>	<u>\$22</u>	22,192	
erials,	ı				\$142,506	\$72,176					
<u>rial</u> Tras	n ,										
ment											
<u>ises</u>		\$92,341 \$144,944					\$ \$23,646 194,403	<u>\$47,717</u>	\$12	21,743	

alties	\$28,702		\$65,350	\$28,713		\$2,313	
ls, C Ca r	'd		\$2,836	\$416			
Permits	•				. -		
ge Expe			D.CEE 1.00	#2 00 2 02	\$0	<u>\$35,843</u>	<u>\$0</u>
L COST			\$677,168	\$209,292	***********		
DS <mark>SOLI</mark>		18/42 16% <u>/ 3</u>	0202 004	0400 (00	\$ \$68,804 <u>631,178</u>	\$250,828	<u>\$343,936</u>
SS PROF	IT \$224,235 \$370,527	16/37 39% <u>/ 7</u>	\$392,001	\$182,688	\$ \$145,410 394,912	<u>\$297,725</u>	\$354,967
<u>ises</u>							
S Softwa	<u>\$6,516</u>		<u>\$5,988</u>	<u>\$7,398</u>		<u>\$5,988</u>	
tomobile	<u>\$19,500</u>		\$45,743	\$4,128		\$18,334	
neral &							
nistrativ	<u>\$115,135</u>		\$127,823	\$201,446		<u>\$107,474</u>	
<u>cal</u>							
eting	<u>\$47,638</u>		<u>\$26,893</u>	\$42,522		\$46,759	
<u>tional</u>							
d Fund	\$14,208		\$19,400	\$11,057		\$14,161	
<u>nt</u>	<u>\$14,331</u>		\$15,300	<u>\$16,414</u>		<u>\$15,511</u>	
<u>valties</u>	\$46,135		\$49,388	\$33,236		\$45,781	
ols, C Ca	<u>ırd</u>						
Permits	\$1,614		<u>\$3,918</u>	\$200		<u>\$511</u>	
<u>L</u>							
NSES	\$265,077	44% / 8	<u>\$294,453</u>	\$316,402		<u>\$254,519</u>	
ER							
RETION	<u>ARY</u>						
ME	<u>\$105,450</u>	61% / 11	\$100,459	<u>(\$18,677)</u>		<u>\$100,448</u>	
ENTAGE							
<u>DTAL</u>							
<u>NUE</u>	<u>14%</u>	<u>50% / 9</u>	<u>10%</u>	<u>-3%</u>		<u>14%</u>	

Table 2

Single Reporting Territories That Have Operated More Than 3 Years

Second Quartile

Average, High, <u>Low and Median and Low Performance Chart</u>
(4 Reporting Operators with 4 Territories)

Based on Total Revenue 18 Reporting Territories

	Single Quartile 2						
	AVERAGE#	-/ <mark>% ABOVE</mark> A	VERAGEHIGHMED	IANLOW 18 Reportin	g <u>Territories</u>		
Revenue:	<u>Average</u>	%/# Above	<u>High</u>	Low	Median		
	\$320,003 \$341,17			\$309,660\$325,97	\$192,612 <u>\$356,13</u>		
Labor Revenue	<u>4</u>		\$477,530 \$389,119	<u>6</u>	<u>0</u>		
<u>Material</u> , <u>Materials</u> Trash,							
Equipment Revenue	\$89,862 \$91,503		\$89,575 \$79,663	\$84,955 <u>\$53,592</u>	\$80,336 <u>\$83,028</u>		
Service Charge Revenue	\$37,671 <u>\$44,630</u>		\$26,212 <u>\$64,511</u>	\$29,944\$47,659	\$20,377 \$49,344		
	\$447,535 \$477,30			\$451,750 \$427,22	\$293,325 \$488,50		
TOTAL REVENUE	7	2/50 39% / 7	\$593,317\$533,293	7	<u>3</u>		
Cost of Goods Sold÷							
	\$178,573 <u>\$163,67</u>			\$176,919 <u>\$127,09</u>	\$131,191 <u>\$175,36</u>		
Craftsmen Personnel Expenses	<u>6</u>		\$250,143 \$288,690	<u>4</u>	<u>2</u>		
Materials, Material Trash,							
Equipment <u>Expenses</u>	\$66,136 \$89,197		\$105,144 <u>\$89,641</u>	\$70,253 <u>\$65,160</u>	\$18,893 <u>\$83,340</u>		
<u>Mileage Expense</u>	<u>\$2,317</u>		<u>\$0</u>	<u>\$0</u>	<u>\$0</u>		
TOTAL COST OF GOODS	\$255,191	56% / 10	\$378,331	\$192,254	\$258,702		
<u>GROSS PROFIT</u>	<u>\$222,116</u>	44%/8	<u>\$154,962</u>	<u>\$234,973</u>	<u>\$229,801</u>		
<u>Expenses</u>							
AHS Software	<u>\$6,262</u>		<u>\$5,988</u>	<u>\$5,988</u>	<u>\$5,988</u>		
<u>Automobiles</u>	<u>\$17,320</u>		\$13,088	<u>\$43,964</u>	<u>\$17,190</u>		

General & Administrative	\$82,254		<u>\$86,573</u>	\$70,904	<u>\$84,655</u>
Local Marketing	<u>\$33,629</u>		<u>\$12,553</u>	<u>\$35,318</u>	<u>\$31,617</u>
National Brand Fund	<u>\$8,178</u>		\$14,953	<u>\$8,653</u>	<u>\$8,883</u>
Rent	<u>\$10,354</u>		<u>\$7,595</u>	<u>\$10,630</u>	<u>\$7,840</u>
Royalties	\$29,592 <u>\$25,573</u>		\$55,785 <u>\$25,575</u>	\$29,730 <u>\$25,427</u>	\$18,261 <u>\$27,191</u>
Tools, C Card Fees, Permits	\$ 1,542 <u>908</u>		\$0	\$ 0 1,242	\$ 6,167 <u>192</u>
TOTAL COST OF GOODS	\$275,843 \$184,47	2/50 61% /		\$268,748 \$202,12	\$174,512 \$183,55
<u>SOLD</u> EXPENSES	<u>7</u>	<u>11</u>	\$411,071 \$166,327	<u>5</u>	<u>7</u>
GROSS PROFITOWNER			\$182,246 (\$11,365		
<u>DISCRETIONARY INCOME</u>	\$171,693 \$37,639	2/50 39% <u>/ 7</u>)	\$170,167 <u>\$32,847</u>	\$118,812 \$46,244
PERCENTAGE OF TOTAL REVENUE	<u>8%</u>	39% / 7	<u>-2%</u>	<u>8%</u>	<u>9%</u>

Table 3

Multi-Territory Single Reporting Operators Territories Third Quartile

That Have Operated between 1 and 3 Years

Average, High, <u>Low and Median and Low Performance Chart</u>
(41 Reporting Operators with 110 Territories)

Based on Total Revenue 18 Reporting Territories

	Single Quartile 3						
	AVERAGE#	// % ABOVE	AVERAGEHIGHMEDI		g <u>Territories</u>		
Revenue:	<u>Average</u>	%/# Above	<u>High</u>	Low	Median		
	\$505,190\\$266,51		\$1,618,181 <u>\$357,19</u>	\$437,465 \$233,17	\$196,443 \$250,90		
Labor Revenue	<u>4</u>		<u>1</u>	<u>3</u>	<u>0</u>		
— Material,Materials Trash,							
Equipment Revenue	\$98,502 \$58,458		\$183,664 <u>\$55,177</u>	\$85,312 \$31,336	\$34,312 <u>\$52,857</u>		
Service Charge Revenue	\$58,028 \$26,360		\$ 9,815 <u>5,375</u>	\$52,710 \$31,470	\$20,665 <u>\$29,474</u>		
	\$661,680 <u>\$351,33</u>	16/39 44% <u>/</u>	\$1,811,660 \$417,74	\$562,662 \$295,97	\$251,420 <u>\$333,23</u>		
TOTAL REVENUE	<u>2</u>	<u>8</u>	<u>3</u>	<u>9</u>	<u>2</u>		
Cost of Goods Sold÷							
	\$215,923 <u>\$104,34</u>			\$197,857 <u>\$125,11</u>			
Craftsmen Personnel Expenses	<u>4</u>		\$595,839 \$150,681	<u>2</u>	\$60,405 <u>\$104,532</u>		
- Materials, Material Trash,							
Equipment Expenses	\$94,293 <u>\$69,469</u>		\$182,482 \$58,391	\$79,374 \$42,758	\$39,207 <u>\$65,446</u>		
Mileage Expense	<u>\$2,148</u>		<u>\$0</u>	<u>\$0</u>	<u>\$0</u>		
TOTAL COST OF GOODS	\$175,961	39% / 7	<u>\$209,073</u>	\$167,870	<u>\$169,978</u>		
<u>GROSS PROFIT</u>	<u>\$175,371</u>	50%/9	\$208,670	<u>\$128,109</u>	\$163,254		
Expenses							
AHS Software	<u>\$4,690</u>		<u>\$5,646</u>	<u>\$4,990</u>	<u>\$5,567</u>		
Automobiles	<u>\$8,285</u>		<u>\$6,598</u>	<u>\$17,602</u>	<u>\$7,477</u>		

General & Administrative	<u>\$73,145</u>		<u>\$67,861</u>	<u>\$59,458</u>	\$60,816
Local Marketing	<u>\$29,791</u>		<u>\$41,634</u>	<u>\$40,749</u>	<u>\$28,671</u>
National Brand Fund	<u>\$5,739</u>		<u>\$7,098</u>	\$6,200	<u>\$6,316</u>
Rent	<u>\$9,643</u>		<u>\$9,096</u>	<u>\$15,549</u>	<u>\$9,167</u>
Royalties	\$38,487 <u>\$21,668</u>		\$98,822 <mark>\$21,337</mark>	\$32,076 <u>\$18,601</u>	\$10,254 <u>\$22,485</u>
Tools, C Card Fees, Permits	\$ 2,273 <u>660</u>		\$ 0 110	\$ 940 1,145	\$ 4,901 <u>62</u>
TOTAL COST OF GOODS	\$348,386 \$153,62	14/34 <u>44</u> % <u>/</u>		\$300,021 \$164,29	\$114,767 \$140,56
<u>SOLD</u> EXPENSES	<u>1</u>	<u>8</u>	\$877,142 <u>\$159,380</u>	<u>4</u>	<u>2</u>
GROSS PROFITOWNER		17/41 22% /		\$255,774 (\$36,185	
<u>DISCRETIONARY INCOME</u>	\$297,384 <u>\$21,750</u>	<u>4</u>	\$934,518 \$49,291)	\$136,653 <u>\$22,691</u>
PERCENTAGE OF TOTAL REVENUE	<u>6%</u>	39% / 7	<u>12%</u>	<u>-12%</u>	<u>7%</u>

Table 4

Multi-Territory Single Reporting Operators Territories Bottom Quartile

That Have Operated More than 3 Years

Average, High, <u>Low and Median and Low Performance Chart</u>
(12 Reporting Operators With 43 Territories)

Based on Total Revenue 18 Reporting Territories

	Single Quartile 4							
	AVERAGE!	AVERAGE# / % ABOVE AVERAGEHIGHMEDIANLOW18 Reporting Territories						
Revenue:	<u>Average</u>	verage %/# Above		Low	Median			
	\$651,492 \$177,95		\$1,849,629 <u>\$215,85</u>	\$480,778 \$163,61	\$240,879 \$168,14			
Labor Revenue	<u>8</u>		<u>6</u>	<u>1</u>	<u>1</u>			
— Material, Materials Trash,								
Equipment Revenue	\$145,807 \$41,378		\$520,960 \$60,584	\$125,541 <u>\$26,765</u>	\$14,332 <u>\$36,735</u>			
Service Charge Revenue	\$49,110 \$22,682		\$134,819 \$15,802	\$47,480\\$10,606	\$16,967 \$20,893			
	\$846,409 <u>\$242,01</u>		\$2,505,407 <u>\$292,24</u>	\$662,496 <u>\$200,98</u>	\$272,178 <u>\$225,77</u>			
TOTAL REVENUE	<u>8</u>	4/33 39% <u>/ 7</u>	<u>1</u>	<u>2</u>	<u>0</u>			
Cost of Goods Sold÷								
Craftsmen Personnel Expenses	\$325,582 \$74,429		\$1,046,176 <u>\$55,688</u>	\$275,053 \$69,847	\$70,418 \$72,624			
<u>Materials, Material</u> Trash,								
Equipment Expenses	\$126,338 <u>\$33,483</u>		\$439,834 <u>\$54,664</u>	\$90,647 <u>\$28,965</u>	\$71,203 \$30,404			
<u>Mileage Expense</u>	\$1,150		\$1,014	\$1,793	<u>\$0</u>			
TOTAL COST OF GOODS	<u>\$109,062</u>	67% / 12	<u>\$111,366</u>	<u>\$100,605</u>	<u>\$103,028</u>			
<u>GROSS PROFIT</u>	<u>\$132,956</u>	44%/8	<u>\$180,875</u>	\$100,377	\$122,742			
<u>Expenses</u>								
AHS Software	\$4,817		<u>\$4,502</u>	<u>\$6,615</u>	<u>\$5,489</u>			
Automobiles	<u>\$6,525</u>		<u>\$28,272</u>	<u>\$6,614</u>	<u>\$5,491</u>			
General & Administrative	\$50,703		\$55,248	\$59,479	<u>\$53,224</u>			

<u>Local Marketing</u>	\$19,314		\$16,332	<u>\$17,487</u>	\$18,427
National Brand Fund	<u>\$2,714</u>		<u>\$0</u>	<u>\$2,751</u>	<u>\$3,305</u>
<u>Rent</u>	<u>\$7,156</u>		<u>\$11,495</u>	\$11,957	\$8,142
Royalties	\$47,984 \$11,264		\$138,911 \$14,292	\$36,039 \$11,996	\$14,199 \$13,567
Tools, C Card Fees, Permits	\$ 1,373 <u>549</u>		\$ 2,505 <u>412</u>	\$ 254 <u>37</u>	\$ 270 170
TOTAL EXPENSES	\$103,042	50%/9	\$130,552	<u>\$116,935</u>	<u>\$107,815</u>
OWNER DISCRETIONARY INCOME	<u>\$29,913</u>	50%/9	\$50,324	<u>-\$16,559</u>	<u>\$14,928</u>
PERCENTAGE OF TOTAL REVENUE	<u>12%</u>	39%/7	<u>17%</u>	<u>-8%</u>	<u>7%</u>

Table 5

Multi-Territory Reporting Groups

Top Quartile

Average, High, Low and Median Performance Chart

Based on Total Revenue

19 Reporting Operators with 62 Reporting Territories

	Multi-Territory Quartile 1 19 Reporting Owners with 62 Reporting Territories					
Revenue	<u>Average</u>	%/# Above	High	Low	<u>Median</u>	
<u>Labor Revenue</u>	\$960,780		\$1,949,362	\$598,900	\$880,396	
Materials Trash Equipment						
<u>Revenue</u>	\$220,913		<u>\$539,846</u>	<u>\$137,120</u>	<u>\$182,112</u>	
Service Charge Revenue	<u>\$96,737</u>		<u>\$146,835</u>	<u>\$69,000</u>	<u>\$93,373</u>	
TOTAL REVENUE	<u>\$1,278,429</u>	47%/9	\$2,636,042	\$805,020	\$1,155,881	
Cost of Goods Sold						
<u>Craftsmen Personnel Expenses</u>	<u>\$433,714</u>		\$892,177	<u>\$276,763</u>	<u>\$376,221</u>	
<u>Material Trash Equipment</u>						
<u>Expenses</u>	<u>\$204,792</u>		<u>\$489,302</u>	<u>\$98,583</u>	<u>\$171,516</u>	
Mileage Expense	<u>\$8,830</u>		<u>\$61,864</u>	<u>\$0</u>	<u>\$0</u>	
	\$501,277 <u>\$647,33</u>	4/33 37% <u>/</u>	\$1,627,426 \$1,443,34	\$404,470 \$375,34	\$156,090 <u>\$547,73</u>	
TOTAL COST OF GOODS SOLD	<u>6</u>	<u> 7</u>	<u>3</u>	<u>5</u>	<u>7</u>	
	\$345,132 \$631,09	4/33 37% /		\$306,556 \$429,67	\$116,088 \$608,14	
GROSS PROFIT	<u>3</u>	7	\$877,981 \$1,192,699	<u>4</u>	<u>4</u>	
<u>Expenses</u>						
AHS Software	<u>\$7,083</u>		<u>\$10,571</u>	<u>\$6,438</u>	<u>\$6,172</u>	
Automobiles	<u>\$33,674</u>		<u>\$59,391</u>	<u>\$24,519</u>	<u>\$24,519</u>	
General & Administrative	<u>\$193,215</u>		\$319,357	<u>\$143,310</u>	<u>\$161,223</u>	
Local Marketing	<u>\$91,747</u>		\$62,037	<u>\$75,250</u>	<u>\$77,406</u>	
National Brand Fund	<u>\$21,561</u>		\$50,716	<u>\$14,219</u>	<u>\$23,995</u>	

Rent	\$22,917		<u>\$71,196</u>	<u>\$29,715</u>	\$22,402
<u>Royalties</u>	\$72,221		<u>\$153,667</u>	<u>\$42,588</u>	<u>\$65,751</u>
Tools, C Card Fees, Permits	<u>\$3,746</u>		\$1,935	<u>\$0</u>	<u>\$758</u>
TOTAL EXPENSES	<u>\$446,166</u>	53% / 10	<u>\$728,871</u>	\$336,039	<u>\$382,226</u>
OWNER DISCRETIONARY INCOME	<u>\$184,927</u>	37% / 7	<u>\$463,828</u>	<u>\$93,636</u>	<u>\$225,918</u>
PERCENTAGE OF TOTAL REVENUE	<u>14%</u>	53% / 10	<u>18%</u>	<u>12%</u>	<u>20%</u>
Number of Territories	<u>3</u>	63% / 12	<u>6</u>	<u>2</u>	<u>3</u>

Table 6

Multi-Territory Reporting Groups Second Quartile

Average, High, Low and Median Performance Chart Based on Total Revenue

19 Reporting Operators with 58 Reporting Territories

Multi-Territory Quartile 2							
	19 Reporting Owners with 58 Reporting Territories						
Revenue	<u>Average</u>	%/# Above	<u>High</u>	Low	<u>Median</u>		
<u>Labor Revenue</u>	<u>\$529,070</u>		<u>\$628,118</u>	<u>\$544,627</u>	<u>\$548,586</u>		
Materials Trash Equipment Revenue	\$118,773		<u>\$111,969</u>	<u>\$95,165</u>	\$100,908		
Service Charge Revenue	<u>\$61,740</u>		<u>\$63,375</u>	<u>\$7,256</u>	\$60,046		
TOTAL REVENUE	<u>\$709,583</u>	42%/8	\$803,462	<u>\$647,048</u>	\$709,540		
Cost of Goods Sold							
<u>Craftsmen Personnel Expenses</u>	\$217,152		\$299,078	\$252,247	\$226,130		
<u>Material Trash Equipment Expenses</u>	<u>\$117,048</u>		<u>\$94,026</u>	\$72,804	<u>\$103,180</u>		
Mileage Expense	\$9,085		<u>\$0</u>	<u>\$0</u>	<u>\$0</u>		
TOTAL COST OF GOODS	\$343,284	47%/9	\$393,104	\$325,051	\$329,311		
GROSS PROFIT	\$366,299	47%/9	\$410,359	\$321,997	\$380,229		
<u>Expenses</u>							
AHS Software	<u>\$5,640</u>		<u>\$8,376</u>	\$9,510	<u>\$5,988</u>		
<u>Automobiles</u>	<u>\$16,110</u>		\$33,441	\$12,623	\$13,629		
General & Administrative	<u>\$102,533</u>		<u>\$153,116</u>	<u>\$86,444</u>	<u>\$92,475</u>		
Local Marketing	<u>\$52,456</u>		\$50,927	\$57,849	\$50,927		
National Brand Fund	<u>\$10,595</u>		<u>\$16,184</u>	\$14,447	\$12,731		
<u>Rent</u>	\$11,277		\$25,287	<u>\$19,568</u>	\$9,635		
<u>Royalties</u>	\$37,593		\$49,155	\$40,013	\$39,112		
Tools, C Card Fees, Permits	<u>\$1,125</u>		\$1,603	<u>\$289</u>	<u>\$701</u>		
TOTAL EXPENSES	\$237,330	42%/8	\$338,090	\$240,744	\$225,196		

OWNER DISCRETIONARY INCOME	\$128,969	53% / 10	\$72,269	\$81,253	<u>\$155,033</u>
PERCENTAGE OF TOTAL REVENUE	<u>18%</u>	53% / 10	<u>9%</u>	<u>13%</u>	<u>22%</u>
Number of Territories	3	53% / 10	7	2	3

Table 7

Multi-Territory Reporting Groups

Third Quartile

Average, High, Low and Median Performance Chart

Based on Total Revenue

19 Reporting Operators with 51 Reporting Territories

Multi-Territory Quartile 3							
	19 Reporting Owners with 51 Reporting Territories						
Revenue	<u>Average</u>	%/# Above	<u>High</u>	Low	<u>Median</u>		
<u>Labor Revenue</u>	<u>\$418,910</u>		<u>\$446,433</u>	<u>\$316,358</u>	<u>\$427,809</u>		
Materials Trash Equipment Revenue	<u>\$75,976</u>		<u>\$139,616</u>	<u>\$57,784</u>	<u>\$66,888</u>		
Service Charge Revenue	\$45,283		<u>\$51,348</u>	\$40,800	<u>\$50,204</u>		
TOTAL REVENUE	<u>\$540,168</u>	47% / 9	\$637,397	<u>\$414,941</u>	<u>\$544,900</u>		
Cost of Goods Sold							
<u>Craftsmen Personnel Expenses</u>	\$166,158		\$237,369	\$99,572	\$167,347		
<u>Material Trash Equipment Expenses</u>	\$69,854		<u>\$113,508</u>	\$47,951	\$61,621		
Mileage Expense	<u>\$2,116</u>		<u>\$0</u>	<u>\$19,738</u>	<u>\$0</u>		
TOTAL COST OF GOODS	\$238,128	47% / 9	\$350,877	<u>\$167,260</u>	<u>\$228,968</u>		
GROSS PROFIT	\$302,040	47% / 9	<u>\$286,520</u>	<u>\$247,681</u>	<u>\$315,932</u>		
<u>Expenses</u>							
AHS Software	<u>\$5,473</u>		<u>\$5,988</u>	\$6,097	<u>\$5,988</u>		
<u>Automobiles</u>	<u>\$18,583</u>		<u>\$15,541</u>	\$272	<u>\$15,541</u>		
General & Administrative	<u>\$106,954</u>		<u>\$148,118</u>	\$86,448	\$107,214		
<u>Local Marketing</u>	\$42,320		\$27,736	\$34,167	\$44,900		
National Brand Fund	\$8,491		<u>\$0</u>	<u>\$8,115</u>	<u>\$9,590</u>		
Rent	<u>\$15,715</u>		<u>\$0</u>	\$22,128	<u>\$15,526</u>		
Royalties	\$30,943		\$1,054	\$23,229	\$30,726		
Tools, C Card Fees, Permits	<u>\$1,048</u>		<u>\$130</u>	<u>\$766</u>	<u>\$366</u>		
TOTAL EXPENSES	\$229,527	58% / 11	<u>\$198,568</u>	\$181,221	\$229,850		

OWNER DISCRETIONARY INCOME	\$72,514	47% / 9	<u>\$87,952</u>	<u>\$66,461</u>	\$86,082
PERCENTAGE OF TOTAL REVENUE	<u>13%</u>	<u>12%</u>	<u>14%</u>	<u>16%</u>	<u>16%</u>
Number of Territories	<u>3</u>	47% / 9	<u>5</u>	2	3

<u>Table 8</u>

<u>Multi-Territory Reporting Groups</u>

<u>Bottom Quartile</u>

<u>Average, High, Low and Median Performance Chart</u>

<u>Based on Total Revenue</u>

19 Reporting Operators with 43 Reporting Territories

Multi-Territory Quartile 4							
	19 Reporting Owners with 43 Reporting Territories						
<u>Revenue</u>	<u>Average</u>	%/# Above	<u>High</u>	Low	<u>Median</u>		
<u>Labor Revenue</u>	\$214,590		<u>\$294,859</u>	<u>\$75,050</u>	\$214,401		
<u>Materials Trash Equipment Revenue</u>	\$55,515		<u>\$74,649</u>	\$8,210	\$52,085		
Service Charge Revenue	\$24,240		\$41,040	<u>\$2,865</u>	\$21,704		
TOTAL REVENUE	\$294,345	58% / 11	<u>\$410,548</u>	<u>\$86,125</u>	\$288,190		
Cost of Goods Sold							
<u>Craftsmen Personnel Expenses</u>	\$100,244		\$169,575	\$23,169	\$101,556		
<u>Material Trash Equipment Expenses</u>	<u>\$55,646</u>		<u>\$62,890</u>	<u>\$11,494</u>	<u>\$57,165</u>		
<u>Mileage Expense</u>	\$882		<u>\$477</u>	<u>\$0</u>	<u>\$0</u>		
TOTAL COST OF GOODS	\$156,772	47% / 9	\$232,942	<u>\$34,663</u>	\$158,721		
GROSS PROFIT	\$137,573	58% / 11	<u>\$177,606</u>	<u>\$51,462</u>	\$129,469		
<u>Expenses</u>							
AHS Software	\$4,972		<u>\$7,321</u>	<u>\$3,493</u>	<u>\$5,988</u>		
<u>Automobiles</u>	<u>\$7,623</u>		<u>\$13,998</u>	<u>\$2,080</u>	\$5,299		
General & Administrative	\$51,937		\$95,803	\$15,168	\$48,550		
<u>Local Marketing</u>	<u>\$26,361</u>		<u>\$25,356</u>	<u>\$1,080</u>	<u>\$25,356</u>		
National Brand Fund	<u>\$3,443</u>		<u>\$8,299</u>	<u>\$2,310</u>	<u>\$2,510</u>		
Rent	\$9,737		<u>\$9,720</u>	<u>\$2,854</u>	\$9,720		
<u>Royalties</u>	<u>\$16,685</u>		<u>\$24,896</u>	<u>\$7,184</u>	<u>\$17,662</u>		
Tools, C Card Fees, Permits	<u>\$571</u>		<u>\$0</u>	<u>\$0</u>	<u>\$2</u>		
TOTAL EXPENSES	\$121,329	47% / 9	\$185,393	<u>\$34,169</u>	\$115,087		

OWNER DISCRETIONARY INCOME	\$16,244	47%/9	<u>-\$7,787</u>	\$17,293	\$13,072
PERCENTAGE OF TOTAL REVENUE	<u>6%</u>	47%/9	<u>-2%</u>	<u>20%</u>	<u>5%</u>
Number of Territories	<u>2</u>	16%/3	<u>4</u>	<u>2</u>	<u>2</u>

The accompanying footnotes are an integral part of these tables and should be read in their entirety for a full understanding of the information contained in them.

FOOTNOTES

- 1. "Total Revenue" means all revenue derived from the Reporting Operators' AHS Businesses <u>in the Reporting Territories</u>, excluding sales tax and <u>net of</u> discounts, in calendar year <u>2023</u>2024. The tables also break the Total <u>Revenues Revenue</u> into Labor Revenue, Material, Trash, Equipment Revenue, and Service Charge Revenues.
- 2. "Craftsmen Personnel Expenses" <u>include include</u> the costs for personnel including craftsmen and subcontractors, insurance, training fees, background checks, drug checks and all payroll <u>benefit</u> expenses paid by the Reporting Operators in calendar year 20232024.
- 3. "Materials, Trash, Equipment <u>Expenses</u>" <u>includes include</u> the costs for all materials, trash removal/dump and equipment rentals paid by the Reporting Operators in the calendar year 20232024.
- 4. "Tools, C Card Fees, Permits" includes the costs for tools, credit card processing fees and permits paid by the Reporting Operators in the calendar year 2023.
- 5. "Royalties" include the 6 percent Royalty Fees paid to us based on the Total Revenues.
- 64. "Total Cost of Goods—Sold" was determined by adding the amounts shown for the Craftsmen Personnel Expenses, the Materials, Trash, Equipment expenses Expenses, Royalties, and the Tools, C Card Fees, Permits expenses and Mileage Expense.
- 75. "Gross Profit" was determined by subtracting the Total Cost of Goods Sold from the Gross Revenues Total Revenue.
- **8**6. "Royalties" referes to the Royalty Fees paid to us based on the Total Revenue.
- 7. "Tools, C Card Fees, Permits" includes the costs for tools, credit card processing fees and permits paid by the Reporting Operators in the calendar year 2024.
- 8. "General & Administrative" includes the costs for office personnel wages, payroll taxes, insurance, telephone, and internet paid by the Reporting Operators in the calendar year 2024.
- 9. "Owner Discretionary Income" is determined by subtracting the Total Expenses from Gross Profit.

 Owner Discretionary Income may include salaries paid to the Reporting Operator or its owner, distributions taken by the Reporting Operator or its owner, and personnel business expenses incurred by the Reporting Operator or its owner. It may not represent the Reporting Operator's net profit.
- 10. Tables 1 and 2 through 4 show the Gross Profit financial results for the Single Territory Reporting Operators—Territories for calendar year 2023. Specifically, each of these tables shows the average and median total Revenues and Gross Profit information of the applicable Single Territory Reporting Operators groups shown in each table for calendar year 2023. Each table 2024, including the average and median Total Revenue, Gross Profit, and Owner Discretionary Income information. It also contains the number and percentage of those groups of Reporting Operators in that Units included in each such table that exceeded the average Total Revenues Revenue, Cost of Goods Sold and Gross Profit of the Reporting Operators

shown in the table and Owner Discretionary Income of that group. Each table also shows the complete financial results for the Reporting Operators shown Units in each applicable table group with the highest Total Revenues Revenue and the lowest Total Revenues, for the full year ending December 31, 2023 2024, in each of the same categories of information.

- 911. Tables 3 and 4.5 through 8 show the Gross Profit financial results for the Multi-Territory Reporting Operators Groups for calendar year 2023. Specifically, each of these tables shows the average and median total Revenues and Gross Profit information of the applicable Multi-Territory Reporting Operators groups shown in each table for calendar year 2023. Each table 2024, including the average and median Total Revenue, Gross Profit, and Owner Discretionary Income information. It also contains the number and percentage of those groups of Reporting Operators in that Units included in each such table that exceeded the average Total Revenues Revenue, Cost of Goods Sold and Gross Profit of the Reporting Operators shown in the table and Owner Discretionary Income of that group. Each table also shows the complete financial results for the Reporting Operators shown Units in each applicable table group with the highest Total Revenues Revenue and the lowest Total Revenues Revenue, for the full year ending December 31, 2023 2024, in each of the same categories of information.
- 12. The results shown in the "High" and "Low" columns of all tables show the financial results of the single Reporting Unit with the highest or lowest Total Revenue in each table. The numbers in that column do not necessary represent the highest or lowest overall result of all of the Reporting Units in the table for each particular line item other than Total Revenue.
- 1013. The "median" for purposes of the tables means the <u>financial</u> results of the Reporting <u>Operator Unit</u> falling in the middle of each group in terms of <u>Gross Revenues Total Revenue</u>, or, where there is an even number of Reporting <u>Operators Units</u>, the average of the <u>financial</u> results of the two Reporting <u>Operators Units</u> falling in the middle of the group. <u>The numbers in that column do not necessary represent the median overall result of all of the Reporting Units in the table for each particular line item other than Total Revenue.</u>
- 4114. The information in the tables were prepared from data obtained from the unaudited reports submitted by the Reporting Operators shown in each table to us from their QuickBooks software for calendar year 2023 2024. We have not audited these reports and cannot guarantee their accuracy.
- 1215. The tables contemplate those costs expressly listed as included in the "Total Cost of Goods—Sold" category of each table. You will have other expenses that are not shown in the tables, including rent, marketing expenses, vehicle expenses, and general administrative—" and "Total Expenses" categories of each table. Expenses vary substantially and are based on particular factors relevant to each Operator. You may incur operating expenses different from those stated above as well as other isolated or recurring expenses. See ITEMS 5, 6, and 7 for fees and other expenses you may incur. Further, the tables do not show any taxes paid by the Reporting Operators, including any sales taxes and income taxes that may be applicable. Taxes vary widely between Territories and from Operator to Operator. You should conduct an independent investigation of the costs and expenses you will incur in operating your AHS Business. Franchisees or former franchisees listed in Attachment I and Attachment I to this Disclosure Document may be one source of this information.

ADDITIONAL NOTES APPLICABLE TO ALL TABLES:

Some AHS Businesses have sold and earned these amounts. Your individual results may differ. There is no assurance that you will sell or earn as much.

Written substantiation for these financial performance representations will be made available to the prospective franchisee upon reasonable request.

Other than the preceding financial performance representations, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing AHS Business, however, we may provide you with the actual records of that AHS Business. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Andy Bell at 390 Union Boulevard Blvd, Suite Ste 520, Lakewood, Colorado 80228, and (303) 984-0177, the Federal Trade Commission, and the appropriate state regulatory agencies.

REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

Table No. 1 Systemwide Outlet Summary For Years 2021 to 2023⁽⁴2022 to 2024⁽¹⁾

Column 1	Column 2	Column 3	Column 4	Column 5
Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change
2021	123	225	+102	
	2022	225	305	+80
Franchised	2023	305	367	+62
Company-Owned ⁽²⁾	2021 2024	7 367	7 369	<u>0+2</u>
	2022	7	7	0
Company-Owned ⁽²⁾	2023	7	12	+5
	2021 2024	130 12	232 18	+ <u>102</u> 6
TOTAL	2022	232	312	+80
OUTLETS	2023	312	379	+67
	<u>2024</u>	<u>379</u>	<u>387</u>	<u>+8</u>

⁽¹⁾ Each year period begins on January 1st and ends on December 31st.

Table No. 2
Transfers of Outlets from Franchisees to New Owners (other than the Franchisor)
For Years 2021 to 2023 (2022 to 2024 (1))

Column 1	Column 2	Column 3	
State	Year	Number of Transfers	
Arizona	2021 2022	<u>01</u>	
	2022	1	
	2023	0	
	2021 <u>2024</u>	0	

65

⁽²⁾ Includes outlets operated by our affiliate, AHHS.

Column 1	Column 2	Column 3
State	Year	Number of Transfers
C-11-	2022	0
<u>Colorado</u> Maryland	2023	<u> 13</u>
Maryland	2021 2024	0
	2022	<u>30</u>
<u>Florida</u>	2023	0
	2021 <u>2024</u>	0 2
	2022	<u> 10</u>
Georgia	2023	0
	<u>2024</u>	3
T 11	2022	0
<u>Indiana</u>	2023	1
	<u>2024</u>	0
17	2022	0
Kansas	2023	<u>01</u>
	<u>2024</u>	<u>0</u> 3
Maryland	2022 2023	<u>3</u> 0
<u>warytanu</u>	2023	0
	2022	0
Michigan	2022	0
Wilchigan	2023	1
	2022	1
New Jersey	2023	$\frac{1}{0}$
<u>Itew sersey</u>	2024	0
North Carolina	2021 2022	<u>0</u> 1
Troitin Guironniu	2022	1
	2023	0
	2024	2
	2022	0
Ohio	2023	1
	2024	0
South Carolina	2021 2022	0 1
	2022	1
	2023	1
	2024	0
	2022	0
South Dakota	2023	0
	2024	1
	<u>2022</u>	0
<u>Tennessee</u>	<u>2023</u>	<u>0</u>
	<u>2024</u>	<u>1</u>
Texas	2021 <u>2022</u>	<u> 30</u>
	2022	0
	2023	4
	<u>2024</u>	<u>2</u>
TOTALS	2021 2022	<u>37</u>
	2022	7
	2023	611
	2024	12

Each year period begins on January 1st and ends on December 31st.

Table No. 3
Status of Franchised Outlets
For Years 2021 to 2023(1)2022 to 2024(1)(2)

Column 1 State	Column 2 Year	Colum n 3 Outlets at Start of Year	Colum n 4 Outlets Opene d	Column 5 Terminations	Column 6 Non- Renewal s	Column 7 Reacquire d by Franchiso r	Column 8 Ceased Operation s- Other Reasons	Colum n 9 Outlets at End of the Year
2021	0 0000	2	0	0	0	0	2	
Alabama	2022	3	1	0	0	0	0	3
	2023 2021 <u>202</u> 4	4 <u>3</u>	0 <u>1</u>	0	0	0	1 4 <u>0</u>	0 <u>4</u>
	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
Arizona	2021 <u>202</u> 2	1 <u>7</u>	<u>65</u>	0	0	0	0	7 12
	<u>2023</u>	<u>12</u>	<u>0</u>	<u>0</u>	0	<u>0</u>	<u>0</u>	<u>12</u>
	2022 202 4	7 <u>12</u>	<u>50</u>	0	0	0	<u>01</u>	12 <u>11</u>
2021	1	0	0	0	0	0	1	
	2023	12	0	0	0	0	0	12
Arkansas	2022	1	0	0	0	0	0	1
California	2023	1	2	0	0	0	0	3
	2021 <u>202</u> 4	<u>53</u>	<u>30</u>	0	0	0	0	<u>83</u>
<u>California</u>	2022	8	7	0	0	0	3	12
	2023	12	6	0	0	0	0	18
	2021 <u>202</u> 4	<u>618</u>	0 2	0	<u>04</u>	0	<u>20</u>	4 <u>16</u>
<u>Colorado</u>	2022	4	2	0	0	0	0	6
	2023	6	0	0	0	3	0	3
Connecticut	2024	<u>3</u>	<u>0</u>	<u>0</u>	0	0	0	<u>3</u>
Connecticut	2021 <u>202</u> 2	<u>21</u>	<u>01</u>	0	0	0	<u> 40</u>	<u> 12</u>
	2022	1	1	0	0	0	0	2
	2023	2	2	0	0	0	0	4
	2024	<u>4</u>	1	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
Delaware	2021 <u>202</u>	<u>01</u>	<u> 40</u>	0	0	0	0	1
	2022 <u>202</u> 3	1	0	<u>01</u>	0	0	0	<u> 40</u>
	2023 <u>202</u> 4	<u> 40</u>	<u>01</u>	<u> 40</u>	0	0	0	<u>01</u>
2021	5	19	2	0	0	0	22	
Florida	2022	22	10	0	0	0	2	30

67

Start of Year 2023 30 11 0 0 0 0 0 0 41	Column 1 State	Column 2 Year	Colum n 3	Colum n 4	Column 5 Terminatio	Column 6 Non-	Column 7 Reacquire d by	Column 8 Ceased Operation	Colum n 9
Coorgia 2024 41 10 0 0 0 0 0 5 46			Start of					Other	of the
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$				11	0	0	0		
2			<u>41</u>	<u>10</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>	<u>46</u>
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Georgia	<u>2</u>		<u>63</u>		0	0	0	7 <u>10</u>
A			<u>10</u>	<u>6</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>16</u>
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		<u>4</u>			0				
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		2023	10	6	0	0	0	0	16
Company Comp	Idaho		0	0 4	0	0	0	0	0 4
Himois		2022	0	4	0	0	0	0	4
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2023	4	0	0	0	0	0	4
Description	Illinois	2021	8	4	0	0	0	0	9
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2022	9	3	0	0	0	0	12
Description Color Color		2023	12	6	7	0	0	1	10
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Indiana	2021	0	4	0	0	0	0	4
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2022	4	2	0	0	0	0	6
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		2023	6	2	0	0	0	0	8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Iowa	2021	0	0	0	0	0	0	0
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2022	0	0	0	0	0	0	0
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2023	0	2	0	0	0	0	2
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Kansas	2021	5	0	0	0	0	0	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2022	5	0	0	0	0	0	5
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		2023	5	0	0	0	4	0	4
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Kentucky								
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		2022	2	2	0	0	0	0	4
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$			4	0	0	0	0	0	4
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		<u>4</u>							
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	<u>Illinois</u>								
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Maine		<u> 112</u>	0 6	<u> </u>	0	0	<u>01</u>	<u> 10</u>
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		<u>4</u>							
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$									
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Maryland	2021 202		_					
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Lowe	_							
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$									
		2021 202							
	Michigan	2022	8 5	2 0	0	0	0	0	105

Column 1	Column 2	Colum n 3	Colum n 4	Column 5	Column 6	Column 7	Column 8	Colum n 9
State	Year	Outlets at Start of	Outlets Opene d	Terminatio ns	Non- Renewal	Reacquire d by Franchiso r	Ceased Operation s- Other Reasons	Outlets at End of the Year
		Year						
<u>Kansas</u>	2023	10 5	<u>20</u>	0	0	<u>01</u>	<u> 10</u>	11 4
	2021 <u>202</u> 4	2 4	9 0	0	0	0	0	11 4
Kentucky	2022	<u>112</u>	<u>92</u>	0	0	0	0	20 4
Minnesota	2023	20 4	0	0	0	0	0	20 4
	2021 <u>202</u> 4	<u>14</u>	<u>20</u>	0	0	0	<u> 12</u>	2
<u>Louisiana</u>	2022	<u>21</u>	<u>50</u>	0	0	0	0	<u>71</u>
Mississippi	2023	<u>71</u>	0	0	0	0	0	<u>71</u>
	2021 <u>202</u> 4	<u>01</u>	<u> 40</u>	0	0	0	<u> 40</u>	<u>01</u>
Maine	2022	<u>02</u>	<u>01</u>	0	0	0	0	0 3
Missouri	2023	<u>03</u>	<u> 10</u>	0	0	0	0	<u> 13</u>
	2021 <u>202</u> 4	3	0	0	0	0	0	3
Maryland	2022	<u>315</u>	<u> 13</u>	0	0	0	0	4 <u>18</u>
Montana	2023	4 <u>18</u>	<u>01</u>	0	0 3	<u> 10</u>	0	<u>316</u>
	2021 <u>202</u> 4	<u>016</u>	0	0	0	0	0	0 16
Massachuset	2022	<u>08</u>	<u>02</u>	0	0	0	0	<u>010</u>
<u>ts</u>	2023	<u>010</u>	2	0	0	0	<u>01</u>	<u>211</u>
Nebraska	2021 <u>202</u> 4	<u> 411</u>	<u>01</u>	0	0	0	0	<u> 112</u>
Michigan	2022	<u> 111</u>	1 9	0	0	0	0	2 20
Nevada	2023	2 20	0	0	0	0	0	2 20
	2021 <u>202</u> 4	<u> 120</u>	0	0	0	0	0	<u> 120</u>
Minnesota	2022	<u> 12</u>	0 5	0	0	0	0	<u> 17</u>
New	2023	<u> 17</u>	0	0	0	0	0	<u> 17</u>
Hampshire	2021 <u>202</u> 4	0 <u>7</u>	0	0	0	0	0	0 <u>7</u>
2021	0	0	0	0	0	0	0	
	2022	0	0	0	0	0	0	0
	2023	0	1	0	0	0	0	1
New Jersey	2021	0	4	0	0	0	0	1
	2022	1	2	0	0	0	0	3
	2023	3	6	0	0	0	0	9
New Mexico	2022	0	0	0	0	0	0	0
<u>Mississippi</u>	2022	0	0	0	0	0	0	0
	2023	0	1	0	0	0	0	1
	2023	0	1	0	0	0	0	1
Misser	<u>2024</u>	1	0	0	0	0	0	1
<u>Missouri</u>	2022 2023	<u>3</u> 4	0	<u>0</u> 0	0	<u>0</u>	0	<u>4</u> <u>3</u>
	2024	3	0	0	0	2	0	<u>3</u>
	<u> 2024</u>	2	<u>U</u>	<u>U</u>	<u>U</u>	<u>4</u>	<u>U</u>	<u>1</u>

Column 1	Column	Colum	Colum	Column 5	Column	Column 7	Column 8	Colum
State	2 Year	n 3 Outlets at Start of Year	n 4 Outlets Opene d	Terminatio ns	6 Non- Renewal s	Reacquire d by Franchiso r	Ceased Operation s- Other Reasons	n 9 Outlets at End of the Year
New	2022	0	0	0	0	0	0	0
York Montan	2023	0	2	0	0	0	0	2
<u>a</u>	2021 202 4	<u> 12</u>	2 0	0	0	0	0	3 2
Nebraska	2022	<u>31</u>	1	0	0	0	0	<u>42</u>
North	2023	<u>42</u>	<u>20</u>	0	0	0	0	<u>62</u>
Carolina	2021 <u>202</u> 4	4 <u>2</u>	<u>50</u>	0	0	0	0	9 2
Nevada	2022	9 1	<u>90</u>	0	0	0	0	<u> 181</u>
North	2023	<u> 181</u>	<u>20</u>	0	0	0	<u> 10</u>	<u> 191</u>
Dakota	2021 <u>202</u> <u>4</u>	<u>01</u>	<u> 40</u>	0	0	0	0	1
New	2022	<u> 40</u>	0	0	0	0	0	<u> 40</u>
<u>Hampshire</u>	2023	<u> 10</u>	<u>01</u>	0	0	0	0	1
	2021 <u>202</u> 4	4 <u>1</u>	<u>30</u>	0	0	0	0	<u>71</u>
New Jersey	2022	<u>71</u>	2	0	0	0	0	9 3
Oklahoma	2023	9 3	<u>26</u>	0	0	0	0	11 9
	2021 <u>202</u> 4	0 9	<u> 10</u>	0	0	0	0	<u> 49</u>
New Mexico	2022	<u> 40</u>	0	0	0	0	0	<u> 40</u>
Oregon	2023	<u> 10</u>	1	0	0	0	0	<u>21</u>
	2021 <u>202</u> 4	<u>21</u>	<u>31</u>	0	0	0	0	<u>52</u>
New York	2022	5 3	<u>01</u>	0	0	0	0	5 4
Pennsylvani	2023	<u>54</u>	2	0	0	0	0	7 6
a	2021 <u>202</u> 4	7 6	<u>20</u>	0	0	0	0	9 6
North	2022	9	<u> 19</u>	0	0	0	0	10 18
<u>Carolina</u>	2023	10 18	<u> 12</u>	0	0	0	<u>01</u>	11 19
Rhode Island	2021 <u>202</u> 4	5 19	<u>01</u>	0	0	0	0 2	<u>518</u>
North	2022	<u>51</u>	0	0	0	0	0	<u>51</u>
<u>Dakota</u>	2023	<u>51</u>	0	0	0	0	0	<u>51</u>
South Carolina	2021 <u>202</u> 4	<u>21</u>	<u>20</u>	0	0	0	0	4 <u>1</u>
Ohio C	2022	4 <u>7</u>	4 <u>2</u>	0	0	0	0	<u>89</u>
South	2023	<u>89</u>	<u>02</u>	0	0	0	0	<u>811</u>
Dakota	2021 <u>202</u> 4	<u>211</u>	<u>01</u>	0	0	0	0	2 12
<u>Oklahoma</u>	2022	<u>21</u>	0	0	0	0	0	<u>21</u>
	2023	<u>21</u>	<u>01</u>	0	0	0	0	2
	2021 <u>202</u> 4	2	<u>20</u>	0	0	0	<u>01</u>	4 <u>1</u>
	2023	6	3	0	0	0	0	9

Column 1	Column 2	Colum n 3	Colum n 4	Column 5	Column 6	Column 7	Column 8	Colum n 9
State	Year	Outlets at Start of Year	Outlets Opene d	Terminatio ns	Non- Renewal s	Reacquire d by Franchiso r	Ceased Operation s- Other Reasons	Outlets at End of the Year
Texas	<u>2022</u>	<u>5</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
Oregon	2022 3	4 <u>5</u>	2	0	0	0	0	6 <u>7</u>
	2021 <u>202</u> <u>4</u>	24 <u>7</u>	15 0	0	0	0	0	39 <u>7</u>
<u>Pennsylvani</u>	2022	39 9	<u>21</u>	0	0	0	0	41 <u>10</u>
<u>a</u>	2023	41 <u>10</u>	<u>71</u>	0	0	0	0	48 <u>11</u>
Utah	2021 <u>202</u> 4	<u>011</u>	4 <u>1</u>	0	0	0	1	<u>311</u>
Rhode Island	2022	<u>35</u>	0	0	0	0	0	3 <u>5</u>
Virginia	2023	<u>35</u>	<u>30</u>	0	0	0	0	6 5
	2021 <u>202</u> 4	<u>25</u>	<u>30</u>	0	0	0	0	5
South	2022	<u>54</u>	2 4	0	2 0	0	0	<u>58</u>
Carolina	2023	<u>58</u>	4 <u>0</u>	0	0	0	0	<u>98</u>
Washington	2021 <u>202</u> 4	<u>28</u>	4 <u>1</u>	0	0	0	<u> 40</u>	5 9
South	2022	<u>52</u>	<u> 10</u>	0	0	0	<u> 10</u>	5 2
<u>Dakota</u>	2023	<u>52</u>	<u> 10</u>	0	0	0	0	<u>62</u>
Wisconsin	2021 <u>202</u> 4	2	<u>20</u>	0	0	0	0	<u>42</u>
<u>Tennessee</u>	2022	4	0 2	0	0	0	0	<u>46</u>
	2023	<u>6</u>	3	0	0	0	0	9
	2024	9	<u>3</u>	0	0	<u>0</u>	<u>0</u>	<u>12</u>
Texas	2023 <u>202</u> 2	4 <u>39</u>	2	0	0	0	0	<u>641</u>
	2023	41	7	0	0	0	0	<u>48</u>
T 14-1	2024	48	2	6	2	3	1	38
<u>Utah</u>	2022	3	0	0	0	0	0	3
	2023 2024	<u>3</u>	<u>3</u> 0	0	0	0	0	<u>6</u>
Virginia	2024	5	2	0	2	0	0	5
, iigiiiu	2023	5	4	0	0	0	0	9
	2024	9	0	0	0	0	0	9
Washington	2022	<u>5</u>	<u>1</u>	0	0	0	1	<u>5</u>
	2023	<u>5</u>	1	0	0	0	0	<u>6</u>
	<u>2024</u>	<u>6</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>6</u>
West	2022	0	0	0	0	0	0	0
<u>Virginia</u>	2023	0	0	0	0	0	0	0
W/:- ·	2024	0	2	0	0	0	0	2
Wisconsin	2022	4	0	0	0	0	0	4
	2023 2024	<u>4</u> 6	<u>2</u> 0	0	0	0	<u>0</u> 1	<u>6</u> 5
Wyoming	2021 202	_		_	_	_	<u>1</u>	
" younnig	<u>2021</u> 202	1	0	0	0	0	0	1

Column 1 State	Column 2 Year	Colum n 3 Outlets at Start of Year	Colum n 4 Outlets Opene d	Column 5 Terminatio ns	Column 6 Non- Renewal s	Column 7 Reacquire d by Franchiso r	Column 8 Ceased Operation s- Other Reasons	Colum n 9 Outlets at End of the Year
	2022 3	1	0	0	0	0	0	1
	2023 <u>202</u> 4	1	0	0	0	0	0	1
2021	123	112	2	0	0	8	225	
TOTALS	2022	225	88	0	2	0	6	305
	2023	305	82	8	3	5	4	367
	<u>2024</u>	<u>367</u>	<u>35</u>	<u>7</u>	<u>6</u>	<u>5</u>	<u>15</u>	<u>369</u>

- Each year period begins on January 1st and ends on December 31st.
- (2) As of December 31, 2024, five of the franchised outlets were Mini AHS Businesses.

Table No. 4
Status of Company-Owned Outlets
For Years 2021 to 2023(1)(2) 2022 to 2024(1)(2)

Column	Column	Column	Column	Column 5	Column	Column 7	Column
1	2	3	4		6		8
				Outlets		Outlets	
State	Year	Outlets	Outlets	Reacquired	Outlets	Sold to	Outlets
		at Start	Opened	From	Closed	Franchisee	at End of
		of Year	•	Franchisee			the Year
Colorado	2021 2022	7	0	0	0	0	7
	2022 2023	7	0	0 3	0	0	7 10
	2023 2024	7 <u>10</u>	0	<u>30</u>	0	0	10
Kansas	2021 2022	0	0	0	0	0	0
	2022	0	0	0	0	0	0
	2023	0	0	1	0	0	1
	<u>2024</u>	<u>1</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
Missouri	2021 2022	0	0	0	0	0	0
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>1</u>	<u>0</u>	<u>0</u>	<u>1</u>
	<u>2024</u>	<u>1</u>	<u>0</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>3</u>
<u>Texas</u>	2022	0	0	0	0	0	0
	2023	0	0	<u> 10</u>	0	0	<u> 10</u>
	<u>2024</u>	<u>0</u>	0	<u>3</u>	<u>0</u>	<u>0</u>	<u>3</u>
Totals	2021 2022	7	0	0	0	0	7
	2022	7	0	0	0	0	7
	2023	7	0	5	0	0	12
	2024	<u>12</u>	1	<u>5</u>	<u>0</u>	0	<u>18</u>

- (1) Each year period begins on January 1st and ends on December 31st.
- ⁽²⁾ Includes outlets operated by our affiliate, AHHS.

Table No. 5
Projected Openings As of December 31, 2023 2024

Column 1	Column 2	Column 3	Column 4
State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlet In The Next Fiscal Year	Projected New Company-Owned Outlet In The Next Fiscal Year
Alabama	θ	4	θ
Arizona	0	2	0
Arkansas Alabama	0	1	0
California	0	4 <u>3</u>	0
Connecticut Florida	<u> 10</u>	<u> 43</u>	0
Delaware	0	1	0
Florida	3	5	θ
Georgia	3	3	0
Idaho Georgia	<u>01</u>	<u> 12</u>	0
Illinois	<u>01</u>	5 3	0
Indiana	0	21	0
Iowa	θ	1	0
Kentucky	0	3 2	0
Louisiana	0	2 1	0
Maryland	θ	1	0
Massachusetts	0	2	0
Michigan	0	2	0
Minnesota	0	2 1	0
Nebraska	θ	1	0
Nevada	0	2 1	0
New Hampshire	0	2 1	0
New Jersey	0 1	2	0
New Mexico	0	1	0
New York	0 1	42	0
North Carolina	0	2 1	0
Ohio	10	2 1	0
Oregon	1	0 1	0
Pennsylvania	1 0	<u>42</u>	0
South Carolina	$+\overline{0}$	2 1	0
Tennessee	1	2	0
Texas	<u>30</u>	4 <u>2</u>	0
Utah	0	2 1	0
Virginia	0	2 1	0
Washington	0	2 1	0
West Virginia	2	0	0
TOTALS	17 5	73 40	0

A list of the names of all franchisees and the addresses and the telephone numbers of their franchised business are listed in Attachment H to this Disclosure Document. A list of the name and last known city, state and telephone number of every franchisee who has had a franchised business terminated, cancelled,

73

not renewed, or otherwise voluntarily or involuntarily ceased to do business under their Franchise Agreement during the most recently completed fiscal year or who has not communicated with us within 10 weeks of the date of this Disclosure Document is listed on <u>Attachment I</u> to this Disclosure Document. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

During the last three fiscal years, no current or former franchisees have signed confidentiality clauses that restrict them from discussing with you their experiences as a franchisee in our franchise system.

There is not currently any advertising council within the Ace Handyman Services System, although we may establish one or more councils in the future on terms we elect in our discretion.

We currently have no other trademark-specific franchisee association within the Ace Handyman Services System, but reserve the right to form them in the future.

ITEM 21 FINANCIAL STATEMENTS

Attached to this disclosure document as <u>Attachment J</u> are the audited consolidated financial statements of Ace Hardware and its subsidiaries as of December <u>28, 2024, December 30, 2023, and December 31, 2022, and January 1, 2022</u> and for the years then ended. We are owned indirectly by Ace Hardware. Ace Hardware and we use a 52-53-week tax year. Therefore, our tax year ends on the Saturday closest to December 31st. The Guaranty of Performance by Ace Hardware of our performance of our obligations under the Franchise Agreement is included in <u>Attachment J.</u>

ITEM 22 CONTRACTS

These are the only contracts we enter into with any franchisee regarding the offering of franchises in this state:

Attachment A - Franchise Agreement and Exhibits

Attachment B - Nondisclosure and Noncompetition Agreement

Attachment C - Statement of Prospective Franchisee

Attachment D - Multi-Territory Addendum to Franchise Agreements

Attachment E - Form of Successor Franchise Rider to Franchise Agreement

Attachment F - Current Form of General Release

Attachment M - AHF Data Collection Agreement

Attachment N - IGX Participation Agreement

ITEM 23 RECEIPTS

The last two pages of this Disclosure Document are receipt pages. Please sign and date each of them <u>as of the date you received this Disclosure Document</u>, detach the second receipt page, and promptly return it to us as specified on that page.

ATTACHMENT A

FRANCHISE AGREEMENT WITH EXHIBITS

ACE HANDYMAN SERVICES® FRANCHISE AGREEMENT

Franchisee:	
Territory:	

ACF HANDYMAN SERVICES® FRANCHISE AGREEMENT TABLE OF CONTENTS

SECTION	PAGE
SECTION	ragi

4	-BACKGROUND AND PURPOSE	1
	1.1.—Background.	
	1.2.—System	
	1.3. —Purpose	
2	GRANT OF FRANCHISE	
	2.1. —Grant of Franchise.	1
	2.2:—Scope of Franchise Operations.	1
3	TERM AND EXPIRATION	
٥.	3.1.—Term	2
	3.2.—Continuation.	2
	3.3.—Successor Franchise.	2
	3.4.—Exercise of Option for Successor Franchise.	
4-	TERRITORY	
т.	4.1.—Business Location.	3
	4.2.—Territory	3
	4.3.—Failure to Respond to Referrals.	
	4.4.—AHF's Reservation of Rights	
5 -	-INITIAL FEES.	
3.	5.1.—Initial Franchise Fee.	5
6 -	TRAINING	
₩.	6.1.—Initial Training Program.	
	6.2.—Length of Training. 6.3.—On-Site Training.	
	6.4.—Additional Training.	0
	6.5.—Release	
7 -	FRANCHISEE'S DEVELOPMENT OBLIGATIONS	
≠•		
	7.1. Computer Equipment and Telephones.	······
	7.2.—Authorized Representatives and Independent Contractors.	
	7.3.—Commencement of Operations.	
0 -	7.4.—Active Operations DEVELOPMENT ASSISTANCE	۷
0.		لا
0	8.1.—AHF's Development Assistance. —OPERATIONS MANUAL	.
y. -		10
	9.1. Operations Manual.	
	9.2.—Confidentiality of Operations Manual Contents. 9.3.—Changes to Operations Manual.	10
10	9.5.—Changes to Operations Manual. -OPERATING ASSISTANCE	1 0
10	10.1.—AHF's Available Services.	
		10
11	10.2.—Additional AHF Services.	12
++	-FRANCHISEE'S OPERATIONAL COVENANTS	
	11.1.—Business Operations.	12
	11.2.—Requirements for Entity Franchisees.	17
10	11.3.—New Products or Services.	17
12	-CONTINUING FEES AND PAYMENTS	18
	12.1.—Royalty Fee	
	12.2.—National Brand Fees.	
	12.3.—Gross Revenues Defined	
	12.4.—Minimum Annual Gross Revenues.	
	12.5.—Software and Internet Fee.	19

	12.6. —	Payments to AHF	20
	12.7. —	-Late Fees; Denied ACH Processing; Interest	21
		-Nonrefundable Fees.	
13		ERTISING	
		Approval of Advertising.	
		-Initial Marketing Campaign.	
		-Minimum Individual Marketing.	
		Local Advertising Group.	
	12.4.	-National Brand Fund	⊆ວ າາ
	12.6	Franchisee Advisory Council	∠∋ 2.4
	13.0.	Franchisee Advisory Council	24
14		LITY CONTROL	
		-Standards and Specifications.	
	14.2. —	Restrictions on Services and Materials.	25
	14.3. —	-Commercial Quality Service Program	26
15			
		-Marks.	
	15.2. —	-No Use of Other Marks	26
	15.3. —	-Licensed Methods.	26
	15.4. —	-AHF's Rights to New Ideas.	27
		-Copyrights.	
	15.6.	-Infringement	27
		Franchisee's Business Name.	
		Change of Marks	
	15.0.	-Data and Other Business Records.	20 28
16		ORTS, RECORDS AND FINANCIAL STATEMENTS	
10	16.1 —	Franchisee Reports.	20
		Books and Records.	
		-Audit of Books and Records.	
		Failure to Comply with Reporting Requirements.	
		Financial Information from Third Parties.	
17		NSFER	
		-Transfer by AHF	
		-Transfer by Franchisee	
		-Pre-Conditions to Franchisee's Transfer	
	17.4. —	-Franchisee's Death or Disability	33
	17.5. —	-AHF's Right of First Refusal.	33
	17.6. —	-Post-Transfer Obligations.	34
18		AULT AND TERMINATION	
		-Termination by AHF-Effective Upon Notice.	
		Termination by AHF with Prior Notice.	
		Termination by AHF for Continuing Force Majeure Events.	
		-AHF's Right to Suspend Services on Franchisee's Default	
		Termination by Franchisee.	
		Obligations of Franchisee Upon Termination or Expiration	
		-Franchisee's Failure to Comply With Post-Termination Obligations	
		-Effects of Termination or Expiration.	
		Outstanding Loan Obligations.	
		-Conflicting Laws.	
		-Remedies Cumulative.	
19		NESS RELATIONSHIP	
		-Business Relationship	
	19.2. —	-Third Party Obligations.	40
	19.3. —	-Indemnification.	40
20	-REST	FRICTIVE COVENANTS	41
		-Non-Competition During Term.	
		-Post-Termination Covenant Not to Compete	
		1	

	20.3.	-No Diversion of Business.	42
		-Non-Disparagement.	
		-Confidentiality of Proprietary Information.	
	20.6. —	-Confidentiality Agreements and Acknowledgements	42
		-Claims Are Not Defenses to Covenants	
21	INSU	RANCE	43
	21.1.	-Insurance Coverage.	43
		-Proof of Insurance	
		-Failure to Maintain Insurance.	
	21.4.	-Contractor and Subcontractor Insurance.	44
22	ARB	TRATION	44
		-Arbitration.	
	22.2. —	-Arbitration Award	45
	22.3. —	-Limitations on Proceedings.	45
		-Injunctive Relief	
	22.5. —	Governing Law/Consent to Jurisdiction/Waiver of Jury Trial	46
	22.6. —	-No Punitive or Consequential Damages	46
	22.7. —	-No Punitive or Consequential DamagesNo Recourse Against Others	47
23	MISC	CELLANEOUS PROVISIONS	47
	23.1.	-Modification.	47
	23.2.	-Entire Agreement	47
		-Delegation by AHF	
		-Consent; Business Judgment.	
		-General Economic Conditions.	
	23.6. —	- <u>Effectiveness</u>	48
	23.7. —	-Limitation on Actions	49
	23.8. —	-Review of Agreement	49
	23.9. —	-Attorneys' Fees.	49
		-No Waiver	
	23.11	-No Right to Set Off	49
	23.12.	-Survival of Terms.	.5 0
		-Invalidity	
	23.14.	-Notices.	.50
		-Force Majeure	
		-Estoppel Certificates.	
		-Charges and Taxes	
		-Cross-Default and Cross Termination Provisions.	
	23.19.	-Manner of Payment	51
		-Approval Within Territory	
		-Translation of Agreement.	52
			52
	23.23.	-Incorporation of Riders.	52
	23.24.	-Counterparts; Electronic Signatures.	.52
			.52
1.	BAC	KGROUND AND PURPOSE	1
	1.1.	Background	
	1.2.	System	
	1.3.	Purpose	
2.	GRA	NT OF FRANCHISE	
	2.1.	Grant of Franchise.	
	2.2.	Scope of Franchise Operations.	
3.	TER	M AND EXPIRATION	2
	3.1.	Term	
	3.2.	Continuation.	
	3.3.	Successor Franchise.	
	3.4.	Exercise of Option for Successor Franchise.	

<u>4.</u>	TER	RITORY	<u></u> 3
	4.1.	Business Location.	<u></u> 3
	4.2.	Territory	3
	4.3.	Failure to Respond to Referrals.	4
	4.4.	AHF's Reservation of Rights	4
5.	INIT	IAL FEES	
	5.1.	Initial Franchise Fee.	
6.		INING	
-	6.1.	Initial Training Program.	
	6.2.	Length of Training.	
	6.3.	On-Site Training.	
	6.4.	Additional Training.	
	6.5.		
7		NCHISEE'S DEVELOPMENT OBLIGATIONS	
7.	7.1.	Computer Equipment and Telephones.	
	7.2.		
	7.3.	*	
	7.3. 7.4.	Active Operations.	
Q		ELOPMENT ASSISTANCE	
0.	8.1.	AHF's Development Assistance.	
9.		RATIONS MANUAL	
9.	9.1.	Operations Manual	<u></u> 10
	9.1.	Confidentiality of Operations Manual Contents.	10
	9.2.	Changes to Operations Manual Contents. Changes to Operations Manual Contents.	10
10	9.5.	RATING ASSISTANCE	1U
10	10.1	ALIE'- Assistance	LI
		AHF's Available Services.	
4.4	10.2.	Additional AHF Services. NCHISEE'S OPERATIONAL COVENANTS	<u></u> 12
Ш			
		Business Operations.	
	<u>11.2.</u>		
10	11.3.	New Products or Services.	18
12		TINUING FEES AND PAYMENTS	
		Royalty Fee.	
		National Brand Fees.	
	12.3.	Gross Revenues Defined.	
		Minimum Annual Gross Revenues.	<u></u> 19
	12.5.		<u></u> 20
	<u>12.6.</u>		
	12.7.	Late Fees; Denied ACH Processing; Interest.	
	12.8.	Nonrefundable Fees.	
<u>13</u>		ERTISING	
	13.1.		
	13.2.	Initial Marketing Campaign.	
	13.3.	Minimum Individual Marketing.	
		Local Advertising Group.	
		National Brand Fund.	
		Franchisee Advisory Council	
<u>14</u>		LITY CONTROL	
	14.1.	Standards and Specifications.	
	14.2.	Restrictions on Services and Materials.	
	14.3.	Commercial Quality Service Program.	
<u>15</u>		DEMARKS, TRADE NAMES AND PROPRIETARY INTERESTS	
		Marks.	
	15.2.	No Use of Other Marks.	<u></u> 27
	15.3.	Licensed Methods.	<u></u> 27
	15.4.	AHF's Rights to New Ideas.	<u></u> 27

		Copyrights.	
		Infringement.	
	15.7.	Franchisee's Business Name.	<u></u> 28
		Change of Marks	
	15.9.	Data and Other Business Records.	<u></u> 29
16.		ORTS, RECORDS AND FINANCIAL STATEMENTS	
	16.1.	Franchisee Reports	<u></u> 29
	16.2.	Books and Records.	30
		Audit of Books and Records.	
	16.4.	Failure to Comply with Reporting Requirements.	31
	16.5.	Financial Information from Third Parties.	31
17.	TRA	NSFER	32
	17.1.	Transfer by AHF.	32
	17.2.	Transfer by Franchisee.	32
	17.3.	Pre-Conditions to Franchisee's Transfer.	32
		Franchisee's Death or Disability.	
	17.5.	AHF's Right of First Refusal.	34
	17.6.	Post-Transfer Obligations.	34
18.	DEF	AULT AND TERMINATION	34
	18.1.	Termination by AHF–Effective Upon Notice.	34
		Termination by AHF with Prior Notice.	
	18.3.	Termination by AHF for Continuing Force Majeure Events.	38
	18.4.	AHF's Right to Suspend Services on Franchisee's Default.	38
	18.5.	Termination by Franchisee.	38
	18.6.	Obligations of Franchisee Upon Termination or Expiration.	38
	18.7.	Franchisee's Failure to Comply With Post-Termination Obligations.	39
		Effects of Termination or Expiration.	
	18.9.	Outstanding Loan Obligations.	40
	18.10.	Conflicting Laws.	40
	18.11.	Remedies Cumulative.	40
19.	BUSI	NESS RELATIONSHIP	40
	19.1.	Business Relationship.	40
	19.2.	Third Party Obligations.	41
	19.3.	Indemnification.	41
20.	REST	FRICTIVE COVENANTS	41
	20.1.	Non-Competition During Term.	41
	20.2.	Post-Termination Covenant Not to Compete.	42
	20.3.	No Diversion of Business.	42
	20.4.	Non-Disparagement.	42
	20.5.	Confidentiality of Proprietary Information.	43
	20.6.		
	20.7.		
21.	INSU	RANCE	43
	21.1.		
	21.2.	Proof of Insurance.	44
	21.3.	Failure to Maintain Insurance.	44
	21.4.	Contractor and Subcontractor Insurance.	45
22.	ARB	ITRATION	
		Arbitration.	
		Arbitration Award.	
		Limitations on Proceedings.	
		Injunctive Relief.	
	22.5.		
		No Punitive or Consequential Damages	
		No Recourse Against Others.	
23.		CELLANEOUS PROVISIONS	47

23.1.	Modification.	<u></u> 47
23.2.	Entire Agreement.	48
23.3.	Delegation by AHF.	
23.4.	Consent; Business Judgment.	
23.5.	General Economic Conditions.	
23.6.	Effectiveness.	
23.7.	Limitation on Actions.	
	Review of Agreement.	
	Attorneys' Fees.	
	No Waiver.	
	No Right to Set Off.	
	Survival of Terms.	
	Invalidity	
	Notices.	
23.15.	Force Majeure.	51
	Estoppel Certificates.	
23.17.	Charges and Taxes.	51
23.18.	Cross-Default and Cross Termination Provisions.	51
23.19.	Manner of Payment.	52
	Approval Within Territory.	
	Translation of Agreement.	
	Communication with AHF	
	Incorporation of Riders.	
	Counterparts; Electronic Signatures.	
	Acknowledgement.	

EXHIBITS

- I. ADDENDUM TO FRANCHISE AGREEMENT
- II. COMMENCEMENT DATE SUPPLEMENT TO FRANCHISE AGREEMENT
- III. GUARANTY AND ASSUMPTION OF FRANCHISEE'S OBLIGATIONS
- IV. STATEMENT OF OWNERSHIP
- V. AUTHORIZATION AGREEMENT FOR DIRECT PAYMENT
- VI. COLLATERAL ASSIGNMENT OF TELEPHONE NUMBERS, TELEPHONE LISTINGS AND INTERNET ADDRESSES
- VII. RIDERS TO FRANCHISE AGREEMENT FOR SPECIFIC STATES OR PROVINCES

ACE HANDYMAN SERVICES® FRANCHISE AGREEMENT

THIS AGREEMENT ("Agreement") is made effective as of the Effective Date set forth on the signature page hereof, between ACE HANDYMAN FRANCHISING, INC., a Colorado corporation ("AHF"), and the franchisee named on the signature page of this Agreement ("Franchisee"), who, on the basis of the following understandings and in consideration of the following promises, agree as follows:

1. BACKGROUND AND PURPOSE

1.1. Background.

AHF has developed and owns a unique system for establishing, operating and marketing businesses that provide a variety of approved handyman, repair, and maintenance services for customers' homes and businesses ("AHS Businesses"), which AHS Businesses provide basic carpentry, plumbing and electrical, drywall, painting, flooring, landscaping and many other miscellaneous repairs and maintenance services (the "Services"), and sales of approved supplies, materials, equipment and other products (the "Products") under the mark ACE HANDYMAN SERVICES and related service marks, trademarks, logos and trade names (collectively the "Marks") and using AHF's unique system for operating the business and related licensed methods of doing business ("Licensed Methods").

1.2. System.

AHF grants the right and license to qualified individuals and entities to use the Marks and Licensed Methods to establish and operate AHS Businesses under its franchise system ("System").

1.3. Purpose.

Franchisee desires to establish an AHS Business and AHF desires to grant Franchisee the right and license to operate AHS Business under the terms and conditions contained in this Agreement.

2. GRANT OF FRANCHISE

2.1. Grant of Franchise.

AHF grants to Franchisee, and Franchisee accepts from AHF, (i) a non-exclusive right and license to open and operate an AHS Business, and (ii) a non-exclusive license to use the Marks solely in connection with the establishment and operation of an AHS Business. The grant hereunder is for the type of franchise set forth in the Addendum, as defined in **Section 2.2** below.

2.2. Scope of Franchise Operations.

Franchisee shall use its best efforts to promote the AHS Business. Franchisee agrees to use the Marks and Licensed Methods, as they may be changed, improved and further developed by AHF from time to time, only in accordance with the terms and conditions of this Agreement, including the Addendum to Franchise Agreement ("Addendum"), which is attached as Exhibit I,

and the other Exhibits to this Agreement, which are attached to, and incorporated into, this Agreement.

3. TERM AND EXPIRATION

3.1. Term.

The term of this Agreement is for a period of 10 years from the date this Agreement is executed by AHF, unless sooner terminated as provided herein. Franchisee agrees to operate the AHS Business for the entire term of this Agreement.

3.2. <u>Continuation</u>.

If Franchisee continues to operate the AHS Business with AHF's express or implied consent following the expiration or termination of this Agreement, the continuation will be on a month-to-month extension of this Agreement. This Agreement will then be terminable by either party on 30 days written notice. Otherwise, all provisions of this Agreement will apply while Franchisee continues to operate the AHS Business.

3.3. Successor Franchise.

Provided Franchisee is not in default at either the time of its notice of exercise of its successor franchise rights or at the time of the grant of the successor franchise rights, at the end of the term of this Agreement, Franchisee will have the option to renew its franchise rights for one additional term of 10 years by acquiring successor franchise rights, provided that Franchisee has met all of the following requirements:

- **a.** Franchisee executes AHF's then current form of Franchise Agreement, which may have terms substantially different than those set forth in this Agreement, within 30 days after Franchisee's receipt of such Franchise Agreement from AHF.
- **b.** Franchisee maintained compliance with all of the provisions of this Agreement during the term, including payment, on a timely basis, of all Royalty Fees, National Brand Fees, Software and Internet Fees, and other payments due hereunder. "**Compliance**" means, at a minimum, that Franchisee has not (i) failed to timely cure any breach of this Agreement specified by AHF in a written notice to Franchisee; or (ii) received any written notification from AHF of breach hereunder more than three times during the 24 month period prior to the expiration of the term of this Agreement, regardless of whether such breaches were timely cured.
- **c.** Franchisee satisfies the then current standards applicable to all new AHF franchisees.
- **d.** Franchisee maintains or modifies the AHS Business and its operations at Franchisee's sole expense (the necessity of which will be in the sole discretion of AHF) to conform to the then current Operations Manual (hereinafter defined).
- **e.** Franchisee executes a successor franchise rider in the form then in use by AHF, which, unless prohibited by law, includes a general release of any and all claims against AHF and its affiliates, and their respective officers, directors, employees and agents.

f. Franchisee pays to AHF a successor franchise fee in the amount of \$5,500, which is due and payable upon execution of AHF's then current Franchise Agreement and will be nonrefundable under all circumstances once paid. Except for the successor franchise fee described in the preceding sentence, an initial franchisee fee will not be charged upon execution of the successor Franchise Agreement and successor franchise rider.

3.4. Exercise of Option for Successor Franchise.

Franchisee may exercise its option for a successor franchise by giving written notice of such exercise to AHF not less than 120 days, but not more than one year, prior to the scheduled expiration of this Agreement. Franchisee's successor franchise rights will become effective upon Franchisee's compliance with **Section 3.3** above. AHF will provide Franchisee with copies of the then current Franchise Agreement and successor franchise rider within 30 days of Franchisee's notice of exercise of its option for a successor franchise. If Franchisee fails to execute and deliver the successor Franchise Agreement and successor franchise rider to AHF within 30 days after Franchisee's receipt thereof from AHF, then Franchisee shall be deemed to have irrevocably declined to exercise its option for a successor franchise and the Franchisee's option for a successor franchise shall terminate as of such date.

4. TERRITORY

4.1. Business Location.

Franchisee will operate its AHS Business from an office ("Business Location") outside of Franchisee's residence or the residence of its Principal Representative or Franchise Manager, as those terms are defined in Section 6.1. Franchisee is solely responsible for selecting and acquiring the premises for its Business Location. The Business Location must be centrally located in the Territory and meet AHF's current standards. Franchisee must obtain AHF's prior written consent to the Business Location. Unless extended by AHF, Franchisee must select and acquire a Business Location within 90 days following the Effective Date of this Agreement. The Business Location is designated in the Addendum. Franchisee may not commence operations of its AHS Business until such time as it has obtained an approved Business Location. The Business Location may not be relocated without the prior written consent of AHF. Franchisee agrees to comply with any additional standards established by AHF from time to time regarding Franchisee's Business Location within 30 days of receipt of written notice from AHF of such standards.

4.2. <u>Territory</u>.

a. AHF has designated a protected territory as set forth in the Addendum (the "Territory") in which AHF agrees it will not operate or franchise a third party to operate an AHS Business, except as set forth in Section 4.3 below. The Territory consists of one or more counties, cities, zip or postal code areas, street boundaries or other designated geographical boundaries as described in the Addendum, and shall contain approximately the number of households set forth in the Addendum. The Territory may not be changed or relocated without the prior written consent of AHF. The technical naming of the Territory will be provided by AHF and that into consideration digital and web marketing optimizations as well as existing and potential future names of adjacent territories. As long as Franchisee is not in default of this Agreement, AHF will not operate, or permit another franchisee of AHF to operate, an AHS Business within the Territory.

b. Franchisee shall limit the operation of its AHS Business to providing Products and Services to customers' homes and businesses located within the Territory. Notwithstanding the foregoing sentence to the contrary, Franchisee may be granted, at AHF's sole discretion, express written permission to sell Products or provide Services to customers located in an unsold territory adjacent to Franchisee's Territory (an "Adjacent Territory"), provided, that Franchisee agrees that when the Adjacent Territory is granted to another franchisee by AHF, Franchisee will, upon receipt of written notice from AHF, cease all of its sales and service efforts within the Adjacent Territory and deliver all customer and prospect lists relating to the Adjacent Territory to AHF within 10 business days of such notice. Franchisee is prohibited from using other channels of distribution, such as the Internet, catalog sales, telemarketing, or other direct marketing, to make sales outside of the Territory, without AHF's prior written consent.

4.3. Failure to Respond to Referrals.

If AHF or another AHF franchisee operating outside of the Territory refers a prospective customer to Franchisee, Franchisee is required to contact that prospective customer within 24 hours of receiving the prospective customer's contact information from AHF or the other franchisee, as applicable. If Franchisee fails to contact the prospective customer within that time period, AHF, in its sole discretion, shall have the right to assign another franchisee the right to contact the prospective customer and perform the services initially requested and all services later requested by the same prospective customer in the future, regardless of the fact that the prospective customer is located within Franchisee's Territory.

4.4. AHF's Reservation of Rights.

Franchisee understands and agrees that AHF has and retains the rights under this Agreement to: (a) develop and establish other franchise systems for similar, the same, or different products or services utilizing proprietary marks not now or hereafter designated as part of the System licensed by this Agreement, and to grant licenses for these other franchise or license systems without providing Franchisee with any rights therein; (b) open, operate, sell, manage and/or franchise AHS Businesses outside the Territory; (c) use, and to license others to use, the Marks and Licensed Methods for the operation of AHS Businesses at any location other than in the Territory; (d) use the Marks and the Licensed Methods in connection with the provision of other services and products or in alternative channels of distribution (including, but not limited to, by Internet, catalog, mail order, telemarketing or other direct marketing sales), without regard to location; (e) pursue any other business concept, or the distribution of proprietary equipment and supplies, if any, to wholesalers or other distribution outlets (other than AHS Businesses) or by Internet, catalog, mail order, telemarketing, other direct marketing sales or otherwise, whether inside or outside the Territory; and (f) operate any Internet sites utilizing a domain name incorporating one or more of the words "Ace Handyman Services," "AHS," "Ace Hardware Handyman," "Ace Home Services," "Handyman Matters," or any other derivative using "Ace" or "Handyman," all without granting Franchisee any rights thereto. AHF is not required to compensate Franchisee for soliciting or accepting orders inside Franchisee's Territory. AHF can develop, acquire or be acquired by, or engage in any other transaction with other businesses, companies and/or units (competitive or not) that are or will be located anywhere, including arrangements where other units are (or are not) converted to the System or other format, or in which company-owned, franchised or other businesses (including AHS Businesses) are (or are not) converted to another format (whether competitive or not), or both, and is maintained as the same concept, as a new concept, or as a separate concept in the Territory. Franchisee must fully cooperate with any of these conversions, at Franchisee's sole expense. Franchisee further acknowledges and agrees that AHF's ultimate parent, Ace Hardware Corporation, or any other

company affiliated with it other than AHF, may use trademarks or service marks similar to the Marks and methods of doing business that are similar to the System and Licensed Methods, in connection with providing or franchising others to provide services and products that may be similar to or include some of the services Franchisee may offer in its AHS Business, including, by way of example but not by limitation, stand alone "Ace Painting" or "Ace Plumbing" businesses, inside or outside of the Territory, and that such use does not violate the protected territory rights of Franchisee.

5. INITIAL FEES

5.1. Initial Franchise Fee.

Franchisee will pay AHF an initial franchise fee in the amount set forth in the Addendum (the "Initial Franchise Fee"), which shall be due and payable as set forth in the Addendum. Franchisee acknowledges and agrees that the Initial Franchise Fee represents payment for the initial grant of the rights to use the Marks and Licensed Methods, that AHF has earned the Initial Franchise Fee upon receipt thereof, and that the Initial Franchise Fee is not refundable to Franchisee once paid.

6. TRAINING

6.1. <u>Initial Training Program.</u>

Franchisee (or if Franchisee is an entity, a managing partner, member or shareholder consented to by AHF (a "Principal Representative")), or a fully trained and qualified operations manager designated by Franchisee in writing to manage the AHS Business on Franchisee's behalf (the "Franchise Manager"), if applicable, plus a qualified office manager that Franchisee is required to employ (the "Office Manager"), shall attend and satisfactorily complete the initial business training program (the "Initial Training Program") prior to opening Franchisee's AHS The Initial Training Program may be conducted in-person or virtually via video Business. Prior to attending the initial business training, Franchisee shall complete the onboarding work and study and Franchisee or its Franchise Manager must obtain the LRRP Certification discussed in Section 11.1.1 for its AHS Business. Franchisee shall be responsible for all wages, travel and living expenses incurred by itself, its Principal Representative, the Franchise Manager, the Office Manager, and for any other employee attending the Initial Training Program. AHF will allow three employees of Franchisee, in addition to Franchisee or its Principal Representative, which may include the Franchise Manager and Office Manager, if applicable, to attend the Initial Training Program without tuition charged. AHF may charge for any additional employees of Franchisee who attend the Initial Training Program. If, under the terms of this Agreement or otherwise, Franchisee needs or desires to have persons attend the Initial Training Program after Franchisee's attendance of the Initial Training Program, then Franchisee must pay the then current tuition charged by AHF for those persons, in addition to all wages, travel and living expenses incurred in connection with their attendance at the Initial Training Program.

6.2. Length of Training.

The onboarding work and study is an approximately 10 to 12 week program, which AHF may adjust at its discretion. The Initial Training Program shall be conducted within 90 days after the full execution of this Agreement, unless Franchisee and AHF agree to a later time, and approximately 30 days prior to the opening of Franchisee's AHS Business. The Initial Training Program will consist of approximately 20 to 29 hours of instruction to be conducted over three

business days at AHF's Support Center facilities in the Denver, Colorado metropolitan area, another location designated by AHF, and/or by video conference. AHF reserves the right to waive all or a portion of the Initial Training Program or to alter the training schedule, if in AHF's sole discretion, Franchisee, its Principal Representative, its Franchise Manager, or its Office Manager, as applicable, has sufficient prior experience.

6.3. On-Site Training.

Franchisee, Franchisee's Principal Representative, or the Franchise Manager, as applicable, and the Office Manager, shall also complete approximately 18 to 24 hours of on-the-job training at the Business Location within the first 90 days of Franchisee's operation of the AHS Business, to be conducted over two to three business days ("On-Site Training"). In AHF's sole discretion, the On-Site Training may be divided into two separate training sessions held at different times. Attendance at the On-Site Training is mandatory for Franchisee, its Principal Representative, or the Franchise Manager or its Office Manager, as applicable. Other employees as Franchisee desires may also attend such training.

6.4. Additional Training.

- a. If AHF determines at any time during the term of this Agreement that Franchisee, its Principal Representative, its Franchise Manager, or its Office Manager requires training in addition to the Initial Training Program and On-Site Training, then AHF will provide notice to Franchisee of such necessary additional training, and AHF will conduct such additional training program(s) at a location designated by AHF. Franchisee will be responsible for paying the travel, lodging and other costs for Franchisee or its Principal Representative, Franchise Manager, Office Manager, or other representatives, and shall pay AHF its standard fees for conducting additional training in accordance with AHF's standard fee schedule, which fees may be changed from time to time upon 30 days' notice to Franchisee.
- b. From time to time, AHF may conduct additional meetings, seminars, conventions, and training programs for the benefit of all franchisees of AHF. Franchisee or its representatives may attend such programs at their own expense and shall attend such programs for which AHF has determined that Franchisee's attendance is required. AHF will not require Franchisee to attend national training programs more often than once every 24 months; provided, however, that this limitation shall not include annual conventions or regional training programs or meetings. AHF will give Franchisee at least 30 days prior written notice of any ongoing seminar, convention, program or meeting being held at which Franchisee's attendance is required. Franchisee or its representatives as designated by AHF will be required to attend AHF's annual convention, which may be held at AHF's Denver, Colorado metropolitan area headquarters or at another location selected by AHF. Franchisee and its representatives will be required to stay at the hotel where the primary functions are held, unless AHF, in its sole discretion, otherwise agrees. Any such meeting, seminar, convention, or training program may be held virtually via video conference.
- c. AHF will not charge a fee or its costs for any additional mandatory meeting, training program, seminar or convention. However, unless AHF agrees otherwise, if Franchisee, its Principal Representative, its Franchise Manager, or its Office Manager fails to attend the applicable mandatory training meeting, training program, seminar or convention, Franchisee will be required to pay a no show fee of \$1,500 per missed event. If Franchisee does not stay for the entire function or does not stay in the designated hotel, the no show fee shall be reduced by 50 percent. Any fees due to missing all or some of a training program, seminar, regional meeting or annual convention or for not staying at the hotel where the primary functions are held, shall be

invoiced to Franchisee within 15 days of completion of the applicable program, seminar, meeting or convention and shall be due within 15 days thereafter. If Franchisee or its applicable representative fails to attend a program at which attendance is deemed mandatory by AHF, AHF may, at its sole discretion, and without waiving any other rights AHF may have hereunder, require Franchisee or its applicable representative to attend and complete a make-up or alternative program at a location determined by AHF. Franchisee will be responsible for paying all wages, travel, lodging and other costs, and for paying Franchisee's registration fee for the program.

d. Franchisee will also be responsible for all wages, travel and living expenses associated with the attendance of Franchisee, its Principal Representative, its Franchise Manager, its Office Manager, as applicable, and all other persons associated with Franchisee at each training program, seminar, regional meeting or annual convention.

6.5. Release.

Franchisee, for itself and its agents, heirs, legal representatives, successors and assigns, forever releases, waives, discharges and holds AHF and any of its affiliated companies, directors, officers, employees and agents harmless from any and all claims, demands, causes of actions, loss, damage or injury, including attorneys' fees and costs, on account of, arising out of or attributable to Franchisee's attendance at or participation in any seminar, convention, program or meeting, or other company function or activity, including but not limited to the Initial Training Program, the On-Site Training, any regional meetings, other required or non-required training programs, or annual conventions, held or sponsored by or for AHF or the travel to or from such programs.

7. FRANCHISEE'S DEVELOPMENT OBLIGATIONS

7.1. Computer Equipment and Telephones.

Franchisee is required to own and operate a computer with software meeting AHF's minimum specifications in the operation of its AHS Business. AHF's minimum specifications currently require a PC computer running Windows 11 or higher as its operating system (although the operating system must be supported by Microsoft and Franchisee will need to upgrade the operating system to a higher system once its system is no longer supported) and a high speed Internet connection, with the Microsoft Office suite standard and QuickBooks online version software. Franchisee is also required to own and use a smart phone or other device agreed to in writing by AHF meeting AHF's standards and specifications that can and will be kept on and used to receive and send e-mails throughout the day from any location within the Territory, including job sites. AHF will provide an e-mail account to Franchisee. Franchisee is required to use only the e-mail account provided by AHF in the operation of its AHS Business, and Franchisee may not use the e-mail account for any purpose not related to the operation of its AHS Business. Franchisee is required to check such e-mail account throughout the business day using the required mobile device for receiving and sending e-mail messages promptly. AHF reserves the right to discontinue providing the e-mail account in the future, in which event Franchisee will be required to maintain an e-mail account to be used in the operation of the AHS Business. Franchisee shall at all times utilize AHF's required software (including AHF's scheduling software) in the operation of its AHS Business. Franchisee shall purchase, at Franchisee's sole cost and expense, all required third-party software and hardware upgrades and shall immediately install the software upgrades on all computer equipment used by Franchisee in operating its AHS Business. Franchisee shall cause Franchisee's employees to have smartphones and or tablets for field operations to operate the scheduling software. If the employees are using personal devices to interact with the scheduling software, Franchisee shall be responsible for ensuring the software and application versions are

current. Franchisee shall add, eliminate, substitute or modify any computer equipment, software, telephones, and other similar equipment upon notice from AHF of changes in AHF's specifications and requirements.

7.2. Authorized Representatives and Independent Contractors.

- Franchisee will be solely responsible for recruiting, appointing, hiring, firing, and a. supervising its employees, agents, Principal Representatives, Franchise Managers, and other authorized representatives of the AHS Business (collectively referred to as "Authorized Representatives"), including its Office Manager who will assist in the answering of its phones, scheduling of jobs with customers, assist with employment applications, paperwork check-in other customer service, and otherwise assist with running the office for Franchisee's AHS Business. These Authorized Representatives will be employees or agents of Franchisee. They are not employees or agents of AHF and AHF is not the joint employer of those persons. Franchisee will have sole authority and control over the day-to-day operations of the AHS Business and its Authorized Representatives. AHF will have no right or obligation to direct Franchisee's Authorized Representatives or to operate the AHS Business. It is Franchisee's responsibility to determine compensation of Authorized Representatives, terms of employment, safety regulations, work assignments, work schedules, and working conditions. Any information regarding any of those issues provided to Franchisee by AHF are mere suggestions and Franchisee shall have the sole discretion to utilize such information or not. Franchisee is solely responsible for implementing training and other programs for employees and other Authorized Representatives related to the legal, safe, and proper performance of their work, regardless of the fact that AHF may provide advice, suggestions, and certain training programs as described in this Agreement. Such advice, suggestions, and training by AHF are provided to protect AHF's brand and the Marks and not to control the day-to-day operation of Franchisee's AHS Business. Franchisee will keep AHF informed of the names, addresses and telephone numbers of all Authorized Representatives. Franchisee shall cause each person holding an ownership interest in Franchisee, if Franchisee is an entity, to execute AHF's standard Nondisclosure and Noncompetition Agreement.
- **b.** Franchisee has sole authority and control over the decision of whether or not to use independent contractors, although AHF does not recommend independent contractors exceed 40 percent of Franchisee's work force. Franchisee will be solely responsible and liable for its decision to use independent contractors, any liability arising from the acts or omissions of its independent contractors, and for complying with all laws, rules and regulations related to the engagement and use of independent contractors in its AHS Business.
- c. Franchisee must conduct a background check of all employees, contractors, and subcontractors hired or engaged by it and utilized in its AHS Business and System including, but not limited to, its Authorized Representatives. In conducting these background checks, Franchisee must abide by all applicable laws, rules and regulations, including, but not limited to, the Fair Credit Reporting Act (and the state equivalents, if any) and/or any employment laws, rules, regulations or ordinances. It is Franchisee's responsibility to make sure that no employee, contractor, or subcontractor enters a customer's home or business if such person has not successfully completed the background check process, as determined in Franchisee's sole discretion.
- **d.** Franchisee is solely responsible for implementing training and other programs for its employees, contractors and other Authorized Representatives related to the legal, safe, and proper performance of their work, regardless of the fact that AHF may provide advice, suggestions, and certain training programs. The advice, suggestions, and training AHF provides

are to protect its brand and the Marks and not to control the day-to-day operation of Franchisee's AHS Business.

7.3. Commencement of Operations.

Unless otherwise agreed to in writing by AHF and Franchisee, Franchisee shall be deemed to have commenced operations on the date that Franchisee books its first job, but in no event later than the date set forth in the Addendum, which shall be 60 days after Franchisee, it Principal Representative, or the Franchise Manager, as applicable, have successfully completed the onboarding study and Initial Training Program.

7.4. <u>Active Operations</u>.

Unless otherwise agreed in writing by AHF and Franchisee, once Franchisee has commenced operations as specified above, Franchisee must actively promote and continue to operate its AHS Business in accordance with the Operations Manual (as defined below) and this Agreement; unless AHF gives its prior written consent to Franchisee to temporarily suspend its operations, which consent may be withheld by AHF for any reason.

8. DEVELOPMENT ASSISTANCE

8.1. AHF's Development Assistance.

Prior to or simultaneously with the opening of the Franchisee's AHS Business, AHF or its designee will provide Franchisee with the following assistance:

- **a.** Designate Franchisee's protected Territory.
- **b.** Provide Franchisee with access to AHF's scheduling software, an e-mail account, an Internet site for prospective customers, and an Intranet site for Franchisee and other franchisees of AHF. AHF reserves the right to discontinue providing the e-mail account and Intranet site at any time in its sole discretion, if technology advancements or solutions make these obsolete or impractical to provide or take the place of these tools. Franchisee acknowledges that the Internet site will also include information for prospective franchisees regarding purchasing an AHS Business franchise. Franchisee must purchase additional required software from suppliers selected by Franchisee.
- **c.** Provide Franchisee with the specifications of all initial and replacement equipment, inventory and supplies required for the operation of its AHS Business, all as specified in the Operations Manual, and a list of all approved suppliers of any Services, Products, equipment, inventory, supplies and other materials that AHF requires Franchise to use.
- **d.** Provide Franchisee with pre-opening and opening activities, conducted as determined by AHF and just before and including the first week of operation of Franchisee's AHS Business. These activities may include providing Franchisee with a list of information about tasks Franchisee will need to complete to establish its business, assisting Franchisee in completing those tasks and assigning a field support TEAM member to assist Franchisee with completion of the startup list.
- **e.** Provide Franchisee access to AHF's proprietary project management start-up onboarding program.

- **f.** Make available to Franchisee one copy of AHF's Operations Manual in accordance with **Article 9** below.
- **g.** Provide the initial business training in accordance with **Sections 6.1** and **6.2** of this Agreement.

9. OPERATIONS MANUAL

9.1. Operations Manual.

AHF agrees to make available to Franchisee one or more books, manuals, technical bulletins, memoranda or other supplemental written materials, in whatever form (including electronic form), prepared by or on behalf of AHF for use by franchisees generally or for Franchisee in particular (all referred to in this Agreement as the "**Operations Manual**") covering the proper operating and marketing techniques of Franchisee's AHS Business. The Operations Manual is designed to protect AHF's reputation and the goodwill of the Marks, it is not designed to control the day-to-day operations of Franchisee's AHS Business. AHF may modify the Operations Manual at AHF's discretion. At AHF's sole discretion, it may make the Operations Manual, or any updates or revisions thereto, available online. Franchisee agrees that it will comply with the Operations Manual as an essential aspect of its obligations under this Agreement.

9.2. <u>Confidentiality of Operations Manual Contents.</u>

Franchisee agrees to use the Marks and Licensed Methods only as specified in the Operations Manual. The Operations Manual is the sole property of AHF, and will be used by Franchisee only during the term of this Agreement and in strict accordance with the terms and conditions hereof. Franchisee will not duplicate the Operations Manual or written materials considered by AHF to be a component of the Operations Manual, nor disclose the contents thereof to persons other than those persons who have an obligation of confidentiality to Franchisee. Franchisee will return the Operations Manual to AHF upon the expiration, termination or assignment of this Agreement, at Franchisee's sole expense.

9.3. Changes to Operations Manual.

AHF reserves the right to revise the Operations Manual from time to time as it deems necessary. Franchisee, within five business days of receiving any updated information or electronic notifications of such updated information, will update its copy of the Operations Manual as instructed by AHF and will conform its operations to the updated provisions. A master copy of the Operations Manual maintained by AHF at its principal office, regardless if maintained electronically or otherwise, will be controlling in the event of a dispute regarding the content of any Operations Manual.

10. OPERATING ASSISTANCE

10.1. AHF's Available Services.

AHF agrees that, during Franchisee's operation of the AHS Business, and provided Franchisee is in compliance with the terms of this Agreement, AHF or its designee will provide to Franchisee the following assistance and services:

- **a.** Conduct the On-Site Training in accordance with **Section 6.3** of this Agreement.
- **b.** From time to time, research new products, services and methods of doing business and provide Franchisee with information concerning developments of this research.
- **c.** Make a representative of AHF available to Franchisee on a reasonable basis via telephone, text, and/or e-mail during AHF's regular business hours to discuss Franchisee's operational questions and experiences.
- **d.** Provide regular communications on current best practices through telephone calls, webinars, video conferencing, e-mail and other electronic communications, internal blogs, and other means.
- **e.** Hold periodic webinars, regional meetings, and annual conferences to discuss sales techniques, new product developments, new service suggestions, accounting, inventory control, performance standards, advertising programs, merchandising procedures and/or other topics deemed relevant by AHF in accordance with **Section 6.4.b** of this Agreement. Any meeting may be held virtually via video conference.
- **f.** Provide Franchisee and its customers periodic newsletters that AHF produces from time to time.
- **g.** If AHF determines that Franchisee requires additional training, conduct the additional training programs in accordance with **Section 6.4.a** of this Agreement.
- **h.** Make available to Franchisee all marketing programs, advertising designs, advertising campaigns, and camera-ready artwork. AHF may charge a fee for these materials.
- In providing standards, specifications, processes, procedures, requirements or i. instructions applicable to Franchisee, assist in local market research and provide guidance and assistance to Franchisee in recommending the prices to be charged by Franchisee for the Services and Products provided by Franchisee. Unless prohibited by applicable law, AHF may periodically set maximum, minimum, or set prices that Franchisee may advertise and charge for Services and Products offered by its AHS Business. If AHF establishes a maximum price for any Services or Products, Franchisee shall not offer or sell those Services or Products at any greater price. If AHF establishes a minimum price for any Services or Products, Franchisee shall not offer or sell those Services or Products at any lesser price. If AHF establishes a set price for any Services or Products, Franchisee shall offer or sell those Services or Products only at that price. If AHF does not establish pricing limits, it may establish suggested prices. In that case, any prices that AHF recommends to Franchisee are merely recommendations and Franchisee may establish its own prices, which may be higher or lower than AHF's recommended prices. Franchisee must abide by Franchisor's advertising policies related to advertising prices. AHF shall have no control over the day-to-day managerial operations of Franchisee's AHS Business.

j. Provide Franchisee with access to AHF's scheduling software and all upgrades, an e-mail account, an Internet site for prospective customers, and an Intranet site for Franchisee and other franchisees of AHF, and to social media accounts for Franchisee's AHS Business that AHF deems relevant for the brand. Franchisee acknowledges that the Internet site will also include information for prospective franchisees regarding purchasing an AHS Business franchise. AHF reserves the right to discontinue providing the Intranet site at any time in its sole discretion.

10.2. Additional AHF Services.

Although not obligated to do so, upon the reasonable request of Franchisee, AHF may make its employees or designated agents available to Franchisee, either at AHF's office or within Franchisee's Territory, for additional advice and assistance in connection with the ongoing operation of the AHS Business governed by this Agreement. If Franchisee requests such additional assistance and AHF agrees to provide it, AHF reserves the right to charge Franchisee for all travel, lodging, living expenses, telephone charges and other identifiable expenses associated with such assistance, plus a fee based on the time spent by each employee on behalf of Franchisee. Any fee will be charged in accordance with the then current published fees being charged by AHF for such assistance.

11. FRANCHISEE'S OPERATIONAL COVENANTS

11.1. Business Operations.

Franchisee acknowledges that it is solely responsible for the successful operation of its AHS Business and that the continued successful operation thereof is, in part, dependent upon Franchisee's compliance with this Agreement and the Operations Manual. In addition to all other obligations contained in this Agreement and in the Operations Manual, Franchisee covenants that:

- **a.** Franchisee shall, consistent with the terms of this Agreement, diligently develop its AHS Business and use its best efforts to market and promote the required Products and Services in the Territory.
- **b.** Franchisee shall strictly comply with all present and future provisions of the Operations Manual and all other policies established by AHF from time to time.
- **c.** Franchisee, or the Franchise Manager, as applicable, shall devote full time and best efforts on a daily basis, in person, to the supervision and conduct of Franchisee's AHS Business, which supervision must not be less than 40 hours per week. Franchisee may only operate another business, including any e-commerce business which operates over the Internet, with AHF's prior approval, which approval may be granted or denied in AHF's sole and absolute discretion.
- **d.** Franchisee, the Franchisee's Principal Representative, or the Franchise Manager, as applicable, and the Office Manager, shall attend and complete the Initial Training Program and the On-Site Training referred to in **Article 6** above. Franchisee, the Franchisee's Principal Representative or the Franchise Manager, as applicable, and the Office Manager shall also attend subsequent mandatory training programs, demonstrations and seminars at locations as AHF may require. Franchisee shall be bound by the terms of **Article 6** of this Agreement in regards to all training programs.

- e. Franchisee shall not sell any service or product except the Services or Products, in conjunction with the operation of its AHS Business, unless Franchisee receives the prior written consent of AHF. For the avoidance of doubt, the Services offered by an AHS Business do not include advanced services in the categories of carpentry, plumbing and electrical, painting, heating, cooling, air conditioning, and other categories of services, as AHF may specify with greater detail from time to time.
- **f.** Franchisee shall only use in its AHS Business advertising and promotional material, services, equipment, supplies, and logoed clothing that meet AHF's standards and specifications. Advertising materials that are produced or approved by AHF for use by Franchisee may be used only in the manner and during the period specified by AHF.
- **g.** Franchisee shall purchase and maintain in good operating condition a computer and software meeting AHF's minimum specifications, a laser printer. Equipment, signs, Services, Products, supplies and other items must be added, eliminated, substituted and modified by Franchisee as soon as practicable in accordance with changes in AHF's specifications and requirements.
 - **h.** All employees must be in logoed clothing, clean, and neat in appearance.
- **i.** Franchisee shall not alter its AHS Business in any manner that materially affects the image of its AHS Business or the System, except at AHF's request or with AHF's written approval, and any alterations must strictly conform to the specifications and requirements established or approved by AHF.
- **j.** All Products and Services provided by Franchisee and its employees, contractors and subcontractors must comply with all applicable federal, state, provincial, or local electric, plumbing and building code laws, ordinances, rules, regulations and other requirements.
- **k.** Franchisee is required to purchase or lease a vehicle, which must be a truck or van, for every three craftsmen Franchisee employs working an average of 30 hours per week or more. For example, Franchisee must purchase one vehicle when it has three craftsmen working an average of 30 hours per week or more, two vehicles when it has six such craftsmen, three vehicles when it has nine such craftsmen, etc. The vehicles must be a truck or van and AHF recommends they be white, gray, red, or black, and have AHF's approved vehicle designs and include the AHF logo. AHF currently does not have standards regarding the particular make or model of any required vehicle nor does it have a preferred or designated supplier from which the vehicle must be purchased, but AHF reserves the right to establish such requirements in the future. AHF will provide Franchisee not less than 30 days' notice prior to imposing any such standards or requirements. AHF may require Franchisee to use a vehicle magnet with the AHF name or logo on any personal vehicle used in the AHS Business that is not one of the required vehicles under this Section.
- I. Franchisee must comply with the "Lead Renovation, Repair, and Painting" rule (40 CFR § 745.226 et seq.; the "LRRP Rule") of the U.S. Environmental Protection Agency ("EPA"), which currently requires in part that Franchisee have both the AHS Business and at least one representative of the AHS Business certified by the EPA under the LRRP Rule (the "LRRP Certification"). Franchisee must obtain the LRRP Certification for the AHS Business and its representative prior to the date of the Initial Training Program provided pursuant to Section 6.1. Franchisee must maintain the LRRP Certification for the AHS Business and at least one individual representative at all times during the term of this Agreement. Franchisee must provide evidence of

its current LRRP Certification to AHF upon initially obtaining the LRRP Certification and thereafter upon any recertification or request of AHF.

- **m.** The number and type of employees and contractors, as well as the amount and type of equipment, supplies, inventory and other items on hand at Franchisee's AHS Business must be at all times sufficient to efficiently meet the anticipated volume of business.
- n. Franchisee shall at all times comply with, and cause its contractors and subcontractors to comply with, all applicable laws, regulations, and ordinances, and shall promptly obtain all licenses and permits, including contractor's licenses, required by any state, provincial, or local licensing authority in all jurisdictions in which Franchisee operates its AHS Business. AHF has no obligation to advise Franchisee of any legislative or other legal developments that may affect its AHS Business. Franchisee is solely responsible for inquiring about and becoming familiar with all applicable laws, regulations, and ordinances, and determining those actions required for compliance. Any information AHF provides to Franchisee regarding applicable laws, regulations, or ordinances does not relieve Franchisee of its responsibility to consult with its own legal advisor and otherwise take appropriate action to inquire about and comply with applicable laws, regulations, and ordinances.
- **o.** Franchisee shall pay when due all debts and taxes arising in connection with Franchisee's AHS Business, except those duly contested in a bona fide dispute.
- **p.** Franchisee shall provide to customers the one-year warranty required by AHF and as set forth in the Operations Manual. All dealings and transactions with customers and suppliers must be fair and honest.
- **q.** Franchisee shall at all times utilize AHF's scheduling software in the operation of its AHS Business. Franchisee shall use the scheduling software and no other software for purposes of scheduling, tracking customers, dispatching employees, and any other functions provided by the software. Franchisee shall purchase, at Franchisee's sole cost and expense, all required third-party software and hardware upgrades and shall immediately install the software upgrades on all computer equipment used by Franchisee in operating its AHS Business.
- r. Franchisee will obtain website development services for one or more local sites specific to Franchisee's AHS Business and other specified search engine optimization and online marketing services from AHF's designated suppliers, and pay AHF the Software and Internet Fee defined in Section 12.5, as it may be changed from time to time, for these services. These services do not cover paid, placed media advertising, which is an additional expense paid to the vendor directly by Franchisee. Franchisee will also use the AHF Intranet site in accordance with AHF's specifications and the requirements of the System. Franchisee will enter into any agreements required by AHF's designated suppliers to obtain the services provided by the applicable supplier.
- s. Franchisee shall use only the e-mail account provided by AHF in the operation of its AHS Business, and Franchisee may not use the e-mail account for any purpose not related to the operation of its AHS Business. Franchisee shall check such e-mail account at least once per business day and respond to e-mail messages (as appropriate) within one business day. AHF reserves the right to discontinue providing this e-mail account if technology advancements make this obsolete or impractical to provide, or charge a fee for providing the e-mail account. Franchisee shall have a high-speed Internet connection, if available.

- t. During the term of this Agreement and for three years after the expiration and termination of this Agreement, Franchisee shall notify AHF of any change to Franchisee's (or its Principal Representative's) home and business addresses and telephone numbers.
- **u.** If Franchisee is an individual, Franchisee or an individual designated by Franchisee and approved by AHF to be the Franchise Manager shall directly supervise his or her AHS Business. If Franchisee is a corporation or other business entity it shall appoint its Principal Representative to be the Franchisee Manager to directly supervise Franchisee's AHS Business. If, in AHF's sole judgment, Franchisee has insufficient experience in a business similar to an AHS Business or insufficient experience in business management in general to operate the AHS Business, Franchisee shall hire a Franchise Manager who shall have direct responsibility for all operations of Franchisee's AHS Business. Any change in the Franchise Manager shall be subject to the prior written approval of AHF, which approval may be granted or denied in AHF's sole and absolute discretion.
- v. Franchisee shall become a member of such franchise, trade or other associations or organizations that, in the opinion of AHF, are useful in the operation of an AHS Business. Franchisee shall have the option to become a member of all benefit programs that are offered from time to time by AHF to all of its franchisees, if any. The costs of participating in such franchise, trade or other associations and benefit programs shall be borne by Franchisee and its employees (if applicable to the employees).
- **w.** Franchisee will comply with all agreements with third parties related to its AHS Business.
- AHS Business authorized hereunder. Upon request of AHF, Franchisee will promptly provide satisfactory proof of such ownership to AHF. Franchisee represents that the Statement of Ownership, attached hereto as Exhibit IV and incorporated by this reference, is true, complete, accurate and not misleading; and, in accordance with the information contained in the Statement of Ownership, the controlling ownership of the AHS Business is held by Franchisee. Franchisee will promptly provide AHF with a written notification if the information contained in the Statement of Ownership changes at any time during the term of this Agreement and will comply with the applicable transfer provisions contained herein. If Franchisee is not an individual, an individual or individuals designated by AHF will execute the Guaranty and Assumption of Franchisee's Obligations attached hereto as Exhibit III and incorporated by this reference.
- y. Except as prohibited or limited by law, Franchisee shall fully participate in all promotional campaigns, prize contests, special offers, and other programs, whether national, regional, or local in nature (including the introduction of new Products or Services, or new franchises or other marketing programs directed or approved by AHF), which are prescribed from time to time by AHF. Franchisee shall be responsible for the costs of such participation. In addition, Franchisee shall honor any coupons, gift certificates or other authorized promotional offers of AHF at Franchisee's sole cost unless otherwise specified in writing by AHF. From time to time a promotion may not benefit all franchisees in the System; and if the promotion is not offered in the region, or another unknown hardship arises, AHF may, at AHF's option, exempt Franchisee and/or other franchisees on a case-by-case basis.
- **z.** Franchisee must offer, honor, and comply with the terms of any subscription package programs that AHF designates as mandatory for residential or commercial customers.

The terms of these subscription programs will be established by AHF, and may be changed, expanded, or discontinued from time to time.

- **aa.** Franchisee will at all times employ a qualified Office Manager to address all customer service concerns for Franchisee's AHS Business. The Office Manager must be appointed by Franchisee within 75 days of the execution of this Agreement. The Office Manager must complete the Initial Training Program and the On-Site Training as described in **Article 6**.
- **bb.** Franchisee will have its AHS Business' phone answered during regular business hours by a live person. Franchisee may use the Office Manager that it hires for this purpose, or it may enlist the services of an answering service.
- **cc.** Franchisee and its Principal Representative, Franchise Manager, Office Manager, employees, and contractors shall at all times present themselves in a professional manner to all customers or clients, and shall refrain from the use of illegal drugs or the consumption of alcohol during normal business hours or during any time that the same are performing services for clients or customers.
- **dd.** Franchisee shall not engage in any activities not covered by Franchisee's liability insurance or workers compensation insurance.
- **ee.** Franchisee shall not engage in any trade, practice or other activity that is harmful to AHF's goodwill or reflects unfavorably on AHF's reputation, or that constitutes deceptive or unfair competition.
- **ff.** Franchisee will comply with all laws prohibiting, or otherwise related to, bribery of or other illegal payments to any government, government agency, public international organization, or political party, or any of their officials, employees, candidates, or other representatives.
- gg. Franchisee must accept credit and debit cards from customers of its AHS Business. Franchisee shall not charge its customers any additional fees or service charges if they elect to pay by credit or debit card. The Payment Card Industry ("PCI") requires all companies that process, store, or transmit credit or debit card information to protect the cardholders' information by complying with the PCI Data Security Standard ("PCI DSS"). Therefore, Franchisee shall be PCI compliant by following and adhering to then-current PCI DSS, currently found at www.pcisecuritystandards.org, or any similar or subsequent standard for the protection of cardholder data throughout the term of this Agreement. Franchisee's AHS Business shall be in compliance with PCI DSS at all times.
- **hh.** Franchisee expressly authorizes AHF and its approved suppliers to contact Franchisee by e-mail, telephone, mail, or any other means related to any aspect of the AHS Business, authorized Products and Services, this Agreement, or the System, for so long as this Agreement remains in effect. Franchisee expressly authorizes AHF to disclose Franchisee's contact information to AHF's approved and designated suppliers to enable such suppliers to contact Franchisee. Franchisee acknowledges that these communications are necessary to facilitate and keep Franchisee updated regarding the ongoing franchise relationship.
- ii. Franchisee agrees that AHF will have the right to use the likeness, including photographs or videos containing images, of: (a) Franchisee; (b) if Franchisee is a business entity, Franchisee's owners; (c) Franchisee's employees; (d) the Business Location premises; and (e) any

operations of Franchisee's AHS Business; each for any purposes relating to the promotion or marketing of the System or Marks. Such likeness may be used in any and all publications or promotional materials of AHF, including but not limited to AHF's website, social media accounts, and printed and digital publications. Franchisee agrees and understands that any such image and resulting promotional material will become the property of AHF and will not be returned. Franchisee agrees and irrevocably authorizes AHF to edit, alter, copy, exhibit, publish or distribute any such images for any lawful purpose. Franchisee agrees and waives any rights to royalties or any other compensation related to our use of any such images. Franchisee agrees to hold harmless and forever discharge AHF from all claims, demands, and causes of action which Franchisee or its owners may have in connection with this authorization.

Representatives must conduct themselves in a dignified, respectful, courteous, and professional manner. No hostile, threatening, or offensive statements or behavior toward any AHF representative is permitted.

kk. Franchisee acknowledges and agrees that certain associations between its AHS Business, the Marks, and/or the System, on the one side, and a political (including a political party or candidate), religious, social, cultural or similar group, member, cause, and/or activities (collectively, a "Political or Social Cause"), on the other side, may result in adverse publicity and other adverse impacts on the Marks and/or System with certain segments of the public. As such, Franchisee agrees that it will not, without AHF's prior written consent, make any statement or take any other action (including posting signs) that could be perceived as promoting, approving, disapproving, publicizing, or otherwise commenting on a Political or Social Cause in any manner that is connected to or associated with the AHS Business, the Marks, and/or the System.

11.2. Requirements for Entity Franchisees.

If Franchisee is a corporation, partnership, limited liability company or other business entity, the following additional conditions must be met, along with any other conditions as may be established by AHF for entity franchisees:

- **a.** Contemporaneously with the business entity acquiring the franchise rights, thereafter upon the issuance or transfer of any ownership interests in the business entity and the appointment or election of any person as director, officer, member or manager of the business entity, and at any other time requested by AHF, the shareholders, members, partners, other owners, directors, officers, or managers (as applicable), and any other individuals designated by AHF will execute the Guaranty and Assumption of Franchisee's Obligations attached hereto as Exhibit III and incorporated herein by reference, personally guaranteeing full payment and performance of Franchisee's obligations to AHF and individually undertaking to be bound, jointly and severally, by all the terms of this Agreement.
- b. No shares in the capital of such corporation or other interest in the business entity shall be issued nor shall Franchisee directly or indirectly, voluntarily or involuntarily, by operation of law or otherwise, sell, assign, transfer, convey, donate, pledge, mortgage or otherwise encumber any such shares or offer or attempt to do so or permit the same to be done without AHF's prior written consent. Such actions shall be deemed a Transfer, as defined in **Section 17.2**, and subject to the requirements of **Article 17** below.

c. The business entity shall maintain stop transfer instructions against the Transfer of ownership on its records subject to the restrictions of this Agreement and shall have all outstanding certificates of ownership endorsed with the following legend printed conspicuously upon the face of each certificate:

The transfer of the shares represented by this certificate is subject to the terms and conditions of a certain Franchise Agreement with Ace Handyman Franchising, Inc..

d. The articles of incorporation or organization and by-laws, operating agreement or other governing documents of the business entity shall provide that its objectives or business is confined exclusively to the operation of the AHS Business as provided for in this Agreement, and recite that the issuance and Transfer of any ownership interest in the business entity is restricted by the terms of this Agreement. Copies thereof shall be furnished to AHF upon request.

11.3. New Products or Services.

On at least three months prior written notice, AHF may specify a new Service or Product as a required Service or Product in Franchisee's market area. The new Service or Product will not be deemed a required Service or Product if Franchisee demonstrates to AHF's satisfaction either of the following situations will exist:

- **a.** Franchisee will incur a substantial capital improvement not contemplated by this Agreement or in the Operations Manual, thereby resulting in a material hardship to Franchisee. For the purposes of this Subsection, the parties agree that a capital improvement in excess of 10 percent of the Franchisee's Gross Revenue for the prior year will be considered a substantial capital improvement; or
- **b.** Franchisee will incur a material reduction in sales or profitability therefrom. For the purposes of this Subsection, the parties agree that an expected 20 percent decrease in sales from the average sales in the prior 12 months will be considered a material reduction in sales, and a 10 percent reduction in profitability from the average profitability during the previous 12 months will be considered a material reduction in profitability.

Subject to the foregoing, any new or additional required Services or Products introduced into the System by AHF must be offered for sale on a continuing basis as part of Franchisee's AHS Business at the time and in the manner required by AHF, and all equipment, products, supplies and other items necessary to add new required Services or Products must be acquired, installed, and utilized as required by AHF.

12. CONTINUING FEES AND PAYMENTS

12.1. Royalty Fee.

Franchisee shall pay to AHF a continuing royalty fee equal to 6 percent of Franchisee's Gross Revenues, as defined below, subject to an annual reconciliation based on the applicable Minimum Annual Gross Revenues, as defined below (the "Royalty Fee"). Franchisee shall not subordinate to any other obligation its obligation to pay the Royalty Fee or any other fee or charge hereunder.

12.2. National Brand Fees.

Franchisee shall remit 2 percent of Franchisee's Gross Revenues, subject to an annual reconciliation based on the applicable Minimum Annual Gross Revenues (the "National Brand Fee") to the Ace Handyman Services National Brand Fund established by AHF for Franchisee's country (the "National Fund").

12.3. Gross Revenues Defined.

"Gross Revenues" means, except as set forth below, the total of all receipts derived from the operation of Franchisee's AHS Business, including all repair, maintenance, and other work done by employees, contractors or sub-contractors, whether the receipts are evidenced by cash, credit, or checks, or exchanged for services, materials, service charges, property or other means of exchange. This includes amounts collected for both labor and materials. Gross Revenue does not include the amount of any tax imposed by any federal, state, provincial, municipal or other governmental authority directly on sales and collected from customers, provided that the amount of any such tax is shown separately and in fact paid by Franchisee to the appropriate governmental authority. Gross Revenues also is net of any discounts given to customers, to the extent the customers do not pay such amounts to Franchisee. Gross Revenues shall be deemed received by Franchisee at the time the Services and Products from which they were derived are delivered or rendered or at the time the relevant sale takes place, whichever occurs first, regardless of whether final payment (e.g., collection on a customer's personal check) actually has been received by Franchisee; provided, however, if a job is partially completed at the end of any month, and the job represents less than 16 hours of labor, the Gross Revenues generated by the job will not have to be reported, and the Royalty Fee and National Brand Fee, as each are defined above, will not have to be paid, until the following month; provided further that any job which is scheduled to extend beyond the second month following commencement of the job will need to be settled with the customer once during each month, the Gross Revenues will have to be reported, and the Royalty Fee and National Brand Fee paid during each month in accordance with the payment terms set forth in Section 12.6.a. Gross Revenues consisting of property or services shall be valued at the retail prices applicable and in effect for the Products and Services at the time that they are received. Cash refunded and credit given to customers, and any receivable uncollected from customers more than 120 days after billing, may be deducted in computing Gross Revenues only to the extent that the receivable was previously included in Gross Revenues on which a Royalty Fee or National Brand Fee was paid.

12.4. Minimum Annual Gross Revenues.

Franchisee must achieve the Gross Revenues requirements set forth in the Addendum on an annual basis (the "Minimum Annual Gross Revenues"). The amount of the Minimum Annual Gross Revenues is determined based on the years of operations of Franchisee. The Minimum Annual Gross Revenues will not change if there is an increase or decrease in population in the Territory during the term of this Agreement. Within 60 days following the end of each Operating Year, as defined below, AHF will compute Franchisee's total Gross Revenues for the just completed Operating Year. If Franchisee fails to achieve the Minimum Annual Gross Revenues in any Operating Year, Franchisee shall pay AHF the Royalty Fee and National Brand Fee based on the Minimum Annual Gross Revenues instead of its actual Gross Revenues. AHF will invoice Franchisee for the difference between the amount of Royalty Fee and National Brand Fee due based on the Minimum Annual Gross Revenues, less the actual Royalty Fee and National Brand Fee paid by Franchisee to AHF during the just completed Operating Year, which amount shall be due within 15 days following invoicing. If Franchisee's actual Gross Revenues are less

than the applicable Minimum Annual Gross Revenue in two consecutive Operating Years, or in any three Operating Years during the term of this Agreement, then AHF shall have the right to (i) terminate this Agreement, (ii) award additional franchises within the Territory or otherwise permit another franchisee to operate within the Territory, (iii) reduce the geographic size of the Territory, or (iv) any combination thereof, after providing notice to Franchisee in accordance with Section 18.2.f below. AHF has the right, in its sole discretion and on a case-by-case basis, to waive the obligation of Franchisee or any other franchisee of AHF to meet the Minimum Annual Gross Revenues requirement or to pay any fees or make expenditures calculated based on the applicable Minimum Annual Gross Revenues. Franchisee acknowledges that AHF is entering into this Agreement with the expectation that it will receive Royalty Fees based on the greater of Franchisees actual Gross Revenues or the Minimum Annual Gross Revenues over the full term of this Agreement. The term "Operating Year" means a calendar year beginning on the commencement date set forth in the Addendum and ending on the anniversary date thereof, and each subsequent calendar year thereafter.

12.5. Software and Internet Fee.

Franchisee shall be assessed a monthly software and Internet fee ("Software and Internet Fee") at AHF's current rate for each month, which shall be set forth in the Operations Manual or in other written notices from AHF, and which may be changed from time to time in AHF's sole discretion upon 30 days prior written notice. The Software and Internet Fee is paid for use of AHF's scheduling software, e-mail, location website, and online marketing services, which includes website development services and search engine optimization. These services do not include paid, placed media advertising, which is an additional expense paid to the vendor directly by Franchisee.

12.6. Payments to AHF.

- a. Franchisee shall pay the Royalty Fees and National Brand Fee to AHF no later than the 23rd 18th day of each month based on the prior month's Gross Revenues, starting with the month following the month of Franchisee's first sale through its AHS Business, or at such other frequency as AHF may determine in its sole discretion upon written notice to Franchisee. The amount of such payments shall be determined based on the Gross Revenues of Franchisee as calculated based on the information provided through AHF's scheduling software over the applicable month. If any of the monthly, quarterly, or annual reports of Gross Revenues required to be submitted by Franchisee pursuant to **Section 16.1.a** below reveal an underpayment of Royalty Fees, National Brand Fee, or both, Franchisee will immediately pay to AHF the amount of such underpayment. If any such reports show an overpayment of Royalty Fees, National Brand Fee, or both, AHF will provide a credit to Franchisee's account in an amount equal to the amount of the overpayment.
- **b.** Franchisee shall pay the Software and Internet Fee once a month in advance on the first day of each calendar month, and as specified by AHF, beginning in the first full calendar month after Franchisee has commenced operations.
- c. Unless directed otherwise, Franchisee shall remit all fees and other amounts due to AHF hereunder via electronic funds transfer in accordance with the Authorization Agreement for Direct Payments, attached as Exhibit V to this Agreement. Franchisee agrees to comply with procedures specified by AHF and/or perform such acts and deliver and execute such documents, including authorization for direct debits from Franchisee's checking or savings accounts, as may be necessary to assist in or accomplish payment by such method. Under this procedure,

Franchisee shall authorize AHF to initiate debit entries and/or credit correction entries to the designated banking account for payments of fees and other amounts payable to AHF and any interest charge due thereon. Franchisee shall make funds available to AHF for withdrawal by electronic transfer no later than the due date for payment therefore, and shall at all times maintain a minimum balance of \$4,000 in the designated account, which shall be available to AHF to cover any required payments to AHF. Franchisee agrees that any time an electronic funds transfer transaction is not honored, Franchisee shall pay AHF a returned electronic funds transfer fee of \$100, in addition to any applicable late fees and interest in accordance with Section 12.7. If Franchisee has not timely reported the Gross Revenues to AHF or submitted information to AHF's scheduling software as required for any applicable period on which a fee is based, then AHF shall be authorized, at AHF's option, to debit Franchisee's account in an amount equal to (a) the fees transferred from Franchisee's account for the last applicable period for which a report of the Gross Revenues was provided to AHF or for which information was submitted through AHF's scheduling software as required hereunder; or (b) the amount due based on information retrieved from the computer system approved or required by AHF. Such payments shall be deemed the minimum amount of fees due for failure to timely remit all fees and file the required reports, and Franchisee shall remain liable for all fees in excess of such amounts once the actual Gross Revenues for the applicable period are determined. AHF may require Franchisee to pay by means other than electronic funds transfer from time to time, whenever AHF deems appropriate, and Franchisee agrees to comply with AHF's alternate payment instructions and to execute any documents in conjunction with such payment means.

d. If Franchisee is unable to operate due to damage or loss to the Business Location caused or created by a casualty, act of God, condemnation, or other condition over which Franchisee has no control, then AHF, in its sole discretion, may elect to waive the Royalty Fee for a period no greater than 120 days commencing with the month in which Franchisee gives AHF notice of the damage or loss.

12.7. Late Fees; Denied ACH Processing; Interest.

To encourage prompt payment and to cover the costs and expenses involved in handling and processing late payments or denied or rejected requests for ACH payments, Franchisee shall also pay to AHF, upon demand, a late payment charge of \$75 per overdue report or fee owed to AHF, and \$100 for each ACH payment request that is denied or rejected by Franchisee's bank or other financial institution, plus interest equal to the lesser of (i) the maximum legal rate of interest then charged on open accounts or (ii) 18 percent per annum, on all payments due to AHF during the period of time said payments are due and unpaid. This same interest rate shall apply as the post-judgment interest rate, regardless of the applicable statutory rate, in the event of any legal actions related to this Agreement.

12.8. Nonrefundable Fees.

Except as specifically contemplated in this Agreement, all fees once paid, shall be nonrefundable in all circumstances.

13. ADVERTISING

13.1. Approval of Advertising.

AHF will make available to Franchisee advertising and promotion materials and programs for its AHS Business that are developed by or for AHF. AHF will provide camera-ready artwork

to Franchisee. Franchisee may also develop advertising materials for Franchisee's own use, at Franchisee's own cost. Before using any promotional and advertising materials, Franchisee will submit to AHF or AHF's designated agency, for AHF's prior written approval, all information pertaining to such promotional materials and advertising developed by Franchisee; including, but not limited to, telephone related materials, print ads, radio and television scripts, Internet advertising, social media advertising, vehicle wraps or vehicle magnets, or any promotional creative materials. In the event written approval or disapproval of any such advertising and promotional material has not been given by AHF to Franchisee within seven business days from the date such information has been received by AHF, the materials will be deemed approved as submitted. Notwithstanding AHF's approval of any advertising created or used by Franchisee, Franchisee acknowledges and agrees that it is responsible for the accuracy of all advertising content used by Franchisee, and its obligation to comply with all of AHF's advertising standards and specifications and all applicable legal requirements relating to advertising and promotional material. In particular, AHF may require Franchisee to acquire some or all advertising services and products only from media or advertising agencies or other vendors designated as approved suppliers by AHF. Franchisee shall display all required promotional materials, signs and other marketing materials in its AHS Business in the manner prescribed by AHF. In all advertising and promotional materials, Franchisee must display its business name only in obvious conjunction with the phrase, "Locally Owned and Independently Operated Franchise of Ace Handyman Franchising, Inc." or similar statement approved by AHF.

13.2. <u>Initial Marketing Campaign</u>.

Franchisee acknowledges that local advertising is required to notify the public of its AHS Business. As such, Franchisee shall advertise and promote the opening of its AHS Business at Franchisee's own expense. However, AHF will recommend the type, manner, and time period for such advertisements, will designate as approved suppliers required or recommended media or advertising agencies or other vendors or suppliers for such advertisements, and will provide other assistance with Franchisee's initial marketing campaign. During the period beginning 30 days prior to opening Franchisee's AHS Business and ending at the end of Franchisee's second full month of operation, Franchisee shall spend a minimum of \$98,000.00 and a maximum of \$12,000.00 on advertising and promotion of its AHS Business, which amount shall be paid directly to vendors developing and providing the initial marketing campaign. In addition to the amount set forth in the preceding sentence, Franchisee is also required to spend between \$21,000.00 and \$4,000.00 for a required grand opening event meeting AHF's specifications.

13.3. Minimum Individual Marketing.

a. Beginning with the third full month of operation and during the remaining term of this Agreement, Franchisee shall spend an amount equal to or greater than the applicable "Minimum Individual Marketing Expenditure" on local marketing and promotion. The Minimum Individual Marketing Expenditure shall be in the amount set forth in the Addendum. These amounts are in addition to the National Brand Fee that Franchisee will pay to AHF. AHF will review Franchisee's spend amounts each six months and if Franchisee is not on pace to meet or exceed the required Minimum Individual Marketing Expenditure, Franchisee must increase its Minimum Individual Marketing Expenditure for the second six months of the applicable 12-month period to meet the required annual Minimum Individual Marketing Expenditure. Any expenditures on online marketing services and products acquired through a third party vendor shall count toward this Minimum Individual Marketing Expenditure. AHF may direct that Franchisee pay all

or a portion of its Minimum Individual Marketing Expenditure to a Local Advertising Group, as defined below.

- b. Franchisee shall submit documentation of Franchisee's marketing expenditures at such times and in such form as AHF designates. If Franchisee fails to spend at least the Minimum Individual Marketing Expenditure on local marketing and promotion for three consecutive months, AHF shall have the right to create a marketing plan on behalf of Franchisee, at the expense of Franchisee. This plan will remain in force for not less than six months at which time Franchisee may be permitted to resume its own marketing pursuant to the requirements set forth in this Agreement. AHF reserves the right to charge Franchisee for the creation and management of the plan. If Franchisee again fails to make any required marketing expenditures, AHF shall have the right to require Franchisee to contribute the amount of any deficiency to the National Fund to be used by AHF for general marketing and promotion or declare Franchisee in default of this Agreement.
- **c.** If there are Ace Suppliers, as defined in **Section 14.2** below, located in or near Franchisee's Territory, AHF encourages, but does not require, Franchisee to do promotions of its AHS Business through those suppliers' stores. Independent Ace Suppliers are not required to participate in any such promotions and there is no assurance they will do so.

13.4. Local Advertising Group.

- **a.** AHF may establish a regional advertising cooperative ("**Local Advertising Group**") in a region that includes Franchisee's Territory. If a Local Advertising Group is established that includes Franchisee's Territory, Franchisee shall join and participate in it. Each of AHF's company-owned and affiliate-owned operations (if any) offering Products and Services similar to an AHS Business within the region for which the Local Advertising Group is established will make contributions to the Local Advertising Group equivalent to the contributions required of Franchisee.
- **b.** If AHF directs that Franchisee join a Local Advertising Group, Franchisee will pay all or part of its Minimum Individual Marketing Expenditure to the Local Advertising Group as directed by AHF.
- c. The rules of the Local Advertising Group must be in writing and established by its members, but must be submitted for prior approval to AHF (and shall be deemed approved 30 days after submission if AHF takes no action). All Local Advertising Groups shall provide quarterly financial reports to AHF.

13.5. National Brand Fund.

- **a.** The National Brand Fee set forth in **Section 12.2** of this Agreement will be deposited in the National Fund. No action taken by Franchisee or any Local Advertising Group shall diminish Franchisee's obligation to pay the National Brand Fee to the National Fund.
- **b.** The creation of advertising materials and provision of other advertising and marketing services will be provided to Franchisee through the National Fund. The National Fund shall be maintained and administered by AHF or AHF's designee, as follows:
- (i) AHF or its designee, with recommendations from the Franchisee Advisory Council, as defined below, will oversee and direct all advertising programs and have sole

discretion over creative concepts, materials and media used in such programs, including the placement and allocation of advertising. AHF or its designee will use the National Fund to conduct system-wide advertising, and/or, if Local Advertising Groups are developed, to conduct regional or local advertising on Franchisee's behalf. However, AHF or its designee cannot and does not ensure that any particular franchisee will benefit directly or pro rata from the placement of advertising. AHF may charge Franchisee for any marketing materials created or produced for its franchisees. In that event, Franchisee shall pay AHF upon receipt for any marketing materials purchased from AHF unless AHF agrees to different terms.

- (ii) For each of AHF's company-owned and affiliate-owned operations (if any) offering Products and Services similar to an AHS Business, AHF will make contributions to the National Fund equivalent to the contributions required of Franchisee.
- (iii) AHF or its designee, with recommendation from the Franchisee Advisory Council, will administer and control the National Fund and will have the absolute and unilateral right to determine how, when and where the monies in the National Fund will be spent. The National Fund will be used to promote the Products and/or Services sold by franchisees and company-owned and affiliate owned operations, to provide media placement for national recruitment of craftsmen, and for such other purposes as determined in AHF's sole discretion to promote the System and brand, including participation in charitable events and programs. The National Fund will not be used to sell additional franchises, except that it may include a brief statement that Ace Handyman Services franchise locations are available and the contact information for acquiring information about Ace Handyman Services franchises. All sums paid by Franchisee to the National Fund shall not be used to defray any of AHF's general operating expenses.
- **(iv)** If AHF places advertising rather than engage an advertising agency for this purpose, AHF shall be entitled to receive a reasonable fee for such services, which will not exceed the highest rate charged for similar services by any recognized advertising agency not owned in whole or part by AHF or its officers, directors, or employees (in addition to reimbursement for costs incurred).
- (v) It is AHF's intent that all contributions to the National Fund will be expended for advertising and promotional purposes during AHF's fiscal year within which contributions are made. However, any monies not expended in the fiscal year in which they were contributed will be applied and used for National Fund expenses in the following year.
- (vi) Although AHF intends the National Fund to be of perpetual duration, AHF has the right to terminate the National Fund. AHF will not terminate the National Fund, however, until all monies in the National Fund have been expended for advertising and promotional purposes.
- **(vii)** An unaudited accounting of the National Fund contributions and expenditures will be prepared annually and will be made available to Franchisee upon request. At AHF's option, any such annual accounting may include an audit of the contributions and expenditures of the National Fund prepared by an independent certified public accountant selected by AHF and prepared at the expense of the National Fund.
- (viii) AHF has no fiduciary obligation to Franchisee in connection with the operation of the National Fund. AHF will not be liable for any act or omission with respect to the

operation of the National Fund or the use of the National Fund that is consistent with this Agreement and is done in good faith.

- (ix) AHF may maintain a separate National Brand Fund for its franchisees located in other countries. AHF reserves the right to combine the National Fund with the National Brand Fund for any other country, to be used to promote the Products and/or Services sold by franchisees on an international basis. The Franchisee Advisory Council may also be combined with the Franchisee Advisory Council established in any other country or region of the world. AHF has the right, in its sole discretion, to choose not to collect a National Brand Fee from franchisees in other countries or regions of the world, and to not maintain an advertising fund for those franchisees. AHF's choice to collect or to not collect a National Brand Fee or to establish or to not establish an advertising fund in other countries or regions of the world will not in any way affect Franchisee's obligation to pay the National Brand Fee.
- **c.** Once Franchisee makes contributions to the National Fund, all such monies will be used as required by this Section and will not be returned to Franchisee.

13.6. Franchisee Advisory Council.

AHF has established a council of current franchisees of AHF to advise and make suggestions to AHF regarding advertising issues or such other issues as AHF may authorize. AHF calls this council the "Franchisee Advisory Council." In AHF's sole discretion, AHF may establish separate Franchisee Advisory Councils for different countries, or different regions of the world. While AHF will consider all recommendations made by the Franchisee Advisory Council, AHF shall retain the right to direct and have sole discretion over such advertising or other decisions. AHF reserves the right to establish all rules and regulations related to the Franchisee Advisory Council, and may terminate or modify the format of the Franchisee Advisory Council, in AHF's sole discretion.

14. QUALITY CONTROL

14.1. Standards and Specifications.

AHF will make available to Franchisee, via the Operations Manual, standards and specifications for materials and services used by, Products sold through, and Services offered through, Franchisee's AHS Business, which standards and specifications AHF reserves the right to change upon 30 days prior written notice to Franchisee. These standards and specifications are designed to protect AHF's reputation and the goodwill of the Marks, they are not designed to control the day-to-day operations of Franchisee's AHS Business. AHF and AHF's representatives will have the right to discuss with Franchisee, or other personnel Franchisee may designate, including the Franchise Manager, all matters that may pertain to compliance with this Agreement and with AHF's standards, specifications, requirements, instructions and procedures. AHF may take photographs of Franchisee's completed work as it relates to its AHS Business. Franchisee shall in all respects cooperate with AHF's rights under this Agreement. AHF also reserves the right to contact any or all of Franchisee's customers, craftsmen, employees, suppliers and other service professionals for quality control, market research and such other purposes as AHF deems appropriate.

14.2. Restrictions on Services and Materials.

Except if AHF has designated or approved suppliers in the Operations Manual, Franchisee shall have the right to purchase directly from any supplier the equipment, services, and Products required for the operation of its AHS Business; provided that the equipment, services, and Products meet all applicable minimum specifications established by AHF. acknowledges that Ace Hardware Corporation and its network of member-retailers (collectively, the "Ace Suppliers") are approved suppliers of certain goods, services, supplies, fixtures, equipment, and inventory used in Franchisee's AHS Business. Franchisee is required to purchase those goods and services, as they are designated from time to time, from an Ace Supplier, only when an item is in stock or readily available, and competitively priced. AHF reserves the right to require approval of any or all of the Products, supplies, equipment, services, and other materials used in connection with Franchisee's AHS Business, or the suppliers of such materials or services. AHF may require that Franchisee submit specifications, information or samples of the items for AHF's review to determine if they meet AHF's specifications and standards set forth in the Operations Manual as to quality, content, composition and service. AHF does not currently require that a fee be paid to secure supplier approval; however, AHF may require a fee in the future, and in its sole discretion, AHF may require the supplier to reimburse AHF for any expenses AHF incurs in determining if the supplier meets its specifications and standards. AHF may revoke its approval of any supplier previously approved by written notice to the supplier and Franchisee, if Franchisee is using that supplier. AHF does, however, highly recommend that Franchisee purchase printed materials, windows, doors, and other specialty products identified from time to time from the suppliers recommended by AHF. Franchisee expressly acknowledges and agrees that AHF may derive revenue from Franchisee's purchases or leases of equipment from current or future suppliers, and may retain such revenue for AHF's use.

14.3. Commercial Quality Service Program.

Franchisee shall be required to participate in AHF's commercial quality service program. Under the commercial quality service program, if Franchisee performs work or services for a commercial customer (as opposed to a residential customer), then such work or services shall be subject to review by AHF, its representatives or the commercial client. If AHF determines, through such reviews, that Franchisee did not provide a satisfactory level and quality of service, in AHF's discretion, then AHF may appoint another franchisee to perform any future work or services for the subject commercial customer, regardless of the location of the commercial customer, or particular work site. In that event, AHF is authorized to grant such other franchisee the right to perform services to that commercial customer in Franchisee's Territory.

15. TRADEMARKS, TRADE NAMES AND PROPRIETARY INTERESTS

15.1. Marks.

Franchisee acknowledges that AHF's ultimate parent company, Ace Hardware Corporation ("Ace Hardware"), is the exclusive owner of the service mark, ACE HANDYMAN SERVICES and certain other Marks, and that Ace Hardware has granted AHF the exclusive license to use its Marks. Franchisee further acknowledges that Franchisee's right to use the Marks is derived solely from this Agreement and is limited to the operation of its AHS Business in accordance with this Agreement. Franchisee agrees that its usage of the Marks and any goodwill established thereby shall inure to the exclusive benefit of AHF. Franchisee shall not use the Marks in any manner calculated to represent that it is the owner of the Marks. Franchisee agrees not to contest or oppose, nor to assist anyone else to contest or oppose, AHF's application for, or

registration of, any of the Marks, or the validity or ownership of the Marks. Franchisee agrees not to directly or indirectly do or cause to be done, whether by commission or omission, any act, that may in any way jeopardize or adversely affect the validity or distinctiveness of the Marks, or the title of Ace Hardware and AHF thereto. Franchisee agrees that it will, without charge to AHF, upon request by AHF or its representatives, do all things and execute all documents that may at any time be necessary or desirable to protect or ensure the validity and distinctiveness of the Marks and to ensure the title of the Marks to Ace Hardware and AHF. Franchisee acknowledges that all references to AHF's Marks includes those Marks licensed to it by Ace Hardware.

15.2. No Use of Other Marks.

Franchisee agrees to use the mark "ACE HANDYMAN SERVICES" as the sole identification of its AHS Business. Franchisee agrees that it shall affix a notice in a conspicuous location in or upon the Business Location with content and format acceptable to AHF, that it is an independent Franchisee of AHF, and as such, an authorized user of the Marks, and that the owner of the Marks is AHF.

15.3. <u>Licensed Methods</u>.

Franchisee acknowledges that AHF owns and controls the distinctive plan for the establishment, operation and promotion of AHS Businesses and all related Licensed Methods. Franchisee acknowledges that much of the information contained in AHF's Operations Manual, and any other manual or nonpublic written information about AHF, and other confidential information provided to Franchisee by AHF, constitutes trade secrets of AHF. Franchisee acknowledges that AHF has valuable rights in and to such trade secrets. Franchisee further acknowledges that it has not acquired any right, title or interest in the Licensed Methods, except for the right to use the Licensed Methods in the operation of the AHS Business as it is governed by this Agreement.

15.4. AHF's Rights to New Ideas.

All enhancements and improvements in the Licensed Methods developed by Franchisee shall be and become the sole and absolute property of AHF. AHF may incorporate such improvements or enhancements into the Licensed Methods and shall have the sole and exclusive right to copyright, register or patent such improvements in AHF's own name and Franchisee shall have no right to use such enhancements and improvements, except as set forth in this Agreement. Franchisee shall promptly disclose all such enhancements and improvements to AHF (whether or not requested by AHF) in such detail as AHF may from time to time request. Franchisee shall, without further consideration, but at the expense of AHF, execute such documents and do such acts as may be necessary for AHF to copyright, register or patent the enhancements or improvements in AHF's own name in any country.

15.5. Copyrights.

Franchisee and AHF acknowledge and agree that: (a) AHF may authorize Franchisee to use certain copyrighted or copyrightable works (the "Copyrighted Works"); (b) the Copyrighted Works are the valuable property of AHF; and (c) Franchisee's rights to use the Copyrighted Works are granted to Franchisee solely on the condition that Franchisee complies with the terms of this Section. Franchisee acknowledges and agrees that AHF owns or is the licensee of the owner of the Copyrighted Works. Such Copyrighted Works include, but are not limited to, AHF's software programs (including the scheduling software program), the Operations Manual, advertisements,

and promotional materials, and may include all or part of the System, trade dress and other portions of an AHS Business. Franchisee acknowledges that this Agreement does not confer any interest in the Copyrighted Works upon Franchisee, other than the right to use them in the operation of its AHS Business in compliance with this Agreement. If AHF authorizes Franchisee to prepare any adaptation, translation or work derived from the Copyrighted Works, or if Franchisee prepares any Copyrighted Works such as advertisements, poster or promotional material, Franchisee agrees that such adaptation, translation, derivative work or Copyrighted Work shall constitute a "work made for hire" as that term is defined in the Copyright Act, 17 U.S.C. § 101 et seq., and shall become the property of AHF, and Franchisee assigns all its right, title and interest therein to AHF (or such other person or entity identified by AHF). Franchisee agrees to execute any documents, in recordable form, which AHF determines are necessary to reflect such ownership. Franchisee shall submit all such adaptations, translations, derivative works and Copyrighted Works to AHF for approval prior to use. Franchisee shall ensure that all Copyrighted Works used hereunder shall bear an appropriate copyright notice as specified by AHF and specifying that AHF is the owner of the copyrights therein.

15.6. <u>Infringement</u>.

Franchisee agrees to notify AHF in writing of any possible infringement or illegal use by others of a trademark, service mark, logo, or other commercial symbol that is the same as or confusingly similar to any of the Marks, or the Copyrighted Works that comes to its attention. Franchisee acknowledges that AHF will have the right, in its sole discretion, to determine whether any action will be taken on account of any possible infringement or illegal use. AHF may commence or prosecute such action in AHF's own name or may join Franchisee as a party to the action, in either event at AHF's expense, if AHF determines it to be necessary for the continued protection and quality control of the Marks, Licensed Methods or Copyrighted Works. Franchisee agrees that, without a fee or other charge to AHF, it shall fully cooperate and participate with AHF in any such litigation.

15.7. Franchisee's Business Name.

Franchisee acknowledges that AHF has a prior and superior claim to the Marks and AHF's corporate name and trade names. Franchisee will not use the designation "ACE HANDYMAN SERVICES," "ACE HARDWARE HANDYMAN SERVICES," "AHS," "ACE HOME SERVICES," "ACE HANDYMAN," "HANDYMAN MATTERS," or any portions thereof, or any other derivative using "Ace" in the legal name of its corporation, partnership or other business entity, nor use any of such names, the Marks or trade names, or portions thereof, as part of an electronic mail address or on any sites on the Internet, without the prior written consent of AHF, which consent may be conditioned upon Franchisee conditionally assigning the name to AHF exercisable upon a default by Franchisee under, or expiration or other termination of, this Agreement. Any sites established by Franchisee on the Internet and any changes subsequently made to those sites must be approved by AHF prior to their establishment or change, which consent may be withheld for any reason. Franchisee also agrees not to register or attempt to register any of the above names, the Marks or the trade names of AHF, or any portions thereof as a trademark, service mark, or domain name on the Internet. During the term of this Agreement, AHF may, however, require that Franchisee post a sign at its Business Location, and include a reference on its letterhead, e-mails, contracts, business cards and/or other items, stating that it is an "authorized franchisee of Ace Handyman Franchising, Inc.," or other language specified by AHF. If local laws require that Franchisee file an affidavit or other registration indicating that it is conducting business under an assumed, fictitious or trade name, Franchisee shall state in such

filing or affidavit that the same is made "as an authorized franchisee of Ace Handyman Franchising, Inc.."

15.8. Change of Marks.

Franchisee shall use and display the Marks and Copyrighted Works only as specified by AHF. If it becomes advisable at any time in the opinion of AHF for Franchisee to modify or discontinue use of any of the Marks or Copyrighted Works, or to use one or more additional or substitute names, Marks or Copyrighted Works, Franchisee agrees to do so at its cost, and the sole obligation of AHF in any event shall be to reimburse Franchisee for its reasonable out-of-pocket costs of changing the main identifying sign and any other significant signage of its AHS Business used to identify the Business Location as an Ace Handyman Services franchise.

15.9. Data and Other Business Records.

Franchisee acknowledges and agrees that AHF owns all data and records ("Business Records") with respect to customers, suppliers and independent contractors, and other service professionals, and/or otherwise related in any way to Franchisee's AHS Business; including, without limitation, all databases (whether in print, electronic or other form) with customer and potential customers, names, addresses, phone numbers, e-mail addresses, and customer purchase records, and all other records contained in the databases. AHF hereby grants Franchisee a license to use such Business Records solely for the purpose of operating the AHS business hereunder. Franchisee further acknowledges and agrees that, at all times during and after the termination, expiration or cancellation of this Agreement, AHF may access such Business Records, and may process, utilize, permit any third party to utilize, transfer, share, or analyze the Business Records as AHF determines to be in the best interest of the System, in AHF's sole discretion, including without limitation, by contacting any or all of Franchisee's customers, contractors, subcontractors, and other service professionals for any purposes AHF deems appropriate. Notwithstanding the foregoing, Franchisee will own all records concerning Franchisee's employees, which are not deemed to be part of the Business Records hereunder.

16. REPORTS, RECORDS AND FINANCIAL STATEMENTS

16.1. Franchisee Reports.

- a. Franchisee will establish and maintain at its own expense bookkeeping and accounting systems that utilizes an accrual method of accounting and otherwise conforms to the specifications that AHF may prescribe from time to time. Franchisee will supply to AHF such reports in a manner and form as AHF may from time to time require, including monthly profit and loss statements in a format prescribed by AHF, and, if requested by AHF, weekly, semi-monthly and/or monthly reports that are to be computer generated and completed on a form and in a format as may be prescribed by AHF. Within 18 days of the end of each calendar month, calendar quarter, and calendar year, during the term of this Agreement, Franchisee shall provide AHF monthly reports, quarterly reports, and annual reports of Gross Revenues, respectively, as prescribed by AHF.
- **b.** Franchisee shall submit to AHF current financial statements and other reports as AHF may request to evaluate or compile research and performance data on any operational aspect of its AHS Business. Franchisee authorizes AHF to utilize this information to prepare an earnings claim, to release this information as necessary to substantiate any earnings claim made by AHF,

and to share such information in summary form as AHF deems necessary or desirable to share with other franchisees at any annual franchise meeting or other franchise business meetings.

- **c.** Upon the request of AHF and within 20 days of such request, Franchisee shall provide AHF an electronic copy of its QuickBooks files as well as the administrative login, password, and such other information to permit AHF to access such files.
- d. Franchisee shall submit to AHF current financial statements and other reports as AHF may request to evaluate or compile research and performance data on any operational aspect of its AHS Business. AHF may also pull this data from other reports and information provided by Franchisee hereunder. Franchisee authorizes AHF to utilize this information to prepare a financial performance representation, to release this information as necessary to substantiate any financial performance representation made by AHF, to share such information in summary form as AHF deems necessary or desirable to share with other franchisees at any annual convention or other franchise business meetings, or in any other manner and with any other parties that AHF deems appropriate without obtaining any further written consent of Franchisee. All financial information transmitted by Franchisee to AHF pursuant to this Agreement shall be owned by AHF as part of the Business Records as defined in **Section 15.9** above, with no duty on the part of AHF to account to Franchisee with respect to the use and exploitation of the same.
- **e.** All reports and financial information to be furnished to AHF must be signed and verified by Franchisee, unless such requirement is waived in writing by AHF, in its sole discretion.

16.2. Books and Records.

In all instances, the accounting and reporting system, and all statements and reports to be submitted by Franchisee, shall conform to U.S. Generally Accepted Accounting Principles, or other applicable Generally Accepted Accounting Principles if Franchisee is located outside of the U.S., applied consistently on a year-to-year basis. Franchisee shall retain all invoices, order forms, payroll records, checks records, bank deposit receipts, sales tax records and returns, cash disbursements journals and general ledgers as specified in the Operations Manual. Franchisee shall advise AHF of the location of all original documents and shall not destroy any records without the written consent of AHF. Franchisee shall prepare on a current basis, complete and accurate records concerning all financial, marketing and other operating aspects of its AHS Business conducted under this Agreement. Franchisee's records shall include tax returns, daily reports, statements of Gross Revenue profit and loss statements, ledgers and balance sheets.

16.3. Audit of Books and Records.

From the date Franchisee and AHF sign this Agreement until three years after the expiration or termination of this Agreement, including any successor franchises, AHF or AHF's authorized agent shall have the right to request, receive, inspect and audit any of the records referred to above and any related records wherever they may be located. Franchisee agrees to keep all records and reports for six years from the date such records are created. AHF may also request, receive, inspect and audit the records of any party affiliated with Franchisee, including but not limited to Franchisee's Principal Representative, Franchise Manager, owners, guarantors, officers, directors, or other Authorized Representatives, any immediate family members of Franchisee or of such affiliated parties, or any companies or entities associated with Franchisee or such affiliated parties, that AHF in its sole discretion determines may be relevant in determining the business results of Franchisee's AHS Business; such as verifying that Franchisee has paid all fees owed to AHF based on the Gross Revenues of Franchisee. Inspections and audits conducted at the

Business Location may take place without prior notice, during normal business hours. AHF may also require at any time the records from Franchisee or its affiliated parties be sent to AHF's offices or another location to permit the inspection or audit of such records to be conducted at AHF's place of business or the other location. If AHF notifies Franchisee that documents are to be sent to a location other than the Business Location for the purpose of conducting an inspection or audit at that location, Franchisee shall provide the requested documents to AHF within the time period set forth in AHF's notice. Franchisee will be responsible for any expenses associated with collecting and delivering any documents requested by AHF for its inspection or audit. Franchisee agrees that AHF will have the right to inspect and audit any records of Franchisee or any affiliated party that AHF determines to be relevant in its sole discretion, which records may include, in addition to those referred to above, (i) tax returns; (ii) quarterly and/or annual financial statements, including profit and loss statements and balance sheets; (iii) copies of checks, check ledgers and bank statements for checking and savings accounts; (iv) all contracts or agreements entered into by Franchisee and any third parties related to its AHS Business, including but not limited to contracts with customers; and (v) any other documents requested by AHF. AHF may inspect and audit documents covering a period beginning with the date on which Franchisee first acquired its AHS Business and ending on the date such audit is concluded. All documents provided for AHF's inspection or audit must be certified by Franchisee and the appropriate affiliated party, if applicable, as true, complete and correct. Should any inspection or audit disclose a deficiency in the payment of any Royalty Fee, Minimum Individual Marketing Expenditure, National Brand Fee, Software and Internet Fee, or other amounts required to be paid or spent under this Agreement, Franchisee shall pay to AHF the deficiency with the late payment charge and interest as set forth in Section 12.7 within 10 business days of notice thereof, without prejudice to any other remedy of AHF under this Agreement. In addition, if the deficiency for any audit period equals or exceeds 2 percent of the correct amount of any Royalty Fee, Minimum Individual Marketing Expenditure, National Brand Fee, Software and Internet Fee, or other amounts due or required to be spent, Franchisee will also pay to AHF the entire cost of the inspection or audit including travel, lodging, meals, salaries and other expenses of the inspecting or auditing personnel within 10 business days of notice thereof. Should the audit disclose an overpayment of any Royalty Fees, National Brand Fees, Software and Internet Fees, or other amounts due AHF, AHF shall pay Franchisee or credit to Franchisee's account, in AHF's sole discretion, the amount of the overpayment within 30 days of AHF's verification of such overpayment by Franchisee. For purposes of this Section, each calendar quarter of each calendar year being audited shall constitute its separate audit period.

16.4. Failure to Comply with Reporting Requirements.

If Franchisee's records and procedures are insufficient to permit a proper determination of Gross Revenues, AHF shall have the right to deliver to Franchisee an estimate, made by AHF, of Gross Revenues for the period under consideration, and Franchisee shall pay to AHF any amount shown thereby to be owing within five business days of the date of the notice. Any such estimate shall be deemed the minimum amount of fees due for the required reports, and Franchisee shall remain liable for all fees in excess of such amounts once the actual Gross Revenues related to such reports are determined. Additionally, Franchisee shall be liable for all late fees and interest set forth in **Section 12.7** of this Agreement for any reports not filed when due.

16.5. Financial Information from Third Parties.

Franchisee authorizes AHF to make inquiries of Franchisee's bank, suppliers and trade creditors concerning Franchisee's AHS Business, and agrees to direct such persons and companies

to provide to AHF such information and copies of documents pertaining to its AHS Business as AHF may request.

17. TRANSFER

17.1. Transfer by AHF.

Franchisee acknowledges that AHF's obligations under this Agreement are not personal, and AHF can unconditionally transfer, assign or convey, on its own discretion, this Agreement to another corporation or any other party, including the operator of a competing franchise system. Franchisee further acknowledges and agrees that AHF may sell its assets, the Marks or the System to any third party of AHF's choice; or may terminate or cease to exist or dissolve, in any such case without Franchisee's consent, and provided the transferee expressly assumes and undertakes to perform AHF's obligations in all material respects, free of any responsibility or liability whatsoever to Franchisee after the transaction occurs. With regard to any such sale, assignment or disposition, Franchisee expressly and specifically waives any claims, demands, or damages against AHF arising from or related to the transfer of the Marks or the System from AHF to any other party.

17.2. Transfer by Franchisee.

Franchisee understands and acknowledges that the rights and duties set forth in this Agreement are personal to Franchisee. Accordingly, AHF will not allow or permit any transfer, assignment, subfranchise or conveyance of this Agreement or any interest in this Agreement or all or any part of the business entity that owns it, or all or a substantial portion of the assets of the AHS Business (in each case, a "**Transfer**"), except in compliance with **Section 17.3**. The term "Transfer," as used in this Agreement, means and includes the voluntary, involuntary, direct or indirect assignment, sale, gift or other similar disposition. Any unauthorized sale, assignment, transfer or other conveyance, by operation of law or otherwise, or any attempt to do so, shall be deemed void and, at the option of AHF, grounds for termination of this Agreement by AHF.

17.3. Pre-Conditions to Franchisee's Transfer.

No Transfer will be approved by AHF or be effective unless and until Franchisee and the transferee obtain AHF's written consent and all the following conditions are satisfied:

- **a.** Franchisee is in full compliance herewith and pays to AHF all outstanding debts or amounts owing to AHF.
- **b.** At AHF's sole discretion, the transferee executes AHF's then current Franchise Agreement (which shall have a term, including extensions, equal to the remainder of Franchisee's term, but which may contain provisions substantially different from those contained herein), and such other documents then customarily used by AHF to grant franchises, and all other documents as may be requested by AHF.
- **c.** The Franchisee or the transferee pays to AHF a transfer fee in the amount of \$10,000 (the "**Transfer Fee**"), unless the transferee is (i) a business entity owned in full and actively managed by Franchisee, in which case no Transfer Fee shall be due, or (ii) a child, parent, sibling, or spouse of Franchisee, in which case the Transfer Fee shall be \$3,500.00. The Transfer Fee is nonrefundable in all circumstances once paid.

- **d.** Except where prohibited by law, Franchisee executes of a general release of AHF, including its shareholders, officers, directors, agents and employees, from all claims and potential claims of Franchisee.
- e. Unless waived by AHF, Franchisee submits to AHF an amount of 1 percent of the greater of Franchisee's Gross Revenues during Franchisee's final 12 months of business or the applicable Minimum Annual Gross Revenues for the last available Operating Year prior to the transfer, which amount shall be held in escrow and used to cover any claims against or expenses to AHF relating to warranty and customer dispute issues arising from Franchisee's AHS Business. This amount shall remain in escrow for the later of 12 months or until all warranties provided by Franchisee to any customer of Franchisee have expired. Any remaining amount left in escrow after the expiration of such period will be returned to Franchisee.
- f. The transferee purchases all of Franchisee's assets used in its AHS Business in accordance with all applicable bulk sales rules and regulations and assumes all of the liabilities of the AHS Business, unless such liabilities have been paid prior to the closing of the transaction or unless the sale is a sale of shares in the capital stock of Franchisee.
- **g.** The transferee completes, to AHF's sole satisfaction, AHF's then current training program established by AHF for franchisees; unless the training is waived by AHF in its sole discretion.
- h. The parties to the proposed transaction have entered into a bona fide binding agreement (a "Purchase Offer"), subject only to the rights of AHF. AHF shall be furnished a copy of this Purchase Offer, and such Purchase Offer shall be subject to AHF's written approval, and the Right of First Refusal reserved to AHF as specified in **Section 17.5** below. Franchisee must advise each prospective transferee of this provision and the other terms of this Agreement.
- i. If the transferee is a corporation, partnership, limited liability company or other legal entity, the transferee and its stockholders, partners, members or owners of a beneficial interest in the transferee have complied with **Section 11.2** above.
- j. The proposed transferee has demonstrated to AHF's satisfaction that it, he or she will meet in all respects AHF's standards applicable to new franchisees regarding experience, personal and financial reputation and stability, willingness and ability to devote its, his or her full time and best efforts to the operation of the AHS Business being transferred, and any other conditions as AHF may apply in evaluating new franchisees. All required conditions will be provided by AHF to the proposed transferee at time of notification of desire to transfer. AHF must be provided all information about the proposed transferee as AHF may require. No Transfer to a competitor of AHF will be permitted.
- **k.** Franchisee agrees that AHF has the right to confer with prospective transferees and furnish them with information regarding Franchisee's AHS Business, this Agreement, and the proposed transfer without being held liable to Franchisee, except for intentional misstatements made to a prospective transferee.

17.4. Franchisee's Death or Disability.

If Franchisee is an individual, upon the death or permanent disability of Franchisee, or if Franchisee is an entity, upon the death or permanent disability of the Franchise Manager, the rights granted by this Agreement may pass to the next of kin or legatees, provided that Franchisee's or

Franchise Manager's legal representatives shall within 120 days from the date of death or permanent disability of Franchisee or the Franchise Manager apply in writing to AHF for the right to transfer to the next of kin or legatee the rights under this Agreement or the ownership interest of the entity. The proposed transferees must meet each of the requirements set forth in this **Article 17** within 30 days of the receipt of a conditional approval for the transfer. For purposes hereof, the term "permanent disability" will mean a mental or physical disability, impairment or condition that is reasonably expected to prevent or actually does prevent Franchisee or the Franchise Manager from supervising the management and operation of the AHS Business for a period of 120 days from the onset of such disability, impairment or condition.

17.5. AHF's Right of First Refusal.

If Franchisee desires to Transfer, in whole or in part, the AHS Business, Franchisee shall obtain a bona fide, executed, written Purchase Offer from a responsible, arms-length, and fully disclosed purchaser for the AHS Business and other assets used by Franchisee in its AHS Business. Franchisee shall submit an exact copy of the Purchase Offer to AHF, which shall, for a period of 30 days from the date of delivery of such offer to AHF, have the right, but not the obligation, exercisable by written notice to Franchisee, to purchase all of the AHS Business and the assets of Franchisee (the "**Right of First Refusal**"), for the price and on the terms set forth in the Purchase Offer, subject to the provisions of this **Article 17** and provided that:

- a. there shall be deducted from the purchase price the amount of any commissions or fees that would otherwise have been payable to any broker, agent or other intermediary in connection with the sale of such property to the offeree; and
- **b.** AHF shall have the right to substitute cash for any other form of consideration specified in the Purchase Offer and to pay in full the entire purchase price at the time of closing.

If the sale to such purchaser is not completed within 60 days after delivery of such offer to AHF, AHF shall again have the Right of First Refusal.

17.6. Post-Transfer Obligations.

With and after each valid Transfer of this Agreement pursuant to this Article 17, the transferee or transferees of Franchisee shall be deemed to be the Franchisee under this Agreement and will be bound by and liable for all of Franchisee's existing and future obligations. No owner in any business entity that becomes Franchisee shall have any rights under this Agreement by reason of his, her or its ownership.

18. DEFAULT AND TERMINATION

18.1. <u>Termination by AHF–Effective Upon Notice.</u>

AHF shall have the right to terminate this Agreement and all rights granted Franchisee hereunder, subject to the provisions of applicable state or provincial law governing franchise termination and renewal, effective upon receipt of notice by Franchisee, upon the occurrence of any of the following events:

a. Unauthorized Disclosure. Franchisee intentionally or negligently discloses to any unauthorized person the contents of, or any part of, AHF's Operations Manual or any other

trade secrets or confidential information of AHF, or fails to take appropriate measures to assure no Authorized Representative discloses such information of AHF.

- **b. Abandonment.** Franchisee voluntarily abandons the AHS Business for a period of 15 consecutive days, or any shorter period that indicates an intent by Franchisee to discontinue operation of its AHS Business; unless such abandonment is due to fire, flood, earthquake or other similar causes beyond Franchisee's control and not related to the availability of funds to Franchisee.
- **c. Insolvency; Assignments.** Franchisee or any guarantor becomes insolvent or is adjudicated a bankrupt; or any action is taken by Franchisee or any guarantor, or by others against Franchisee or any guarantor under any insolvency, bankruptcy or reorganization act; or Franchisee or any guarantor makes an assignment for the benefit of creditors, or a receiver is appointed for Franchisee or any guarantor.
- d. Unsatisfied Judgments; Levy; Foreclosure. Any material judgment (or several judgments which in the aggregate are material) is obtained against Franchisee and remains unsatisfied or of record for 30 days or longer (unless a supersedeas or other appeal bond has been filed); or execution is levied against the AHS Business or any of the property used in the operation of the AHS Business and is not discharged within five days; or the real or personal property of the AHS Business is sold after levy thereupon by any sheriff, marshal or constable.
- **e. Criminal Conviction.** Franchisee or any owner of Franchisee's entity is convicted of a felony, a crime involving moral turpitude, a crime related to its AHS Business, or any crime or offense that is likely, in the sole opinion of AHF, to materially and unfavorably affect the System, Marks, goodwill or reputation thereof.
- **f.** Repeated Noncompliance. Franchisee receives three notices of default with respect to Franchisee's obligations hereunder from AHF during the term of this Agreement, regardless of whether the defaults were cured by Franchisee.
- **g.** Unauthorized Transfer. Franchisee sells, transfers or otherwise assigns the AHS Business, an interest in its franchise or the Franchisee entity, this Agreement, the AHS Business or a substantial portion of the assets of the AHS Business owned by Franchisee without complying with the provisions of this Agreement.
- h. Under-Reporting of Gross Revenues. Franchisee submits on three or more occasions during the term of this Agreement, or any successor franchise term, a report, financial statement, tax return, schedule or other information or supporting record which understates its Gross Revenues by more than 2 percent, or has a variance of more than 10 percent between reports generated through the scheduling software and Franchisee's regular accounting reports, unless Franchisee demonstrates to AHF's satisfaction that such understatement or variance resulted from inadvertent error.
- i. Failure to Deliver Reports. Franchisee submits reports more than five days late on three or more occasions during the term of this Agreement, or during the term of any successor franchise, unless due to circumstances beyond the control of Franchisee.
- j. Condemnation or Loss of Business Location. Franchisee loses possession or the right of possession of all or a significant part of the Business Location through condemnation,

casualty, lease termination or mortgage foreclosure and the AHS Business is not relocated or reopened within 60 days of such loss of possession or condemnation or casualty.

- **k.** Contesting Ownership of Marks. Franchisee contests in any court or proceeding the validity of, or AHF's ownership of, the Marks.
- **l.** Unauthorized Entity Action. Franchisee is a corporation or other business entity and any action is taken which purports to merge, consolidate, dissolve or liquidate such entity without AHF's prior written consent.
- **m.** Failure to Complete Training. Any of Franchisee, Franchisee's Principal Representative, or the Franchise Manager, as applicable, fail to successfully complete AHF's Initial Training Program.
- **n.** Improper Business Practices. AHF determines that Franchisee: engaged in an act of fraud with respect to its rights or obligations under this Agreement; engaged in false advertising; failed to, or intentionally underreported, sales or other financial information to AHF; made a misrepresentation or gave any false information in any reports or other information provided to AHF; or failed to comply with applicable laws, regulations and ordinances.
- o. Sexual Harassment or Discrimination. AHF receives credible evidence, which it verifies to its satisfaction, that Franchisee, its Principal Representative, its Franchise Manager, or any other management level employee of Franchisee, has sexually harassed or intimidated any individual or intentionally engaged in any racial, ethnic, religious, sexual, or other offensive discrimination against any individual or group.
- **p.** Sales Activity in the Territory of Another AHS Business. Franchisee provides, markets, or sells any of the Products or Services, or otherwise services any customers, within the territory of any AHS Business owned by AHF, any affiliate of AHF or any other franchisee of AHF, without the express written consent of AHF.
- q. Violation of Covenant Not to Compete or Other Restrictive Covenant. Franchisee or any of the Franchisee Affiliates (as defined in Section 20.1) violates the covenant not to compete or any other restrictive covenant contained in Article 20 below.
- r. Executive Order 13224; Patriot Act. Franchisee, or any officer, director, member, manager, or partner of Franchisee (as applicable), or the Franchise Manager, violates or becomes subject to United States Executive Order 13224 or The Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001 (the "Patriot Act").
- **s. Breach of Other Agreement.** Franchisee breaches the terms of any other agreement between AHF and Franchisee and fails to cure said breach during any applicable cure period provided in the other agreement.
- **t. Inadequate Guaranties.** Any guaranty of this Agreement fails to be a continuing obligation fully enforceable against the guarantor signing the guaranty, or there is any inadequacy of the guaranty or guarantor and the guarantor is unable to provide adequate assurances as required by AHF.

18.2. <u>Termination by AHF with Prior Notice</u>.

AHF shall have the right to terminate this Agreement and all rights granted Franchisee hereunder, subject to the provisions of applicable state or provincial law governing franchise termination and renewal, effective after the specified number of days after delivery of written notice by AHF to Franchisee:

- **a. Unauthorized Sales.** Franchisee sells or offers for sale any unauthorized merchandise, product or service after 30 days after notification from AHF.
- **b. Failure to Make Payments.** Franchisee fails to pay any amounts due AHF or affiliates, including the Initial Franchise Fee, the Royalty Fee, the National Brand Fee, the Software and Internet Fee, and all other fees or sums owed to AHF or its affiliates, or any amounts payable to AHF, within 10 days after receiving notice that such fees or amounts are overdue.
- **c. Misuse of Marks.** Franchisee misuses or fails to follow AHF's directions and guidelines concerning use of the Marks and fails to correct the misuse or failure within 10 days after notification from AHF.
- **d.** Failure to Submit Requested Information. Franchisee fails, or refuses, to submit any report, financial statement, tax return, schedule or other information or supporting records required herein within 10 days after notification from AHF.
- **e. Failure to Answer Business Phone.** Franchisee fails to employ a live person, who may be either an employee of Franchisee or an employee of an answering service business, to answer its published business telephone number between the hours of 8:00 a.m. and 5:00 p.m., Monday through Friday, holidays excepted, within 10 days after notification from AHF.
- **f. Failure to Meet Minimum Annual Gross Revenues.** Franchisee fails to generate Gross Revenues for any 12-month period which equals or exceeds the Minimum Annual Gross Revenues for the same period after 10 days notification from AHF.
- **g. Filing Non-Compliant Legal Action.** Franchisee or any of the Franchisee Affiliates files or otherwise commences litigation, arbitration, or any other legal action against AHF or any of the AHF Affiliates, as defined in **Section 22.1**, that is not in compliance with the dispute resolution terms agreed upon in **Article 22** as may be modified by any applicable rider in Exhibit VII, and fails to dismiss such action within seven days after notification from AHF.
- h. All Other Defaults Under Agreement. In addition to the foregoing termination rights, AHF shall have the right to terminate this Agreement (subject to any state or provincial laws to the contrary, where state or provincial law shall prevail), effective upon 30 days written notice to Franchisee, if Franchisee breaches any other provision of this Agreement and fails to cure the default during such thirty 30-day period. In that event, at AHF's sole discretion this Agreement will terminate without further notice to Franchisee, effective upon expiration of the 30-day period. Notwithstanding the foregoing, if the breach is curable, but is of a nature which cannot be reasonably cured within such 30-day period and Franchisee has commenced and is continuing to make good faith efforts to cure the breach during such 30-day period, Franchisee shall be given an additional reasonable period of time to cure the breach.

18.3. <u>Termination by AHF for Continuing Force Majeure Events.</u>

AHF shall have the right to terminate this Agreement and all rights granted Franchisee hereunder, subject to the provisions of applicable state or provincial law governing franchise termination and renewal, effective upon receipt of notice by Franchisee, in the event of a force majeure event as described in **Section 23.15** below that continues for a period of six consecutive months or longer and which prevents AHF from performing its obligations hereunder.

18.4. AHF's Right to Suspend Services on Franchisee's Default.

If AHF has provided Franchisee with a notice of any default pursuant to this **Article 18,** in addition to AHF's other remedies, AHF reserves the right, on behalf of itself and the AHF Affiliates, to suspend any services to be provided by AHF or any AHF Affiliate or the sales of any products to Franchisee by AHF or any AHF Affiliate until such time as Franchisee cures the default. The services that may be suspended include but are not limited to any services related to advertising or promotion of Franchisee's AHS Business such as the listing of Franchisee's AHS Business on any website and any scheduling services. The suspension may continue until Franchisee has cured each default identified in the default notice from AHF and Franchisee is deemed to be in good standing. Franchisee is not relieved of any obligation to pay any fees during the term of any suspension. The rights afforded to AHF in this **Section 18.4** are in addition to any other rights of AHF upon a default by Franchisee.

18.5. <u>Termination by Franchisee</u>.

Franchisee shall have the right to terminate this Agreement as the result of a material breach of this Agreement by AHF, provided Franchisee provides AHF with written notice of the breach within 30 days of the breach and a reasonable opportunity to cure such breach, which shall in no event be less than 90 days. Notwithstanding the foregoing, if the breach is curable, but is of a nature which cannot be reasonably cured within a 90-day or other given period and AHF has commenced and is continuing to make good faith efforts to cure the breach during the given period, AHF will be given an additional reasonable period of time to cure the breach. If Franchisee terminates this Agreement pursuant to this **Section 18.5**, Franchisee shall remain responsible for complying with the post termination obligations set forth in this Agreement, including in **Section 18.6** below.

18.6. Obligations of Franchisee Upon Termination or Expiration.

Franchisee agrees that upon termination or expiration of this Agreement Franchisee shall do all of the following:

a. Pay within five days of the effective date of termination or expiration of this Agreement all amounts owed to AHF, the landlord of the Business Location (if applicable) and Franchisee's trade and other creditors that are then unpaid. In the event of a termination due to a default by Franchisee, the amounts owed to AHF shall include an amount equal to the Royalty Fees that would have been payable based on the Minimum Annual Gross Revenues for each year from the date of termination until the expiration date that would apply to this Agreement had it not been terminated. All periodic payments to AHF shall be deemed to accrue daily, shall be adjusted accordingly, and shall include interest at the rate of 18 percent per annum or the highest rate permitted by law, whichever is lower. This same interest rate shall apply as the post-judgment interest rate, regardless of the applicable statutory rate, in the event of any legal actions related to this Agreement.

- **b.** Submit to AHF an amount of 1 percent of the greater of Franchisee's Gross Revenues during Franchisee's final 12 months of business or the applicable Minimum Annual Gross Revenues for the last available Operating Year, which amount shall be held in escrow and used to cover any claims against or expenses to AHF relating to warranty and customer dispute issues arising from Franchisee's AHS Business. This amount shall remain in escrow for the later of 12 months or until all warranties provided by Franchisee to any customer of Franchisee have expired. Any remaining amount left in escrow after the expiration of such period will be returned to Franchisee.
- **c.** Immediately discontinue the use of all Marks, signs, structures, forms of advertising, telephone listings and service, the Operations Manual, and all materials and Products and Services of any kind which are identified or associated with the System, and return all these materials and Products to AHF, at Franchisee's sole cost and expense.
- **d.** Immediately notify the telephone company, all listing agencies, Internet service providers, and social media website operators of the termination or expiration of Franchisee's right to use any telephone number, classified or other telephone directory listing, domain name, or social media website or account associated with the Marks, and authorize the transfer of them to AHF or any new franchisee as directed by AHF. Franchisee acknowledges as between AHF and Franchisee, AHF has the sole rights to, and interest in, all telephone numbers, directory listings, web addresses, domain names, and social media websites and accounts used by Franchisee to promote its AHS Business and/or associated with the Marks. Franchisee irrevocably appoints AHF, with full power of substitution, as its true and lawful attorney-in-fact, which appointment is coupled with an interest, to execute such directions and authorizations as may be necessary or prudent to accomplish the foregoing. Such appointment is evidenced by Exhibit VI to this Agreement.
- **e.** Make no representation nor state that Franchisee is in any way approved, endorsed or licensed by AHF or associated or identified with AHF or the System in any manner.
- **f.** Immediately take all steps necessary to amend or terminate any registration or filing of any d/b/a or business name or fictitious name or any other registration or filing containing the Marks, so as to delete the Marks and all references to anything associated with the System.
- **g.** Immediately shut down any Internet site operated by Franchisee to promote the AHS Business and assign and transfer all web addresses used by Franchisee for the same purpose.
- **h.** Comply with the provisions of this Agreement that survive termination, or expiration including in particular, the restrictive covenants in **Article 20**.

18.7. Franchisee's Failure to Comply With Post-Termination Obligations.

If, within 30 days after termination or expiration of this Agreement, Franchisee fails to:

a. Remove all displays of the Marks from Franchisee's AHS Business that are identified or associated with the System, AHF may enter the Franchisee's Business Location to effect removal, except if prohibited by law. In this event, AHF will not be charged with trespass nor be accountable or required to pay for any displays or materials.

b. Take all steps necessary to amend or terminate any registration or filing of any business name or d/b/a or any other registration or filing containing the Marks, then Franchisee irrevocably appoints AHF, with full power of substitution, as Franchisee's true and lawful attorney-in-fact, which appointment is coupled with an interest, and in Franchisee's name, place and stead and on Franchisee's behalf, to take action as may be necessary to amend or terminate all registrations and filings if Franchisee fails to timely take such action.

18.8. Effects of Termination or Expiration.

Termination or expiration of this Agreement shall not affect, modify or discharge any claims, rights, causes of action or remedies which AHF may have against Franchisee, whether such claims or rights arise before or after such termination or expiration. All obligations of the parties hereto which expressly or by their nature survive the expiration or termination of this Agreement shall continue in full force and effect notwithstanding such expiration or termination. In particular, but without limiting the generality of the foregoing, the provisions of **Sections 11.1.t** and **19.3** and **Articles 20** and **22** of this Agreement shall survive termination or expiration hereof.

18.9. Outstanding Loan Obligations.

In the event that this Agreement expires or is terminated for any reason whatsoever and AHF is the lender under any loan agreement ("Loan") or the holder of any promissory note ("Note") or the holder of any personal property, security interest, chattel mortgage, debenture or mortgage of any nature whatsoever (the "Security Interest") from Franchisee concerning assets used at any time by Franchisee in its AHS Business or which are situated on the Business Location, such Loan, Note or Security Interest shall, upon the effective date of termination or expiration, immediately become fully due and payable as to all principal and interest so loaned and secured.

18.10 Conflicting Laws.

THE PARTIES ACKNOWLEDGE THAT IN THE EVENT THAT THE TERMS OF THIS AGREEMENT REGARDING TERMINATION OR EXPIRATION ARE INCONSISTENT WITH APPLICABLE STATE, PROVINCIAL, OR FEDERAL LAW, SUCH LAW SHALL GOVERN FRANCHISEE'S RIGHTS REGARDING TERMINATION OR EXPIRATION OF THIS AGREEMENT.

18.11 Remedies Cumulative.

All rights and remedies conferred upon AHF by this Agreement and by law shall be cumulative of each other, and neither the exercise nor the failure to exercise any right or remedy shall preclude the exercise of any other right or remedy.

19. BUSINESS RELATIONSHIP

19.1. <u>Business Relationship</u>.

Franchisee acknowledges that it is an independent contractor and is not an agent, partner, joint venturer or employee of AHF, and Franchisee agrees not to hold itself out as such. This Agreement does not create a fiduciary or other special relationship between the parties. Neither party is liable or responsible for the other's debts or obligations, nor shall either party be obligated for any damages to any person or property directly or indirectly arising out of the operation of the

other party's business authorized by or conducted pursuant to this Agreement. It is further agreed that Franchisee has no authority to create or assume in AHF's name or on behalf of AHF, any obligation, express or implied, or to act or purport to act as agent or representative on behalf of AHF for any purpose whatsoever. All employees, contractors and subcontractors hired by or working for Franchisee shall be the employees, contractors and subcontractors of Franchisee and shall not, for any purpose, be deemed employees, contractors and subcontractors of AHF or subject to AHF's control. Neither this Agreement nor the course of conduct between AHF and Franchisee is intended, nor may anything in this Agreement (or the course of conduct) be construed, to state or imply that AHF is the employer or joint employer of Franchisee's Authorized Representatives, or vice versa. Each of the parties agrees to file its own tax, regulatory, payroll and other reports with respect to its respective employees, contractors and subcontractors, and its operations. Notwithstanding any other provisions in this Agreement, AHF shall not be responsible for supervising the activities of Franchisee's AHS Business or ensuring that the AHS Business is operated in compliance with applicable laws.

19.2. Third Party Obligations.

AHF will have no liability for Franchisee's obligations, or to pay or otherwise fulfill any of Franchisee's obligations to any third parties.

19.3. Indemnification.

Franchisee agrees to indemnify, defend, release and hold AHF, its subsidiaries and affiliates (if any), and their respective shareholders, directors, officers, members, managers, partners, employees, agents, successors and assignees, as applicable, (the "Indemnified Parties") harmless against, and to reimburse them for all Claims, (as defined below), any and all third party obligations described above, and any and all claims, obligations and liabilities directly or indirectly arising out of the operation of the AHS Business, out of the use of the Marks and Licensed Methods in any manner not in accordance with this Agreement, out of the use of any advertising or promotional images or content in any manner not in accordance with this Agreement, or out of the violation of any laws including, but not limited to, labor and employment laws, advertising and marketing laws, and consumer protection laws. This indemnity includes any Claims arising from the acts or omissions of Franchisee's Authorized Representatives and independent contractors. For purposes of this Agreement, "Claims" include all obligations, actual and consequential damages and costs reasonably incurred in the defense of any claim against the Indemnified Parties, including, without limitation, reasonable accountants', attorneys' and expert witness fees, costs of investigation and proof of facts, court costs, other litigation expenses and travel and living expenses. AHF will have the right to defend any such Claim against it. This indemnity will continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.

20. RESTRICTIVE COVENANTS

20.1. Non-Competition During Term.

Franchisee acknowledges that, in addition to the license of the Marks hereunder, AHF has also licensed commercially valuable information which comprises and is a part of the Licensed Methods, including without limitation, operations, marketing, advertising and related information and materials, and that the value of this information derives not only from the time, effort and money which went into its compilation, but from the usage of the information and materials by all franchisees of AHF using the Marks and Licensed Methods. Franchisee therefore agrees that other

than the AHS Business licensed herein, neither Franchisee, its Principal Representative, nor any of Franchisee's other owners or principals, as applicable (collectively, the "Franchisee Affiliates"), nor any member of his or their immediate families, will during the term of this Agreement:

- **a.** have any direct or indirect controlling interest as a disclosed or beneficial owner in a "Competitive Business," as defined below; or
- **b.** perform services as a director, officer, manager, employee, consultant, representative, agent or otherwise for a Competitive Business.

The term "Competitive Business" as used in this Agreement means any business operating, or granting franchises or licenses to others to operate, a business that is similar to an AHS Business, including a business that provides handyman, repair, maintenance, and/or remodeling services for homes and businesses. However, Franchisee will not be prohibited by this Agreement from owning securities in a Competitive Business if such securities are listed on a stock exchange or traded on the over-the-counter market and represent 2 percent or less of that class of securities issued and outstanding.

20.2. Post-Termination Covenant Not to Compete.

Upon termination or expiration of this Agreement for any reason, or the Transfer of the rights under this Agreement, Franchisee and the Franchisee Affiliates agree that, for a period of 24 months commencing on the effective date of termination, expiration, or Transfer, or the date on which Franchisee ceases to conduct business, whichever is later, neither Franchisee nor the Franchisee Affiliates will have any direct or indirect interest (through a member of any immediate family or otherwise) as a disclosed or beneficial owner, investor, partner, director, officer, employee, consultant, representative or agent or in any other capacity in any Competitive Business within 25 miles from the closest point to Franchisee's Territory or any territory of any AHS Business owned by AHF, any affiliate of AHF or any other franchisee of AHF. Franchisee and the Franchisee Affiliates expressly acknowledge that they possess business and career skills and abilities of a general nature and have other opportunities for exploiting such skills. Consequently, enforcement of the covenants made in this Section will not deprive them of their personal goodwill or ability to earn a living. If Franchisee or any former Franchisee Affiliate breaches this Section, the 24-month period shall start on the date that such person is enjoined from competing or stops competing, whichever is later.

20.3. No Diversion of Business.

During the term of this Agreement, and for a period of 24 months after the termination or expiration of this Agreement, Franchisee and the Franchisee Affiliates agree not to divert or attempt to divert any business related to AHF's business or any other business of a franchisee of AHF to any Competitive Business, by direct inducement or otherwise.

20.4. Non-Disparagement.

Franchisee agrees that neither it nor any of the Franchisee Affiliates shall, at any time during the term of this Agreement or afterwards, make any oral or written statements that directly or indirectly criticize, disparage, denigrate, or defame AHF or any of its affiliates or franchisees, or any of its or their officers, directors, employees, agents, representatives, partners, affiliates, successors, or assigns or the business, or any of the Licensed Methods, products, or services, of

AHF, or that would otherwise tend to be injurious to the reputation or goodwill of AHF or its Marks.

20.5. Confidentiality of Proprietary Information.

Franchisee will treat, and shall cause its Franchisee Affiliates and the Authorized Representatives to treat, all information it receives that comprises or is a part of the Licensed Methods as proprietary and confidential, and will not use or duplicate such information in an unauthorized manner or disclose the information to any unauthorized person, including in any business that may be competitive with AHF, without first obtaining AHF's written consent. Franchisee acknowledges, and will cause each of the Franchisee Affiliates and the Authorized Representatives to acknowledge, that the Marks and the Licensed Methods have valuable goodwill attached to them, that the protection and maintenance thereof is essential to AHF and that any unauthorized use or disclosure of the Marks and Licensed Methods will result in irreparable harm to AHF.

20.6. Confidentiality Agreements and Acknowledgements.

AHF reserves the right to require that Franchisee cause each of its Franchisee Affiliates and any member of their immediate families, to execute a Nondisclosure and Noncompetition Agreement in a form approved by AHF containing the restrictive covenants of this Agreement, and any other Authorized Representative to execute a confidentiality agreement containing the covenants in Section 20.5 of this Agreement. If AHF requires any immediate family member to execute a Nondisclosure and Noncompetition Agreement subsequent to the execution of this Agreement by Franchisee, Franchisee must use its best efforts to cause that immediate family member to execute the Nondisclosure and Noncompetition Agreement. Franchisee will provide to AHF a copy of each Nondisclosure and Noncompetition Agreement signed by any Franchisee Affiliate and any confidentiality agreement signed by any Authorized Representative, or any member of their immediate families, immediately following its execution and thereafter upon AHF's request.

20.7. Claims Are Not Defenses to Covenants.

Franchisee expressly agrees that the existence of any claim it may have against AHF, whether or not arising from this Agreement, shall not constitute a defense to the enforcement by AHF of the covenants of this **Article 20**. Franchisee further agrees that AHF shall be entitled to set off from any amount owed by AHF to Franchisee any loss or damage to AHF resulting from Franchisee's breach of this **Article 20**.

21. INSURANCE

21.1. Insurance Coverage.

Franchisee shall, upon commencement of the term of this Agreement, purchase and at all times maintain in full force and effect all of the following coverages:

a. Workers' compensation insurance with employers liability coverage, with a broad form all-states endorsement coverage sufficient to meet the requirements of law; (or, if such coverage is not required by applicable law, then with limits of at least \$1 million and otherwise meeting the requirements designated by AHF);

- **b.** Comprehensive general liability insurance including product liability, property damage, personal injury liability, bodily injury coverage, and premises/operation liability for the Business Location and its operations, with a limit of not less than \$1 million per occurrence and \$2 million in the aggregate, or such greater limit as may be required as part of any lease agreement for the Business Location; with a crime policy endorsement with a limit of not less than \$25,000;
- **c.** Comprehensive automobile liability insurance including hired and non-owned coverage with a limit of not less than \$1 million per occurrence and \$2 million in the aggregate;
 - **d.** Umbrella commercial excess coverage with a limit of not less than \$1 million;
- e. All-risk business property insurance covering all equipment, tenant improvements, inventory and other personal property owned or leased by Franchisee in an amount equal to at least 100 percent of the replacement costs of the contents and tenant improvements located at the Business Location, less a reasonable deductible which cannot exceed \$5,000; and
- **f.** Such additional insurance as may be required by the terms of any lease or mortgage for Franchisee's AHS Business Location.

Cyber liability insurance in an amount of not less than \$250,000 is optional but highly recommended.

The liability insurance afforded by the policy or policies shall not be limited in any way by reason of any insurance that may be maintained by AHF. AHF reserves the right to increase or decrease the amounts of insurance Franchisee must purchase by providing Franchisee with 10 days advance written notice of any changes in coverage amounts. All policies of insurance required under this Section will be with responsible companies qualified to do business and in good standing in the state or province where the Franchisee's AHS Business is located, and shall be in a form satisfactory to AHF. All liability insurance policies shall name AHF as an additional insured to the extent of claims arising out of the operations of Franchisee's AHS Business. To the extent AHF's preferred insurance provider is licensed and qualified to do business in the state or province where Franchisee's AHS Business is located, Franchisee shall obtain the above-required insurance coverage from Franchisee's preferred insurance provider, unless AHF approves an alternate insurance provider for Franchisee.

21.2. Proof of Insurance.

Prior to opening for business, Franchisee shall furnish to AHF or a third-party administrator used by AHF to monitor insurance compliance, certificates issued by each of Franchisee's insurers indicating that all premiums due have been paid, that all required insurance is in full force and effect, and that the insurance will not be terminated or changed without at least 30 days prior written notice from the insurer to AHF. New certificates evidencing renewal of insurance shall be furnished at least 30 days prior to the date of expiration of each policy. Within five business days of any request by AHF, Franchisee shall deliver a copy of all insurance policies to AHF for examination.

21.3. Failure to Maintain Insurance.

If Franchisee fails to obtain or maintain adequate insurance, in addition to any other remedies available to AHF under this Agreement, AHF may obtain insurance for and in

Franchisee's name. Within five days of any written request by AHF, Franchisee shall pay all costs of obtaining adequate insurance.

21.4. Contractor and Subcontractor Insurance.

Franchisee agrees not to permit any third party contractor or subcontractor to perform any work or offer any services on behalf of Franchisee in respect of the AHS Business unless such contractor or subcontractor maintains insurance coverage in such amounts and types as Franchisee is required to maintain under the provisions of this **Article 21**, with the specific addition that contractors and subcontractors cannot exclude principals from its Workers' Compensation coverage and that liability policies name Franchisee and AHF as an additional insured on a primary and non-contributory basis, for ongoing and completed operations. Franchisee agrees to maintain evidence that such insurance by its contractors and subcontractors is in effect and to provide such proof of insurance as AHF may require, in its sole discretion, from time to time.

22. ARBITRATION

22.1. Arbitration.

All controversies, disputes, claims, causes of action and/or alleged breaches or failures to perform between AHF, its subsidiaries and affiliated companies or their shareholders, officers, directors, members, managers, partners, agents, employees and attorneys (in their representative capacity) (collectively, the "AHF Affiliates") and Franchisee and the Franchisee Affiliates (as defined in Section 20.1 above) arising out of or related to: (1) this Agreement; (2) the relationship of the parties; (3) the validity of this Agreement; or (4) any Licensed Methods, will be submitted for binding arbitration to either the Judicial Arbiter Group ("JAG") or the American Arbitration Association ("AAA"), as selected by the party submitting the demand; except for actions brought which are related to or based on the Marks or the copyrights of AHF or to enforce the provisions of Article 20 of this Agreement, which actions AHF, at its option, may bring either in a court of competent jurisdiction or in arbitration. Notwithstanding the language above, if the action is based on a separate agreement or instrument between Franchisee or the Franchisee Affiliates and AHF or the AHF Affiliates (such as a promissory note or lease), the dispute resolution procedure in that agreement or instrument will control, rather than this Section; provided, that, at AHF's sole option, any claim of any AHF Affiliate against a Franchisee Affiliate based on such other agreement or instrument may be brought in arbitration in conjunction with a dispute between the parties that is subject to arbitration under this Section, regardless of any provisions to the contrary contained in that other agreement or instrument. Arbitration proceedings will be conducted in Denver, Colorado, U.S.A. and will be heard by one arbitrator in accordance with the then current rules of AAA that apply to commercial arbitration. The decision as to whether a claim is subject to mandatory arbitration shall be made by an arbitrator, not a court, except that the decision whether the arbitration may proceed as a class action shall be made by a court. The arbitrator shall be a resident of the State of Colorado, U.S.A. knowledgeable of Colorado law and fluent in English. The arbitration proceeding and all other hearings shall be conducted in English only, although Franchisee shall have the right, at Franchisee's option and sole expense, to have a translator present at the proceeding or other hearings. The expense of a translator shall not be considered a cost or expense related to an action pursuant to Section 23.9 of this Agreement. The parties further agree that, in connection with any arbitration proceeding, each will file any compulsory counterclaim (as defined by Rule 13 of the U.S. Federal Rules of Civil Procedure) within 30 days after the date of the filing of the claim to which it relates. Any party to an arbitration proceeding may apply to the arbitrator for reasonable discovery from the other. In this Agreement, "reasonable discovery" means a party may submit no more than ten interrogatories, including

subparts, 25 requests for admission, 25 document requests, and three depositions per side of the dispute. The foregoing discovery rights and limitations shall control over any contradictory discovery rules of AAA, unless the parties agree otherwise.

22.2. Arbitration Award.

Subject to **Sections 22.6** and **22.7** below, the arbitrator will have the right to award or include in the award any relief available and appropriate under the applicable law (as set forth in **Section 22.5**) and this Agreement. Any award shall be based on established law and shall not be made on broad principles of justice and equity. The award and decision of the arbitrator will be conclusive and binding upon all parties, and judgment upon the award may be entered in any court of competent jurisdiction. This provision will continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.

22.3. <u>Limitations on Proceedings.</u>

- a. AHF and Franchisee agree that arbitration will be conducted on an individual basis only. Neither party shall commence any arbitration with a third party against the other, or join with any third party in any arbitration involving AHF and Franchisee. Further, neither AHF nor Franchisee shall attempt to consolidate or otherwise combine in any manner an arbitration proceeding involving AHF and Franchisee with another arbitration of any kind, nor shall AHF or Franchisee attempt to certify a class or participate as a party in a class action against the other.
- **b.** The foregoing notwithstanding, in the event Franchisee controls, is controlled by, or is in active concert with another franchisee of AHF, or there is a guarantor of some or all of the Franchisee's obligations to AHF, then the joinder of those parties to any arbitration between AHF and Franchisee shall be permitted, and in all events, the joinder of an owner, director, officer, manager, partner or other representative or agent of Franchisee shall be permitted.
- **c.** Franchisee agrees that no claims may be brought on its behalf or on behalf of any of the Franchisee Affiliates by any third party, including but not limited to any association representing Franchisee.

22.4. Injunctive Relief.

Notwithstanding anything to the contrary contained in this Agreement, AHF and Franchisee will each have the right in a proper case to obtain temporary or preliminary injunctive relief from a court of competent jurisdiction. Each party agrees that the other may have such temporary or preliminary injunctive relief, without bond, but upon due notice, and with the sole remedy in the event of the entry of such injunctive relief being the dissolution of such injunctive relief, if warranted, upon hearing duly held (all claims for damages by reason of the wrongful issuance of such injunction being expressly waived by each party). Any such action will be brought as provided below.

22.5. Governing Law/Consent to Jurisdiction/Waiver of Jury Trial.

The United States Federal Arbitration Act shall govern all questions about the enforceability of the dispute resolution procedures in this Agreement and the confirmation of any arbitration awards pursuant to such procedures, and no arbitration issues are to be resolved pursuant to any state or provincial statutes, regulations or common law. Otherwise, except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections

1051 et seq.) or applicable international trademark law, this Agreement shall be interpreted under the laws of the State of Colorado, U.S.A. and any dispute between the parties shall be governed by and determined in accordance with the internal substantive laws (and not the laws of conflict) of the State of Colorado, U.S.A., which laws shall prevail in the event of any conflict of law. Notwithstanding the foregoing, the parties agree that the Colorado Consumer Protection Act (COLO. REV. STAT. ANN Section 6-1-101, et seq.) shall not apply to this Agreement or any disputes between the parties. Franchisee and AHF have negotiated regarding a forum in which to resolve any disputes that may arise between them and have agreed to select a forum in order to promote stability in their relationship. Therefore, if a claim is asserted in any legal proceeding not subject to mandatory arbitration, as specified in Section 22.1 above, involving Franchisee and/or the Franchisee Affiliates and AHF and/or the AHF Affiliates, the parties agree that the exclusive venue for disputes between them shall be in the state and federal courts of Colorado, U.S.A., and each waive any objection either may have to the personal jurisdiction of or venue in the state and federal courts of Colorado, U.S.A. Notwithstanding the foregoing, any legal proceeding by AHF or any AHF Affiliate not subject to mandatory arbitration may be brought in any court of competent jurisdiction in the country, state, province, or other geographic area in which the AHS Business is located or in which Franchisee or any Franchisee Affiliate resides or owns assets. IF A CLAIM MAY BE BROUGHT IN COURT, THEN AHF, THE AHF AFFILIATES, FRANCHISEE, AND THE FRANCHISEE AFFILIATES EACH WAIVE THEIR RIGHTS TO A TRIAL BY JURY.

22.6. No Punitive or Consequential Damages.

Except as specifically permitted elsewhere in this Agreement, neither AHF or any of the AHF Affiliates, on the one side, nor Franchisee or any of the Franchisee Affiliates, on the other side, shall be liable to the other for punitive or other damages not measured by the other party's actual damages, except as may be required by statute, in any action between the parties, whether of the type subject to mandatory arbitration under **Section 22.1** or otherwise, and whether such action is brought in arbitration, litigation, or any other legal proceeding.

22.7. No Recourse Against Others.

Franchisee agrees that its sole recourse for claims (whether in contract or in tort, in law or in equity, or granted by statute) arising between the parties shall be against AHF or its successors and assigns. Franchisee agrees that the shareholders, officers, directors, members, managers, partners, agents, and employees of AHF and its affiliates (the "Nonparty Affiliates") shall not be personally liable nor named as a party in any action between AHF and Franchisee. To the maximum extent permitted by law, Franchisee waives any such claims against such Nonparty Affiliates.

23. MISCELLANEOUS PROVISIONS

23.1. Modification.

- a. This Agreement may only be modified upon execution of a written agreement between AHF and Franchisee or, at AHF's option, upon notice of the approval of a Super-Majority as defined in **Section 23.1.b** below.
- **b.** This Agreement may be modified by AHF at its option whenever AHF and a Super-Majority, as hereinafter defined, of franchisees of AHF agree to any such modification. A "**Super-Majority**" of AHF franchisees shall consist of the owners of at least 75 percent of all

AHS Businesses, or, if only a portion of AHS Businesses are affected by the modification, at least 75 percent of those AHS Businesses affected by the modification. Whenever a modification is approved by a Super-Majority, AHF may elect to treat the modification as effective to all franchisees or the applicable group thereof, including Franchisee, to the same extent and in the same manner as if the modification was unanimously approved by them, and regardless of whether Franchisee may or may not desire to be bound by the modification. AHF shall provide Franchisee with notice of any modification to this Agreement based on a Super-Majority approval at least 30 days prior to the date such modification is to be effective. By signing this Agreement, Franchisee appoints the officers of AHF as its attorneys in fact with irrevocable power and authority to execute any such modification so approved.

- c. Franchisee acknowledges that AHF may modify its standards and specifications and operating, marketing, and other policies and procedures set forth in the Operations Manual unilaterally under any conditions and to the extent in which AHF, in its sole discretion, deems necessary, and Franchisee shall be bound by such modifications. These modifications may include regional and local variations. Franchisee may be obligated to invest additional capital in Franchisee's AHS Business and incur higher operating costs based on these periodic modifications.
- **d.** AHF has the right to vary the franchise agreement and any standards, specifications, policies, and procedures for a particular AHF franchisee based on the circumstances related to the franchisee, its territory, or any other condition. Franchisee shall not be entitled to require AHF to grant Franchisee a similar variation.

23.2. Entire Agreement.

This Agreement (which includes the Addendum and Exhibits expressly incorporated herein) contains the entire agreement between the parties and supersedes any and all prior agreements concerning the subject matter hereof. AHF does not authorize and will not be bound by any representation of any nature other than those expressed in this Agreement. Franchisee acknowledges and agrees that no representations have been made to it by AHF or its representatives regarding projected sales volumes, market potential, revenues or profits of Franchisee's AHS Business, or operational assistance other than as stated in this Agreement or in any franchise disclosure document or advertising or promotional materials provided by AHF in connection herewith. Additionally, Franchisee hereby acknowledges and agrees that, in entering into this Agreement, it is not relying on the existence or non-existence of any particular fact or matter not set forth in this Agreement or in the franchise disclosure document provided to Franchisee. Franchisee agrees and understands that AHF will not be liable or obligated for any oral representations or commitments made prior to the execution hereof, for claims of negligent or fraudulent misrepresentation based on any such oral representations or commitments, or for claims of negligent or fraudulent omissions or nondisclosure of facts or information. Nothing in this Agreement or in any related agreement is intended to disclaim any representations made by AHF in the franchise disclosure document provided to Franchisee.

23.3. Delegation by AHF.

From time to time, AHF will have the right to delegate the performance of any portion or all of its obligations and duties under this Agreement to third parties, whether they are employees of AHF or independent contractors that AHF has contracted with to provide such services. Franchisee agrees in advance to any such delegation by AHF of any portion or all of its obligations and duties hereunder.

23.4. Consent; Business Judgment.

Wherever AHF's consent or approval is required in this Agreement, unless the provision specifically indicates otherwise, AHF has the right to withhold its approval at its option, in its business judgment, taking into consideration its assessment of the long-term interests of the System overall. AHF may withhold any and all consents or approvals required by this Agreement if Franchisee is in default or breach of this Agreement. AHF's approvals and consents will not be effective unless given in writing and signed by one of its duly authorized representatives. In no event may Franchisee make any claim for money damages based on any claim that AHF has unreasonably withheld or delayed any consent or approval to a proposed act by Franchisee under the terms of this Agreement. Franchisee's sole remedy for the claim will be an action or proceeding to enforce the provisions of this Agreement by specific performance or by declaratory judgment.

23.5. General Economic Conditions.

Neither a general economic downturn or conditions nor Franchisee's financial inability to perform the terms of this Agreement will be a defense to an action by AHF for Franchisee's breach of this Agreement.

23.6. Effectiveness.

This Agreement will not be effective until accepted by AHF as evidenced by signing by an authorized officer of AHF.

23.7. <u>Limitation on Actions</u>.

Notwithstanding anything contained in this Agreement to the contrary, any and all claims and actions arising out of or relating to this Agreement, the relationship between Franchisee and AHF, or Franchisee's operation of the AHS Business shall be commenced within one year from the occurrence of the facts giving rise to the claim or action.

23.8. Review of Agreement.

Franchisee acknowledges that it had a copy of this Agreement in its possession for a period of time not less than that required by applicable law and has been given sufficient time to seek, and to submit this Agreement for, professional review and advice of Franchisee's choosing prior to freely executing this Agreement.

23.9. Attorneys' Fees.

- **a.** Subject to **Section 23.9.b** below, Franchisee shall reimburse AHF for its costs and expenses, including, without limitation, attorneys' fees, which AHF incurs in pursuit of its rights following a breach or event of default of or by Franchisee whether or not the pursuit of rights involves litigation or arbitration.
- **b.** The prevailing party in any action arising out of, or related to this Agreement (including an action to compel arbitration) is entitled to recover from the other party all costs and expenses related to the action, including reasonable attorneys' fees, and all costs of collecting monies owed. If both parties are awarded a judgment in any dollar amount, the court or arbitrator,

as applicable, shall determine the prevailing party taking into consideration the merits of the claims asserted by each party, the amount of the judgment received by each party and the relative equities between the parties.

23.10. No Waiver.

No waiver of any condition or covenant contained in this Agreement or failure to exercise a right or remedy by AHF or Franchisee will be considered to imply or constitute a further waiver by AHF or Franchisee of the same or any other condition, covenant, right, or remedy.

23.11. No Right to Set Off.

Franchisee will not be allowed to set off amounts owed to AHF for Royalty Fees, National Brand Fees, Software and Internet Fees, or other amounts due hereunder, against any monies owed to Franchisee, which right of set off is expressly waived by Franchisee. No endorsement or statement on any check or payment of any sum less than the full sum due to AHF shall be construed as an acknowledgment of payment in full or an accord and satisfaction, and AHF may accept and cash such check or payment without prejudice to its right to recover the balance due or pursue any other remedy provided herein or by law. AHF may apply any payments made by Franchisee against any past due indebtedness of Franchisee as AHF may see fit. AHF may set off against any payment due to Franchisee hereunder any outstanding debts of Franchisee to AHF, and may, at AHF's option, pay Franchisee's trade creditors out of any sum otherwise due to Franchisee.

23.12. Survival of Terms.

Every article and section of this Agreement that by its terms is intended to survive expiration and/or termination of this Agreement shall survive the expiration or termination of this Agreement for any reason and shall apply to a transferor upon a Transfer.

23.13. Invalidity.

In the event that any arbitrator or court of competent jurisdiction determines that any provision of this Agreement, including but not limited to any of the restrictive covenants contained in **Article 20** hereof, are unenforceable as written for any reason, including for purposes of the restrictive covenants, reasons that the areas of restriction exceed the reasonable maximum time period, geographic area or scope, then the parties hereby request and authorize the arbitrator or court to "blue pencil" such provision so as to make it enforceable and to best carry out the intent of the parties, or to deem such provision severed from this Agreement if it cannot be so modified. The holding, declaration or pronouncement shall not adversely affect any other provisions of this Agreement, which shall otherwise remain in full force and effect.

23.14. **Notices**.

All notices required to be given under this Agreement will be given in writing, by <u>personal delivery</u>, <u>by</u> registered or certified mail, return receipt requested, <u>by e-mail</u>, or by an overnight delivery service providing documentation of receipt, at the address set forth below the signatures of AHF and Franchisee respectively on the signature page hereto or at such other addresses as AHF or Franchisee may designate from time to time; and will be effectively given when deposited in the mails, postage prepaid, or when received via overnight delivery, as may be applicable. <u>Any such notice shall be deemed to have been delivered and received (a) in the case of personal</u>

delivery, on the date of such delivery, (b) in the case of mailing by registered or certified mail, on the third business day after being deposited in the United States mail, with proper address and postage prepaid, (c) in the case of e-mail, on the date delivered by e-mail to the proper address, or (d) in the case of delivery by overnight delivery service, one day after being deposited with the overnight delivery service.

23.15. Force Majeure.

Neither AHF nor Franchisee shall be liable for loss or damage or deemed to be in breach of this Agreement if a failure or delay in performance results from: (a) transportation shortage, inadequate supply or unavailability from manufacturers or suppliers of equipment, food products, supplies, labor, material or energy, or the voluntary surrender of the right to acquire or use any of the foregoing in order to accommodate or comply with any order, request, regulation, recommendation or instruction of any federal, state or municipal government; (b) compliance with any law, ruling, order, regulation, requirement or instruction of any federal, state or municipal government; (c) viral or bacterial epidemic, pandemic, or other public health crisis; (d) Act of God; (e) fire, strike embargo, war, terrorism, riot, hurricane, tornado, earthquake; or (f) other similar event or cause beyond the control of the party whose performance was prevented or delayed. Any delay resulting from any of these causes shall extend the time for performance or excuse performance, as may be reasonable, except that the causes shall not excuse payments of amounts owed at the time of the occurrence or payment of any amount due thereafter. A party seeking relief under this provision shall as soon as practicable notify the other party of its inability to perform.

23.16. Estoppel Certificates.

Franchisee agrees at any time and from time to time within 10 days after notice from AHF, to execute, acknowledge and deliver to AHF a statement in writing, form and substance acceptable to AHF, verifying that this Agreement is unmodified and in full force and effect (or if there have been modifications, that the Agreement is in full force and effect as modified and stating the modifications), and whether or not there exists any default in the performance of any term, condition or covenant of this Agreement and, if so, specifying each such default, and such other matters related to this Agreement as AHF shall request, it being intended that any such statement delivered pursuant hereto may be relied upon by AHF and by any lenders of AHF, or any prospective purchasers of all or any part of AHF's business.

23.17. Charges and Taxes.

All provisions in the Agreement stating that Franchisee will pay or be responsible for any costs, charges or taxes includes all customs or duty charges, foreign currency purchase levies, import and export fees and levies, and other similar costs, charges and taxes.

23.18. Cross-Default and Cross Termination Provisions.

a. A default by Franchisee under this Agreement will be deemed a default of all agreements between Franchisee and/or any company(ies) affiliated with Franchisee, on the one hand, and AHF and/or any company(ies) affiliated with AHF, on the other hand (the "Other Agreements"). A default by Franchisee and/or any company(ies) affiliated with Franchisee under any of the Other Agreements will be deemed a default under this Agreement. A default by any guarantor(s) of this Agreement or of any of the Other Agreements will be deemed a default of this Agreement.

b. If this Agreement is terminated as a result of a default by Franchisee, AHF may, at its option, elect to terminate any or all of the Other Agreements. If any of the Other Agreements is terminated as a result of a default by Franchisee and/or any company(ies) affiliated with Franchisee, AHF may, at its option, elect to terminate this Agreement. It is agreed that an incurable or uncured default under this Agreement or any of the Other Agreements will be grounds for termination of this Agreement and/or any and all of the Other Agreements without additional notice or opportunity to cure.

23.19. Manner of Payment.

All references in this Agreement to the term "Dollars" or the symbol "\$" refer to United States Dollars, and all payments made to AHF, unless otherwise noted, must be paid in United States Dollars net of any taxes or withholdings. The exchange rate for calculating payments due hereunder shall be the exchange rate published in The Wall Street Journal on the day the payment is due. If The Wall Street Journal is not published on the date a payment is due, the applicable exchange rate shall be that rate published in The Wall Street Journal on the nearest date of publication prior to the payment date, or by a successor or equivalent publication to be designated by AHF in the event The Wall Street Journal ceases to be published or ceases to publish the applicable exchange rates. AHF may designate and change payment instructions at any time on prior written notice to Franchisee. Franchisee shall be solely responsible for the payment of any costs and charges incurred in connection with the transfer and exchange of currency over and above any fees due or paid.

23.20. Approval Within Territory.

Any approval of this Agreement by the appropriate authorities in the Territory that is required to enable Franchisee to enter into this Agreement, perform under the terms of this Agreement, do business with AHF, or to make payments to AHF in United States Dollars in the United States of America will be the sole responsibility of Franchisee.

23.21. Translation of Agreement.

The English language will be regarded as the authoritative and official text of this Agreement; however, this Agreement may be translated into the language in dominant use in the Territory, at Franchisee's expense, in the event that translation is necessary for the purpose of registration of the Agreement with the applicable governmental authority. Nevertheless, in the event that any discrepancies exist between the English and the translated text, the English text will be considered the official text of this Agreement.

23.22. Communication with AHF.

All communication by Franchisee with AHF, whether written or oral, must be in English.

23.23. Incorporation of Riders.

To the extent that any of the Riders to Franchise Agreement for Specific States or Provinces attached as <u>Exhibit VII</u> is applicable, such rider is incorporated herein and this Agreement is modified accordingly. The provisions in any applicable rider are included as a condition to registration or use in certain jurisdictions, and AHF is not precluded from contesting

the validity, enforceability, or applicability of such provisions in any action relating to this Agreement or its rescission or termination.

23.24. Counterparts; Electronic Signatures.

This Agreement and any riders and addenda hereto may be executed in any number of identical counterparts and via electronic signatures, and each such counterpart shall be deemed a duplicate original hereof.

23.25. Acknowledgement.

BEFORE SIGNING THIS AGREEMENT, FRANCHISEE SHOULD READ IT CAREFULLY AND DISCUSS ITS PROVISIONS WITH ITS LEGAL COUNSEL. FRANCHISEE ACKNOWLEDGES ALL OF THE FOLLOWING:

- A. FRANCHISEE OR ITS PRINCIPAL REPRESENTATIVE HAS BEEN AFFORDED THE OPPORTUNITY TO ASK QUESTIONS AND REVIEW MATERIALS OF AHF THAT FRANCHISEE OR ITS PRINCIPAL REPRESENTATIVE DEEMS RELEVANT IN ORDER TO MAKE A DECISION TO ENTER INTO THIS AGREEMENT AND ACQUIRE A FRANCHISE HEREUNDER.
- B. FRANCHISEE OR ITS PRINCIPAL REPRESENTATIVE HAS SUCH KNOWLEDGE AND EXPERIENCE IN FINANCIAL, TAX AND BUSINESS MATTERS AND HAS PRIOR EXPERIENCE SO AS TO ENABLE FRANCHISEE OR ITS PRINCIPAL REPRESENTATIVE TO UTILIZE THE INFORMATION MADE AVAILABLE TO FRANCHISEE AND FULLY UNDERSTAND SUCH INFORMATION.
- C. THE SUCCESS OF THE BUSINESS VENTURE CONTEMPLATED HEREIN INVOLVES SUBSTANTIAL RISKS AND DEPENDS UPON FRANCHISEE'S ABILITY AS AN INDEPENDENT BUSINESS PERSON AND ITS ACTIVE PARTICIPATION IN THE DAILY AFFAIRS OF THE BUSINESS.
- D. NO ASSURANCE OR WARRANTY, EXPRESS OR IMPLIED, HAS BEEN GIVEN TO FRANCHISEE OR ITS PRINCIPAL REPRESENTATIVE AS TO THE POTENTIAL SUCCESS OF SUCH BUSINESS VENTURE OR THE EARNINGS LIKELY TO BE ACHIEVED.
- E. NO STATEMENT, REPRESENTATION OR OTHER ACT, EVENT OR COMMUNICATION, EXCEPT AS SET FORTH IN THIS DOCUMENT AND IN ANY DISCLOSURE DOCUMENT SUPPLIED TO FRANCHISEE, IS BINDING ON AHF IN CONNECTION WITH THE SUBJECT MATTER OF THIS AGREEMENT, AND, IN ENTERING INTO THIS AGREEMENT, FRANCHISEE IS NOT RELYING ON THE EXISTENCE OR NON-EXISTENCE OF ANY FACT OR MATTER NOT SET FORTH IN THIS AGREEMENT OR IN A DISCLOSURE DOCUMENT SUPPLIED TO FRANCHISEE.
- F. NEITHER FRANCHISEE, NOR ANY FRANCHISEE AFFILIATE, IS SUBJECT TO UNITED STATES EXECUTIVE ORDER 13224 OR THE PATRIOT ACT. IF FRANCHISEE OR ANY FRANCHISEE AFFILIATE BECOMES SUBJECT TO UNITED STATES EXECUTIVE ORDER 13224 OR THE PATRIOT ACT, FRANCHISEE OR THAT FRANCHISEE AFFILIATE SHALL NOTIFY AHF IMMEDIATELY THEREOF.

[SIGNATURES FOLLOW ON NEXT PAGE]

The parties have	executed this Agreer	nent to be made	e effective as	s of the	day of
, 202	_(the "Effective Date	").			

AHF: ACE HANDYMAN FRANCHISING, INC., a Colorado corporation	FRANCHISEE: IF AN INDIVIDUAL
By:	Signature
Address for Notice: 390 Union Boulevard, Suite 520 Lakewood, Colorado 80228, U.S.A. Fax No: 303-984-0133	Print Name: IF A PARTNERSHIP, LIMITED LIABILITY COMPANY, CORPORATION, OR OTHER ENTITY:
E-mail Address: With a copy to:	Print Company Name
2915 Jorie Blvd. Oak Brook, Illinois 60523 Attn: Legal Department E-mail Address:	By:

ADDENDUM TO FRANCHISE AGREEMENT

This Addendum to the Franchise Agreement (the "Addendum"), dated as of the date set forth below, modifies and amends that certain Ace Handyman Services Franchise Agreement (the "Agreement"), by and between Ace Handyman Franchising, Inc., hereinafter "AHF," and the undersigned franchisee, hereinafter "Franchisee." This Addendum modifies the terms of the Agreement and in the event of a conflict in terms between the Agreement and this Addendum, the terms of this Addendum shall be controlling.

The parties agree as follows:

1			
	siness Location. The Business I	Location, referenced in Section 4.1	of the Agreement,
geographical areas	s described as follows:	ced in Section 4.2 of the Agree thibit A-1 attached hereto, with a	, as
3. Ini 5.1 , shall be \$		Franchise Fee payable to AHF, ref	erenced in Section
commenced busine Training Program of unknown, the part	ess no later than the following da currently scheduled for Franchise ties will subsequently execute to	Agreement, Franchisee shall be ate, which is 60 days following the e (if known): the Commencement Date Supple n order to specify the commencement	e end of the Initial If currently ment to Franchise
		The Minimum Annual Gross Revenue, and shall be	
	Period	Minimum Annual	

Period	Minimum Annual
	Gross Revenues
	(Standard Franchise)
First Year	\$
Second Year	\$
Third Year and thereafter	\$

	nced in Section 13.3.a of the Ag	Expenditure. The Minimum Individual Marketing reement, shall commence on,
	Period	Minimum Individual Marketing Expenditure (Standard Franchise)
	Beginning on third full month through the end of First Year of operations	\$
	Second Year of operations Third Year of operations and thereafter	\$ \$
7. Otl	ner Provisions:	
Fully executed this	day of	. 202
AHF:		FRANCHISEE:
ACE HANDYMA a Colorado corpora		IF AN INDIVIDUAL:
By: Date:		Signature
		Print Name:
		IF A PARTNERSHIP, LIMITED LIABILITY COMPANY, CORPORATION, OR OTHER ENTITY:
		Print Name of Company
		By:
		Print Name and Title: Date:
		By:

Date: _____

TERRITORY ZIP OR POSTAL CODES

As used in this Agreement, the "**Territory**" means the geographic area in , comprised of the zip or postal codes set forth below as of the date of this Agreement. For the avoidance of doubt, the Territory shall also include any other zip or postal codes that may cover the geographic area described by the zip or postal codes set forth below if and to the extent that the government subdivides or otherwise modifies the zip or postal codes set forth below.

EXHIBIT II TO FRANCHISE AGREEMENT

COMMENCEMENT DATE SUPPLEMENT TO FRANCHISE AGREEMENT

This is a Supplement (the "Supplement") to the Franchise Agreement (the "Agreement") by and between Ace Handyman Franchising, Inc., hereinafter "AHF," and the undersigned franchisee, hereinafter "Franchisee."

1. Commencement Date. In paragraph 4 of the Addendum to the Agreement, AHF and
Franchisee did not list the date on which Franchisee is deemed to have commenced Franchisee's AHS
Business under the Agreement. AHF and Franchisee have now determined the date of the
commencement of the business. Therefore, pursuant to Section 7.3 of the Agreement, Franchisee shall b
deemed to have commenced business no later than the following date, which is 60 days following the end
of Franchisee's Initial Training Program:

2. Conflicts; Ratification. The terms of this Supplement shall control over the conflicting terms of the Agreement, including the Addendum. In all other respects, the Agreement and the Addendum are ratified, affirmed and confirmed, and shall remain in full force and effect in their original form. All capitalized terms in this Supplement that are not defined herein shall have the meaning ascribed to them in the Agreement.

The parties have executed this Supplement as of the date set forth below each of their signatures below, to be made effective as of the Effective Date of the Agreement, regardless of the date signed.

[SIGNATURES APPEAR ON THE FOLLOWING PAGE]

Fully executed this day of	_, 202
AHF:	FRANCHISEE:
ACE HANDYMAN FRANCHISING, INC., a Colorado corporation	IF AN INDIVIDUAL:
Ву:	
Date:	_ Signature
	Print Name:
	IF A PARTNERSHIP, LIMITED LIABILITY COMPANY, CORPORATION, OR OTHER ENTITY:
	Print Name of Company
	By:
	Signature
	Print Name and Title: Date:
	2
	By:
	Signature
	Print Name and Title: Date:

EXHIBIT III TO FRANCHISE AGREEMENT

GUARANTY AND ASSUMPTION OF FRANCHISEE'S OBLIGATIONS

GUARANTY AND ASSUMPTION OF FRANCHISEE'S OBLIGATIONS

In consideration of, and as an inducement to, the execution of Franchise Agreement (the "Franchise Agreement") executed on <u>or about</u> the date set forth below is by and between each of the undersigned and ACE HANDYMAN FRANCHISING, INC., a Colorado corporation, having its head office at 390 Union Boulevard, Suite 520, Lakewood, Colorado 80228, U.S.A. ("AHF"), each of the undersigned personally and unconditionally:

- 1. Guarantees to AHF and its successors and assigns, for the Term, including renewals and successor franchise terms thereof, that the franchisee named on the signature page hereof ("**Franchisee**") shall punctually pay and perform each and every undertaking, agreement and covenant set forth in the Franchise Agreement.
- 2. Agrees to be personally bound by, and personally liable for the breach of, each and every provision in the Franchise Agreement, including but not limited to, the terms of the articles and sections pertaining to non-competition during and after the Term, confidentiality and the Marks and Copyrighted Works of the AHF.
 - 3. Each of the undersigned waives all of the following:
 - (a) Acceptance and notice of acceptance by AHF of the foregoing undertaking.
 - (b) Notice of demand for payment of any indebtedness or nonperformance of any obligations hereby guaranteed.
 - (c) Protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed.
 - (d) Any right he or she may have to require that any action be brought against Franchisee or any other person as a condition of liability.
 - (e) Notice of any amendment, modification, deletion or addition of any term or condition of or to any of the obligations hereby guaranteed.
 - (f) Notice of any termination as to future liability of any other guarantor.
 - (g) Any and all other notices and equitable defenses to which he or she may be entitled.
 - 4. Each of the undersigned consents and agrees that:
 - (a) His or her direct and immediate liability under this Guaranty shall be joint and several.
 - (b) He or she shall render any payment or performance required under the Franchise Agreement upon demand if Franchisee fails or refuses punctually to do so.
 - (c) Such liability shall not be contingent or conditioned upon pursuit by AHF of any remedies against Franchisee or any other person.

- (d) Such liability shall not be diminished, relieved or otherwise affected by any extension of time, credit or other indulgence which AHF may from time to time grant to Franchisee or to any other person; including without limitation the acceptance of any partial payment or performance, or the compromise or release of any claims, none of which shall in any way modify or amend this Guaranty, which shall be continuing and irrevocable during the Term, including renewals and successor franchise terms thereof.
- (e) He or she shall be bound by the restrictive covenants, confidentiality provisions, audit provisions, and indemnification provisions contained in the Franchise Agreement.
- (f) AHF may, at its option, without notice to or further consent of him or her, take any of the following actions:
 - (i) retain the primary or secondary liability of any other party with respect to all or any part of the obligations hereby guaranteed.
 - (ii) release or compromise any liability of any other guarantor or any other party with respect to the obligations hereby guaranteed.
 - (iii) amend, modify, delete, or add any term or condition of or to any of the obligations hereby guaranteed, which may include the creation of new obligations.
- 5. No delay or neglect on the part of AHF in the exercise of any right or remedy existing under law or by virtue of this Guaranty shall operate as a waiver thereof, but such rights and remedies shall continue in full force and effect until specifically waived or released by an instrument in writing executed by AHF and designated as a waiver or release; and no single or partial exercise by AHF of any right or remedy shall preclude further exercise thereof or the exercise of any right or remedy.
- 6. The arbitration, injunctive relief, governing law and jurisdiction provisions contained in the Franchise Agreement shall govern this Guaranty, and such provisions are incorporated into this Guaranty by this reference.
 - 7. This Guaranty may be executed via electronic signature.

[SIGNATURES APPEAR ON THE FOLLOWING PAGE]

the day of, 202	each of the undersigned has affixed his or her signature effective on .
Name of Franchisee:	GUARANTOR(S)
(Location Name)	Signature
	Print Name
	Address
	Telephone Number
	Signature
	Print Name
	Address
	Telephone Number
	Signature
	Print Name
	Address
	Telephone Number

EXHIBIT IV TO FRANCHISE AGREEMENT

STATEMENT OF OWNERSHIP

STATEMENT OF OWNERSHIP

Franchisee:
(Print Company Name or Individual Franchisee's Name)
Trade Name (if different from above):
Trade Name (ii directin nom above).
Complete the applicable section below based on your form of ownership and entity:
☐ INDIVIDUAL/SOLE PROPRIETORSHIP
Country/state/province of residence:
Date of birth:
PARTNERSHIP
Name and address of each partner, showing percentage ownership of each:
State whether each partner identified above is or will be active in the business:
Country/state/province of formation:
Date of formation:
Attach a copy of the Partnership Agreement and any other formation documentation.
COPPOR LIVER
CORPORATION
Name and address of each shareholder, showing percentage of stock owned by each:
Name and address of each officer and director:
Country/state/province of incorporation:
Date of incorporation:
Attach a copy of the Articles of Incorporation certified by the Secretary of State or other official in the place of incorporation.

LIMITED LIABILITY COMPANY	
Name and address of each member, showing per	centage ownership of each:
Country/state/province of formation:	
Date of formation:	
Secretary of State or other official in the place Agreement.	ion or other formation document certified by the ce of formation and a copy of the signed Operating
☐ OTHER TYPE OF ENTITY	
Name and address of each owner, showing perce	entage ownership of each:
Name and address of each manager and officer:_	
Date of formation:	
Attach a copy of the articles or other formation other official in the place of formation.	ion document certified by the Secretary of State or
Franchisee acknowledges that this Sta Services Franchise authorized under Franchise A	atement of Ownership applies to the Ace Handyman greement.
Use additional sheets if necessary. A reported to AHF in writing.	ny and all changes to the above information must be
Date	Signature:
	Print Name:
Date	Signature:
	Print Name:

	EXHIBIT	V
TO FRANCHISE	AGREEMEN	T

AUTHORIZATION AGREEMENT FOR DIRECT PAYMENT



AUTHORIZATION AGREEMENT for DIRECT PAYMENT

ACE HANDYMAN FRANCHISING, INC.

I (we) authorize the Company to initiate variable entries to my (our) account described below:		
Checking Account No:		
Routing No (including zeros):		
- , - ,		
<u>or</u> Savings Account No:		
-		
Routing No (including zeros):		
Business Name:		
Name(s) on Account:		
Address Associated w/Account:		
Address Associated W/Account.		
,		
Financial Institution's Name:		
Financial Institution's Address:		
i maneiai mettation s Address.		
,		
Please attach a voided check or savings deposit slip for the account listed above.		
This authority is to remain in full force and offe	ctive until the Company has received written notification from	
•	ch time and manners as to afford the Company a reasonable	
opportunity to act on it.	on time and maintere de to arrord the company a redoctable	
spranner, to see an in	(For Joint Accounts)	
E-Signature*	E-Signature*	
B	D: W	
Printed Name	Printed Name	
Data Click or ton to enter a data	Data Cliek arten to enter a data	
Date Click or tap to enter a date.	Date Click or tap to enter a date.	
Territory #:	Territory #:	
Territory ".	remory ".	
Telephone No. () -	Telephone No. () -	
,		
*Dutania vous page on this fame vous are in the	is agreement electronically. Voyages	
*By typing your name on this form, you are signing the legal equivalent of your manual signature on this agr	is agreement electronically. You agree your electronic signature is the	
regai equivalent or your manual signature on this agr	ecment.	
For Company Use: Representative	Location:	

EXHIBIT VI
TO FRANCHISE AGREEMENT

COLLATERAL ASSIGNMENT OF TELEPHONE NUMBERS, TELEPHONE LISTINGS, INTERNET ADDRESSES, AND SOCIAL MEDIA WEBSITES

COLLATERAL ASSIGNMENT OF TELEPHONE NUMBERS, TELEPHONE LISTINGS, INTERNET ADDRESSES, AND SOCIAL MEDIA WEBSITES

THIS COLLATERAL ASSIGNMENT OF TELEPHONE NUMBERS, TELEPHONE LISTINGS, INTERNET ADDRESSES, AND SOCIAL MEDIA WEBSITES (the "Assignment") is entered into on the day and date set forth on the signature page hereof, by and between Ace Handyman Franchising, Inc., a Colorado corporation ("AHF") and the undersigned franchisee ("Franchisee"). This Assignment is executed in accordance with the terms of that certain Ace Handyman Franchising, Inc. Franchise Agreement (the "Franchise Agreement") under which AHF granted Franchisee the right to own and operate an Ace Handyman Services Franchise located at the business location set forth on the signature page hereof (the "Franchise Business").

FOR VALUE RECEIVED, Franchisee assigns to AHF (1) those certain telephone numbers and regular, classified or other telephone directory listings (collectively, the "Telephone Numbers and Listings"), (2) those certain Internet Website Addresses ("URLs"), and (3) those certain social media websites or accounts (collectively, the "Social Media Sites"), that are associated with AHF's trade and service marks and used from time to time in connection with the operation of Franchise Business. This Assignment is for collateral purposes only and, except as specified herein, AHF shall have no liability or obligation of any kind whatsoever arising from or in connection with this Assignment, unless AHF shall notify the telephone company and/or the listing agencies with which Franchisee has placed telephone directory listings (all such entities are collectively referred to herein as "Telephone Company"), Franchisee's Internet service provider ("ISP"), or the relevant social media website operator ("Social Media Operator") to effectuate the assignment pursuant to the terms hereof.

Upon termination or expiration of the Franchise Agreement (without renewal or extension), AHF shall have the right and is hereby empowered to effectuate the assignment of the Telephone Numbers and Listings, the URLs, and the Social Media Sites, and, in such event, Franchisee shall have no further right, title or interest in the Telephone Numbers and Listings, URLs, and Social Media Sites, and shall remain liable to the Telephone Company, the ISP, and the Social Media Operator for all past due fees owing to the Telephone Company, the ISP, and the Social Media Operator on or before the effective date of the assignment hereunder.

Franchisee agrees and acknowledges that as between AHF and Franchisee, upon termination or expiration of the Franchise Agreement, AHF shall have the sole right to and interest in the Telephone Numbers and Listings, URLs, and Social Media Sites, and Franchisee appoints AHF as Franchisee's true and lawful attorney-in-fact to direct the Telephone Company, the ISP, and the Social Media Operator to assign them to AHF, and execute such documents and take such actions as may be necessary to effectuate the assignment. Upon such event, Franchisee shall immediately notify the Telephone Company, the ISP, and the Social Media Operator to assign the Telephone Numbers and Listings, URLs, and Social Media Sites to AHF. If Franchisee fails to promptly direct the Telephone Company, the ISP, and the Social Media Operator to assign the Telephone Numbers and Listings, URLs, and Social Media Sites to AHF, AHF shall direct the Telephone Company, the ISP, and the Social Media Operator to effectuate the assignment contemplated hereunder to AHF. The parties agree that the Telephone Company, the ISP, and the Social Media Operator may accept AHF's written direction, the Franchise Agreement or this Assignment as conclusive proof of AHF's exclusive rights in and to the Telephone Numbers and Listings, URLs, and Social Media Sites upon such termination or expiration and that such assignment shall be made automatically and effective immediately upon Telephone Company's, ISP's, and Social Media Operator's receipt of such notice from AHF or Franchisee. The parties further agree that if the Telephone Company, the ISP, or the Social Media Operator requires that the parties execute the Telephone Company's, the ISP's, or the Social Media Operator's assignment forms or other documentation at the time of termination or expiration of Franchise Agreement, AHF's execution of such forms or documentation on behalf of Franchisee shall effectuate Franchisee's consent and agreement to the assignment. The parties agree that at any time after the date hereof they will perform such acts and execute and deliver such documents as may be necessary to assist in or accomplish the assignment described herein upon termination or expiration of Franchise Agreement.

IN WITNESS WHEREOF, the, 202	parties have entered into this Assignment on the day or
Business Location:(Location Name)	
ASSIGNEE	ASSIGNOR
Ace Handyman Franchising, Inc.	(Print Company or Individual Franchisee Name)
By:	By:Signature
Print Name:	C
Title:	Title:
	By:
	Signature Print Name:
	T'AL.

RIDERS TO FRANCHISE AGREEMENT FOR SPECIFIC STATES OR PROVINCES

If any one of the following Riders to the Franchise Agreement for Specific States or Provinces ("Riders") is checked as an "Applicable Rider" below, then that Rider shall be incorporated into the Franchise Agreement entered into by Ace Handyman Franchising, Inc. and the undersigned Franchisee. To the extent any terms of an Applicable Rider conflict with the terms of the Franchise Agreement, the terms of the Rider shall supersede the terms of the Franchise Agreement.

APPLICABLE RIDER:

<u>Unite</u>	d States					
	California		Maryland			Virginia
	Hawaii		Minnesota			Wisconsin
	Illinois		North Dakota			
	Indiana		Rhode Island			
ACE HANDYMAN FRANCHISING, INC. (Print Company or Individual Franchisee)						vidual Franchisee Name)
				By:		
Signature Title:				Signature Title:		
Date:						
				By:Signature		
				Title:		
				Date:		

UNITED STATES RIDERS TO THE FRANCHISE AGREEMENT

CALIFORNIA RIDER TO THE FRANCHISE AGREEMENT

1. **Section 22.5** is deleted and replaced with the following language:

The United States Federal Arbitration Act shall govern all questions about the enforceability of the dispute resolution procedures in this Agreement and the confirmation of any arbitration awards pursuant to such procedures, and no arbitration issues are to be resolved pursuant to any state or provincial statutes, regulations or common law. Otherwise, except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 et seq.) or applicable international trademark law, this Agreement shall be interpreted under the laws of the State of Colorado, U.S.A. and any dispute between the parties shall be governed by and determined in accordance with the internal substantive laws (and not the laws of conflict) of the State of Colorado, U.S.A., which laws shall prevail in the event of any conflict of law. Notwithstanding the foregoing, the parties agree that the Colorado Consumer Protection Act (COLO. REV. STAT. ANN Section 6-1-101, et seq.) shall not apply to this Agreement or any disputes between the parties. If a claim is asserted in any legal proceeding not subject to mandatory arbitration, as specified in Section 22.1 above, involving Franchisee and/or the Franchisee Affiliates and AHF and/or the AHF Affiliates, the parties consent to jurisdiction and venue for disputes between them in the state and federal courts of Colorado, U.S.A., and each waive any objection either may have to the personal jurisdiction of or venue in the state and federal courts of Colorado, U.S.A. Notwithstanding the foregoing, any legal proceeding by AHF or any AHF Affiliate not subject to mandatory arbitration may be brought in any court of competent jurisdiction in the country, state, province, or other geographic area in which the AHS Business is located or in which Franchisee or any Franchisee Affiliate resides or owns assets. IF A CLAIM MAY BE BROUGHT IN COURT, THEN AHF, THE AHF AFFILIATES, FRANCHISEE, AND THE FRANCHISEE AFFILIATES EACH WAIVE THEIR RIGHTS TO A TRIAL BY JURY.

2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

HAWAII RIDER TO THE FRANCHISE AGREEMENT

1. The following language is added at the end of **Sections 3.3.e**, **6.5**, and **17.3.d**:

Any release executed herewith will not apply to any claims that Franchisee may have that have arisen under the Hawaii Franchise Investment Law.

2. The following paragraph is added to **Article 18**:

Section 482E-6(3) of the Hawaii Revised Statutes provides that upon termination or refusal to renew the Franchise, AHF is obligated to compensate Franchisee for the fair market value, at the time of the termination or expiration of Franchise, of Franchisee's

inventory, supplies, equipment and furnishings purchased from AHF or a supplier designated by AHF; provided that personalized materials which have no value to AHF need not be compensated for. If AHF refuses to renew a Franchise for the purpose of converting Franchisee's business to one owned and operated by AHF, AHF, in addition to the remedies provided above, shall compensate Franchisee for the loss of goodwill. AHF may deduct from such compensation reasonable costs incurred in removing, transporting and disposing of Franchisee's inventory, supplies, equipment and furnishings pursuant to this requirement, and may offset from such compensation any monies due AHF.

3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

ILLINOIS RIDER TO THE FRANCHISE AGREEMENT

- 1. **Section 22.5** is amended to provide that Illinois law governs this Agreement.
- 2. Section 4 of the Illinois Franchise Disclosure Act of 1987 (the "Illinois Act") provides that any provision in a franchise agreement that designates jurisdiction or venue in a forum outside of the State of Illinois is void provided that a franchise agreement may provide for arbitration in a forum outside of the State of Illinois.
- 3. Section 41 of the Illinois Act provides that any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of the Illinois Act or any other law of the State of Illinois is void. Section 41 does not prevent any person from entering into a settlement agreement or executing a general release regarding a potential or actual lawsuit filed under any of the provisions of the Illinois Act, nor does it prevent the arbitration of any claim pursuant to the provision of Title 9 of the United States Code.

INDIANA RIDER TO THE FRANCHISE AGREEMENT

The following modifications are made to the Franchise Agreement only to the extent required by the Indiana Franchises Act, IND. CODE § 23-2-2.5, and the Indiana Deceptive Franchise Practices Act, IND. CODE § 23-2-2.7:

- 1. The following language is added to **Section 4.4**:
 - Indiana law prohibits AHF from establishing an AHF-owned outlet engaged a substantially identical business within Franchisee's Territory.
- 2. The last sentence of **Section 14.2** is supplemented with the following language:
 - Provided, however, that any amounts retained by AHF will be for compensation for services provided.

- 3. **Section 20.2** is revised to delete any restriction on operating or otherwise being involved with a Competitive Business within 25 miles of any territory of any AHS Business owned by AHF, any affiliate of AHF or any other franchisee of AHF.
- 4. **Section 22.5** is deleted and replaced with the following language:

The United States Federal Arbitration Act shall govern all questions about the enforceability of the dispute resolution procedures in this Agreement and the confirmation of any arbitration awards pursuant to such procedures, and no arbitration issues are to be resolved pursuant to any state statutes, regulations or common law. Otherwise, except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 et seq.) or applicable international trademark law, disputes related to a violation of the Indiana Franchises Act or the Indiana Deceptive Franchise Practices Act shall be governed thereby, and all other matters regarding this Agreement shall be governed by the laws of the State of Colorado. Notwithstanding the foregoing, the parties agree that the Colorado Consumer Protection Act (COLO. REV. STAT. ANN Section 6-1-101, et seq.) shall not apply to this Agreement or any disputes between the parties. Subject to the foregoing, Franchisee and AHF have negotiated regarding a forum in which to resolve any disputes which may arise between them and have agreed to select a forum in order to promote stability in their relationship. Therefore, if a claim is asserted in any legal proceeding not subject to mandatory arbitration, as specified in Section 22.1 above, involving Franchisee or the Franchisee Affiliates and AHF or the AHF Affiliates, both parties agree that the exclusive venue for disputes between them shall be in the state and federal courts of Colorado, U.S.A., and each waive any objection either may have to the personal jurisdiction of or venue in the state and federal courts of Colorado. U.S.A. Notwithstanding the foregoing, any legal proceeding by AHF or any AHF Affiliate not subject to mandatory arbitration may be brought in any court of competent jurisdiction in the country, state, province, or other geographic area in which the AHS Business is located or in which Franchisee or any Franchisee Affiliate resides or owns assets. IF A CLAIM MAY BE BROUGHT IN COURT, THEN AHF, THE AHF AFFILIATES, FRANCHISEE AND THE FRANCHISEE AFFILIATES EACH WAIVE THEIR RIGHTS TO A TRIAL BY JURY.

MARYLAND RIDER TO THE FRANCHISE AGREEMENT

1. The following language is added at the end of **Sections 3.3.e**, **6.5**, and **17.3.d**:

Any release executed in connection herewith will not apply to any claims that may arise under the Maryland Franchise Registration and Disclosure Law.

2. The following information is added at the end of **Section 18.1.c**:

Termination of this Agreement upon bankruptcy may not be enforceable under federal bankruptcy law.

3. The following language is added at the end of **Article 22**:

Notwithstanding the language above, Colorado law applies to this Agreement (except for claims arising under the Maryland Franchise Registration and Disclosure Law). Any claims arising under the Maryland Franchise Registration and Disclosure Law must be

brought within three years after the grant of the franchise. If the Franchise Agreement permits Franchisee to file a lawsuit, rather than requiring arbitration, Franchisee may file that suit in Maryland.

4. **Section 23.7** is amended by adding the following thereto:

The foregoing shall not act to reduce the three year statute of limitations afforded a franchisee for bringing a claim arising under the Maryland Franchise Registration and Disclosure Law.

5. **Section 23.25** is deleted and replaced with the following:

23.25. Acknowledgement.

- A. NO STATEMENT, REPRESENTATION OR OTHER ACT, EVENT OR COMMUNICATION, EXCEPT AS SET FORTH IN THIS DOCUMENT AND IN ANY DISCLOSURE DOCUMENT SUPPLIED TO FRANCHISEE, IS BINDING ON AHF IN CONNECTION WITH THE SUBJECT MATTER OF THIS AGREEMENT. THE ACKNOWLEDGMENTS OR REPRESENTATIONS OF FRANCHISEE WHICH DISCLAIM THE OCCURRENCE AND/OR ACKNOWLEDGE THE NON-OCCURRENCE OF ACTS THAT WOULD CONSTITUTE A VIOLATION OF THE MARYLAND FRANCHISE REGISTRATION AND DISCLOSURE LAW ARE NOT INTENDED TO NOR SHALL THEY ACT AS A RELEASE, ESTOPPEL, OR WAIVER OF ANY LIABILITY INCURRED UNDER THE MARYLAND FRANCHISE REGISTRATION AND DISCLOSURE LAW.
- B. FRANCHISEE ACKNOWLEDGES THAT NEITHER FRANCHISEE, NOR ANY FRANCHISEE AFFILIATE, IS SUBJECT TO UNITED STATES EXECUTIVE ORDER 13224 OR THE PATRIOT ACT. IF FRANCHISEE OR ANY FRANCHISEE AFFILIATE BECOMES SUBJECT TO UNITED STATES EXECUTIVE ORDER 13224 OR THE PATRIOT ACT, FRANCHISEE OR THAT FRANCHISEE AFFILIATE SHALL NOTIFY AHF IMMEDIATELY THEREOF.
- 6. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

MINNESOTA RIDER TO THE FRANCHISE AGREEMENT

1. **Articles 3, 17, and 18** are modified by the following language:

AHF will comply with Minnesota Statutes, Section 80C.14, Subds. 3, 4 and 5, which require (except in certain specified cases) (1) that Franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of this Agreement, and (2) that consent to the transfer of the franchise will not be unreasonably withheld.

2. The following language is added at the end of **Sections 3.3.e** and **17.3.d**:

Any release executed in connection herewith will not apply to any claims that may arise under the Minnesota Franchise Act. Rule 2860.4400(D) prohibits a franchisor from requiring a franchisee to assent to a general release from liability imposed by Minnesota Statutes, Sections 80C.01 to 80C.22, provided, it does not bar the voluntary settlement of disputes.

3. **Section 15.6** is modified by the following language:

As it deems necessary, AHF will protect Franchisee's right to use AHF's Marks. The Minnesota Department of Commerce requires AHF to indemnify Franchisees against liability to third parties resulting from claims by third parties that the Franchisee's use of AHF's Marks infringes upon the trademark rights of the third party.

4. **Section 22.4** is modified by the following language:

Pursuant to Minnesota Rule 2860.4400(J), a franchisee cannot consent to a franchisor obtaining injunctive relief. The franchisor may seek injunctive relief. Also, a court will determine if a bond is required.

5. **Section 22.5** is modified by the following language:

Minnesota Statutes, Section 80C.21 and Minnesota Rule 2860.4400(J) prohibit AHF from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring Franchisee to consent to liquidated damages, termination penalties, or judgment notes. In addition, nothing in the Franchise Disclosure Document or this Agreement can abrogate or reduce (1) any of Franchisee's rights as provided for in Minnesota Statutes, Chapter 80C, or (2) Franchisee's rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction. The above language has been included in this Agreement as a condition to registration. AHF and Franchisee do not agree with the above language and believes that each of the provisions of the Agreement are fully enforceable. AHF and Franchisee intend to fully enforce all of the provisions of the Agreement and all other documents signed by them, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.

6. The following statement is added at the end of **Section 23.7**:

Minnesota law provides that no action may be commenced pursuant to Minnesota Statute Section 80C.17 more than three years after the cause of action accrues. Minnesota Statutes, Section 80C.17, Subd. 5.

NORTH DAKOTA RIDER TO THE FRANCHISE AGREEMENT

- 1. The North Dakota Securities Commissioner has held the following to be unfair, unjust or inequitable to North Dakota franchisees:
 - A. Restrictive Covenants: Franchise Disclosure Documents which disclose the existence of covenants restricting competition contrary to Section 9-08-06, N.D.C.C., without further disclosing that such covenants will be subject to this statute.
 - B. Situs of Arbitration Proceedings: Franchise agreements providing that the parties must agree to the arbitration of disputes at a location that is remote from the site of the franchisee's business.
 - C. Restrictions on Forum: Requiring North Dakota franchisees to consent to the jurisdiction of courts outside of North Dakota.
 - D. Liquidated Damages and Termination Penalties: Requiring North Dakota franchisees to consent to liquidated damages or termination penalties.
 - E. Applicable Laws: Franchise agreements which specify that they are to be governed by the laws of a state other than North Dakota.
 - F. Waiver of Trial by Jury: Requiring North Dakota franchisees to consent to the waiver of a trial by jury.
 - G. Waiver of Exemplary & Punitive Damages: Requiring North Dakota franchisees to consent to a waiver of exemplary and punitive damages.
 - H. General Release: Franchise agreements that require the franchisee to sign a general release upon renewal of the franchise agreement.
 - I. Limitation of Claims: Franchise agreements that require the franchisee to consent to a limitation of claims. The statute of limitations under North Dakota law applies.

RHODE ISLAND RIDER TO THE FRANCHISE AGREEMENT

1. The following paragraph is added at the end of **Section 22.1**:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

The above language has been included in this Agreement as a condition to registration. AHF and Franchisee do not agree with the above language and believe that each of the provisions of the Agreement, including all choice of law provisions, are fully enforceable. AHF and Franchisee intend to fully enforce all of the provisions of the Agreement and all other documents signed by them, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.

VIRGINIA RIDER TO THE FRANCHISE AGREEMENT

1. The following is added to the end of **Section 23.18**:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in this Agreement does not constitute "reasonable cause," as that term is defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

WISCONSIN RIDER TO THE FRANCHISE AGREEMENT

1. The following paragraph is added to the end of **Article 18**:

The conditions under which this Agreement can be terminated or not renewed may be effected by the Wisconsin Fair Dealership Law, Wisconsin Statutes 1981-82, Title XIV-A, Chapter 135.

ATTACHMENT B

NONDISCLOSURE AND NONCOMPETITION AGREEMENT

NONDISCLOSURE AND NONCOMPETITION AGREEMENT

This Nondisclosure and Noncompetition Agreement (the "Agreement") is made effective on the date set forth on the signature page hereof, by and between ACE HANDYMAN FRANCHISING, INC., a Colorado corporation ("AHF"), the undersigned franchisee (the "Company"), and the undersigned ("Owner").

RECITALS

- A. AHF is engaged in the business of operating and selling franchises for the operation of handyman, repair, and maintenance businesses ("AHS Businesses"), associated with the mark "ACE HANDYMAN SERVICES," and related marks and utilizing AHF's unique system for operating the businesses and related licensed methods of doing business ("Licensed Methods");
- B. AHF and its affiliates have developed proprietary methods for establishing, operating and promoting the AHS Businesses utilizing certain confidential information relating to Company and its operations as more fully described herein, and have established substantial goodwill and an excellent reputation with respect to the quality of the services available through AHS Businesses, which goodwill and reputation have been and will continue to be of major benefit to AHF;
- C. Company is a franchisee under an effective agreement with AHF (the "Franchise Agreement");
- D. Owner is or will become involved with Company in the capacity of an owner or Principal Representative (as defined in the Franchise Agreement) (such capacities collectively referred to as "Affiliation") or is related to a person who has an Affiliation with the Company, and will become privileged as to certain confidential information related to AHF, its operations, the AHS Businesses and/or the Licensed Methods; and
- E. Owner, Company and AHF have reached an understanding and agreement with regard to nondisclosure by Owner of confidential information and with respect to noncompetition by Owner with AHF and Company.

NOW THEREFORE, in consideration of the foregoing, the mutual promises contained herein and other good and valuable consideration, the receipt and sufficiency of which are acknowledged, Owner, Company and AHF, intending legally to be bound, agree as follows:

AGREEMENT

1. <u>Confidential Information</u>. Owner recognizes and agrees that certain proprietary information relating to AHF and its operations ("Confidential Information") is owned by and treated as confidential by AHF and Company, including without limitation, (1) all proprietary information concerning AHF's business and the AHS Businesses; (2) all AHF's financial information other than financial information filed with any government regulatory agency; (3) franchise sales methods; (4) sales methods used to sell services by AHS Businesses; (5) all nonpublic statistical information; (6) the strategic plan, budgets and projections for AHF; (7) all information concerning negotiations of any kind conducted by AHF whether pending or completed; (8) all marketing research data and marketing plans, including information relating to marketing of services through AHS Businesses; (9) all information contained in AHF's operations manuals, and any other manual or other nonpublic written information; (10) internal lists of AHF's franchisees and clients of AHF and its franchisees; and (11) all other

information that may be considered a trade secret or proprietary and such Confidential Information that may be further developed from time to time by AHF.

- Use and Disclosure of Confidential Information. Owner acknowledges that, in connection with Owner's Affiliation with Company, AHF or Company will disclose in strict confidence certain Confidential Information necessary for the operation of an AHS Business or for sale of services through such AHS Businesses. Owner specifically acknowledges that the Confidential Information is valuable, unique and comprises a substantial portion of the assets of AHF; and Owner agrees that he or she will not utilize all or any portion of the same for Owner's personal benefit during the term of Owner's Affiliation with Company, nor in any manner use the same subsequent to the termination of Owner's Affiliation with Company or the termination or expiration of the Franchise Agreement, nor disclose any of the same to any person, firm, corporation or other entity whatsoever, including but not limited to a Competitive Business, as defined below, at any time for any reason or purpose, without the prior written consent of AHF. Owner shall not copy, publish or otherwise duplicate the Confidential Information or permit others to do so and shall return all Confidential Information to AHF or Company upon termination of Owner's Affiliation with Company. Owner may disclose to other employees, agents, or representatives of AHF or Company the Confidential Information only to the extent necessary for such employees, agents or representatives to carry out their intended function.
- 3. Prohibitions as to Employing Personnel. During the term of Owner's Affiliation with the Company, and for a period of 24 months after the earlier of (i) the effective date of termination or expiration of Owner's Affiliation with the Company, or (ii) the effective date of termination or expiration of the Company's Franchise Agreement, Owner shall not directly or indirectly solicit or attempt to solicit for the purposes of employment or consultation outside Company any present or former employee, agent or representative of AHF, Company or any other franchisee of AHF, without the prior written consent of AHF and Company.
- 4. 3. Noncompetition Covenant. Owner covenants and agrees that, during the term of his or her Affiliation, except while conducting Company's business in a manner authorized by AHF and Company, Owner shall not, either directly or indirectly through any member of Owner's immediate family, separate business entity or otherwise:
- (a) have any direct or indirect controlling interest as a disclosed or beneficial owner in a "Competitive Business," as defined below; or
- (b) perform services as a director, officer, manager, employee, consultant, representative, agent or otherwise for a Competitive Business.

The term "Competitive Business" as used in this Agreement means any business operating, or granting franchises or licenses to others to operate, a business that is similar to an AHS Business, including a business that provides handyman, repair, maintenance, and/or remodeling services for homes and businesses. However, Owner will not be prohibited by this Agreement from owning securities in a Competitive Business if such securities are listed on a stock exchange or traded on the over-the-counter market and represent 2% or less of that class of securities issued and outstanding.

4. Post-Termination Covenant Not to Compete. During the term of Owner's Affiliation with the Company, and for a period of 24 months after the earlier of (i) the effective date of termination or expiration of Owner's Affiliation with the Company, or (ii) the effective date of termination or expiration of the Company's Franchise Agreement, neither Owner, nor any member of Owner's immediate family, shall have any direct or indirect interest, as a disclosed or a beneficial owner,

investor, partner, director, officer, employee, consultant, representative or agent or in any other capacity, in any Competitive Business within 25 miles from the closest point to Franchisee's Territory (as defined in Company's Franchise Agreement with AHF) or any territory of any AHS Business owned by AHF, any affiliate of AHF or any other franchisee of AHF. If Owner or any member of Owner's immediate family breaches either of Sections 54 or 65, then the 24-month period under each of these Sections shall start on the date that Owner or its family member, as applicable, is enjoined from such activity or ceases such activity, whichever is later. Owner expressly acknowledges that he or she possesses business and career skills and abilities of a general nature and has other opportunities for exploiting such skills and abilities. Consequently, enforcement of this covenant will not deprive Owner of his or her personal goodwill or ability to earn a living.

- 6. 5. No Diversion of Business. During the term of Owner's Affiliation with the Company, and for a period of 24 months after the earlier of (i) the effective date of termination or expiration of Owner's Affiliation with the Company, or (ii) the effective date of termination or expiration of the Company's Franchise Agreement, Owner agrees not to divert or attempt to divert any business related to the Company, AHF or any other franchisee of AHF, or any client or account of the Company, AHF or any other franchisee of AHF, by direct inducement or otherwise.
- Audit of Business Records. AHF or its authorized agent may request, receive, inspect, 76. and audit any business records, financial or otherwise, of Owner, Owner's immediate family members, or any party affiliated with Owner or its immediate family members, including any companies or entities associated with Owner or its immediate family members, that AHF in its sole discretion determines may be relevant in determining Owner's compliance with the terms of this Agreement or Franchisee's business results in its AHS Business. The records subject to this audit include (i) tax returns; (ii) quarterly and/or annual financial statements, including profit and loss statements and balance sheets; (iii) copies of checks, check ledgers, and bank statements for checking and savings accounts; (iv) all contracts or agreements entered into with any third parties related to the AHS Business, including but not limited to contracts with customers; and (v) any other documents requested by AHF. Any such inspection or audit shall be conducted in accordance with the audit provisions set forth in the Franchise Agreement, which are deemed incorporated herein. Inspections and audits conducted at Owner's business location or other location where the records are held may take place without prior notice, during normal business hours. AHF may also require at any time the records from Owner or its affiliated parties be sent to AHF's offices or another location to permit the inspection or audit of such records to be conducted at AHF's place of business or the other location. If AHF notifies Owner that documents are to be sent to a location other than Owner's business location for the purpose of conducting an inspection or audit at that location, Owner shall provide the requested documents to AHF within the time period set forth in AHF's notice. AHF may audit and inspect documents covering a period beginning with the date on which Owner's Affiliation commenced and ending on the date such audit is concluded. All documents provided for AHF's inspection or audit must be certified by Owner and the appropriate affiliated party, if applicable, as true, complete and correct. Inspections and audits may be conducted following the expiration or termination of Owner's Affiliation for any reason.
- **87.** <u>Injunction</u>. Owner hereby acknowledges and agrees that in the event of any breach or threatened breach of this Agreement, AHF or Company shall be authorized and entitled to seek, from any court of competent jurisdiction, preliminary and permanent injunctive relief in addition to any other rights or remedies to which AHF or Company may be entitled.
- 98. Assignment. Company or AHF may assign all or part of this Agreement and the rights which inure to either of them hereunder without the consent of Owner, provided that any assignment by

the Company shall require the written consent of AHF. This Agreement shall not be assignable by Owner.

- 109. Effect of Waiver. The waiver by Owner, Company or AHF of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach thereof.
- **1110. Binding Effect.** This Agreement shall be binding upon and inure to the benefit of Owner, Company and AHF and their respective heirs, executors, representatives, successors and assigns.
- **1211.** Entire Agreement. This instrument contains the entire agreement of Owner, Company and AHF relating to the matters set forth herein. It may not be changed orally, but only by an agreement in writing, signed by the party against whom enforcement of any waiver, change, modification, extension or discharge is sought. Further, both Owner and Company agree that no change to this Agreement shall be made without the written consent of AHF having first been obtained.
- 1312. Governing Law. If AHF is a party to any action, this Agreement shall be governed by and construed under the laws of the State of Colorado, U.S.A. Otherwise, this Agreement shall be governed by and construed under the laws of the state or province where Company is located.
- 1413. Arbitration. Any and all controversies, disputes or claims between AHF, its subsidiaries and affiliated companies or their shareholders, officers, directors, agents, employees and attorneys (in their representative capacity); Company, its shareholders, officers, directors, agents and employees; and/or Owner arising out of or related to: (1) this Agreement; (2) the relationship of the parties; or (3) the validity of this Agreement shall be submitted for binding arbitration; except for actions for injunctive relief pursuant to Section \$7 above, which actions AHF and/or Company at their option may bring either in a court of competent jurisdiction or in arbitration. If AHF is a party to any controversy, dispute or claim, such arbitration proceedings shall be conducted in Denver, Colorado, U.S.A., will be submitted to either the Judicial Arbiter Group or the American Arbitration Association ("AAA"), as selected by the party submitting the arbitration demand, and will be heard by one arbitrator in accordance with the then current rules of AAA applicable to commercial arbitration. The arbitrator shall be a resident of the State of Colorado, U.S.A. knowledgeable of Colorado law and fluent in English. The arbitration proceeding and all other hearings shall be conducted in English only, although Owner shall have the right, at Owner's option and sole expense, to have a translator present at the proceeding or other hearings. Any party to such arbitration may apply to the arbitrator for reasonable discovery from the other. For purposes of the foregoing sentence, the term "reasonable discovery" shall mean a party may submit no more than ten interrogatories, including subparts, 25 requests for admission, 25 document requests, and three depositions per side of the dispute. With respect to any arbitration proceedings to which AHF is a party, the foregoing discovery rights and limitations shall control over any contradictory discovery rules of AAA, unless the parties agree otherwise. If AHF is not a party to such controversy, dispute or claim, such arbitration proceedings shall be conducted within the Territory (as defined in the Franchise Agreement between Company and AHF) of the Company and will be heard by one arbitrator in accordance with the then current commercial arbitration rules of AAA if the Territory is located in the U.S.A., or of any other arbitration group acceptable to Company and Owner, if the Territory is located outside the U.S.A. The decision as to whether a claim is subject to mandatory arbitration shall be made by an arbitrator, not a court.
- 1514. Severability. If any provision of this Agreement shall be held, declared or pronounced void, voidable, invalid, unenforceable or inoperative for any reason, by any court of competent jurisdiction, government authority or otherwise, the parties authorize and request such court or governmental authority to modify the provision held to be void, voidable, invalid, unenforceable or

inoperative to contain such lesser covenants that impose the maximum duty permitted by law so that the provision is upheld as valid, and the parties agree to be bound by such modified provision. Further such holding, declaration or pronouncement shall not affect adversely any other provisions of this Agreement which shall otherwise remain in full force and effect.

- **1615.** Attorneys' Fees. If AHF or Company must enforce any of the provisions or rights under this Agreement in any action at law or in equity and if the Company and/or AHF is successful in such litigation as determined by the court in a final judgment or decree taking into consideration the merits of the claims asserted by each party, then the Owner shall pay AHF or Company, as applicable, all costs, expenses and reasonable attorneys' fees incurred by AHF and/or Company (including without limitation such costs, expenses and fees on any appeals), and if AHF and/or Company shall recover judgment in any such action or proceeding, such costs, expenses and attorneys' fees shall be included as part of such judgment.
- 1716. <u>Definitions</u>. All capitalized terms not defined in this Agreement have the respective meanings set forth in the effective Franchise Agreement between Company and AHF.
- **1817.** Cross Default. A default by Owner under this Agreement will be deemed a default of all agreements between Franchisee and AHF, unless waived by AHF in writing.
- **1918.** Counterparts; Electronic Signature. This Agreement may be executed in counterparts and via electronic signatures.

ISIGNATURES APPEAR ON FOLLOWING PAGE

The parties have signed this Agreement on the date set forth below.

"AHF"	"COMPANY"
ACE HANDYMAN FRANCHISING, INC., a Colorado corporation	a
By:	By:
Its:	
Address: 390 Union Boulevard, Suite 520 Lakewood, Colorado 80228, U.S.A.	Address:
	"OWNER"
	Name:
	Date:
	Address:

RIDERS TO THE NONDISCLOSURE AND NONCOMPETITION AGREEMENT FOR SPECIFIC STATES

If any one of the following Riders to the Nondisclosure and Noncompetition Agreement for Specific States ("Riders") is checked as an "Applicable Rider" below, then that Rider shall be incorporated into the Nondisclosure and Noncompetition Agreement entered into by Ace Handyman Franchising, Inc. and the undersigned Owner and Company. To the extent any terms of an Applicable Rider conflict with the terms of the Nondisclosure and Noncompetition Agreement, the terms of the Applicable Rider shall supersede the terms of the Nondisclosure and Noncompetition Agreement.

	APPI	LICABLE RIDER	
		INDIANA	
		WASHINGTON	
"AHF	***		"COMPANY"
ACE HANDYMAN FRANCHISING, INC., a Colorado corporation			a
By: Its:			By:
			"OWNER"
			Name:

INDIANA RIDER TO THE NONDISCLOSURE AND NONCOMPETITION AGREEMENT

- 1. **Section 54** is reduced in scope to apply only to the Territory of Company.
- 2. The first sentence of **Section 1312** is deleted and replaced with the following language:
 - Governing Law. Except to the extent governed by the Indiana Franchise Disclosure Law or the Indiana Deceptive Franchise Practices Law, this Agreement shall be governed as to validity, interpretation, construction, effect and in all other respects by the laws and decisions of the State of Colorado.

WASHINGTON RIDER TO THE NONDISCLOSURE AND NONCOMPETITION AGREEMENT

1. The Agreement is subject to the terms of Washington statute RCW 49.62.005 et seq., which imposes certain limitations and prohibitions on the enforcement of noncompetition covenants against employees and independent contractors.

ATTACHMENT C STATEMENT OF PROSPECTIVE FRANCHISEE

ACE HANDYMAN FRANCHISING, INC. STATEMENT OF PROSPECTIVE FRANCHISEE

(Note: Dates and Answers Must be completed in the Prospective Franchisee's Own Handwriting.)

Since the prospective franchisee (also called "me," "our", "us", "we", and/or "I" in this document) and Ace Handyman Franchising, Inc. (also called "AHF", "you", or "your") both have an interest in making sure that no misunderstanding exist between each of us, and to verify that no violations of law might have occurred, and understanding that AHF is relying on the statements I/we make in this document, I/we advise AHF as follows:

	The following dates and information are true and correct:						
	1. The date of our first face-to-face meeting with any person to discuss the possible purchase of an AHS Business.						
	2. The date on which I/we received a Franchise Disclosure Document providing me/us with information regarding the purchase of an AHS Business.						
	3. The date when I/we received a fully completed copy (other than signatures) of the Franchise Agreement and all other documents I/we later signed.						
	4. The earliest date on which I/we signed the Franchise Agreement or any other binding document (not including any Receipt evidencing our receipt of the Franchise Disclosure Document).						
	5. The earliest date on which I/we delivered cash, a check or other consideration to AHF, or any other person or company.						
	Representations and Other Matters:						
	1. No oral, written, visual or other promises, agreements, commitments, or representations of any type, including, but not limited to, any which expanded upon or were inconsistent with the Franchise Disclosure Document or the Franchise Agreement, have been made to me/us with respect to any matter nor have I/we relied in any way on such, except as expressly set forth in the Franchise Agreement or a written Addendum thereto signed by me/us and the President or CEO of AHF, except as follows:						

2. No oral, written, visual or other claim, guarantee or representation (including, but not
limited to, charts, tables, spreadsheets or mathematical calculations to demonstrate actual or
possible results based on a combination of variables, such as multiples of price and quantity to
reflect gross sales, or otherwise), which stated or suggested any specific level or range of actual
or potential sales, costs, income, expenses, profits, cash flow, tax effects or otherwise (or from
which such items might be ascertained), from franchised or non-franchised units, was made to
me/us by any person or entity, nor have I/we relied in any way on any such, except for the
information expressly set forth in the Franchise Disclosure Document, if any, except as follows:
-

(If none, write NONE in your own handwriting).

3. No contingency, prerequisite, reservation or other condition exists with respect to any matter (including, but not limited to, my/our obtaining any financing, my/our selection, purchase, lease or otherwise of a location, any operational matters or otherwise) or my/our fully performing any of my/our obligations, nor am I/we relying on AHF or any other entity to provide or arrange financing of any type, nor have I/we relied in any way on such, except as expressly set forth in the Franchise Agreement or a written Addendum thereto signed by me/us and the President or CEO of AHF, except as follows:

(If none, write NONE in your own handwriting).

- 4. I/we understand that the information contained in ITEM 19 of the Franchise Disclosure Document, if any, is not intended to express or infer an estimate, projection or forecast of revenues, sales, expenses, income or earnings to be derived in connection with any particular franchise. I/we understand that AHF makes no representation to whether I/we will ever be able to sell any products or services, or the length of time it will take me/us to realize any gross revenues, net income or any other financial results. I/we understand that my/our actual financial results are likely to differ from the figures presented. I/we understand that AHF does not represent that I/we can expect to attain the revenues or limit my/our expenses to those contained in ITEM 19 of the Franchise Disclosure Document, if any, or that I/we can do as well as the outlets included therein. If I/we rely on those figures, I/we accept the risk of not doing as well. I/we acknowledge that my/our ability to achieve any level of income will depend upon factors not within AHF's control, including the occurrence of certain start-up and operating expenses and the amount of those expenses, and my/our level of expertise.
- 5. If the prospective franchisee is a business entity, the individuals signing for the "Prospective Franchisee" constitute all of the executive officers, members, managers, partners, shareholders, investors and/or principals (as applicable) of the Prospective Franchisee and each of such individuals has received the Franchise Disclosure Document and all attachments and carefully read, discussed, understands and agrees to the Franchise Agreement and each written attachment, addendum, or exhibit.
- 6. I/we have had an opportunity to consult with an independent professional advisor, such as an attorney or accountant, prior to signing any binding documents or paying any sums, and AHF

has strongly recommended that I/we obtain such independent professional advice. I/we have also been advised by AHF to discuss my/our proposed purchase of, or investment in, an AHS Business with one or more existing Ace Handyman Services franchisees prior to signing any binding documents or paying any sums and I/we have been supplied with a list of existing Ace Handyman Services franchisees.

7. I/we understand that entry into any business venture necessarily involves certain risk of loss or failure, that the purchase of an AHS Business (or any other franchise) is a speculative investment, that investment beyond the amounts outlined in the Franchise Disclosure Document may be required to succeed, that there exists no guaranty against possible loss or failure in this or any other business and that the most important factors in the success of any AHS Business, including the one to be operated by me/us, are my/our personal business, marketing, sales, management, judgment and other skills.

If there are any matters inconsistent with the statements in this document, or if anyone has suggested that I sign this document without all of its statements being true, correct and complete, I/we will make a written statement regarding such next to my signature below so that AHF may address and resolve any such issue(s) at this time and before either party goes forward.

I/we understand and agree to all of the foregoing and represent and warrant that all of the above statements are true, correct and complete.

Date:	
PROSPECTIVE FRANCHISEE:	
By:	_
Print Name:	- -
By:	_
Print Name:	-
By:	_
Print Name:	-
All of the above is true, correct and complete to the	e best of my knowledge.
Franchise Marketing Representative:	
Reviewed by:	(AHF)
President/CEO:	Franchise Agreement Number:

ATTACHMENT D MULTI-TERRITORY ADDENDUM TO FRANCHISE AGREEMENTS

MULTI-TERRITORY ADDENDUM TO FRANCHISE AGREEMENTS

This Multi-Territory Addendum to Franchise Agreements (the "Multi-Territory Addendum"), dated as of the date set forth below, modifies and amends those certain Franchise Agreements (as defined below), each of which has been entered into by and between Ace Handyman Franchising, Inc., hereinafter "AHF," and the undersigned franchisee, hereinafter "Franchisee." This Multi-Territory Addendum modifies the terms of the Franchise Agreements and in the event of a conflict in terms between the Franchise Agreements and this Multi-Territory Addendum, the terms of this Multi-Territory Addendum shall be controlling.

Background. Franchisee is either contemporaneously with the execution of this Multi-

The parties agree as follows:

Businesses wit	th Territ	ories that ar	e adjacent to	each oth	ner, or has previous	sly ent	ered i	nto a franci	hise
agreement to a	cquire ar	n AHS Busine	ess and is cont	emporan	eously with the exec	ution	of this	Multi-Terri	tory
Addendum ent	ering int	o one or mor	e further franc	chise agr	eements to operate	one or	more	additional A	HS
Businesses wit	th Territ	ories adjacer	nt to the Terri	tory und	er the original franc	chise a	igreen	nent. AHF	and
Franchisee agr	ee that t	the franchise	agreements b	etween	them that are amen	ded by	y this	Multi-Terri	tory
Addendum are	as follo	ws (collective	ely, the "Fran	chise Ag	greements"):				
	a.	Franchise	Agreement	dated		for	the	Territory	of
	b.	Franchise	Agreement	dated		for	the	Territory	of

Territory Addendum entering into two or more franchise agreements to acquire and operate multiple AHS

Collectively, the AHS Businesses being operated pursuant to the terms of the Franchise Agreements shall be referred to herein as the "Franchised AHS Businesses," and the Territories designated under all of the Franchise Agreements shall be referred to herein as the "Franchised Territories."

- 2. <u>Single Franchise Manager</u>. Section 11.1.c of each Franchise Agreement is hereby modified to permit Franchisee to designate a single Franchise Manager for multiple or all of the Franchise Agreements and Franchised AHS Businesses, provided that the Franchise Manager devotes full time and best efforts on a daily basis, in person, to the supervision and conduct of Franchisee's Franchised AHS Businesses as a whole, which supervision must not be less than 40 hours per week.
- 3. <u>Single Purpose Entity</u>. Each Franchise Agreement is hereby modified to permit Franchisee to operate all of the Franchised AHS Businesses as a single business entity.
- 4. <u>Single Business Location</u>. Section 4.1 of each Franchise Agreement is hereby modified to provide that Franchisee may operate all of the Franchised AHS Businesses from a single Business Location centrally located within the Franchised Territories as a whole (rather than requiring a separate Business Location for each AHS Business located in each individual Franchised Territory), subject to the following limitations and conditions:
- a. The single selected Business Location must have a Google "pin" that enables the Franchised AHS Businesses to appear with reasonable priority in a Google search performed throughout the entire extent of the Franchised Territories. If the Franchised AHS Businesses do not appear with reasonable priority in Google search results for "handyman services" or similar terms throughout the

entirety of the Franchised Territories (as determined by AHF in its sole discretion), then Franchisee must either (i) acquire one or more additional Business Location(s) (up to a maximum of one Business Location for each Franchise Agreement, each of which must comply with the requirements of the Franchise Agreements) until such desired search results are achieved; and (ii) if agreed to by AHF in its discretion, arrange for additional pay-per-click advertising (which expenses shall be the sole responsibility of Franchisee and shall be in addition to the Minimum Individual Marketing Expenditure or Minimum Individual Advertising Expenditure), on terms acceptable to AHF, to allow for appropriate Google search engine optimization throughout the Franchised Territories.

- b. In the event that future developments in Google and other online advertising and business listings cause the Franchised AHS Business to not be searchable or to have reduced visibility or presence online throughout the entirety of the Franchised Territories based on the use of a single Business Location, as determined by AHF in its discretion, then, unless another arrangement is agreed upon by AHF and Franchisee to mitigate such effects, AHF may require Franchisee to acquire one or more additional Business Locations (up to a maximum of one Business Location for each Franchise Agreement) meeting the requirements of each of the Franchise Agreements.
- 5. <u>Software and Internet Fee.</u> In lieu of the Software and Internet Fee for each Franchise Agreement, Franchisee shall pay each month for all of the Franchise Agreements combined (i) a software fee (the "Software Fee") for access to the scheduling software, e-mail, and online marketing services, and (ii) an internet Fee (the "Internet Fee"), for Franchisee's location website(s), which includes website development services and search engine optimization and shall be based on the number of location websites Franchisee shall have, in the following amounts:

\$ Software Fee
\$ - Internet Fee.

All references to the Software and Internet Fee set forth in the applicable Franchise Agreements shall refer to the Software Fee and the Internet Fee separately, as applicable. The Software Fee and/or the Internet Fee may be changed from time to time in AHF's sole discretion upon 30 days prior written notice.

6. <u>Minimum Annual Gross Revenues</u>. Franchisee and AHF agree that in lieu of the Minimum Annual Gross Revenues or Minimum Monthly Gross Revenues to be met as set forth in **Section 12.4** of each of the Franchise Agreements, Franchisee must achieve Gross Revenues combined for all of the Franchised AHS Businesses under all of the Franchise Agreements, in the following amounts:

Period	Aggregate Minimum Annual Gross Revenues
to	\$
to	\$
and thereafter	\$

This table shall be deemed to supersede and replace the table of Minimum Annual Gross Revenues or Minimum Monthly Gross Revenues set forth in the addenda to the Franchise Agreements. Franchisee and AHF agree that the consequences and rights of AHF set forth in the most recent of the Franchise Agreements shall apply for all of the Franchise Agreements and Franchised Territories in the event there is a failure to meet the foregoing Minimum Annual Gross Revenues.

7. <u>Minimum Individual Marketing</u>. AHF agrees that Franchisee may combine the marketing efforts of the Franchised AHS Businesses, and to enable this, AHF will agree that to a combined

Minimum Individual Marketing Expenditure or Minimum Individual Advertising Expenditure that covers all of the Franchised AHS Businesses to be used on local marketing in the Franchised Territories together. Therefore, **Section 13.3.a** of each Franchise Agreement is hereby amended to provide that Franchisee shall spend a combined Minimum Individual Marketing Expenditure or Minimum Individual Advertising Expenditure for the local marketing of all of the Franchised AHS Businesses in the following amounts:

Period	Aggregate Minimum Individual Marketing Expenditure
to	\$
to	\$
and thereafter	\$

Fully executed this day of	, 202
AHF:	FRANCHISEE:
ACE HANDYMAN FRANCHISING, INC., a Colorado corporation	IF AN INDIVIDUAL:
By:	
Date:	Signature
	Print Name:
	IF A PARTNERSHIP, LIMITED LIABILITY COMPANY, CORPORATION, OR OTHER ENTITY:
	Print Name of Company
	By:
	Signature
	Print Name and Title:

ATTACHMENT E

FORM OF SUCCESSOR FRANCHISE RIDER TO FRANCHISE AGREEMENT

FORM OF SUCCESSOR FRANCHISE RIDER TO FRANCHISE AGREEMENT

ACE HANDYMAN FRANCHISING, INC., a Colorado corporation ("AHF"), and the undersigned franchisee ("Franchisee") entered into that certain Ace Handyman Services Franchise Agreement ("Agreement") contemporaneously herewith, and desire to supplement and amend certain terms and conditions of such Agreement by this Successor Franchise Rider to Franchise Agreement ("Rider"). The parties therefore agree as follows:

1. <u>Term.</u> Section 3.1 of the Agreement is hereby deleted in its entirety with the following substituted in its place:

3.1. <u>Term</u>.

The term of this Agreement shall commence on ________, 20____ (the "Commencement Date") and is for a period of ______ years from the Commencement Date, unless sooner terminated as provided herein. Franchisee agrees to operate the AHS Business for the entire term of this Agreement.

2. Business Location. Section 4.1 of the Agreement is hereby deleted in its entirety with the following substituted in its place:

4.1. **Business Location.**

Franchisee will operate its AHS Business from an office ("Business Location") outside of Franchisee's residence or the residence of its Principal Representative or Franchise Manager, as those terms are defined in Section 6.1. Franchisee is solely responsible for selecting and acquiring the premises for its Business Location. The Business Location must be centrally located in the Territory and meet AHF's current standards. Franchisee must obtain AHF's prior written consent to the Business Location. The Business Location is designated in the Addendum. The Business Location may not be relocated without the prior written consent of AHF. Franchisee agrees to comply with any additional standards established by AHF from time to time regarding Franchisee's Business Location within 30 days of receipt of written notice from AHF of such standards.

3. <u>Initial Franchise Fee.</u> Section 5.1 of the Agreement and Paragraph 3 of the Addendum to Franchise Agreement ("Addendum") are hereby deleted in their entirety, with the following substituted in the place of Section 5.1:

5.1. Successor Franchise Fee.

Franchisee	will	pay	to	AHF	a	successor	franchise	fee	of
\$, wl	nich sh	all b	e due an	d pa	yable as foll	ows:		

Franchisee acknowledges and agrees that the successor franchise fee represents payment for the grant of successor franchise rights, that AHF has earned the successor franchise fee upon receipt thereof, and that the successor franchise fee is not refundable to Franchisee once paid.

- 4. <u>Initial Training.</u> Sections 6.1, 6.2, 6.3, and 10.1.a of the Agreement are hereby deleted in their entirety. The first sentence of Section 11.1.d of the Agreement is also hereby deleted. Franchisee specifically acknowledges that it has already received training from AHF and it is not entitled to the Initial Training Program or the On-Site Training, as defined in Sections 6.1 and 6.3 of the Agreement, as a result of this exercise of successor franchise rights. If, under the terms of the Agreement or otherwise, Franchisee needs or desires to have persons attend AHF's Initial Training Program, then Franchisee must pay the then current tuition charged by AHF for those persons, in addition to all wages, travel and living expenses incurred in connection with their attendance at the Initial Training Program. Notwithstanding the foregoing, the definitions of "Principal Representative," "Franchise Manager," "Office Manager," "Initial Training Program," and "On-Site Training," as provided in Sections 6.1 and 6.3 shall remain effective and have the meaning set forth therein.
- 5. <u>Commencement of Operations and Active Operations</u>. Sections 7.3 and 7.4 of the Agreement are hereby deleted in their entirety, with the following substituted in the place of Section 7.4:

7.4. <u>Active Operations.</u>

Unless otherwise agreed in writing by AHF and Franchisee, Franchisee must actively promote and continue to operate its AHS Business in accordance with the Operations Manual (as defined below) and this Agreement; unless AHF gives its prior written consent to Franchisee to temporarily suspend its operations, which consent may be withheld by AHF for any reason.

- 6. AHF's Development Assistance. Article 8 of the Agreement is hereby deleted in its entirety. Franchisee acknowledges that it has previously received the development assistance set forth in Article 8, and it is not entitled to additional development assistance as a result of this exercise of successor franchise rights.
- 7. <u>Operations Manual</u>. Franchisee acknowledges that it has already received a copy of AHF's Operations Manual and it will not be entitled to another copy thereof as a result of this exercise of successor franchise rights. **Section 9.1** of the Agreement is hereby modified accordingly.
- 8. <u>Minimum Annual Gross Revenues</u>. Section 12.4 of the Agreement and Paragraph 5 of the Addendum are hereby deleted in their entirety, with the following substituted in the place of Section 12.4:

12.4. Minimum Annual Gross Revenues.

Franchisee must achieve Gross Revenues of \$______ on an annual basis (the "Minimum Annual Gross Revenues"). Within 60 days following the end of each Operating Year, as defined below, AHF will compute Franchisee's total Gross Revenues for the just completed Operating Year. If Franchisee fails to achieve the Minimum Annual Gross Revenues in any Operating Year, Franchisee shall pay AHF the Royalty Fee and National Brand Fee based on the Minimum Annual Gross Revenues instead of its actual Gross Revenues. AHF will invoice Franchisee for the difference between the amount of Royalty Fee and National Brand Fee due based on the Minimum Annual Gross Revenues, less the actual Royalty Fee and National Brand Fee paid by Franchisee to AHF during the just completed Operating Year, which amount shall be due within 15 days following

invoicing. If Franchisee's actual Gross Revenues are less than the applicable Minimum Annual Gross Revenue in two consecutive Operating Years, or in any three Operating Years during the term of this Agreement, then AHF shall have the right to (i) terminate this Agreement, (ii) award additional franchises within the Territory or otherwise permit another franchisee to operate within the Territory, (iii) reduce the geographic size of the Territory, or (iv) any combination thereof, after providing notice to Franchisee in accordance with Section 18.2.f below. AHF has the right, in its sole discretion and on a case-by-case basis, to waive the obligation of Franchisee or any other franchisee of AHF to meet the Minimum Annual Gross Revenues requirement or to pay any fees or make expenditures calculated based on the applicable Minimum Annual Gross Revenues. Franchisee acknowledges that AHF is entering into this Agreement with the expectation that it will receive Royalty Fees based on the greater of Franchisees actual Gross Revenues or the Minimum Annual Gross Revenues over the full term of this Agreement. The term "Operating Year" means a calendar year beginning on the commencement date set forth in the Addendum and ending on the anniversary date thereof, and each subsequent calendar year thereafter.

- 9. <u>Initial Marketing Campaign</u>. Section 13.2 of the Agreement is hereby deleted in its entirety.
- 10. <u>Minimum Individual Marketing</u>. Section 13.3.a of the Agreement is hereby deleted in its entirety with the following substituted in its place:
 - a. Franchisee shall spend on local marketing and promotion a minimum of \$50,000 per year as the "Minimum Individual Marketing Expenditure" during the term of this Agreement. These amounts are in addition to the National Brand Fee that Franchisee will pay to AHF. AHF will review Franchisee's spend amounts each six months and if Franchisee is not on pace to meet or exceed the required Minimum Individual Marketing Expenditure, Franchisee must increase its Minimum Individual Marketing Expenditure for the second six months of the applicable 12-month period to meet the required annual Minimum Individual Marketing Expenditure. Any expenditures on online marketing services and products acquired through a third party vendor shall count toward this Minimum Individual Marketing Expenditure. AHF may direct that Franchisee pay all or a portion of its Minimum Individual Marketing Expenditure to a Local Advertising Group, as defined below.
- Release. Franchisee for itself, its employees, officers, directors, shareholders, members, partners, agents, representatives, successors and assigns, hereby fully and forever unconditionally releases and discharges AHF and its employees, officers, directors, shareholders, members, partners, agents, representatives, successors and assigns (collectively referred to as the "AHF Affiliates") from any and all claims, demands, obligations, actions, liabilities and damages of every kind and nature whatsoever, in law or in equity, whether known or unknown to it, which it may now have against AHF or the AHF Affiliates, or which may hereafter be discovered, in connection with, as a result of, or in any way arising from, any relationship or transaction with AHF or the AHF Affiliates, however characterized or described, from the beginning of time until the date of this Agreement.

Fully executed this day of	, 202
AHF:	FRANCHISEE:
CE HANDYMAN FRANCHISING, INC., Colorado corporation	IF AN INDIVIDUAL:
By: Date:	Franchisee, Individually
	Print Name:
	IF A PARTNERSHIP, LIMITED LIABILITY COMPANY, CORPORATION, OR OTHER ENTITY:
	By:
	Print Name:

Effectiveness of Agreement. The terms and conditions of this Rider are in addition

12.

ATTACHMENT F CURRENT FORM OF GENERAL RELEASE

THE FOLLOWING FORM OF GENERAL RELEASE AGREEMENT IS A SAMPLE OF OUR CURRENT FORM OF GENERAL RELEASE AGREEMENT. THIS AGREEMENT IS OFTEN MODIFIED TO CONFORM TO THE FACTS SURROUNDING THE EVENT OR INCORPORATED INTO A LARGER AGREEMENT WHICH MORE PRECISELY ADDRESSES THE EVENT. WE MAKE NO REPRESENTATION OR GUARANTY THAT THE GENERAL RELEASE AGREEMENT YOU MAY BE REQUIRED TO SIGN WILL BE IDENTICAL TO THE GENERAL RELEASE AGREEMENT SET FORTH BELOW.

GENERAL RELEASE AGREEMENT
THIS GENERAL RELEASE AGREEMENT (this " Agreement ") is made as or , 202 by and between ACE HANDYMAN FRANCHISING, INC., a
Colorado corporation ("AHF") and
RECITALS
A. AHF and Franchisee entered into that certain Franchise Agreement dated, 202, (the "Franchise Agreement").
B. Franchisee desires to its rights and obligations under Franchise Agreement.
C. As a condition to the of Franchisee's rights and obligations under the Franchise Agreement, AHF requires Franchisee to execute this Agreement.

AGREEMENT

NOW, THEREFORE, in consideration of the terms and conditions set forth below, and other good and valuable consideration, the receipt, adequacy and sufficiency of which are acknowledged, the parties hereto, intending to be legally bound, agree as follows:

- 1. <u>Release</u>. Franchisee, for itself, its principals, owners, directors, officers, employees, heirs, assigns, agents and representatives, fully and forever unconditionally releases and discharges AHF, and its shareholders, directors, officers, employees, successors, assigns, agents and representatives (collectively referred to as "AHF Affiliates") from any and all claims, demands, obligations, actions, liabilities and damages of every kind and nature whatsoever, in law or in equity, whether known or unknown to it, which it may now have against AHF or the AHF Affiliates or which it may discover hereafter, in connection with, as a result of, or in any way arising from, any relationship or transaction with AHF or the AHF Affiliates, however characterized or described, from the beginning of time until the date of this Agreement.
- 2. <u>Notice</u>. Any notice, request, demand, statement or consent made under this Agreement shall be in writing and shall be personally delivered or sent by registered or certified mail, return receipt requested, and shall be deemed given when personally delivered or three days after deposit in the United States Mail, postage prepaid, and properly addressed to the other party at its address as set forth below. Each party may designate a change of address by notice to the other party in accordance with this Section.

If to Franchisee:	
If to AHF:	Ace Handyman Franchising, Inc. 390 Union Boulevard, Suite 520 Lakewood, Colorado 80228 Attention: Andy Bell
3. <u>Colorado Laws</u> . This Agreement shall Should any provision of this Agreement be found to vi Colorado or of the United States, that provision shall be to such statutes or court decisions to affect the intent of the	deemed to be amended to comply with and conform
4. <u>Binding Effect</u> . This Agreement shall successors, assigns, trustees, receivers, personal represent	l be binding upon and inure to the benefit of the tatives, legatees and devisees of the parties.
5. <u>Attorneys' Fees</u> . Each party shall be resexpenses incurred in the preparation of this Agreement. parties based upon an alleged breach or default in their rangement, the prevailing party in the action shall be entered the non-prevailing party(ies).	respective obligations to be fulfilled pursuant to this
6. <u>Entirety</u> . This Agreement embodies the parties and supersedes all prior agreements and understan	ne entire agreement and understanding between the adings related to the subject matter hereof.
FOR WASHINGTON FRANCHISEES ONLY:	
A release or waiver of rights executed by a franchis Franchise Investment Protection Act, RCW 19.100, or a pursuant to a negotiated settlement after the agreement independent counsel.	any rule or order thereunder except when executed
IN WITNESS WHEREOF, the parties have exwritten.	xecuted this Agreement as of the date first above
AHF:	FRANCHISEE:
ACE HANDYMAN FRANCHISING, INC.	
By: Andy Bell, CEO	By:

Date: _____

Date:

ATTACHMENT G OPERATIONS MANUAL TABLE OF CONTENTS



Operations Manual Table of Contents

Introduction to the Manual.	Chapter 1
Chapter 1 Page 1	Ownership of the Manual
Chapter 1 Page 1	Purpose of the Manual
Chapter 1 Page 1	Importance of Confidentiality
Chapter 1 Page 2	Content Updates
Chapter 1 Page 2	Submitting Suggestions
Chapter 1 Page 2	Disclaimer
Introduction to Your Franchi	ise System Chapter 2
Chapter 2 Page 1	Welcome
Chapter 2 Page 2	History of the Company
Chapter 2 Page 2	Vision, Mission, and Values
Chapter 2 Page 3	T.E.A.M.
Chapter 2 Page 4	Franchise Support
Chapter 2 Page 4	Pre-Opening Assistance
Chapter 2 Page 4	Handyman Matters University
Chapter 2 Page 4	Grand Opening & Post-Opening Onsite Assistance
Chapter 2 Page 4	Ongoing Training & Support
Chapter 2 Page 5	Conferences
Chapter 2 Page 5	Communication Protocol
Chapter 2 Page 6	Profitable Partnerships (Purchasing Power for
	Profit)
Chapter 2 Page 7	Visits from the HMFC Support Center TEAM
Chapter 2 Page 7	Franchisee Responsibilities
Chapter 2 Page 8	Grow a Base of Happy Customers
Chapter 2 Page 8	Support the Franchisor's Systems and Brand Values
Chapter 2 Page 8	Run a Profitable Business
Chapter 2 Page 9	Responsibilities to Customers
Chapter 2 Page 9	Responsibilities to Your Employees
Chapter 2 Page 10	Responsibilities to the Franchisor
Chapter 2 Page 10	Responsibilities to Your Fellow Franchisees
Chapter 2 Page 10	Franchise Candidate Validation Calls
Financials	Chapter 3
Chapter 3 Page 1	Financial Basics
Chapter 3 Page 2	The CPA Relationship
Chapter 3 Page 3	Taxes
Chapter 3 Page 3	Understanding Basic Accounting



Chapter 3 Page 3 **Financial Statements** Chapter 3 Page 5 Efficiency of the Cycle Chapter 3 Page 6 Working Capital Cycle Chapter 3 Page 6 **Budgeting** Chapter 3 Page 7 Management Chapter 3 Page 7 **Analysis** Chapter 3 Page 8 Breakeven Chapter 3 Page 9 Price-Cost-Volume Relationship Chapter 3 Page 10 Credit Chapter 3 Page 10 Lines of Credit Chapter 3 Page 10 Credit Cards for Employees Chapter 3 Page 11 Fees & Expenses Chapter 3 Page 11 Royalties and Brand Fund Contribution Chapter 3 Page 12 Software Fees Chapter 3 Page 12 **Potential Additional Expenses** Chapter 3 Page 13 **Payroll Processing** Chapter 3 Page 14 **Credit Card Processing** Administrative Chapter 4 Chapter 4 Page 1 **Independently Owned and Operated Professional Advisors** Chapter 4 Page 1 Chapter 4 Page 2 Technology Chapter 4 Page 2 Requirements Chapter 4 Page 4 Data Management Chapter 4 Page 5 Compliance Chapter 4 Page 5 **EPA Lead Abatement Certification** Chapter 4 Page 5 **Insurance Requirements** Chapter 4 Page 5 **Operational Letters** Personnel...... Chapter 5 Chapter 5 Page 1 **Employee Relations** Chapter 5 Page 1 **Employment Law Basics** Chapter 5 Page 1 **Employee Rights & Employer Responsibilities** Chapter 5 Page 1 Workplace Safety & OSHA Chapter 5 Page 2 Sexual Harassment Chapter 5 Page 3 Drug & Alcohol Policy Chapter 5 Page 3 State & Federal Employment Law Resources Chapter 5 Page 4 Staffing Chapter 5 Page 4 **Employee Roles and Responsibilities** Chapter 5 Page 6 **Job Descriptions** Chapter 5 Page 8 Recruiting Chapter 5 Page 8 Sources Chapter 5 Page 8 Advertisements



Chapter 5 Page 9 Requirements to Advertise Open Positions Chapter 5 Page 9 Offering Benefits Chapter 5 Page 9 **Applicants** Chapter 5 Page 9 **Application Form** Chapter 5 Page 9 Confidentiality Chapter 5 Page 10 Interviewing Chapter 5 Page 12 **Conditional Offers of Employment** Chapter 5 Page 12 **Background Checks** Chapter 5 Page 13 **Drug Screening** Chapter 5 Page 13 **Probationary Period** Chapter 5 Page 13 At Will Employment Chapter 5 Page 14 **New Employee Paperwork** Chapter 5 Page 14 **Personnel Policies** Chapter 5 Page 15 Non-Compete Chapter 5 Page 15 **Paying Craftsmen** Chapter 5 Page 15 Wages Chapter 5 Page 15 **Travel Time** Chapter 5 Page 16 **Evening Before Calls** Chapter 5 Page 16 Redos & Warranty Work Chapter 5 Page 17 **TEAM Meetings** Chapter 5 Page 17 **Orientation & Training** Chapter 5 Page 17 Orientation Chapter 5 Page 17 The 4-Step Training Process Chapter 5 Page 18 On the Job Training Chapter 5 Page 18 Motivation Chapter 5 Page 19 **Evaluation, Discipline, & Termination** Chapter 5 Page 19 **Evaluations** Chapter 5 Page 19 Discipline Chapter 5 Page 20 Termination & Resignation Chapter 5 Page 21 Handyman Matters Recommended Organizational Structure Chapter 5 Page 23 Disclaimer Operations...... Chapter 6 Chapter 6 Page 1 Introduction Chapter 6 Page 1 **Revenue Streams** Chapter 6 Page 1 Residential Chapter 6 Page 2 Commercial Chapter 6 Page 2 **Key Performance Indicators (KPIs)** Chapter 6 Page 3 **Hours of Operation** Chapter 6 Page 3 Equipment

Tools

Chapter 6 Page 3



Chapter 6 Page 3 Office Equipment

Chapter 6 Page 4 **Telephone Standards**Chapter 6 Page 5 **Managing "Inventory"**

Chapter 6 Page 5 Service Path

Chapter 6 Page 5 Creating a Service Path Culture

Chapter 6 Page 6 Charitable Contributions
Chapter 6 Page 6 Customer Complaints

Chapter 6 Page 7 Importance of Online Customer Reviews

Chapter 6 Page 7 Call Blueprint

Chapter 6 Page 7 **TEAM Meetings**

Chapter 6 Page 8 Cash Flow
Chapter 6 Page 8 Cash

Chapter 6 Page 8 Checks

Chapter 6 Page 8 Credit Card Payments
Chapter 6 Page 9 NSF & Late Payments

ATTACHMENT H

LIST OF FRANCHISEES

LIST OF FRANCHISEES As of December 31, 2023 2024

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Steve Hartman	3516 Vann Road, Ste 108	Birmingham NE	Alabama	35235	205-263-3961	2
Richard & Anne Compton	14 Live Oak Street, Unit D	Gulf Breeze	Alabama	32561	850-455-6600	1
Cole Guthrie*	851 Watts Lane	Nashville	Tennessee	37209	651-581-2288	1
Glen & Stephanie Steinbrunner	1176 E Warner Road, Bldg C Suite 102	Gilbert	Arizona	85296	480-716-9800	4
David Blair	10210 N 32 nd Street Bldg B Ste 4	Phoenix	Arizona	85208	480-275-5000	4
Scott Joseph	9102 E. Holmes St	Tucson	Arizona	85710	520-207-1335	1
Rob Salvino	2706 N Silverbell Rd. Suite 3110	Tucson South	Arizona	85745	520-524-1423	2
Earnest Perkins	11129 West Alabama Ave.	Youngtown	Arizona	85363	623-624-6373	4
James Dyer	200 Cornerstone Boulevard, Ste 234	Hot Springs	Arkansas	71913	501-229-5550	1
Steve Modi	1019 Main Street	Little Rock	Arkansas	72202	501-451-3266	1
John Green	3824 Cawood Lane, Suite "A"	Springdale	Arkansas	72762	479-633-0022	1
Aaron Rogelstad	8900 Brentwood Blvd, Suite C	Brentwood	California	94513	925-684-4966	1
Chris Arnold	1101 California Ave, Suite #11	Corona	California	92881	951-737-8000	<u> 42</u>
Chris Castor	873 Embarcadero Dr. Unit 4	El Dorado Hills	California	95762	530-503-9244	1
Kareem Al Mudhafar	11919 Valley View St., Ste B	Garden Grove	California	92845	714-902-1440	1
Blake Keiser	16065 Emerald City Drive	Valley Center	California	92082	760-456-7801	1
Kevin Jenkins	2725 Central Avenue, Ste A	McKinleyville	California	95519	707-839-1545	1
Theresa Mitchell	2915 Kitchener Court	Oakland	California	94602	510-531-4300	4
Scott Weller	1018 Artesia Blvd	Redondo Beach	California	90278	424-408-0081	3
Eric Hassett	282 Woodside Plaza, Unit A	Redwood City	California	94061	650-284-4025	2
Kent Cusick	801 Riverside Ave, Suite 104	Roseville	California	95678	916-633-2140	1
Mark Ivary	3201 W Benjamin Holt Drive, Ste 156	Stockton	California	95219	209-951-8050	1
Josh Lindsay	1909 E Main Street	Ventura	California	93001	805-410-1122	2

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Doug Ralph	487 Windchime Place	Colorado Springs	Colorado	80919	719-391-9491	1
Glenn Brink	220 Smith Street	Fort Collins	Colorado	80525	970-460-4900	2
Jim Daigle	9 Shady Lane	Norwich	Connecticut	06360	860-603-0000	1
John Ordonez	233 Boston Post Road, #2A	Orange	Connecticut	06477	475-209-9700	1
Griffin Wolff	690 Main Street South, Ste B	Woodbury	Connecticut	06798	203-981-9266	1
Rodrigo Narajno	428A Main Street, 2nd Floor S	-Greenwich	Connecticut	10504	914-219-5172	1
Steve & Christy Smith Andy Grant	378 Center Point Circle # 123240 Boston Post Road, Ste 20	Waterford Altam onte Springs	Connecticut Flo	32701 <u>63</u> 85	407-710- 2700 <u>860-300-</u> 4548	<u>21</u>
Bill Ciskowski	31038 Country Gardens, Ste D-1	<u>Dagsboro</u>	<u>Delaware</u>	<u>19939</u>	302-604-9389	1
Steve Adler	1055 Yamato Rd. #112	Boca Raton	Florida	33498	561-240-6000	3
Michael Wynn	9148 Bonita Beach Rd, Suite 207	Bonita Springs	Florida	34135	239-366-5039	3 6
<u>Darren Rodgers</u>	214 West University Avenue, Ste H	<u>Gainesville</u>	<u>Florida</u>	32601	352-888-6680	1
Bob and Sue Bowles	6950 Philips Hwy, Suite 27	Jacksonville	Florida	32216	904-902-4223	4
Robert Morshorak	2225 E Edgewood Drive Ste 14	Lakeland	Florida	33803	863-777-2343	4
Joel Freedland	21326 Coakley Ln.	Land O Lakes	Florida	34639	813-678-2600	1
Bill Worrall	230 Commercial Blvd, Ste B	Lauderdale By the Sea	Florida	23430	954-776-5812	1
Janel Boshoff	2189 N. Lecanto Hwy	Lecanto	Florida	34461	352-249-7218	2
Vic Hyder	12150 SW 128th Ct. #131	Miami	Florida	33186	305-964-7612	1
Rudy Jara	5201 Blue Lagoon Drive	Miami	Florida	33126	305-359-8064	1
Austin Burkett	4409 Grand Blvd, Ste 250	New Port Richey	<u>Florida</u>	34652	727-330-3200	3
Drew Bruff	1900 Coral Way, Ste 300	Miami	Florida	33145	786-686-6337	1
Michael Ball	1402 Cat Mar Rd Ste D	Niceville	Florida	32578	850-299-4253	1
Kyle O'Connor	637 N Tamiani Trail	Nokomis	Florida	34275	941-621-3919	2
Joyce & Desmond Fernando	3801 Avalon Park East Blvd Ste 240	<u>Orlando</u>	<u>Florida</u>	32828	321-300-3686	1
Richard & Anne Compton	2190 Berg Street	Pensacola	Florida	32514	850-455-6600	<u>24</u>
Kurt & Angela George	701 Market St, Suite #111	St Augustine	Florida	32095	904-436-5977	2

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Jonathan Porcelli	4636 Central Ave	St Petersburg South	Florida	33711	727-873-7268	2
Mark Oldfield	2930 W County Highway 30A, Ste 201	Santa Rosa Beach	<u>Florida</u>	32459	850-558-3201	1
Quinn Hoffman	17860 SE 109th Avenue, Ste 604	Summerfield	Florida	34491	352-706-2367	4 <u>1</u>
John Rigney	2361 Vista Pkwy Unit 3	West Palm Beach	Florida	33411	561-709-2351	3
Roger Blair	10026 S. Federal Highway	Port St Lucie	<u>Florida</u>	<u>34952</u>	<u>754-234-7455</u>	1
Lucas & Carla Elias	12430 Langstaff Dr.	Windermere	Florida	34786	407-881-3771	1
Tom Thayer Amy Gambrell	5344 9th Street 3271 US Highway 84, Ste 103 100	Blackshear Zeph yrhills	<u>Georgia</u> Florida	3151633 542	912-807- 7767813-923- 1223	1
Mike Fisk Jason Jackinski	3715 Northside Parkway1954 Airport Road, Ste 400-710260	Atlanta Chamble e	Georgia	30327 <u>30</u> 341	678-916- 6400 <u>580-945-</u> 1675	<u>14</u>
Matt CochranKen Washington	3271 US Hwy 84, Suite 100 301 Green Street NW Ste 200-i	GainesvilleBlackshear	Georgia	30501 ³¹ 516	770-530- 2519 <mark>912-807-</mark> 7767	1
Alicia Todisco	5150 Stilesboro Road, #510	Kennesaw	Georgia	30152	770-627-7770	2
Darryl Sizemore	283 Swanson Drive, Bldg 100, Suite #101	Lawrenceville	Georgia	30043	404-381-1117	1
Bill Ward	288 S Main Street Ste 103	Madison	Georgia	30650	706-553-4222	1
Ed Vickers	5950 Crooked Road, Ste 170 N	Peachtree Corners	Georgia	30092	470-704-9758	1
Tom Glenn	100 Helpful Place	Ringgold	Georgia	30736	423-716-6475	1
David Mountain	11 John Davenport Drive, Ste A	Rome	Georgia	30165	706-314-0448	1
Phillip Norrell	2350 Prince Ave. Unit 11	Athens	Georgia	<u>30606</u>	706-363-3936	1
Bob KelleyBen Richardson	45 W Crossville Rd Ste 51B	Roswell	Georgia	30075	678-941-4500	2
Randy West	5698 Ogeechee Rd. Unit B	Savannah	Georgia	31405	912-777-5861	1
Rudy Richards	2795 Main St W # 21-A	Snellville	Georgia	30078	470-567-0809	2
Mike Davis	4480 Valnorth Drive, Ste D	Valdosta	Georgia	31602	229-740-6911	1
Stacey & Melissa Hanson	281 Carl Vinson Pkwy Ste L	Warner Robbins	Georgia	31088	478-334-2385	4
Nick Coe	1111 S Broadway Ave, Suite	Boise	Idaho	83709	208-900-5090	1
Brian Henebry	9460 W Fairview Ave, Suite 145	Boise	Idaho	83704	208-593-6939	1
Daniel Moldenhauer	662 N 4128 E #3	Rigby	Idaho	83442	208-757-5888	<u>23</u>

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Chris Duncan	495 Jetway Drive, Unit D	Coeur D'Alene	<u>Idaho</u>	<u>59714</u>	406-599-1153	1
Jerris Duncan	2441 Village Green Place, Ste D	Champaign	Illinois	61822	217-417-8403	1
Dan Kern	707 Davis Road	Elgin	Illinois	60123	847-930-5053	1
Kyle Sheahan	2760 Skokie Valley Road	Highland Park	Illinois	60035	847-432-0029	4
Mike Schmidt	4155 W Orleans Street	McHenry	Illinois	60050	847-586-8700	1
Chris Meadows	12410 W Horseshoe Drive, Unit #2	New Lenox	Illinois	60451	815-418-6336	1
Kim Porter	2019 S Route 59	Plainfield	Illinois	60586	779-341-1234	1
Matthew Garrett	401 S Carlton Ave, Suite 101	Wheaton	Illinois	60187	630-765-5070	1
Sid Canchester	207 N Harlem Ave, Ste 50	Oak Park	Illinois	60302	708-773-0218	1
Vinay Iyer	231 S Rangeline Rd Ste F	Caramel	Indiana	46033	317-819-6696	1
Brad Freeman	510 S Green River Road	Evansville	Indiana	47715	812-631-5858	1
Bryan Harmon	190 Commerce Drive Ste 7	Franklin	Indiana	46131	317-668-3947	2
Joe Bardense	110 N Randolph St.	Garrett	Indiana	46738	260-465-0555	3
Rebecca Simon	921 E 86 th Street Ste 150	Indianapolis	Indiana	46240	317-708-7320	1
Turner & Jade Schenzel	2175 BW 86th Street Ste 6A	Clive	Iowa	50325	531-306-9922	1
Edward Gadelha	285 7th Avenue	Marion	Iowa	52302	319-777-1199	1
Zachary Vinduska	8406 West Maple	Wichita	Kansas	67209	316-773-0303	4
Eric Moore	8445 US Hwy 42 Suite A	Florence	Kentucky	41042	859-869-8400	2
Travis Foster	2525 Harrodsburg Rd Suite #220	Lexington	Kentucky	40504	859-399-6191	2
David Abadie	2713 Athania Pkwy, #200	Metairie	Louisiana	70005	504.372-1369	1
Jordan Brigham	250 US Route 1	Scarborough	Maine	4074	207-819-4585	3
Rod Wheeler	1761 A Richie Station Ct	Capitol Heights	Maryland	20743	301-390-2882	1
Brian Wigutow	6235 Sykesville Road	Eldersburg	Maryland	21784	410-549-9696	5
Phil Savard	413 Headquarters Drive, Suite 4F	Millersville	Maryland	21108	443-645-0721	3
Ifetunde Oyelowo	30 Courthouse Square	Rockville	Maryland	20850	240-331-5995	4

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Donald Gradwell	1208 Butterworth Ct.	Stevensville	Maryland	21666	410-793-4499	3
Darin Krum	4205 State Highway, Suite 8	Eastham	Massachusetts	02642	774-207-0519	1
Bob Samson	401 Lowell Street, Suite 2	Lexington	Massachusetts	02420	781-369-9557	7
Eric Smoller	665 Salem St Ste 210	Malden	Massachusetts	02148	781-333-4118	1
Max & Allan Houle	86 Lafayette Street, Ste 250	<u>Salem</u>	<u>Massachusetts</u>	01970	973-435-8812	1
Ken Fernandes	299 Depot Street, Ste B	South Easton	Massachusetts	02375	508-682-1709	2
Chris Toomey	2075 E West Maple Rd Ste B- 203	Commerce Township	Michigan	48390	248-983-3900	<u>67</u>
Josh Schnakenberg	1345 Walker Village Drive NW	Grand Rapids	Michigan	49534	616-320-5800	3
Terry Steffen	2650 Horizon Dr. SE, Suite 235	Grand Rapids	Michigan	49546	616-805-3670	1
Gerard Perczenkiuk	1520 S Lapeer Rd, Ste 212	Lake Orion	Michigan	48360	248-940-1550	1
Scott Hedges	307 W 6 th Street Ste 202A	Royal Oak	Michigan	48067	248-716-8111	4
Paul Manion	29623 Northwestern Highway	Southfield	Michigan	48034	248-406-4989	2
Dan & Lindsay Smith	876 Robinwood Court , Unit A	Traverse City	Michigan	49686	231-247-0820	2
Emily Johnson	3263 E Colby St. Ste D	Whitehall	Michigan	49461	231-981-8601	1
Tim Owata	9680 Foley Blvd NW, Suite 2	Coon Rapids	Minnesota	55433	763-762-6578	2
Mike Sitek	6224 Halifax Avenue South	Edina	Minnesota	55424	651-784-3777	1
Elena Nelson	3809 Nicollet Ave S.	Minneapolis North	Minnesota	55409	612-822-9952	2
Callie Kuyava	14165 James Road, Suite #200	Rogers	Minnesota	55374	763-330-0920	1
Beth Young	901 Twelve Oaks Center Dr # 930	Wayzata	Minnesota	55391	952-522-3888	1
Toleman Mills	104 Business Park, Ste E	Dr Ridgeland	Mississippi	39157	601-499-4802	1
Marty Smith	301 Sovereign Court, Suite 103	Ballwin	Missouri	63011	314-887-1238	1
David Pierce	8002 North Oak Traffieway, Suite #134	Kansas City	Missouri	64118	816-734-1370	2
Chris Duncan	495 Jetway Drive, Unit D	Belgrade	Montana	59714	406-599-1153	1
Mariah Joos	6490 US Highway 93 S, Ste 100	Whitefish	Montana	59937	406-730-1770	1
Turner & Jade Schenzel	10824 Old Mill Rd, #10-9	Omana	Nebraska	68104	402-702-7274	2

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Brad Hess	8689 W Sahara Ave, #250	Las Vegas	Nevada	89117	702-323-6431	1
Karen Henderson	5 Depot Street, Ste 2	Goffstown	New Hampshire	03045	603-384-2862	1
Tom Durney	17 Linden Place	Red Bank	New Jersey	07701	732-737-8677	2
David Palmer	940 Kinderkamack Road	River Edge	New Jersey	07661	201-262-0900	1
Chris Maglin	940 S Avenue W, Unit D	Westfield	New Jersey	07090	908-543-8900	3
Tamiour Hassin	5340 S Verde Mesa	Allentown	New Jersey	07728	303-359-0458	3
Victor Ulibari	5100 Juan Tabo Blvd NE	Albuquerque	New Mexico	87111	505-569-0003	<u> 12</u>
Rodrigo Narajno	428A Main Street, 2nd Floor S	Armonk	New York	10504	914-219-5172	1
Nate Creery	1435 Upper Front Street, #3	Binghamton	New York	13901	607-323-0766	1
Stephen Lutter	298 Northhampton St Suite 201	Buffalo	New York	14208	716-427-0427	1
Owen Peterson	65 Lakehill Road, Ste 14	Burnt Hills	New York	12027	518-344-6079	1
Mike Steppe	2954 Rt 54A	Penn Yan	New York	14527	315-536-1411	2
Dylan Dechant	115 Elk Mountain Rd Ste B	Asheville	North Carolina	28804	828-900-0900	1
Randy Saunders	3141 Airport Road	Carthage	North Carolina	28327	910-939-8449	1
Alexander Arguello	228 Westinghouse Blvd Unit 108	Charlotte	North Carolina	28273	980-375-0700	1
Corey Evans	9700 Research Drive	Charlotte	North Carolina	28262	704-461-2772	1
Nate Stobbe	2800-26 Raeford Road	Fayetteville	North Carolina	28303	910-500-0550	1
Tim Howard	7345 W Friendly Ave, Suite D	Greensboro	North Carolina	27410	336-245-4485	2
John & Erica Holland	1231 Halltown Road	Jacksonville	North Carolina	28546	910-333-1060	2
Brandon Rutherford	4112 N Croatan Hwy Unit A	Kitty Hawk	North Carolina	27949	252-297-8900	1
Wess Cason	1312 Matthews Mint Hill Rd	Matthews	North Carolina	28105	704-443-7153	2
Dave Wigfield Tyler Elliott	106 Chasestone Lane 125 E Plaza Drive, Ste 116	Mooresville Mooresville	North Carolina	28117	704-765-6800	1
Todd Manzer	2408 Madison Drive	North Myrtle Beach	North Carolina	29582	843-491-9800	1
Artem Lopatchenko	1818 St. Albans Dr. Ste 103A	Raleigh	North Carolina	27609	919-999-3334	4 <u>5</u>
Mike Geisler	2195 Hwy 64 East	Sapphire	North Carolina	28774	828-547-0455	<u> 12</u>

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Matt Denevan	405 W Main Sve Ste 4R	Fargo	North Dakota	58078	701-532-2230	1
Sheri Adkins	840 Rothrock Road, Ste 101	Copley	Ohio	44321	330-598-1618	<u>12</u>
Chuck Shive	3143 Western Row Rd	Maineville	Ohio	45039	513-982-4988	1
Darin Fleming	157 A Mansfield Avenue	Shelby	Ohio	44875	419-989-6633	1
David Drickhammer	795 Sharon Drive Ste 226	West Lake	Ohio	44145	440-925-5222	3
Paul & Laura Hammer	101 East College Ave, #2-C	Westerville	Ohio	43081	614-212-4422	5
Nathan Acker	100 NW 150th St Ste C 2	Edmond	Oklahoma	73013	405-861-8300	1
Tom Radditz	6705 E 82nd Street, Ste 140	Tuls	Oklahoma	74133	918-743-0099	1
Chris Holmes	2111 Front Street NE, Ste 3- 206	Salem	Oregon	97301	503-798-9868	2
Cory Boatwright	7350 SW Landmark Lane	Tigard	Oregon	97224	614-212-4422	5
<u>David Lovely</u>	5790 Easton Road, Ste 5766	Plumsteadville	Pennsylvania	18949	215-766-1649	1
Jerry Oliver	236 E Main St.	Evans City	Pennsylvania	16033	724-432-3166	1
Erin Woods	4219 Derry Street	Harrisburg	Pennsylvania	17111	717-566-6604	2
Steve Eisel	350 Scarlett Road, Ste 60	Kenneth Square	Pennsylvania	19348	610-880-0055	1
Brian McMahon	1610 Fourth Street	Monongahela	Pennsylvania	15063	724-258-4884	4
Nathan Kight	103 N Meadows Dr Ste 231	Wexford	Pennsylvania	15090	412-273-8550	1
Rob & Michele Orlandini	323 Wyoming Ave	Wyoming	Pennsylvania	18644	570-299-5951	2
Louis Paiva	284 Phenix Ave	Cranston	Rhode Island	2920	401-946-1800	5
Scott Willis	3015 Dunes West Blvd, Suite 504	Mt. Pleasant	South Carolina	29466	839-204-1002	3
Todd Manzer	2408 Madison Drive	North Myrtle Beach	South Carolina	29582	843-491-9800	2
Brooks & Rion Salley	1613 Paris Ave.	Port Royal	South Carolina	29935	843-929-7300	1
Karin Gierman	3209 Wade Hampton Blvd #5	Taylors	South Carolina	29687	864-900-6111	2
Gary Thrift	1643 Sandifer Blvd	Seneca	South Carolina	<u>29678</u>	864-882-3341	1
John Stoysich	811 E St. Andrews	Rapid City	South Dakota	57701	605-791-4791	1
Chad Burleson Kevin Nyberg	1600 W. Russell St. 330 W 41st Street, Ste 100	Sioux Falls	South Dakota	57104 <u>57</u> 105	605-271-0150	1

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Tom GlennBarry Frizzell	5959 Shallowford2328B Georgetown Rd,#2055 NW	ChattanoogaCle veland	Tennessee	37421 <u>37</u> 311	423- 551- 8892 <u>813-3007</u>	<u>14</u>
James DeLaurentis Tom Glenn	2328B Georgetown Rd NW620 W Poplar Ave.	Cleveland Collie rville	Tennessee	37311 <u>38</u> 017	423-813- 3007 <u>901-386-</u> 1100	<u>32</u>
Logan Yandell	500 W Main Street, Ste 270	Hendersonville	Tennessee	37075	615-912-9025	1
Drew Mann	5204 Kingston Pike Ste 27-28	Knoxville	Tennessee	37919	865-338-3753	1
Heidi Bertram	1432 W Main St. Ste 300	Lebanon	Tennessee	37087	615-257-1933	1
Cole & Renee Guthrie	851 Watts Lane	Nashville	Tennessee	37209	651-581-2288	2
Andrew Bush Scott and Lori Crowe	1210 W. McDermott, Suite #101410 S High Street	AllenWaynesbor o	Texas Tennesse e	75013 <u>38</u> 485	972-346- 3041 <u>931-744-</u> 8162	<u>21</u>
Guy Carroll	6423 Western St	Amarillo	Texas	79110	806-412-1848	1
Freddie Perales	26919 US-380, Ste 204	Aubrey	Texas	76227	469-678-7242	1
Clifton Robinson	3930 Bee Caves Road	Austin	Texas	78746	737-263-4500	3
Todd & Joni Concienne	2626 Calder Ave Ste 203	Beaumont	Texas	77702	409-422-5223	1
John Berringer	28720 Interstate Hwy 10, Suite #420	Boerne	Texas	78006	210-951-2006	4
Kevin Harless	755 N Burleson Boulevard	Burleson	Texas	76028	817-782-9575	2
Mustafa Sunlwala	1611 I-35 E, Ste 416	Carrollton	Texas	75206	469-986-0053	2
Nick Meyer	1808 Davisville Drive	Providence Village	<u>Texas</u>	<u>76227</u>	817-996-9698	1
Blake Robison	10500 State Highway 30, Ste 600	College Station	Texas	77845	979-464-5090	1
Craig Young	6226 Colleyville Blvd Ste B	Colleyville	Texas	76034	817-809-6138	2
David Alexander	4531 Ayers Street, Ste 218	Corpus Christi	Texas	78415	361-201-9002	1
Stacy Huston	5735 Kenwood Ave	Dallas	Texas	75206	972-308-6035	1
Brett Jeffreys	3421 South Grove St	Fort Worth	Texas	76110	817-560-4070	4
Nathan Jones	1711 N Llano	Fredericksburg	<u>Texas</u>	<u>78624</u>	830-997-0703	1
Chance Burton	8045 FM 359 Ste 210	Fulshear	Texas	77411	281-346-8227	1
Terre & Amy Albert	32618 Tamina Rd Ste C	Houston	Texas	77385	936-235-4124	3
Randy Diefenthal	710 N Post Oak Road, Suite #120	Houston	Texas	77024	281-595-9056	1
Mauricio & Carlos Portugal	22503 Katy Fwy, Suite #61	Katy	Texas	77450	832-661-5706	1

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Robert Mabry	1701 Northpark Drive, Suite #6	Kingwood	Texas	77339	281-358-6666	1
Ronald Willis	1851 Gulf Freeway South, #31	League City	Texas	77573	832-534-4255	6
Steve Cortez	1702 S State Hwy 121 STE 604	Lewisville	Texas	75067	972-440-0700	1
Charlotte Waterhouse	105 Schumanns Beach Road, Ste 1102C	New Braunfels	Texas	78130	830-825-0050	1
Luis Guerrero	11200 Broadway Ste 2743	Pearland	Texas	77584	346-395-5775	<u>34</u>
Dave Barker Darryl Ross	20700 Hwy 71 West 109 York Street, Ste A203	Spicewood Weat herford	Texas	78669 <u>76</u> 086	512-306- 3121817-475- 7458	<u>31</u>
Greg Krowski	18207 N Eldridge Pkwy Suite A3	Tomball	Texas	77377	832-698-1237	1
Eric and Denise Miller	3709 W Way St Suite E	Tyler	Texas	75703	903-352-3980	1
Daniel Moldenhauer	1090 W 200 S	Heyburn	Utah	83336	208-900-2234	1
Kevin Buckner	10918 N Towne Center Blvd	Highland	Utah	84003	385-245-2225	1
Bryant Curtis	6340 N Salt Creek Road	Honeyville	Utah	84314	435-282-0678	1
Brent Anderson	2682 S Highland Drive, Ste 104	Salt Lake City	Utah	84016	406-702-3991	2
Darren Brown	157 E Riverside Dr Ste 2Z	St George	Utah	84790	702-423-6426	1
Bernie Watson	10370 Leadbetter Road	Ashland	Virginia	23005	804-547-7676	2
David Guglielmi	5641 Burke Center Pkwy Ste 109	Burke	Virginia	22015	703-818-2323	2
Ted Rollins	319C Main St.	Smithfield	Virginia	23430	757-974-0409	1
Steve Blackwell and Rachel Jones	22636 Glenn Drive # 304	Sterling	Virginia	20164	703-563-4124	2
Chris Lackovic	2200 Opitz Blvd Suite 340	Woodbrige	Virginia	22191	517-513-3777	2
Charles Masters	18933 59th Ave NE	Arlington	Washington	98223	360-926-0926	3
Stanley Sanchez	4040 Wheaton Way, Suite 209	Bremerton	Washington	98310	360-602-2876	1
Jason Chambers	7325 W Deschutes Ave, Ste B	Kennewick	Washington	93366	502-219-6300	1
Joel Stangeland	10000 NE 7th Avenue, Suite #120	Vancouver	Washington	98685	360-576-7100	1
Chris Dewesee	7500 MacCorkle Ave SE, Ste 100	Charleston	West Virginia	<u>25304</u>	304-936-6772	1
Mark Queen	6268 Country Road, Ste 107	Proctorville	West Virginia	<u>45669</u>	304-972-0400	1
Brian Eavey	8223 66th Street	Kenosha	Wisconsin	53142	262-842-5100	1

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Kyle Mallie	7818 Big Sky Drive, Ste 107	Madison	Wisconsin	53719	608-673-6100	<u>21</u>
Doug Prahst	101 E Pier St Ste 23	Port Washington	Wisconsin	53074	262-235-1003	1
Caryn Bartelt	121 Skelley Street	Schofield	Wisconsin	54476	715-841-9988	1
Dan Neuwirth	104 E Summit Ave	Wales	Wisconsin	53183	262-232-6099	1
John Stoysich	253 S Lowell Street	Casper	Wyoming	82601	307-337-3377	1

^{*} This franchisee is located in Tennessee but operates within one Territory located in the State of Alabama. This franchisee is included in this list with the Alabama franchisees for purposes of that Territory.

LIST OF FRANCHISEES WHO HAVE SIGNED A FRANCHISE AGREEMENT FOR OUTLETS NOT YET OPEN

As of December 31, 2023 <u>2024</u>

Name	Contact Address	City	State	Zip	Phone	Territories Owned
Andy Grant	TBD	Waterford	Connecticut	-	-	1
Michael Wynn	9148 Bonita Beach Rd, Ste 207	Bonita Springs	Florida	34135	239-992- 0454	3
Phillip Norrell	TBD	<u>Cleveland</u> Athens	Georgia		<u>706-363-</u> <u>3936</u>	1
Jason Jackoniski Remington Riccolo	TBD	Brookhaven Newark	Georgia Illinois			<u>21</u>
Sheri Adkins Tom Filopoulos	TBD840 Rothrock Rd, Ste 101	Copley Montclair	Ohio New Jersey	44321_	330-498- 1618_201- 547-4223	1
Tim Hayes	TBD	The Hamptons	New York	-	<u>404-661-</u> <u>6117</u>	1
Cory Boatwright	7350 SW Landmark Lane	Tigard	Oregon	97224	614-212- 4422	1
Justin Werner	TBD	Schyulkill	Pennsylvania	-	-	4
Gary Thrift	TBD	Seneca	South Carolina	-	-	1
Drew Mann	5204 Kingston Pike Ste 27-28	Knoxville	Tennessee	37919	865-338- 3753	1
Darryl Ross	TBD	Weatherford	Texas	-	-	1
Jeremy French	TBD	Ennis	Texas	-	-	1
Addison Gee	TBD	Midland	Texas	-	-	1
Chris Deweese	TBD	Charleston	West Virginia	-	-	4
Mark Queen	-TBD	Huntington	West Virginia	-	-	4

ATTACHMENT I

LIST OF FRANCHISEES WHO HAVE LEFT THE SYSTEM

LIST OF FRANCHISEES WHO HAVE LEFT THE SYSTEM

Listed below is the name and last known city, state and telephone number of every Ace Handyman Franchisee who has had an outlet terminated, cancelled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under their respective Franchise Agreement, including transfers, during the most recently completed fiscal year, or who has not communicated with us within 10 weeks of the Issuance Date of this Disclosure Document.

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

Name	City	State	Phone	Territories Owned	Transfer
John McDonough Earnest Perkins	Huntsville Youngtown	Alabama Arizona	256-937-3855 <u>623-624-</u> 6373	1	
Taimor Hussein Theresa Mitchell	Aurora Oakland	Colorado California	720 370 8008510-531- 4300	<u>34</u>	
Steve & Christy Smith	Altamonte Springs	<u>Florida</u>	407-710-2700	<u>2</u>	
Nick Pieller Michael Ball	Dover <u>Niceville</u>	Delaware Florida	302-899-7300 <u>850-299-</u> 4253	1	<u>X</u>
Charles Cole Quinn Hoffman	<u>Chicago</u> Summerfield	Illinois Florida	773-235-0900 <u>352-706-</u> 2367	7 <u>3</u>	
Jody Weintraub Tom Thayer	Northbrook Zephyrhills	<u>Illinois</u> Florida	847- 813- 9000 <u>923-1223</u>	1	<u>X</u>
Matthew Graham Mike Fisk	Fort Wayne Atlanta	Indiana Georgia	630-990-6791 <u>678-916-</u> 6400	1	X
Laurel Thompson Matt Cochran	Olathe Blackshear	Kansas Georgia	913-825-6205 912-807- 7767	1	X
Ed Salmon Vickers	Frederick Peachtree Corners	Maryland Georgia	301-371-0110 470-704- 9758	<u>31</u>	X
Bruce BaltzStacey & Melissa Hanson	FranklinWarner Robbins	Massachusetts Georgia	508-307-8274 <u>478-334-</u> 2385	1	
Nick Coe	Boise	<u>Idaho</u>	208-900-5090	<u>1</u>	
<u>Travis Foster</u>	Lexington	<u>Kentucky</u>	859-399-6191	<u>2</u>	
Gerard Perczenkiuk	Lake Orion	<u>Michigan</u>	248-940-1550	<u>1</u>	X
Mike Highfill David Pierce	Kansas City	Missouri	816- 945-6383 <u>734-1370</u>	<u> 12</u>	
Tom Segreto Dylan Dechant	Apex Asheville	North Carolina	919-874-5400 <u>828-900-</u> 0900	1	<u>X</u>
Tim Howard	Greensboro	North Carolina	336-245-4485	<u>2</u>	
Dave Wigfield	Mooresville	North Carolina	<u>704-765-6800</u>	<u>1</u>	<u>X</u>
Nathan Acker	<u>Edmond</u>	<u>Oklahoma</u>	<u>405-861-8300</u>	<u>1</u>	
Steve Eisel	Kenneth Square	<u>Pennsylvania</u>	<u>610-880-0055</u>	<u>1</u>	
Chad Burleson	Sioux Falls	South Dakota	605-271-0150	<u>1</u>	<u>X</u>
Barry Frizzell	Chattanooga	Tennessee	423-551-8892	<u>1</u>	<u>X</u>
Andrew Bush	Allen	<u>Texas</u>	972-346-3041	<u>2</u>	
Michael DelpierRandy Diefenthal	Conroe Houston	Texas	936-523-7051 281-595- <u>9056</u>	1	X
Ronald Willis	<u>League City</u>	<u>Texas</u>	832-534-4255	<u>6</u>	
Steve Cortez	<u>Lewisville</u>	<u>Texas</u>	972-440-0700	<u>1</u>	
Hazem ElSayed Dave Barker	Round RockSpicewood	Texas	512- 665-5345 <u>306-3121</u>	2 3	X

Jessie MesserliEric and Denise Miller	Shady Hollow Tyler	Texas	512-557-5115 903-352- <u>3980</u>	1	X
Kyle Mallie	Madison	Wisconsin	608-673-6100	<u>1</u>	

There was one franchisee who transferred his Territory in one state and acquired another Territory in another state and therefore did not leave the system. Therefore, that franchisee is not shown in the chart above.

ATTACHMENT J

FINANCIAL STATEMENTS

ACE HARDWARE CORPORATION 2024 Annual Report



ACE HARDWARE CORPORATION INDEX TO CONSOLIDATED FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

Report of Independent Auditors
Consolidated Balance Sheets as of December 28, 2024 and December 30, 2023
Consolidated Statements of Income for the years ended December 28, 2024, December 30, 2023 and December 31, 2022
Consolidated Statements of Comprehensive Income for the years ended December 28, 2024, December 30, 2023 and December 31, 2022
Consolidated Statements of Equity for the years ended December 28, 2024, December 30, 2023 and December 31, 2022
Consolidated Statements of Cash Flows for the years ended December 28, 2024, December 30, 2023 and December 31, 2022
Notes to Consolidated Financial Statements
Management's Discussion and Analysis of Financial Condition and Results of Operations
Five Year Summary of Earnings and Distributions
Management's Responsibility for Financial Statements

Report of Independent Auditors

The Board of Directors Ace Hardware Corporation

Opinion

We have audited the consolidated financial statements of Acc Hardware Corporation (the Company), which comprise the consolidated balance sheets as December 28, 2024 and December 30, 2023, and the related consolidated statements of income, comprehensive income, equity and cash flows for each the three fiscal years in the period ended December 28, 2024, and the related notes (collectively referred to as the "financial statements").

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company at December 28, 2024 and December 30, 2023, and the results of its operations and its cash flows for each of the three fiscal years in the period ended December 28, 2024 in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under the standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepte in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation financial statements that are free of material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that rais substantial doubt about the Company's ability to continue as a going concern for one year after the date that the financial statements are available to be issued

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free of material misstatement, whether due to frau or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omission misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud or error, and design and perform aud procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in th
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

Other Information

Management is responsible for the other information. The other information comprises the management's discussion and analysis of financial condition and results of operations and five year summary of earnings and distributions but does not include the financial statements and our auditor's report thereon. Ou opinion on the financial statements does not cover the other information, and we do not express an opinion or any form of assurance thereon.

In connection with our audit of the financial statements, our responsibility is to read the other information and consider whether a material inconsistency exists between the other information and the financial statements, or the other information otherwise appears to be materially misstated. If, based on the world performed, we conclude that an unrecorded material misstatement of the other information exists, we are required to describe it in our report.

Chicago, Illinois

February 11, 2025

ACE HARDWARE CORPORATION CONSOLIDATED BALANCE SHEETS (In millions, except share data)

	December 28, 2024	December 3 2023
Assets		
Cash and cash equivalents Marketable securities	\$ 36.3	\$ 31
	42.7	49
Receivables, net of allowance for doubtful accounts of \$10.9 and \$12.1, respectively	727.8	674
Inventories	1,197.9	1,235
Prepaid expenses and other current assets	87.3	69
Total current assets	2,092.0	2,060
Property and equipment, net	594.6	541
Operating lease right-of-use assets, net	798.2	668
Finance lease right-of-use assets, net	185.7	123
Goodwill and other intangible assets	455.6	410
Other assets	166.7	153
Total assets	\$ 4,292.8	\$ 3,957
Liabilities and Equity		
Current maturities of long-term debt	\$ 80.6	\$ 49
Accounts payable	1,215.2	1,081
Patronage distributions payable in cash	137.2	137
Patronage refund certificates payable	15.0	29
Current operating lease liabilities	94.4	88
Current finance lease liabilities	37.9	25
Accrued expenses		
Total current liabilities	313.1	292
	1,893.4	1,702
ong-term debt	246.5	359
atronage refund certificates payable	205.8	149
ong-term operating lease liabilities	728.4	603
ong-term finance lease liabilities	145.0	100
Other long-term liabilities	151.8	120
Total liabilities	3,370.9	3,036
Member Retailers' Equity: Class A voting common stock, \$1,000 par value, 10,000 shares authorized, 2,586 and 2,615		
issued and outstanding, respectively	26	2
Class C nonvoting common stock, \$100 par value, 20,000,000 shares authorized, 7,528,693	2.6	2.5
	752.0	700
and 7,226,176 issued and outstanding, respectively	752.9	722.
class C nonvoting common stock, \$100 par value, issuable to Retailers for patronage	9330	9550
distributions, 825,206 and 897,317 shares issuable, respectively	82.5	89.
ontributed capital	19.3	19.
detained earnings	58.5	78.
accumulated other comprehensive (loss) income	(1.9)	0.
Equity attributable to Ace member Retailers	913.9	912.
quity attributable to noncontrolling interests	8.0	8.
Total equity	921.9	921.

ACE HARDWARE CORPORATION CONSOLIDATED STATEMENTS OF INCOME (In millions)

	Years Ended					
	December 28, 2024	December 30, 2023	December 3,			
	(52 Weeks)	(52 Weeks)	(52 Weeks)			
Revenues:			- 1			
Wholesale revenues	\$ 8,666.0	\$ 8,344.7	\$ 8,372.3			
Retail revenues	825.4	786.4	797.5			
Total revenues	9,491.4	9,131.1	9,169.8			
Cost of revenues:	.,,,,,,,	,,,,,,,,,,	3,10,10			
Wholesale cost of revenues	7,495.6	7,223.1	7,368.8			
Retail cost of revenues	450.7	430.1	439.6			
Total cost of revenues	7,946.3	7,653.2	7,808.4			
Gross profit:						
Wholesale gross profit	1,170.4	1,121.6	1,003.5			
Retail gross profit	374.7	356.3	357.9			
Total gross profit	1,545.1	1,477.9	1,361.4			
Wholesale operating expenses:						
Distribution operations expenses	288.6	281.5	262.3			
Selling, general and administrative expenses	342.4	314.7	244.6			
Retailer success and development expenses	294.3	234.5	223.2			
Retail operating expenses	297.7	280.1	287.5			
Retail pre-opening expenses	4.8	6.0	1.9			
Total operating expenses	1,227.8	1,116.8	1,019.5			
Operating income	317.3	361.1	341.9			
nterest expense	(48.9)	(38.4)	(19.2)			
nterest income	6.9	5.4	3.2			
Other income (expense), net	36.5	(3.7)	23.7			
ncome tax benefit (expense)	2.3	(1.0)	(9.0)			
let income	314.1	323.4	340.6			
Less net loss attributable to noncontrolling interests	(0.9)	(2.7)	(1.1)			
let income attributable to Ace Hardware Corporation	\$ 315.0	\$ 326.1	\$ 341.7			
Patronage distributions accrued	\$ 357.0	\$ 356.5	\$ 344.5			
Patronage distributions accrued for third party Retailers	\$ 335.6	\$ 335.7	\$ 323.6			

ACE HARDWARE CORPORATION CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (In millions)

	Years Ended					
	December 28, 2024 (52 Weeks)	December 30, 2023 (52 Weeks)	December 3 2022 (52 Weeks			
Net income	\$ 314.1	\$ 323.4	\$ 340.6			
Other comprehensive (loss) income, net of tax:						
Foreign currency translation (loss) gain	(2.8)	2.0	0.5			
Unrealized gain (loss) on investments		1.1	(3.8)			
Total other comprehensive (loss) income, net	(2.8)	3.1	(3.3)			
Comprehensive income	311.3	326.5	337.3			
Less comprehensive loss attributable to noncontrolling interests	(1.4)	(2.2)	(1.0)			
Comprehensive income attributable to Ace Hardware Corporation	\$ 312.7	\$ 328.7	\$ 338.3			

ACE HARDWARE CORPORATION CONSOLIDATED STATEMENTS OF EQUITY

(In millions)

					Share	holders of A	ce Ha	rdware	Corp	oration								
	Cla	Capital		lass C	Issuat for	ass C Stock ble to Retailers r Patronage Dividends	S	ditional tock scribed	100000000000000000000000000000000000000	tributed apital		etained arnings	Compr	nulated her chensive c (Loss)	Noncontrolling Interests	Total Equ	al Equity	
Balances at January 1, 2022	\$	2.7	s	583.8	\$	120.9	s		s	19.1	s	68.9	\$	1.0	s	11.7	\$	808.1
Net income		-										341.7				(1.1)		340.6
Other comprehensive loss						-								(3.3)				(3.3)
Net payments on subscriptions						-		1.6				-				-		1.6
Stock issued		0.1		121.0		(120.9)		(1.6)		-		0.2		-				(1.2)
Change in noncontrolling interests				-						(0.1)		100		-		0.2		0.1
Stock repurchased		(0.1)		(48.7)		-				-								(48.8)
Patronage distributions issuable						118.1												118.1
Patronage distributions payable				-		-				-		(323.6)				-		(323.6)
Other				_		-		-		0.2								0.2
Balances at December 31, 2022	\$	2.7	\$	656.1	\$	118.1	\$	•	S	19.2	S	87.2	\$	(2.3)	S	10.8	S	891.8
Net income												326.1				(2.7)		323.4
Other comprehensive income						-						-		3.1				3.1
Net payments on subscriptions		-						1.0				-						1.0
Stock issued				116.9		(118.1)		(1.0)		-		0.6		-		-		(1.6)
Change in noncontrolling interests				-						(0.1)				(0.5)		0.6		
Stock repurchased		(0.1)		(50.4)		-				-		-		-		-		(50.5)
Patronage distributions issuable		-				89.7						-		-		-		89.7
Patronage distributions payable												(335.7)				-		(335.7)
Other						-				0.1						0.1		0.2
Balances at December 30, 2023	\$	2.6	\$	722.6	S	89.7	\$		S	19.2	S	78.2	S	0.3	\$	8.8	\$	921.4
Net income		*		-						-		315.0				(0.9)		314.1
Other comprehensive loss														(2.8)				(2.8)
Net payments on subscriptions								1.5				-				-		1.5
Stock issued		0.1		89.2		(89.7)		(1.5)				0.6				-		(1.3)
Change in noncontrolling interests						-				0.2		-		0.6		0.1		0.9
Stock repurchased		(0.1)		(58.9)										-				(59.0)
Patronage distributions issuable						82.5						-				-		82.5
Patronage distributions payable												(335.6)						(335.6)
Other										(0.1)		0.3						0.2
Balances at December 28, 2024	S	2.6	S	752.9	\$	82.5	\$		S	19.3	S	58.5	\$	(1.9)	\$	8.0	\$	921.9

ACE HARDWARE CORPORATION CONSOLIDATED STATEMENTS OF CASH FLOWS (In millions)

			Year	s Ended		
	2	nber 28, 024 Weeks)	December 30, 2023 (52 Weeks)			ember 31 2022 2 Weeks)
Operating Activities						
Net income	\$	314.1	\$	323.4	S	340.6
Adjustments to reconcile net income to net cash provided by operating activities:						
Depreciation and amortization		99.7		92.2		86.6
Non cash lease expense		32.2		26.9		17.1
Amortization of deferred financing costs		0.9		1.3		0.4
(Gain) loss on disposal of assets, net		(16.0)		5.5		(19.8)
Provision for doubtful accounts		2.0		6.9		3.0
Other, net		0.6		1.2		5.8
Changes in operating assets and liabilities, exclusive of effect of acquisitions:			16.7			
Receivables		(78.5)		(55.4)		(159.6)
Inventories		54.9		74.7		(77.3)
Other current assets		(17.4)		(9.1)		4.1
Other long-term assets		(18.6)		(13.5)		(9.1)
Accounts payable and accrued expenses		149.7		(104.7)		80.4
Other long-term liabilities		26.1		14.5		13.4
Deferred taxes		(1.6)		(0.1)		(3.1)
Net cash provided by operating activities		548.1		363.8		282.5
nvesting Activities						
Purchases of investment securities		(24.6)		(39.9)		(60.3)
Proceeds from sale of investment securities		30.0		43.2		47.7
Purchases of property and equipment	((155.4)		(164.6)		(92.8)
Cash paid for acquired businesses, net of cash acquired		(58.6)		(309.7)		(12.8)
Increase in notes receivable, net		(37.3)		(31.8)		(19.7)
Proceeds from sale of assets		22.7		1.3		29.4
Other, net		0.4	_	(0.8)		(0.6)
Net cash used in investing activities	(222.8)		(502.3)		(109.1)
inancing Activities						
Net (payments) borrowings under revolving lines of credit		(85.3)		362.0		(4.9)
Principal payments on long-term debt		(21.1)		(17.8)		(7.1)
Principal portion of finance lease payments		(40.3)		(28.3)		(16.7)
Payments of debt issuance costs				(4.4)		
Payments of cash portion of patronage distribution	(137.1)		(133.1)		(121.0)
Payments of patronage refund certificates		(22.9)		(26.7)		(26.1)
Repurchase of stock		(15.1)		(10.8)		(8.0)
Purchase of noncontrolling interests		(0.9)		(0.4)		(0.4)
Other, net		1.8		1.4		1.7
Net cash (used in) provided by financing activities	(320.9)		141.9		(182.5)
icrease (decrease) in cash and cash equivalents		4.4		3.4	-	(9.1)
ash and cash equivalents at beginning of period		31.9		28.5		37.6
ash and cash equivalents at end of period	\$	36.3	\$	31.9	\$	28.5
upplemental disclosure of cash flow information:					-	
Interest paid	\$	39.4	\$	29.7	\$	14.5
Income taxes paid	\$	7.9	\$	5.6	\$	11.5

(1) Summary of Significant Accounting Policies

The Company and Its Business

Ace Hardware Corporation ("the Company") is a wholesaler of hardware, paint and other related products. The Company als provides to its retail members ("Retailers") value-added services such as advertising, marketing, merchandising and store location and design services. The Company's goods and services are sold predominately within the United States, primarily to retailers that operar hardware stores and with whom the Company has a retail membership agreement. As a retailer-owned cooperative, the Compand distributes substantially all of its patronage sourced income in the form of patronage distributions to Retailers based on their volume of merchandise purchases.

Ace Hardware International Holdings, Ltd. ("AIH"), is a majority-owned and controlled subsidiary of the Company with a 21. percent noncontrolling interest held by AIH customers. AIH shareholders do not own shares of stock in the Company nor receiv patronage dividends.

Ace Retail Holdings LLC ("ARH") is the owner of the 180-store Westlake Ace Hardware ("Westlake") and the 79-store Great Lakes Ace Hardware ("GLA") retail chains. As a result, the Company is also a retailer of hardware, paint and other related products.

Ace Wholesale Holdings LLC ("AWH") owns and operates Emery Jensen Distribution ("EJD"). EJD is a wholesaler of hardware, paint and other related products to non-member retailers.

Ace Services Holdings LLC ("ASH"), through its subsidiaries, operates Ace Hardware Home Services ("AHHS"), a provide of residential plumbing, heating, cooling and electric ("PHCE") services, and Ace Handyman Services, a franchisor of hom improvement, maintenance and repair services.

Basis of Presentation

The accompanying consolidated financial statements have been prepared in accordance with U.S. generally accepte accounting principles ("GAAP"). The Company's fiscal year ends on the Saturday nearest December 30. Accordingly, fiscal year 2024, 2023 and 2022 ended on December 28, 2024, December 30, 2023 and December 31, 2022, respectively. Unless otherwise noted all references herein for the years 2024, 2023 and 2022 represent fiscal years ended December 28, 2024, December 30, 2023 and December 31, 2022, respectively. Fiscal years 2024, 2023, and 2022 consisted of 52 weeks.

Subsequent events have been evaluated through February 11, 2025, the date these statements were issued.

Principles of Consolidation

The accompanying consolidated financial statements include the accounts of the Company and its majority-owned subsidiaries All intercompany transactions have been eliminated.

Use of Estimates

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumption that affect the reported amounts of assets and liabilities at the date of the financial statements and the reported amounts of revenues are expenses during the reporting period. Actual results could differ from those estimates.

Cash, Cash Equivalents and Marketable Securities

In the normal course of business, the Company has outstanding checks that exceed the cash balances in the Company's band accounts, which create a book overdraft that are recorded as a liability. As of December 28, 2024 and December 30, 2023, the Company had outstanding checks in excess of bank balances totaling \$93.6 million and \$78.4 million, respectively, which have been included in accounts payable in the accompanying consolidated balance sheets. These outstanding amounts were subsequently funded through cash receipts and borrowings under the Company's debt facilities during the following fiscal year.

The Company classifies all highly liquid investments with original maturities of three months or less as cash equivalents.

The Company determines the appropriate classification of its investments in marketable securities, which are predominate held by the Company's New Age Insurance, Ltd. ("NAIL") subsidiary, at the time of purchase and evaluates such designation at each balance sheet date. All marketable securities have been classified and accounted for as available for sale. The Company may hold de securities until maturity. In response to changes in the availability of and the yield on alternative investments as well as liquidit requirements, securities are occasionally sold prior to their stated maturities. Debt securities with maturities beyond twelve months at viewed by the Company as available to support current operations and are therefore classified as current assets in the accompanyin Consolidated Balance Sheets. Marketable securities are carried at fair value based on quoted market prices, with unrealized gains an losses, net of taxes, reported as a component of Accumulated other comprehensive income ("AOCI") for debt securities, while equit securities' unrealized gain and losses are recorded as a component of Other income, net. Realized gains and losses on securities at determined using the specific identification method.

In the normal course of NAIL's operations, standby letters of credit totaling \$17.2 million and \$33.3 million at December 2024 and December 30, 2023, respectively, were issued in favor of the insurance companies that reinsure a portion of NAIL's los exposure. At December 28, 2024, NAIL has pledged substantially all of its cash and cash equivalents and marketable securities a collateral for these letters of credit. The Company maintains a multiple employer welfare arrangement ("MEWA") that offers medical insurance and other welfare benefits to employees of the Company and Retailers and their employees at participating Ace retail locations. Ace has provided a \$15.0 million standby letter of credit to the State of Vermont in connection with the MEWA. See Note 12 for mor information.

Revenue Recognition

Revenue is recognized when performance obligations under the terms of contracts with our customers are satisfied; generally this occurs with the transfer of control of merchandise or services. Revenue is measured as the amount of consideration we expect t receive in exchange for transferring goods or providing services. The Company excludes sales and usage-based taxes collected an recognizes revenues net of expected returns. Provisions for sales returns are provided at the time the related sales are recorded base on historic returns activity.

The Company's warehouse merchandise revenue originates with a single performance obligation to ship the products, and therefore the Company's performance obligations are satisfied when control of the products is transferred to the customer per the arranged shipping terms. The customer takes ownership and assumes risk of loss for warehouse merchandise upon delivery. Although products are generally shipped FOB shipping point, the Company effectively retains the responsibilities of ownership until the good reach the customer. The Company considers shipping and handling as activities to fulfill its performance obligation for warehous merchandise revenues. Billings for freight are accounted for as Revenues and shipping and handling costs are accounted for in Cost or revenues.

The Company has direct shipment arrangements with various vendors to deliver products to its customers without having to physically hold the inventory at the Company's warehouses, thereby increasing efficiency and reducing costs. The Company recognize revenue for direct shipment arrangements upon delivery to the customer with contract terms that typically specify FOB destination. The Company recognizes the revenue and cost of goods sold from these arrangements on a gross basis as the principal in the transaction a the Company is primarily responsible for fulfilling the promise to customers to provide merchandise at negotiated prices with the vendors, assumes inventory risk if the product is returned by the customers, and assumes all the credit risk for the vendors with the customers.

Service revenues (advertising activities, brand building initiatives and fees for other services provided primarily to domestic Retailers) are recognized when the service is complete as this is when the Retailer has the ability to direct the use of and obtain the benefits from the service. Service revenues from the Company's residential home service businesses are recognized when the repair of installation is complete and the consumer has tendered payment. Service revenues are included in Wholesale revenues on the Consolidated Statements of Income.

The Company's e-commerce revenues come from Ace Hardware's website. For e-commerce transactions, customers choose whether to have merchandise delivered to them or to collect their merchandise from an Ace store ("in-store pick up"). For items delivered directly to the customer, control passes and revenue is recognized when delivery has been completed to the customer, as title has passed and we have transferred possession to the customer. For in-store pick up, control passes and revenue is recognized once the customer has taken possession of the merchandise. Any fees charged to customers for delivery are a component of the transaction price and are recognized when delivery has been completed. Payment terms for retail and e-commerce sales are at the point of sale. E-commerce revenues are included in Wholesale revenues on the Consolidated Statements of Income.

Retail revenues from retail locations owned and operated by the Company and e-commerce revenues are recognized when the customer takes ownership of the products sold and assumes the risk of loss. The customer takes ownership and assumes risk of loss generally at the point of sale in our owned retail locations. The Company offers its Retailers various incentive programs which provide the Retailers with certain sales allowances. The Company offers these incentive programs in anticipation of future sales to the Retailers

that participate in these programs. Since the sales allowances provide future economic benefit to the Company, they are capitalized in Other assets and amortized as a reduction of revenue on a straight-line basis over the period of expected future sales to these Retailer which is generally five years.

The following table provides a summary of revenues by sales category:

	Years Ended				
	December 28, 2024	December 30, 2023			
Warehouse merchandise	\$ 6,162.7	\$ 6,060.7			
Direct shipment merchandise	1,556.9	1,491.0			
ARH revenues	825.4	786.4			
E-commerce E-commerce	382.3	334.0			
Service revenues	564.1	459.0			
Total revenues	\$ 9,491.4	\$ 9,131.1			

Trade Receivables

Trade receivables from customers include amounts invoiced for the sale of merchandise, services and equipment used in thoperation of customers' businesses.

Notes Receivable

The Company makes available to its Retailers various lending programs whose terms exceed one year. The notes bear interes at various rates based on market rates, the loan program or the Retailer's credit quality and are recorded at face value. Interest i recognized over the life of the note on the effective interest method. Loan origination fees were not material for any period presented.

Allowance for Expected Credit Losses

The Company estimates an allowance for credit losses related to future expected losses by using relevant information such a historical collection statistics, current customer credit information, the current economic environment, the aging of receivables, the evaluation of compliance with lending covenants and the offsetting amounts due to Retailers for stock, notes, interest and anticipated but unpaid patronage distributions. Significant past due accounts are reserved on a specific identification basis. For the remainder of the accounts, the allowance is evaluated on a pool basis for accounts with similar risk characteristics and a loss-rate for each pool is determined using current conditions and historical credit loss statistics. The Company considers accounts and notes receivable past due if invoices remain unpaid past their due date and provides for the write-off of uncollectible receivables after exhausting all commercially reasonable collection efforts.

Inventories

Wholesale inventories are valued at the lower of cost or net realizable value. Cost is determined primarily using the last-in first-out ("LIFO") method for all inventories.

Inventories at retail locations operated by ARH are valued at the lower of cost or net realizable value. Inventory cost is determined using the moving average method, which approximates the first-in, first-out ("FIFO") method.

Vendor Funds

The Company receives funds from vendors in the normal course of business principally as a result of purchase volumes, sales, early payments or promotions of vendors' products. Based on the provisions of the vendor agreements in place, management develops accrual rates by estimating the point at which the Company will have completed its performance under the agreement and the amount agreed upon will be earned. Amounts accrued throughout the year can be impacted if actual purchase volumes differ from projected annual purchase volumes. At year-end, the accrual reflects actual purchases made throughout the year.

Vendor funds are treated as a reduction of inventory cost, unless they represent a reimbursement of specific, incremental and identifiable costs incurred by the Company to sell the vendor's product, in which case the costs are netted. The majority of the vendor funds that the Company receives do not meet the specific, incremental and identifiable criteria. Therefore, the Company treats a majority of these funds as a reduction in the cost of inventory as the amounts are accrued and recognizes these funds as a reduction of cost of revenues when the inventory is sold.

Property and Equipment

Property and equipment are stated at cost less accumulated depreciation and amortization. Expenditures for maintenance repairs and renewals of relatively minor items are generally charged to expense. Significant improvements or renewals are capitalized

Depreciation expense is computed on the straight-line method based on estimated useful lives as follows:

Buildings	15 - 40 years
Leasehold improvements	Lease life
Warehouse equipment	3 - 20 years
Transportation equipment	5 - 10 years
Office equipment	3 - 10 years

Leasehold improvements are generally amortized on a straight-line basis over the lesser of the lease term or the estimated usefulife of the asset.

The Company evaluates long-lived assets, such as property and equipment, for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. Recoverability of assets to be held and used is measured by a comparison of the carrying amount of an asset group to estimated undiscounted future cash flows expected to be generated by the asset group. If the carrying amount of an asset group exceeds its estimated future cash flows, an impairment charge is recognized in the amount by which the carrying amount of the asset group exceeds its fair value.

Goodwill and Other Intangible Assets

Goodwill represents the excess of the cost of an acquired business over the fair value of the net tangible assets acquired and identified intangible assets. Goodwill is not amortized but is tested for impairment at a reporting unit level on an annual basis as of the first day of the fourth quarter, or more frequently, if circumstances change or an event occurs that would more likely than not reduce the fair value of a reporting unit below its carrying amount.

The Company's other intangible assets primarily relate to trademarks, tradenames and customer relationships. The intangible are amortized over their estimated useful lives.

See Note 8 for additional information.

Internal-Use Software

Included in fixed assets is the capitalized cost of internal-use software. The Company capitalizes costs incurred during the application development stage of internal-use software and amortizes these costs over its estimated useful life. Costs incurred related to design or maintenance of internal-use software are expensed as incurred.

Leases

At the commencement date of a lease, the Company recognizes a liability to make lease payments and an asset representing the right to use the underlying asset during the lease term. The lease liability is measured at the present value of lease payments over the lease term, including variable fees that are known or subject to a minimum floor. The Company generally does not include options to extend or terminate leases unless it is reasonably certain that the option will be exercised. The Company has lease agreements that contain both lease and non-lease components. The Company elected to not separate non-lease components from lease components for all categories. The Company utilizes the practical expedient to use a risk-free discount rate at commencement date based upon the lease term in determining the recognition of the operating and finance lease assets and liabilities. The initial valuation of the right-of-use ("ROU") asset includes the initial measurement of the lease liability, lease payments made in advance of the lease commencement date and initial direct costs incurred by the Company and excludes lease incentives.

Real estate operating leases also typically require payment of real estate taxes, common area maintenance and insurance. These components comprise the majority of variable lease costs and are excluded from the present value of lease obligations. Operating lease assets also include prepaid or accrued lease payments and are reduced by lease incentives.

Certain retail lease agreements also provide for contingent rentals based on sales performance in excess of specified minimums or on changes in the consumer price indexes. Contingent rentals, which are based upon future performance or changes in indices, are excluded from the determination of lease payments and were not material.

Leases with an initial term of 12 months or less are classified as short-term leases and are not recorded on the Consolidated Balance Sheets. The lease expense for short-term leases is recognized on a straight-line basis over the lease term.

Advertising Expense

The Company expenses advertising costs when incurred. Gross advertising expenses amounted to \$336.9 million, \$297. million, and \$263.6 million in fiscal 2024, 2023 and 2022, respectively.

Gift Cards

The Company sells gift cards to customers through its Retailers, the Company website and select third parties. The gift card do not expire. A liability is initially established for the value of the gift card when sold. Gift card breakage income is recognize proportionately as redemption occurs at the approximate monthly breakage rate and represents the balance of gift cards for which the Company believes the likelihood of redemption by the customer is remote. The breakage income calculation takes into account an legal obligation to remit the unredeemed portion to relevant jurisdictions. During fiscal years 2024, 2023 and 2022, the Compan recognized gift card breakage income of \$4.3 million, \$3.3 million and \$3.3 million, respectively. The Company does not believe there is a reasonable likelihood that there will be a material change in the future estimates or assumptions that we use to record breakage.

Retirement Plans

The Company sponsors health benefit plans for its retired officers and a limited number of retired non-officer employees. The Company and its subsidiaries also sponsor defined contribution plans for substantially all employees. The Company's contribution under these plans are determined annually by the Board of Directors and charged to expense in the period in which it is earned b employees.

Income Taxes

The Company accounts for income taxes using the asset and liability method. Under this approach, deferred taxes ar recognized for the future tax consequences of differences between the financial statement and income tax bases of existing assets an liabilities and measured based upon enacted tax laws and rates.

Self-Insurance

NAIL, the Company's wholly-owned subsidiary, operates as a captive insurance company. NAIL provides the reinsurance of property and casualty insurance policies for some Retailers and is the direct insurer for certain property and casualty insurance policies of the Company. These insurance programs are subject to varying retention levels of self-insurance. Such self-insurance relates to losses and liabilities primarily associated with property, general liability, workers' compensation, and auto liability insurance programs. Losses are accrued based upon the Company's estimates of the aggregate liability for claims incurred using certain actuarial assumption based on Company experience and insurance industry metrics.

Concentration of Credit Risk

Credit risk pertains primarily to the Company's trade and notes receivables. The Company extends credit to its customers a part of its day-to-day operations. Management believes that as no specific receivable or group of receivables comprises a significar percentage of total trade accounts, its concentration of credit risk with respect to trade receivables is limited. Additionally, management believes that its allowance for doubtful accounts is adequate with respect to overall customer credit risks. Also, the Company's certificate of incorporation and by-laws specifically provide that the Company may set-off its obligation to make any payment to Retailer for such Retailer's stock, notes, interest and declared and unpaid distributions against any obligation owed by the Retailer the Company. The Company, but not the Retailer, may at its sole discretion exercise these set-off rights when any such funds become due to former Retailers with outstanding accounts and notes receivable owed to the Company and current Retailers with past due receivables owed to the Company.

Impact of New Accounting Standards

There are no recently issued applicable accounting pronouncements that are not yet effective that are expected to have a materia impact on consolidated results of operations, consolidated financial condition, or consolidated cash flows.

(2) Acquisitions

The Company accounts for all business combinations using the acquisition method to record a new cost basis for the asse acquired and liabilities assumed. The difference between the purchase price and the fair value of the assets acquired and liabilitie assumed has been recorded as goodwill in the financial statements. The results of operations are reflected in the consolidated financi statements of the Company from the dates of acquisition.

During the year ended December 28, 2024, Unique Indoor Comfort, LLC ("UIC"), a subsidiary of ASH, acquired five PHC businesses for consideration of \$36.6 million, which is inclusive of \$6.1 million in contingent earnouts and prior to final working capit adjustments. These acquisitions resulted in UIC recording \$34.1 million of goodwill. The Company believes these acquisitions wi help advance the Company's strategic plan to be a leading provider of home preservation services. Goodwill has an indefinite life and therefore, is not amortized. The goodwill is expected to be deductible for tax purposes.

In addition, during the year ended December 28, 2024, ARH acquired retail stores for consideration of \$28.0 million, prior t working capital adjustments. These acquisitions resulted in ARH recording \$11.7 million of goodwill. The goodwill is expected to be deductible for tax purposes.

During the year ended December 30, 2023, ASH, through its subsidiaries, acquired PHCE businesses for total consideration of \$313.7 million (including final working capital adjustments) and the Company recorded \$272.5 million of goodwill and \$37.3 million for the fair value of other intangible assets. The goodwill allocated to UIC's disregarded entities is expected to be deductible for tapurposes. The following table summarizes the consideration paid and the final purchase price allocation:

Cash	\$	
Receivables		(
Inventories		. 5
Other current assets		(
Property and equipment		1
Goodwill		272
Other intangibles		37
Other long-term assets		(
Accounts payable		(4
Other current liabilities		(18
Other long-term liabilities		(6
Acquisition purchase price	\$ 3	313

During the year ended December 30, 2023, ARH acquired retail stores for consideration of \$2.8 million, prior to working capital adjustments. These acquisitions resulted in ARH recording \$1.5 million of goodwill. The goodwill is expected to be deductible for tax purposes.

(3) Receivables, net

Receivables, net include the following amounts:

	Dec	cember 28, 2024	December 30, 2023		
Trade	\$	560.0	\$	513.0	
Vendor		120.2		120.3	
Other		23.0		25.0	
Notes receivable – current portion		35.5		28.2	
Less allowance for doubtful accounts		(10.9)		(12.1)	
Receivables, net	\$	727.8	\$	674.4	

The components of changes to the Receivable allowance for doubtful accounts for the years ended December 28, 2024 ard December 30, 2023 were as follows:

	December 2024			mber 30, 2023
Beginning balance	\$ 12	2.1	S	7.0
Charges to provision	1	1.6		5.3
Accounts written off, net of recoveries	(2	2.8)		(0.3)
Other				0.1
Ending balance	\$ 10).9	\$	12.1

(4) Inventories

Inventories consist of wholesale merchandise inventories held for sale to customers and retail merchandise inventory held for resale at ARH retail locations. Substantially all of the Company's wholesale inventories are valued on the LIFO method. The excess of replacement cost over the LIFO value of inventory was \$251.4 million and \$271.0 million at December 28, 2024 and December 30, respectively. Inventories at ARH retail locations are valued at the lower of cost or net realizable value. Inventory cost determined using the moving average method, which approximates the FIFO method. The Company regularly reviews its inventor and establishes a reserve for excess and obsolete inventory based on a number of factors, including historical sales, sales forecast obsolescence due to technology changes and defective goods.

Inventories consisted of:

	Dec	ember 28, 2024	December 30 2023		
Wholesale merchandise inventory (LIFO)	S	932.9	S	1,000.4	
Retail merchandise inventory at ARH stores (FIFO)		265.0		234.7	
Inventories	S	1,197.9	\$	1,235.1	

(5) Property and Equipment, net

Property and equipment, net is summarized as follows:

		December 28, 2024		mber 30, 2023
Land		12.6	\$	13.3
Buildings and improvements		285.5		294.7
Warehouse equipment		311.0		301.1
Computer hardware and software and other office equipment		564.5		509.5
Transportation equipment	The state of the state of	46.1		45.9
Leasehold improvements		92.3		81.2
Assets under construction		76.4		31.3
Property and equipment, gross		1,388.4		1,277.0
Accumulated depreciation and amortization		(793.8)	BOUTEN	(736.0)
Property and equipment, net	S	594.6	\$	541.0
	and the same of th			

Depreciation and amortization expense related to property and equipment for fiscal years 2024, 2023 and 2022 was \$94.1 million, \$88.1 million and \$79.9 million, respectively.

During the year ended December 28, 2024, the Company sold its Little Rock, Arkansas retail support center ("RSC") for \$23.5 million and recorded a pre-tax gain of \$17.8 million which is presented in Other income, net in the Consolidated Statements of Income The Company has leased-back this RSC on a short-term basis until construction of the replacement RSC in Kansas City, Missouri is complete.

(6) Leases

The Company leases certain warehouse and distribution space, office space, retail locations, equipment and vehicles under finance and operating leases.

The components of total lease expense for the years ended December 28, 2024 and December 30, 2023, were as follows:

	Income Statement Classification	December 28, 2024			nber 30, 2023
Finance lease amortization	Cost of revenues	S	32.9	\$	27.6
Finance lease interest	Interest expense		3.7		2.9
Operating lease cost, net of sublease income	Cost of revenues, operating expenses		122.1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	117.5
Short-term lease cost	Cost of revenues, operating expenses		8.0		12.2
Variable lease cost	Cost of revenues, operating expenses	444	29.3	TO HE ST	27.5
Net lease cost		\$	196.0	\$	187.7

The future minimum undiscounted lease payments under operating and finance leases at December 28, 2024 are as follows:

Fiscal Year	Operating Leases	Finance Leases
2025	\$ 127.4	\$ 50.8
2026	110.9	42.5
2027	97.0	35.7
2028	83.0	26.4
2029	84.0	24.4
Thereafter	440.5	20.1
Total future undiscounted lease payments	942.8	199.9
Less imputed interest	(120.0)	(17.0)
Total reported lease liability	822.8	182.9
Less current portion	(94.4)	(37.9)
Long-term lease liabilities	\$ 728.4	\$ 145.0

Additionally, the Company has excluded approximately \$21.6 million of payments (undiscounted basis) for leases that have no yet commenced. These leases will commence between the first and fourth quarters of 2025 with terms of up to fifteen years and includ new stores for ARH.

Other information related to operating and finance leases was as follows:

, ,		cember 28, 2024	December 30, 2023	
Cash paid for amounts included in the measurement of lease liabilities:	THE REAL PROPERTY.		OF THE STATE OF	100
Operating cash flows from operating leases	S	(115.9)	\$	(112.2)
Operating cash flows from finance leases	\$	(36.4)	5	(25.9)
Lease assets obtained in exchange for new lease liabilities:				
Operating leases	5	230.9	S	134.8
Finance leases	S	93.9	S	66.6
Weighted average remaining lease term for operating leases	91,720,6	9.9 years		9.2 years
Weighted average discount rate for operating leases		2.60%		2.44%
Weighted average remaining lease term for finance leases		4.9 years		4.9 years
Weighted average discount rate for finance leases		3.48%		2.79%

(7) Notes Receivable, net

The Company makes available to its Retailers various lending programs whose terms exceed one year. At December 28, 202 and December 30, 2023, the outstanding balance of the notes was \$36.8 million and \$30.2 million, respectively, of which the curren portion of \$35.5 million and \$28.2 million, respectively, was recorded in Receivables, net and the long-term portion of \$1.3 million an \$0.4 million, respectively, was recorded in Other assets.

		ember 28, 2024	December 30, 2023	
Notes receivable, gross	S	74.6	\$	66.4
Less accrued patronage to be applied		(37.8)		(36.2)
Net of accrued patronage		36.8		30.2
Less current portion of notes receivable included in Receivables, net		(35.5)		(28.2)
Less allowance for doubtful accounts				(1.6)
Notes receivable, net	S	1.3	\$	0.4

For substantially all of the Company's Notes receivable, the amounts due are generally expected to be collected through th non-cash portion of the annual patronage distribution and are therefore classified as current. In the event a Retailer cancels it membership with the Company, any outstanding loans are transferred from Notes receivable to Trade receivables and are du immediately. As the non-cash portion of the patronage distribution is used to settle the Notes receivable, there are no loans that are currently past due. The patronage distribution for each Retailer can vary from year to year based on the Company's financial performance, as well as the volume of patronage-based merchandise that each Retailer purchases from the Company. The contractual maturities, assuming no patronage deductions, of the Notes receivable are as follows:

	December 28, 2024	December 30, 2023
0 – 4 years	\$ 25.5	\$ 14.2
5 – 8 years	27.1	31.8
9 – 10 years	22.0	20.4
Total	\$ 74.6	\$ 66.4

The Company evaluates risk on its loan portfolio by categorizing each loan into an internal risk category. The Company's ris categories include:

Low - The Retailer possesses a strong financial position, above average payment record to both Ace and other vendors, and the business is well established.

Moderate - The Retailer possesses an average financial position, an average payment record to both Ace and other vendors and the business is somewhat established.

High - The Retailer possesses a weak financial position, a substandard payment record to Ace or other vendors, or the busines is somewhat new.

Based upon these criteria, the Company has classified its loan portfolio as follows:

Corporate Credit Exposure:	December 28, 2024	December 30, 2023		
Low risk	\$ 29.1	S	26.8	
Moderate risk	27.1		22.1	
High risk	18.4	E COL	17.5	
Total	\$ 74.6	\$	66.4	

Pursuant to the Company's Amended and Restated Certificate of Incorporation and the Company's by-laws, notes receivable (like all obligations owed to the Company by the Company's Retailers) are secured by the Company stock owned by the Retailers. Fo this reason, as well as the fact that substantially all of the notes receivable are repaid from the following year's annual patronage distribution, the actual risk of collection is low for the notes receivable portfolio despite many Retailers being classified as Moderat and High risk.

The Company applies a consistent practice of establishing an allowance for notes that it feels may become uncollectible be monitoring the financial strength of its Retailers. The collectability of certain notes is evaluated on an individual basis, while the remaining notes are evaluated on a collective basis. Due to the nature of the notes and the collateral held by the Company, virtually a outstanding notes were collectively evaluated for impairment.

The Company has evaluated the collectability of the notes and has established an allowance for doubtful accounts of \$2 million and \$1.6 million for December 28, 2024 and December 30, 2023, respectively. The allowance for 2024 was reclassified to Receivables, net as substantially all notes are classified as short-term. Management records the allowance for doubtful accounts base on the above information as well as judgments made considering a number of factors, primarily historical collection statistics, currer Retailer credit information, the current economic environment and the offsetting amounts due to Retailers for stock, notes, interest and declared and unpaid patronage distributions. The components of changes to the Notes receivable allowance for doubtful accounts for 2024 and 2023 were as follows:

		mber 28, 024	December 30, 2023		
Beginning balance	\$	1.6	\$	1	
Charges to provision		0.4		0.6	
Reclassifications (to) from accounts receivable allowance for doubtful accounts	after the	(2.0)	3. C [2]	1.0	
Ending balance	\$		\$	1.6	

Notes bear interest at various rates and are recorded at face value. Interest is recognized over the life of the note based on th outstanding balance and stated interest rate, which approximates the effective interest method. During fiscal years 2024, 2023 and 2022 \$5.9 million, \$4.6 million and \$2.2 million respectively, were recorded as interest income related to the notes.

Generally, in the event a Retailer cancels their membership with the Company, any outstanding Notes receivable, and relate allowance for doubtful accounts, are transferred to trade receivables and the Retailer is billed for any unpaid principal and interes balances. In fiscal 2024 and 2023, \$14.5 million and \$8.5 million, respectively, of Notes receivable were transferred to Trade receivable as an event occurred which made the notes due immediately.

(8) Goodwill and Other Intangible Assets

Changes in the carrying value of goodwill were as follows:

\$ 101.5
271.8
373.3
3.4
46.3
\$ 423.0

Impairment exists when a reporting unit's carrying value exceeds its fair value. The Company tests reporting units for impairment annually as of the first day of the fourth quarter, or more frequently if events or circumstances indicate it is more likely than not that the fair value of a reporting unit is less than its carrying value.

Identifiable intangible assets were as follows:

Trademarks and tradenames		mber 28, 024	December 30, 2023		
		9.2	\$	9.2	
Customer relationships	39.1		39.1		
Total identifiable intangible assets, gross		48.3	48.3		
Less accumulated amortization		(15.7)		(10.7)	
Identifiable intangible assets, net	\$	32.6	S	37.6	

The trademarks and tradenames are being amortized over 2-20 years. The customer relationships are being amortized over 10 years. Net amortization expense related to all intangible assets was \$5.0 million, \$3.8 million and \$6.5 million for fiscal years 2024, 2023 and 2022, respectively. The estimated total amortization expense over the next five fiscal years is \$18.7 million.

(9) Patronage Distributions and Refund Certificates Payable

The Company operates as a cooperative organization and has paid or may pay patronage distributions to Retailers on a portio of patronage-based income derived from business done with such Retailers. Patronage distributions are allocated in proportion to the volume of purchases by Retailers during the period. The cash portion of the patronage distribution was 40 percent for all years presented.

The accrued patronage distributions composition is summarized as follows:

			Ye	ars Ended		
Cash portion	De	2024	December 30, 2023		De	cember 31, 2022
	\$	137.2	\$	137.1	S	133.1
Class C stock		82.5		89.7		118.1
Patronage refund certificates		73.5		68.9		42.4
Patronage financing deductions		42.4		40.0		30.0
Total patronage distributions accrued for third party Retailers	5	335.6	S	335.7	S	323.6

Patronage distributions are allocated on a fiscal year basis with issuance in the following year.

In those instances where the Class C stock requirements have been met, the non-cash portion of the patronage distribution i distributed in the form of patronage refund certificates with a five-year term and bearing interest at 4 percent.

The patronage refund certificates outstanding at December 28, 2024 are payable as follows:

2025	2	15.0
2026		24.3
2027	ZIN	41.1
2028		66.9
2029	118	73.5
Total patronage refund certificates payable		220.8
Less current portion	No. of	(15.0)
Long-term patronage refund certificates payable	\$	205.8

(10) Accrued Expenses

Accrued expenses include the following components:

		mber 28, 2024		mber 30, 2023
Salaries and wages	S	95.2	\$	64.1
Insurance reserves		34.5		28.9
Deferred income		8.3	STANI	8.7
Vendor funds		16.7		36.9
Taxes		25.3	100	24.3
Profit sharing		23.7		25.5
Gift card	MIGHT	25.9	ALC:	23.7
Interest		9.3		7.8
Advertising and marketing		26.5		29.9
Other		47.7		42.2
Accrued expenses	S	313.1	S	292.0

(11) Debt

The Company has a \$1.1 billion line of credit facility. The facility is expandable to \$1.4 billion through a \$300.0 million accordion that is exercisable without the consent of existing lenders provided that the Company is not in default of the credit agreement and further provided that none of the existing lenders are required to provide any portion of the increased facility. At the Company's discretion, borrowings under the credit facility bear interest at a rate of either 25 to 100 basis points over the prime rate or 125 to 200 basis points over the Secured Overnight Financing Rate ("SOFR"), depending on the Company's leverage ratio. The credit facility was priced at SOFR plus 150 basis points at December 28, 2024. Fees on the unused portion of the line of credit range from 17.5 to 27.5 basis points depending on the Company's leverage ratio. The credit facility expires on January 24, 2028. As of December 28, 2024 the Company was in compliance with its covenants and had \$216.5 million in borrowings outstanding under the credit facility.

The credit facility includes a \$175.0 million sublimit for the issuance of standby and commercial letters of credit. As o December 28, 2024, a total of \$37.3 million in letters of credit were outstanding.

The credit facility allows the Company to make revolving loans and other extensions of credit to AIH in an aggregate principa amount not to exceed \$100.0 million at any time. As of December 28, 2024, there were no loans or other extensions of credit provided to AIH.

The Company's Westlake subsidiary has a \$125.0 million asset-based revolving credit facility ("Westlake Facility") that expires on August 3, 2026. The facility is expandable to \$150.0 million through a \$25.0 million accordion that is exercisable without the consent of existing lenders provided that the Company is not in default of the credit agreement and further provided that none of the existing lenders are required to provide a portion of the increased facility. Under this facility, Westlake has the right to issue letters of credit up to a maximum of \$7.5 million. Borrowings under this facility bear interest at a rate of either the prime rate plus an applicable spread of 25 to 50 basis points or SOFR plus an applicable spread of 135 to 160 basis points, depending on Westlake's average availability under the Westlake Facility as measured on a trailing 12-month basis. The Westlake Facility was priced at SOFR plus 13: basis points at December 28, 2024.

The Westlake Facility is collateralized by substantially all of Westlake's tangible and intangible assets. Borrowings under the facility are subject to a borrowing base calculation consisting of certain advance rates applied to eligible collateral balances (primarily consisting of certain receivables and inventories). This agreement requires maintenance of certain financial covenants including a minimum fixed charge coverage ratio. As of December 28, 2024, Westlake was in compliance with its covenants and had \$60.1 million in borrowings outstanding under the Westlake Facility.

The Westlake Facility includes a lender-controlled cash concentration system that results in all of Westlake's daily available cash being applied to the outstanding borrowings under their facility. Pursuant to Financial Accounting Standards Board ("FASB" Accounting Standards Codification Section 470-10-45, "Classification of Revolving Credit Agreements Subject to Lock-Box Arrangements and Subjective Acceleration Clauses," all borrowings under the Westlake Facility are classified as a Current maturity o long-term debt.

Total debt outstanding is comprised of the following:

	Dece	December 30, 2023		
Revolving Credit Facility	S	216.5	S	331.5
Westlake Facility		60.1		30.5
Installment notes with maturities through 2028 at a fixed rate of 6.00%		50.5		46.3
Total debt		327.1		408.3
Less maturities within one year		(80.6)	1101190	(49.1)
Long-term debt	\$	246.5	S	359.2

The aggregate scheduled maturities of total debt at December 28, 2024 are as follows:

Fiscal Year	Amount
2025	\$ 80.6
2026	13.5
2027	11.7
2028	221.3
Total debt	\$ 327.1

(12) Commitments and Contingencies

Contingencies

The Company has certain contingent liabilities resulting from litigation and claims incident to the ordinary course of business Management believes that the probable resolution of such contingencies will not materially affect the financial position, results of operations, or liquidity of the Company. The Company expenses legal fees as they are incurred.

Other guarantees

In the normal course of business, the Company enters into commercial commitments including standby letters of credit and guarantees that could become contractual obligations. Letters of credit are issued generally to insurance agencies and financial institutions in direct support of the Company's corporate and Retailer insurance programs as well as to international vendors for imported inventory purchases. As of December 28, 2024, the Company had outstanding standby letters of credit of \$17.2 million issued in the normal course of NAIL's operations and commercial letters of credit of \$37.3 million issued by the Company's credit facility.

The Company maintains the Ace Hardware Corporation Cooperative Group Health Plan ("AHP"), which is a MEWA that provides medical insurance and other welfare benefits to employees of the Company and their families and employees and families of participating Retailers.

The AHP is insured by Ace Group Insurance, Inc. ("AGI"), a Vermont-based captive insurance company. AGI is owned by funded welfare trust ("Trust") that serves as the mechanism for the AHP to collect, hold, invest and pay funds deposited with the AHI for the conduct of AHP operations. The Trust is not owned by the Company or any of its subsidiaries.

In order to establish and provide initial capitalization of the AHP, AGI and Trust, NAIL maintained a \$15.0 million standby letter of credit in favor of AGI. The letter of credit is held by the Vermont Department of Financial Regulations ("VDFR") and can be drawn upon at VDFR's direction if AGI becomes illiquid. The Company believes that the likelihood of the letter of credit being drawn is remote. On March 31, 2023, the \$15.0 million standby letter of credit provided by NAIL was terminated and reissued under the Company's credit facility.

(13) Fair Value

Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. There is a three-level hierarchy for disclosure to show the extent and level of judgment used to estimate fair value measurements

Level 1 – Investments included in this category are the Company's Fixed income mutual funds, Equity mutual funds, Mone market funds and U.S. government notes. The Company's valuation techniques used to measure the fair values of Level 1 investment are derived from quoted market prices for identical instruments, as active markets for these instruments exist. The Company ha classified its investments used to support the Company's deferred compensation plan as Other assets as these investments are restricted solely for this purpose.

Level 2 – Investments included in this category are the Company's Corporate fixed income securities, Mortgage-backer securities and Other debt securities. Other debt securities primarily consist of taxable municipal bonds, corporate asset-backed securities and U.S. Agency fixed rate notes and bonds. The Company's valuation techniques used to measure the fair values of Level 2 securities are derived from the following: non-binding market consensus prices that are corroborated by observable market data, quoted market prices for similar instruments, or pricing models, such as discounted cash flow techniques, with all significant inputs derived from o corroborated by observable market data.

Level 3 – Uses inputs that are unobservable and are supported by little or no market activity and reflect the use of significan management judgment. These values are generally determined using pricing models for which the assumptions utilize management' estimates of market participant assumptions.

The tables below set forth, by level, the Company's financial assets that were accounted for at fair value as of December 28 2024 and December 30, 2023. The tables do not include cash on hand and also do not include assets and liabilities that are measured a historical cost or any basis other than fair value. The carrying values for other current financial assets and liabilities, such as account receivable and accounts payable, approximate fair value due to the short maturity of such instruments. Long-term notes receivable approximate fair value because the Company charges its Retailers an interest rate and a significant portion of the notes have the Company's stock as collateral.

Carrying Value Measured at Fair Value

Items measured at fair value on a recurring basis	December		L	evel 1	L	evel 2	Lev	vel 3
Assets:		Michael I	TAX SO	STEIN	THE WATER	MA TELL	NUMBER OF STREET	1
Cash equivalents:								
Money market funds	\$	2.7	S	2.7	S	E KANTS	\$	10.0
Marketable securities:		1200	Table 1		17507	I SOPE	300	30.53
Corporate fixed income securities	\$	15.3	\$		\$	15.3	\$	
Mortgage-backed securities		11.3			6213	11.3	SME	(1940)
U.S. government notes		8.7		8.7				
Other	MAN TO SHARE	7.4		UP TO	EANER	7.4		1974
Total marketable securities	\$	42.7	S	8.7	\$	34.0	S	
Other assets:	2 A STATE	(New York Inc.		EGVEN	HEREN.	A SPORTS AND	Miles	100000
Money market funds	\$	2.7	\$	2.7	\$	1	5	1
Fixed income mutual funds		15.5		15.5				
Equity mutual funds		15.7		15.7		200	BY STEEL	1
Total other assets	S	33.9	\$	33.9	S		S	-

Carrying Value Measured at Fair Value

		L	evel 1	L	evel 2	Le	vel 3
	1137	The by		100	Page 12	A STATE OF	12/3
\$	2.5	S	2.5	\$	STATE OF	S	
		SAGE	ACHERO S	15.50			Safe:
\$	22.4	\$		S	22.4	\$	
	14.8		N. A. T.	15/25	14.8	NE ALL	17.
	4.4		4.4		-		
	7.8	SOLVE			7.8	CHAIR!	14.
\$	49.4	\$	4.4	\$	45.0	\$	
A STATE OF THE STA	No.	200					
S	2.6	\$	2.6	\$	ECOLUM	\$	
	13.0		13.0		-		
	14.7		14.7	1 2	S. Ann All	Re 120	180
\$	30.3	\$	30.3	\$		\$	
	S S	\$ 22.4 14.8 4.4 7.8 \$ 49.4 \$ 2.6 13.0 14.7	\$ 2.5 \$ \$ \$ 2.5 \$ \$ \$ 4.4 \$ 7.8 \$ \$ 49.4 \$ \$ \$ 13.0 \$ 14.7	December 30, 2023 Level 1 \$ 2.5 \$ 2.5 \$ 22.4 \$ - 14.8 - 4.4 4.4 7.8 - \$ 49.4 \$ 4.4 \$ 2.6 \$ 2.6 13.0 13.0 14.7 14.7	December 30, 2023 Level 1 L \$ 2.5 \$ 2.5 \$ \$ 22.4 \$ - \$ 14.8 - - 4.4 4.4 - 7.8 - - \$ 49.4 \$ 4.4 \$ \$ 2.6 \$ 2.6 \$ 13.0 13.0 13.0 14.7 14.7 14.7	December 30, 2023 Level 1 Level 2 \$ 2.5 \$ 2.5 \$ - \$ 22.4 \$ - \$ 22.4 14.8 - 14.8 4.4 4.4 - 7.8 - 7.8 \$ 49.4 \$ 4.4 \$ 45.0 \$ 2.6 \$ 2.6 \$ - 13.0 13.0 - 14.7 14.7 -	December 30, 2023 Level 1 Level 2 Level 2 \$ 2.5 \$ 2.5 \$ - \$ \$ 22.4 \$ - \$ 22.4 \$ 14.8 - 14.8 4.4 4.4 - 7.8 - 7.8 \$ 49.4 \$ 4.4 \$ 45.0 \$ 2.6 \$ 2.6 \$ - \$ 13.0 13.0 - 14.7 14.7 -

The fair value of the Company's debt and equity securities exceeded cost by \$1.5 million at December 28, 2024, but was below cost by \$0.8 million at December 30, 2023. The realized and unrealized gains and losses on equity securities are recorded to Other income, net and the realized gain or loss on debt securities are recorded in the period in which the gain or loss occurs to Other income net in the Consolidated Statement of Income. Unrealized gains and losses on debt securities are recorded as a component of Other comprehensive income (loss), net of tax. The Company has not recorded an allowance due to the unrealized losses being driven by the market factors related to increasing interest rates as opposed to credit factors. All debt securities are expected to reach par value at their maturity dates.

Gross proceeds from the sale of money market, debt and equity securities and the related realized gains and losses for the fisc years ended December 28, 2024, December 30, 2023 and December 31, 2022 were as follows:

			Yea	rs Ended		
	Dec	ember 28, 2024	Dec	ember 30, 2023	Dec	ember 31, 2022
Gross proceeds	\$	30.0	\$	43.2	\$	47.7
Gross realized gains		0.3		0.5		1.2
Gross realized losses		(0.8)		(1.2)		(3.2)

Gross realized gains and losses were determined using the specific identification method. For the fiscal year ended December 28, 2024, the Company reclassified \$0.1 million of unrealized gains and \$0.8 million of unrealized losses on marketable securities that were recorded in AOCI as of December 30, 2023 into realized income. These amounts were recorded to Other income, net in the Consolidated Statements of Income.

The following table summarizes the contractual maturity distributions of the Company's debt securities at December 28, 2024. Actual maturities may differ from the contractual or expected maturities since borrowers may have the right to prepay obligations without prepayment penalties.

Fair value of available-for-sale debt securities	One	ie in Year Less	On the	e After e Year rough Five 'ears	Five thr	After Years ough Years		e After		Total
Corporate fixed income securities	S	1.0	S	3.6	\$	5.1	5	5.6	\$	15.3
Mortgage-backed securities				-				11.3		11.3
U.S. government notes	328.74	0.3		7.7		0.6		0.1	358	8.7
Other		1.5		0.3		1.7		3.9		7.4
Total	S	2.8	\$	11.6	\$	7.4	\$	20.9	S	42.7

The Company's debt instruments are recorded at cost on the Consolidated Balance Sheets. The fair value of the Company's debt approximated carrying value at December 28, 2024. The estimated fair value of long-term debt is based on estimated rates fo similar instruments and discounted cash flow analysis using the Company's weighted-average interest rate and is, therefore, classifier as Level 3 within the fair value hierarchy.

(14) Income Taxes

Income tax benefit (expense) includes the following components:

		Years Ended		
Current:	December 28, 2024	December 30, 2023	December 31, 2022	
Federal	\$ (0,4)	\$ 0.1	\$ (9.1)	
State	(0.9)	(0.9)	(2.6)	
Foreign	(1.7)	(1.2)	(1.0)	
Current income tax expense	(3.0)	(2.0)	(12.7)	
Deferred:				
Federal	4.3	0.9	3.6	
State	1.0	0.1	0.1	
Deferred income tax benefit	5.3	1.0	3.7	
Total income tax benefit (expense)	\$ 2.3	\$ (1.0)	\$ (9.0)	

Income tax differs from the amount computed by applying the statutory U.S. Federal income tax rate of 21 percent for fiscal year ended December 28, 2024, December 30, 2023 and December 31, 2022 to pre-tax income because of the effect of the following items

		Years Ended						
	December 28, 2024		December 30, 2023		December 31, 2022			
Expected tax at U.S. Federal income tax rate	\$ (65.5)	\$	(68.1)	S	(73.4)		
Patronage distribution deductions	70.5			70.5		67.9		
Other, net expense	(2,7)		(3.4)		(3.5)		
Income tax benefit (expense)	\$ 2.3		\$	(1.0)	\$	(9.0)		

Deferred income taxes reflect the tax effects of temporary differences between the carrying amounts of existing assets an liabilities for financial reporting purposes and the amounts used for income tax purposes. Significant components of the Company' deferred tax assets and liabilities are as follows:

Foreign tax credits and other tax credit carryforwards \$ 11.7 \$ 12.9 Net operating loss carryforwards 35.8 16.8 Unearned insurance premium and loss reserves 1.9 1.7 Allowance for doubtful accounts 2.3 2.1 Inventory reserves 4.7 5.5 Deferred vendor rebates 17.7 19.0 Accrued compensation and benefits expense 25.3 21.1 Net unrealized losses 0.4 0.4 Cleased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2 Net deferred tax assets \$ 37.2 \$ 32.0	Deferred tax assets:	December 28, 2024	December 30, 2023
Unearned insurance premium and loss reserves 1.9 1.7 Allowance for doubtful accounts 2.3 2.1 Inventory reserves 4.7 5.5 Deferred vendor rebates 17.7 19.0 Accrued compensation and benefits expense 25.3 21.1 Net unrealized losses 0.4 0.4 Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax liabilities: 92.9 81.2 eferred tax liabilities: 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Foreign tax credits and other tax credit carryforwards	\$ 11.7	\$ 12.9
Allowance for doubtful accounts 2.3 2.1 Inventory reserves 4.7 5.5 Deferred vendor rebates 17.7 19.0 Accrued compensation and benefits expense 25.3 21.1 Net unrealized losses 0.4 0.4 Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax liabilities: 92.9 81.2 eferred tax liabilities: 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Net operating loss carryforwards	35.8	16.8
Inventory reserves 4.7 5.5 Deferred vendor rebates 17.7 19.0 Accrued compensation and benefits expense 25.3 21.1 Net unrealized losses 0.4 0.4 Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Unearned insurance premium and loss reserves	1.9	1.7
Deferred vendor rebates 17.7 19.0 Accrued compensation and benefits expense 25.3 21.1 Net unrealized losses 0.4 0.4 Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: Depreciation and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Allowance for doubtful accounts	2.3	2.1
Accrued compensation and benefits expense 25.3 21.1 Net unrealized losses 0.4 0.4 Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: Depreciation and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Inventory reserves	4.7	5.5
Net unrealized losses 0.4 0.4 Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: Depreciation and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Deferred vendor rebates	17.7	19.0
Leased assets 6.8 6.9 Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: 25.1 26.4 Amortization and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Accrued compensation and benefits expense	25.3	21.1
Other reserves 5.1 13.3 Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: 25.1 26.4 Amortization and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Net unrealized losses	0.4	0.4
Total deferred tax assets 111.7 99.7 Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 eferred tax liabilities: Depreciation and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Leased assets	6.8	6.9
Less valuation allowance (18.8) (18.5) Deferred tax assets 92.9 81.2 Eferred tax liabilities: Depreciation and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Other reserves	5.1	13.3
Deferred tax assets 92.9 81.2	Total deferred tax assets	111.7	99.7
Depreciation and deferred gains on property and equipment 25.1 26.4	Less valuation allowance	(18.8)	(18.5)
Depreciation and deferred gains on property and equipment 25.1 26.4 Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Deferred tax assets	92.9	81.2
Amortization of intangibles 8.0 1.3 Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	eferred tax liabilities:		
Prepaid expenses and deferred income 5.9 3.6 Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Depreciation and deferred gains on property and equipment	25.1	26.4
Inventory valuation 16.7 17.9 Deferred tax liabilities 55.7 49.2	Amortization of intangibles	8.0	1.3
Deferred tax liabilities 55.7 49.2	Prepaid expenses and deferred income	5.9	3.6
	Inventory valuation	16.7	17.9
Net deferred tax assets \$ 37.2 \$ 32.0	Deferred tax liabilities	55.7	49.2
	Net deferred tax assets	\$ 37.2	\$ 32.0

Net deferred tax assets are included in Other assets on the Consolidated Balance Sheets.

At December 28, 2024, the Company has federal and state net operating loss carryforwards of \$164.6 million available fo offset against future taxable income. The federal net operating loss can be carried forward indefinitely. For those states not conforming to the indefinite federal carryforward provisions, the state net operating losses can, generally, be carried forward through the 2044 tax year.

At December 28, 2024, the Company has foreign net operating loss carryforwards of \$44.2 million available for offset agains future taxable income. A valuation allowance of \$11.0 million has been established against the foreign net operating losses as it is more likely than not that the benefit of the net operating losses will not be realized.

At December 28, 2024, the Company has foreign tax credits of \$3.9 million available to offset future tax expense. The foreign tax credits may be carried forward through the 2034 tax year.

At December 28, 2024, the Company has state tax credit carryforwards of \$7.8 million available to offset future state incomtax expense. The state tax credits may be carried forward to tax years 2025 through 2027. A valuation allowance of \$7.8 million has been established against certain state tax credits as it is more likely than not that the benefit of the tax credits will not be realized.

The federal income tax returns of the consolidated group are subject to examination by the Internal Revenue Service ("IRS") generally for three years after the returns are filed. The 2020 through 2024 tax years remain subject to examination by the IRS. Fo state purposes, the 2019 through 2024 tax years remain subject to examination.

The Company recognizes interest and penalties related to uncertain tax positions in Income tax expense. Accrued interest and penalties included in the reserve for uncertain tax positions and recognized in income tax expense are immaterial. It is reasonably possible that the total amount of unrecognized tax benefits will increase or decrease within the next twelve months. The Company currently estimates that such increases and decreases will not be significant.

(15) Capital Stock

The Company's classes of stock are described below (not in millions):

Number of Shares at		
December 28, 2024	December 30, 2023	
10,000	10,000	
2,586	2,615	
20,000,000	10,000,000	
7,528,693	7,226,176	
825,206	897,317	
	December 28, 2024 10,000 2,586 20,000,000 7,528,693	

No dividends can be declared on any shares of any class of the Company's stock.

Upon termination of the Company's membership agreement with any retail outlet, all shares of stock of the Company held by the Retailer owning or controlling such outlet must be sold back to the Company, unless a transfer of such shares is made to anothe party accepted by the Company as a Retailer with respect to the same outlet. A single Class A share is issued to a Retailer only when the share subscribed has been fully paid and Class C shares are issued only when all shares subscribed with respect to a retail outlet hav been fully paid. Additional stock subscribed in the accompanying consolidated financial statements represents the paid portion of stock subscribed for stores that have not opened. All shares of stock are currently issued and repurchased at par value.

(16) Retirement Plans

The Company and its subsidiaries maintain profit sharing and 401k retirement plans for substantially all employees. Amount expensed under these plans totaled \$46.2 million, \$46.6 million and \$47.4 million during fiscal 2024, 2023 and 2022, respectively.

(17) Supplemental Disclosures of Cash Flow Information

During fiscal 2024, 2023 and 2022, accrued patronage distributions of \$60.4 million, \$42.2 million and \$31.6 million respectively, were offset against Trade receivables and Notes receivable owed to the Company by its Retailers with no net impact in th Consolidated Statements of Cash Flows. In addition, the Company had \$12.9 million in patronage refund certificates which wer included in Accounts payable pending payment as of December 28, 2024 and had no net impact in the Consolidated Statements of Cash Flows.

During fiscal 2024, 2023 and 2022, non-cash repurchases of stock from Retailers of \$44.4 million, \$39.7 million and \$40. million, respectively, were offset against Trade receivables of \$4.1 million, \$5.2 million and \$16.9 million, respectively, and Note receivable of \$14.9 million, \$8.5 million and \$9.6 million, respectively. The remaining \$25.4 million, \$26.0 million and \$14.3 million respectively, were primarily issued as notes payable with no net impact in the Consolidated Statements of Cash Flows.

The Company received \$6.9 million, \$9.8 million and \$11.3 million of property and equipment for fiscal years 2024, 2023 and 2022, respectively, prior to year-end and accrued for these items as no cash payments were made. These capital expenditures were not included in the Purchases of property and equipment in the Consolidated Statement of Cash Flows for fiscal years 2024, 2023 and 2022. The Company paid \$9.8 million, \$11.3 million and \$7.8 million for property and equipment for fiscal years 2024, 2023 and 2022 that was purchased and accrued during the years ended December 30, 2023, December 31, 2022 and January 1, 2022. These capital expenditures were included in the Purchases of property and equipment in the Consolidated Statement of Cash Flows for fiscal year 2024, 2023 and 2022.

(18) Cybersecurity Event

On October 29, 2023, Ace was the subject of a targeted cybersecurity incident. The Company shut down most of its operatin systems globally while managing the safety of its overall global systems environment and had limited ability to conduct operations including arranging for shipments of freight and distributing product to its member Retailers.

For the year-ended December 30, 2023, the Company incurred \$8.6 million of expenses related to this cybersecurity inciden to investigate and remediate the matter. During the year-ended December 28, 2024, the Company received an \$8.0 million insuranc recovery related the 2023 incident, partially offsetting the expenses incurred.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis summarizes the significant factors affecting Ace Hardware Corporation's ("th Company's") consolidated operating results and financial condition during the three-year period ended December 28, 2024 (th Company's fiscal years 2024, 2023 and 2022). Fiscal years 2024, 2023 and 2022 contain 52 weeks of operating results. Unles otherwise noted, all references herein for the years 2024, 2023 and 2022 represent fiscal years ended December 28, 2024, December 30, 2023 and December 31, 2022, respectively. This discussion and analysis should be read in conjunction with the consolidated financial statements and the related notes included in this annual report that have been prepared in accordance with U.S. generally accepte accounting principles ("GAAP").

Company Overview

The Company is a wholesaler of hardware and other related products and provides services and best practices for its member Retailers ("Retailers"). The overall home improvement industry is estimated to be approximately \$526 billion and consists of a broad range of products and services, including lawn and garden products, paint and sundries, certain building supplies and general merchandise typically used in connection with home and property improvement, remodeling, repair and maintenance. The industry is fragmented and competition exists between the large home improvement centers, retail hardware stores and other chains offering hardware merchandise. In addition, we face growing competition from online and multichannel retailers.

The Company's Retailers generally compete in the \$65 billion "convenience hardware" segment which is characterized be purchased primarily of products related to home maintenance and repair, including paint and related products, lawn and garde equipment, and those products less focused on large-scale building, renovation and remodeling projects. The Company believes that the following competitive strengths distinguish it from its peers and contribute to its success in the convenience hardware market: (1 strong consumer recognition of the Ace Brand; (2) well-regarded for exceptional customer service and convenience; (3) strength of distribution operations; (4) consolidated purchasing power; (5) differentiated and localized product and service offerings; and (6) diversified network of independent retailers.

The Company strives to be the best provider of products, services and operating methods for convenience hardware retailers. The four main drivers that support that goal and the Company's efforts to grow the business are improving the store model, accelerating new store openings, increasing store projects that drive store sales and profitability and reducing the number of store closings.

Ace Hardware International Holdings, Ltd. ("AIH"), is a majority-owned and controlled subsidiary of the Company with a 21. percent noncontrolling interest held by AIH customers. AIH shareholders do not own shares of stock in the Company nor receiv patronage dividends. AIH has wholesale distribution capabilities in Ningbo, China and Cuautitlán Izcalli, Mexico. AIH customer operate 822 stores in approximately 60 countries.

Ace Retail Holdings LLC ("ARH") is the owner of the 180-store Westlake Ace Hardware ("Westlake") and the 79-store Great Lakes Ace Hardware ("GLA") retail chains. As a result, the Company is also a retailer of hardware, paint and other related products.

Ace Wholesale Holdings LLC ("AWH") owns and operates Emery Jensen Distribution ("EJD"). EJD is a wholesaler of hardware, paint and other related products to non-member retailers.

Ace Services Holdings LLC ("ASH"), through its subsidiaries, operates Ace Hardware Home Services ("AHHS"), a provide of residential plumbing, heating, cooling and electric ("PHCE") services, and Ace Handyman Services, a franchisor of hom improvement, maintenance and repair services.

Worldwide Store Count

The number of worldwide Ace retail outlets during the past three fiscal years is summarized as follows:

	Fiscal Years					
	2024	2023	2022			
Retail outlets at beginning of period	5,870	5,746	5,583			
New retail outlets	230	222	232			
Retail outlet cancellations	(134)	(98)	(69)			
Retail outlets at end of period	5,966	5,870	5,746			

Worldwide Customer Count

The number of wholesale customers served worldwide during the past three fiscal years is summarized as follows:

	Fiscal Years				
	2024	2023	2022		
Ace retail outlets	5,966	5,870	5,746		
Non-member outlets	2,994	2,985	3,007		
Total outlets served at end of period	8,960	8,855	8,753		

Results of Operations

Comparison of the Year Ended December 28, 2024 to the Year Ended December 30, 2023

The following data summarizes the Company's performance in fiscal 2024 as compared to fiscal 2023 (in millions):

	2	024	2	023	Favorable/ (Unfavorable)		
	s	% of Revenues*	s	% of Revenues*	s	%	
Revenues:	AVAILED ST		The officer			(2) (5) (5) (1)	
Wholesale revenues	8,666.0	91.3%	8,344.7	91.4%	321.3	3.9%	
Retail revenues	825.4	8.7%	786.4	8.6%	39.0	5.0%	
Total revenues	9,491.4	100.0%	9,131.1	100.0%	360.3	3.9%	
Gross profit:			111111111111111111111111111111111111111	We are stored	VALUE SA	SON THE REAL PROPERTY.	
Wholesale gross profit	1,170.4	13.5%	1,121.6	13.4%	48.8	4.4%	
Retail gross profit	374.7	45.4%	356.3	45.3%	18.4	5.2%	
Total gross profit	1,545.1	16.3%	1,477.9	16.2%	67.2	4.5%	
Operating expenses:			STATE OF THE			10176	
Wholesale operating expenses	925.3	10.7%	830.7	10.0%	(94.6)	(11.4%	
Retail operating expenses	297.7	36.1%	280.1	35.6%	(17.6)	(6.3%	
Retail pre-opening expenses	4.8	0.6%	6.0	0.8%	1.2	20.0%	
Total operating expenses	1,227.8	12.9%	1,116.8	12.2%	(111.0)	(9.9%	
Operating income	317.3	3.4%	361.1	4.0%	(43.8)	(12.1%	
Interest expense	(48.9)	(0.5%)	(38.4)	(0.4%)	(10.5)	(27.3%	
Other income, net	46.6	0.4%	3.4	0.0%	43.2	1270.6%	
Net income attributable to Ace Hardware Corporation	315.0	3.3%	326.1	3.6%	(11.1)	(3.4%	

^{*}Wholesale gross profit and expenses are shown as a percentage of wholesale revenues. Retail gross profit and expenses are shown as a percentage of retail revenues. Non-operating items are shown as a percentage of total revenues.

A reconciliation of consolidated revenues follows (in millions):

	Amount	% Change vs. 2023
2023 Revenues	\$ 9,131.1	
Wholesale Merchandise Revenues change based on new and cancelled domestic Ace stores:		
Revenues increase from new stores added since January 2023	238.0	2.6%
Revenues decrease from stores cancelled since January 2023	(32.9)	(0.4%)
Decrease in wholesale merchandise revenues to comparable domestic Ace stores	(35.2)	(0.4%)
Increase in wholesale service revenues from domestic Ace stores	66.4	0.7%
Increase in AWH wholesale revenues	22.0	0.3%
Decrease in AIH wholesale revenues	(1.0)	(0.0%)
Increase in ASH service revenues	46.1	0.5%
Increase in Acehardware.com revenues	48.3	0.5%
Increase in ARH retail revenues	39.0	0.4%
Other revenue changes, net	(30.4)	(0.3%)
2024 Revenues	\$ 9,491.4	3.9%

Consolidated revenues for the year ended December 28, 2024 totaled \$9.5 billion, an increase of \$360.3 million, or 3.9 percer as compared to the prior year.

Total wholesale revenues were \$8.7 billion for fiscal 2024, an increase of \$321.3 million or 3.9 percent, as compared to the prior year. Increases were seen across many departments with outdoor power equipment, grilling, power tools and generators showing the largest gains. New domestic Ace stores are defined as stores that were activated from January 2023 through December 2024. It is 2024, the Company had an increase in wholesale merchandise revenues from new domestic Ace stores of \$238.0 million. This increase was partially offset by a decrease in wholesale merchandise revenues due to domestic Ace store cancellations of \$32.9 million. As result, the Company realized a net increase in wholesale merchandise revenues of \$205.1 million related to the impact of both new Ac stores affiliated with the Company and from stores that cancelled their membership in 2023 and 2024. Wholesale merchandise revenue to comparable domestic Ace stores decreased \$35.2 million compared to the prior year. Warehouse sales represented 77.2 percent of wholesale merchandise revenue in 2024 compared to 78.2 percent in 2023, while direct ship sales were 22.8 percent, up from 21. percent in 2023.

AWH wholesale revenues were \$480.5 million during fiscal 2024. This is an increase of \$22.0 million from fiscal 2023, which was primarily due to higher sales to e-commerce and pro lumber customers.

AIH wholesale revenues were \$215.0 million during fiscal 2024. This is a decrease of \$1.0 million from fiscal 2023.

ASH wholesale revenues were \$171.8 million during fiscal 2024. This is an increase of \$46.1 million from fiscal 2023 due to the inclusion of a full year of revenues from UIC during fiscal 2024. UIC was acquired on April 12, 2023.

Revenues from Acehardware.com were \$382.3 million during fiscal 2024. This is an increase of \$48.3 million from fiscal 2023.

Total retail revenues from ARH were \$825.4 million, an increase of \$39.0 million, or 5.0 percent, as compared to the price year. The GLA chain experienced a 0.7 percent increase in same-store-sales, while the Westlake chain experienced a 0.7 percent decrease in same-store-sales in fiscal 2024. The new stores opened by ARH in 2023 and 2024 contributed \$42.5 million of incrementar revenues during fiscal 2024.

Wholesale gross profit for fiscal 2024 was \$1.2 billion, an increase of \$48.8 million from fiscal 2023. The wholesale gross margin percentage was 13.5 percent of wholesale revenues in fiscal 2024, up slightly from 13.4 percent in fiscal 2023.

Retail gross profit for fiscal 2024 was \$374.7 million, an increase of \$18.4 million from fiscal 2023. The retail gross margi percentage was 45.4 percent of retail revenues in fiscal 2024, up slightly from 45.3 percent in fiscal 2023. For ARH, retail gross prof as reported in the Ace financial statements is based on the Ace wholesale acquisition cost of product, not ARH's acquisition cost whic includes a markup from the Company.

Wholesale operating expenses increased \$94.6 million, or 11.4 percent in fiscal 2024, as compared to fiscal 2023. This increas is due to a \$59.8 million increase in marketing and advertising spending, higher compensation benefit expense and a full year of operatin expenses from UIC in 2024 compared to a partial year in 2023. As a percentage of wholesale revenues, wholesale operating expense increased to 10.7 percent of wholesale revenues in fiscal 2024 from 10.0 percent of wholesale revenues in fiscal 2023.

Retail operating expenses increased \$17.6 million, or 6.3 percent in fiscal 2024, as compared to fiscal 2023. The increase i driven by expenses incurred related to new stores added in 2023 and 2024. Retail operating expenses as a percentage of retail revenu increased to 36.1 percent of retail revenues in fiscal 2024 from 35.6 percent in fiscal 2023.

Interest expense increased \$10.5 million compared to fiscal 2023 due to higher average borrowings.

Other income increased \$43.2 million compared to fiscal 2023 due to a \$17.8 million gain on sale of the Company's RSG located in Little Rock, Arkansas and an insurance recovery of approximately \$8.0 million of the \$8.6 million of expenses the Company recorded in 2023 related to the cybersecurity incident.

Comparison of the Year Ended December 30, 2023 to the Year Ended December 31, 2022

The following data summarizes the Company's performance in fiscal 2023 as compared to fiscal 2022 (in millions):

	2	023	2	022	Favorable/ (Unfavorable)		
	\$	% of Revenues*	\$	% of Revenues*	s	%	
Revenues:							
Wholesale revenues	8,344.7	91.4%	8,372.3	91.3%	(27.6)	(0.3%	
Retail revenues	786.4	8.6%	797.5	8.7%	(11.1)	(1.4%	
Total revenues	9,131.1	100.0%	9,169.8	100.0%	(38.7)	(0.4%	
Gross profit:			A CHICATON	EC AND THE			
Wholesale gross profit	1,121.6	13.4%	1,003.5	12.0%	118.1	11.8%	
Retail gross profit	356.3	45.3%	357.9	44.9%	(1.6)	(0.4%	
Total gross profit	1,477.9	16.2%	1,361.4	14.8%	116.5	8.6%	
Operating expenses:							
Wholesale operating expenses	830.7	10.0%	730.1	8.7%	(100.6)	(13.8%	
Retail operating expenses	280.1	35.6%	287.5	36.1%	7.4	2.6%	
Retail pre-opening expenses	6.0	0.8%	1.9	0.2%	(4.1)	(215.8%	
Total operating expenses	1,116.8	12.2%	1,019.5	11.1%	(97.3)	(9.5%	
Operating income	361.1	4.0%	341.9	3.7%	19.2	5.6%	
Interest expense	(38.4)	(0.4%)	(19.2)	(0.2%)	(19.2)	(100.0%	
Other income, net	3.4	0.0%	19.0	0.2%	(15.6)	(82.1%	
Net income attributable to Ace Hardware Corporation	326.1	3.6%	341.7	3.7%	(15.6)	(4.6%	

^{*}Wholesale gross profit and expenses are shown as a percentage of wholesale revenues. Retail gross profit and expenses are shown as a percentage of total revenues. Non-operating items are shown as a percentage of total revenues.

A reconciliation of consolidated revenues follows (in millions):

	Amount	% Change vs. 2022
2022 Revenues	\$ 9,169.8	CONTRACTOR OF STREET
Wholesale Merchandise Revenues change based on new and cancelled domestic Ace stores.		
Revenues increase from new stores added since January 2022	207.2	2.3%
Revenues decrease from stores cancelled since January 2022	(43.5)	(0.5%)
Decrease in wholesale merchandise revenues to comparable domestic Ace stores	(333.8)	(3.7%)
Increase in wholesale service revenues from domestic Ace stores	14.2	0.2%
Decrease in AWH wholesale revenues	(20.1)	(0.2%)
Decrease in AIH wholesale revenues	(7.6)	(0.1%)
Increase in ASH service revenues	109.0	1.2%
Increase in Acehardware.com revenues	31.5	0.3%
Decrease in ARH retail revenues	(0.6)	(0.0%)
Other revenue changes, net	5.0	0.1%
2023 Revenues	\$ 9,131.1	(0.4%)

Consolidated revenues for the year ended December 30, 2023 totaled \$9.1 billion, a decrease of \$38.7 million, or 0.4 percent as compared to the prior year.

Total wholesale revenues were \$8.3 billion for fiscal 2023, a decrease of \$27.6 million or 0.3 percent, as compared to the prio year. Decreases were seen across many departments with plumbing, grilling and outdoor power equipment showing the largest declines New domestic Ace stores are defined as stores that were activated from January 2022 through December 2023. In 2023, the Company had an increase in wholesale merchandise revenues from new domestic Ace stores of \$207.2 million. This increase was partially offse by a decrease in wholesale merchandise revenues due to domestic Ace store cancellations of \$43.5 million. As a result, the Company realized a net increase in wholesale merchandise revenues of \$163.7 million related to the impact of both new Ace stores affiliated with the Company and from stores that cancelled their membership in 2022 and 2023. Wholesale merchandise revenues to comparable domestic Ace stores decreased \$333.8 million compared to the prior year. Warehouse sales represented 78.2 percent of wholesale merchandise revenue in 2023 compared to 78.3 percent in 2022, while direct ship sales were 21.8 percent, up from 21.7 percent in 2022

AWH wholesale revenues were \$458.5 million during fiscal 2023. This is a decrease of \$20.1 million from fiscal 2022, which was primarily due to lower sales to paint, hardware and e-commerce customers.

AIH wholesale revenues were \$216.0 million during fiscal 2023. This is a decrease of \$7.6 million from fiscal 2022 du entirely to a decline in sales to AIH's largest customer.

ASH wholesale revenues were \$125.7 million during fiscal 2023. This is an increase of \$109.0 million from fiscal 2022 du to the acquisition of UIC during the second quarter of 2023, which contributed \$99.5 million of incremental revenues during fiscal 2023

Revenues from Acehardware.com were \$334.0 million during fiscal 2023. This is an increase of \$31.5 million from fiscal 2022.

Total retail revenues were \$786.4 million, a decrease of \$11.1 million, or 1.4 percent, as compared to the prior year. Retail revenues were negatively impacted by the closure of The Grommet during the third quarter of 2022, resulting in the absence of revenue from Ace Ecommerce Holdings ("AEH") in 2023, compared with \$10.5 million in the prior year. ARH revenues decreased \$0.6 million or 0.1 percent from the prior year. The Westlake chain experienced a 1.8 percent decrease in same-store-sales, while the GLA chain experienced a 4.2 percent decrease in same-store-sales in fiscal 2023. The new stores opened by ARH in 2022 and 2023 contribute \$17.8 million of incremental revenues during fiscal 2023.

Wholesale gross profit for fiscal 2023 was \$1.1 billion, an increase of \$118.1 million from fiscal 2022. The wholesale gross margin percentage was 13.4 percent of wholesale revenues in fiscal 2023, up from 12.0 percent in fiscal 2022. The increase in wholesal gross margin percentage was primarily due to the reversal of last-in, first-out ("LIFO") from an expense in fiscal 2022 when vendo prices were rising, to LIFO income in fiscal 2023 when inflation moderated and inventory levels declined.

Retail gross profit for fiscal 2023 was \$356.3 million, a decrease of \$1.6 million from fiscal 2022. The retail gross margin percentage was 45.3 percent of retail revenues in fiscal 2023, up from 44.9 percent in fiscal 2022. This increase in gross margin percentage primarily resulted from vendor funds earned related to new store openings. For ARH, retail gross profit as reported in the Ace financial statements is based on the Ace wholesale acquisition cost of product, not ARH's acquisition cost which includes a markut from the Company.

Wholesale operating expenses increased \$100.6 million, or 13.8 percent in fiscal 2023, as compared to fiscal 2022. Thi increase is primarily due to the acquisition of UIC during the second quarter of 2023, which contributed \$49.5 million of incrementa operating expenses in fiscal 2023, increased lease expense driven by additional warehouse space, increased marketing and advertising expenses, increased depreciation expense and higher health insurance expenses. As a percentage of wholesale revenues, wholesal operating expenses increased to 10.0 percent of wholesale revenues in fiscal 2023 from 8.7 percent of wholesale revenues in fiscal 2022

Retail operating expenses decreased \$7.4 million, or 2.6 percent in fiscal 2023, as compared to fiscal 2022. This decrease wa driven by the closure of The Grommet during the third quarter of 2022, resulting in the absence of operating expenses from AEH i fiscal 2023. Retail operating expenses as a percentage of retail revenue decreased to 35.6 percent of retail revenues in fiscal 2023 from 36.1 percent in fiscal 2022.

Retail pre-opening expenses increased \$4.1 million in fiscal 2023 due to an increase in new stores opened by ARH in 202 compared to 2022.

Interest expense increased \$19.2 million compared to fiscal 2022 due to increased borrowings for the acquisition of UIC an higher interest rates.

Other income, net decreased \$15.6 million compared to fiscal 2022 due to a \$21.7 million gain on sale of the Company's RSC located in Gainesville, Georgia during 2022 and \$8.6 million of expenses incurred in 2023 related to a cybersecurity incident; partially offset by an \$8.0 million decrease in income tax expense.

Liquidity and Capital Resources

The Company believes that existing cash balances, along with the existing lines of credit and long-term financing, will be sufficient to finance the Company's working capital requirements, debt service, patronage distributions, capital expenditures, share redemptions from Retailer cancellations and growth initiatives for at least the next 12 months.

The Company's borrowing requirements have historically arisen from, and are expected to continue to arise from, seasona working capital needs, debt service, capital improvements and acquisitions, patronage distributions and other general corporate purposes. In the past, the Company has met its operational cash needs using cash flows from operating activities and funds from its revolving credit facilities. The Company currently estimates that its cash flows from operating activities and working capital, together with it lines of credit, will be sufficient to fund its short-term liquidity needs. Actual liquidity and capital funding requirements depend on numerous factors, including operating results, general economic conditions and the cost of capital.

The Company has a \$1.1 billion line of credit facility. The facility is expandable to \$1.4 billion through a \$300.0 million accordion that is exercisable without the consent of existing lenders provided that the Company is not in default of the credit agreement and further provided that none of the existing lenders are required to provide any portion of the increased facility. At the Company's discretion, borrowings under the credit facility bear interest at a rate of either 25 to 100 basis points over the prime rate or 125 to 200 basis points over the Secured Overnight Financing Rate ("SOFR"), depending on the Company's leverage ratio. The credit facility was priced at SOFR plus 150 basis points at December 28, 2024. Fees on the unused portion of the line of credit range from 17.5 to 27.5 basis points depending on the Company's leverage ratio. The credit facility expires on January 24, 2028. As of December 28, 2024 the Company was in compliance with its covenants and had \$216.5 million in borrowings outstanding under the credit facility.

The credit facility includes a \$175.0 million sublimit for the issuance of standby and commercial letters of credit. As o December 28, 2024, a total of \$37.3 million in letters of credit were outstanding.

The credit facility allows the Company to make revolving loans and other extensions of credit to AIH in an aggregate principa amount not to exceed \$100.0 million at any time. As of December 28, 2024, there were no loans or other extensions of credit provided to AIH.

The Company's Westlake subsidiary has a \$125.0 million asset-based revolving credit facility ("Westlake Facility") that expires on August 3, 2026. The facility is expandable to \$150.0 million through a \$25.0 million accordion that is exercisable without the consent of existing lenders provided that the Company is not in default of the credit agreement and further provided that none of the existing lenders are required to provide a portion of the increased facility. Under this facility, Westlake has the right to issue letters or credit up to a maximum of \$7.5 million. Borrowings under this facility bear interest at a rate of either the prime rate plus an applicable spread of 25 to 50 basis points or SOFR plus an applicable spread of 135 to 160 basis points, depending on Westlake's average availability under the Westlake Facility as measured on a trailing 12-month basis. The Westlake Facility was priced at SOFR plus 135 basis points at December 28, 2024.

The Westlake Facility is collateralized by substantially all of Westlake's tangible and intangible assets. Borrowings under the facility are subject to a borrowing base calculation consisting of certain advance rates applied to eligible collateral balances (primarily consisting of certain receivables and inventories). This agreement requires maintenance of certain financial covenants including a minimum fixed charge coverage ratio. As of December 28, 2024, Westlake was in compliance with its covenants and had \$60.1 million in borrowings outstanding under the Westlake Facility.

The Westlake Facility includes a lender-controlled cash concentration system that results in all of Westlake's daily available cash being applied to the outstanding borrowings under their facility. Pursuant to FASB Accounting Standards Codification Section 470-10-45, "Classification of Revolving Credit Agreements Subject to Lock-Box Arrangements and Subjective Acceleration Clauses," all borrowings under the Westlake Facility are classified as a Current maturity of long-term debt.

As of December 28, 2024, total debt was \$327.1 million and was comprised of \$276.6 million borrowed on lines of credit and \$50.5 million owed to former Retailers. As of December 30, 2023, total debt of \$408.3 million was comprised of \$362.0 million borrowed on lines of credit and \$46.3 million owed to former Retailers.

Cash Flows

The Company had \$36.3 million and \$31.9 million of cash and cash equivalents at December 28, 2024 and December 30, 2023 respectively. Following is a summary of the Company's cash flows from operating, investing and financing activities for fiscal year 2024 and 2023, respectively (in millions):

2024	2023	
\$ 433.5	\$ 457.4	
114.6	(93.6)	
548.1	363.8	
(222.8)	(502.3)	
(320.9)	141.9	
\$ 4.4	\$ 3.4	
	\$ 433.5 114.6 548.1 (222.8) (320.9)	

The Company's operating activities generated \$548.1 million of cash in fiscal 2024 compared to \$363.8 million in fiscal 2023. Excluding the impact of net changes in assets and liabilities, cash provided by operating activities decreased from \$457.4 million is fiscal 2023 to \$433.5 million in fiscal 2024. The \$23.9 million decrease was driven by the \$9.3 million decrease in net income and non-operating gain recognized in fiscal 2024.

The net change in assets and liabilities provided \$114.6 million of cash in fiscal year 2024 compared to \$93.6 million used i fiscal year 2023. The net change was driven by an increase in accounts payable and accrued expenses; partially offset by an increase i accounts receivable. Accounts payable and accrued expenses increased \$149.7 million during fiscal 2024 compared to a decrease of \$104.7 million during fiscal 2023. The increase in accounts payable and accrued expenses was due to higher vendor purchases drive by higher sales volumes. Accounts receivable increased \$78.5 million during fiscal 2024 compared to an increase of \$55.4 million during fiscal 2023. The higher increase in accounts receivable was due to higher sales volumes.

Net cash used for investing activities was \$222.8 million in fiscal 2024 compared to \$502.3 million in fiscal 2023. Investin activities in fiscal 2024 consisted of \$155.4 million in capital expenditures, \$58.6 million paid to acquire businesses and a \$37.3 million increase in notes receivable for Retailer loan programs; partially offset by \$22.7 million of proceeds from the sale of the Company' RSC located in Little Rock, Arkansas. Investing activities in fiscal 2023 consisted of \$309.7 million in cash paid for acquisitions to support the growth of the Ace Home Services business and \$164.6 million in capital expenditures.

Net cash used in financing activities was \$320.9 million in fiscal 2024 compared to \$141.9 million provided in fiscal 2023 During 2024, the Company paid \$137.1 million for the cash portion of the 2023 patronage distribution, paid down borrowings under it revolving credit facilities by \$85.3 million, made principal payments on finance leases of \$40.3 million, paid \$22.9 million on patronag refund certificates and made payments on long-term debt of \$21.1 million. During 2023, the Company had \$362.0 million of ne borrowings under its revolving credit facilities, paid \$133.1 million for the cash portion of the 2022 patronage distribution, mad principal payments on finance leases of \$28.3 million, paid \$26.7 million on patronage refund certificates and made payments on long term debt of \$17.8 million.

Off-Balance Sheet Arrangements

The Company has certain other guarantees, as further described in the Notes to the Consolidated Financial Statements – Not 12 – Commitments and Contingencies. The Company believes the likelihood of any such payment under these guarantees is remote.

Application of Critical Accounting Policies and Estimates

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumption that affect the reported amounts of assets, liabilities, revenues and expenses in the consolidated financial statements. On an ongoin basis, the Company evaluates its estimates and judgments based on historical experience and various other factors that are believed to be reasonable under the circumstances. Actual results may differ from these estimates, and these estimates would vary under different assumptions or conditions. Management believes these estimates and assumptions are reasonable.

The Company annually reviews its financial reporting and disclosure practices and accounting policies to ensure that the provide accurate and comprehensive information relative to the current economic and business environment. The Company's significan accounting policies are described in the Notes to the Consolidated Financial Statements. The following represents those critica accounting policies which involve a relatively higher degree of judgment, estimation and complexity and where materially different amounts could be reported under different conditions or using different assumptions.

Valuation of Inventories When necessary, the Company provides allowances to adjust the carrying value of inventories to th lower of cost or market, including costs to sell or dispose of surplus or damaged/obsolete inventory, and for estimated shrinkage Estimates of the future demand for the Company's products are key factors used by management in assessing the net realizable value of the inventories. While management believes that the estimates used are appropriate, an unanticipated decline in revenues at retail outlet or a significant decline in demand for products in selected product categories could result in valuation adjustments.

Vendor Funds The Company receives funds from vendors in the normal course of business principally as a result of purchas volumes, revenues or promotions of vendors' products. Based on the provisions of the vendor agreements in place, management develops accrual rates by estimating the point at which the Company will have completed its performance under the agreement and the amount agreed upon will be earned. Amounts accrued throughout the year can be impacted if actual purchase volumes differ from projected annual purchase volumes, especially in the case of programs that provide for increased funding when graduated purchase volumes are met. At year-end, the accrual reflects actual purchases made throughout the year.

Vendor funds are treated as a reduction of inventory cost, unless they represent a reimbursement of specific, incremental an identifiable costs incurred by the customer to sell the vendor's product, in which case the costs would be netted. The majority of th vendor funds that the Company receives do not meet the specific, incremental and identifiable criteria. Therefore, the Company treat a majority of these funds as a reduction in the cost of inventory as the amounts are accrued and recognized as a reduction of cost of revenues when the inventory is sold.

Allowance for Expected Credit Losses The Company estimates an allowance for credit losses related to future expected losse by using relevant information such as historical collection statistics, current customer credit information, the current economic environment, the aging of receivables, the evaluation of compliance with lending covenants and the offsetting amounts due to Retailer for stock, notes, interest and anticipated but unpaid patronage distributions. Significant past due accounts are reserved on a specific identification basis. For the remainder of the accounts, the allowance is evaluated on a pool basis for accounts with similar rist characteristics and a loss-rate for each pool is determined using current conditions and historical credit loss statistics. The Compan considers accounts and notes receivable past due if invoices remain unpaid past their due date and provides for the write-off of uncollectible receivables after exhausting all commercially reasonable collection efforts. While the Company believes it has appropriately considered known or expected outcomes, its customers' ability to pay their obligations, including those to the Company could be adversely affected by declining revenues at retail resulting from such factors as contraction in the economy or competitive conditions in the wholesale and retail industry including increased competition from omni-channel retailers, discount stores, chain store and other mass merchandisers.

Insurance Reserves Insurance reserves for claims related to the Company's self-insured property, general liability, workers compensation and auto liability insurance programs are dependent on assumptions used in calculating such amounts. These assumption include projected ultimate losses and confidence levels of the reserve requirement and consider historical loss levels and other factors. While management believes that the assumptions used are appropriate, differences in actual claims experience or changes in assumption may affect the Company's insurance reserves.

Goodwill Goodwill represents the excess of the cost of an acquired business over the fair value of net tangible assets acquire and identified intangible assets. Goodwill is not amortized but is tested for impairment at a reporting unit level on an annual basis of more frequently, if circumstances change or an event occurs that would more likely than not reduce the fair value of a reporting un below its carrying amount.

Qualitative factors may be assessed to determine whether it is more likely than not that the fair value of the reporting unit less than its carrying amount. If an election is made to not perform the qualitative assessment, or the qualitative assessment indicate that the carrying amount is more likely than not higher than the fair value, goodwill is tested for impairment. If the fair value of reporting unit exceeds its carrying amount, goodwill of the reporting unit is considered not impaired. If the carrying amount of reporting unit exceeds its fair value, an impairment loss shall be recognized in an amount equal to that excess.

Significant judgment is applied when goodwill is assessed for impairment. This judgment includes developing cash flow projections, selecting appropriate discount rates, identifying relevant market comparables, incorporating general economic and market conditions and selecting an appropriate control premium. The income approach is based on discounted cash flows, which are derive from internal forecasts and economic expectations for each respective reporting unit.

The Company tests reporting units for impairment annually as of the first day of the fourth quarter, or more frequently if event or circumstances indicate it is more likely than not that the fair value of a reporting unit is less than its carrying value. No goodwi impairment charge was recorded during fiscal years 2024 and 2023.

Impact of New Accounting Standards

There are no recently issued applicable accounting pronouncements that are not yet effective that are expected to have a materia impact on consolidated results of operations, consolidated financial condition, or consolidated cash flows.

Qualitative and Quantitative Disclosure About Market Risk

Inflation and Changes in Prices The Company's business is not generally governed by contracts that establish price substantially in advance of the receipt of goods or services. As vendors increase or decrease their prices for merchandise supplied to the Company, the Company generally increases or decreases the price to its customers in an equal amount plus the normal handlin charge on such amounts. The U.S. economy experienced a significant increase in inflation as the prices of raw materials and freight negatively impacted the costs the Company pays to acquire products. For the most part, the Company was able to offset the effect of increased costs by passing along cost increases to the Company's customers. Although inflation has slowed, the Company is unable to

ensure that a future increase in inflation will be able to be passed on to its customers through increased prices. If these increases canno be passed to our customers, our financial condition, results of operations, liquidity, and cash flows may be negatively affected.

Foreign Currency Although the Company has international operating entities, its exposure to foreign currency rate fluctuation is not significant to its financial condition and results of operations.

Customer Credit Risk The Company is exposed to the risk of financial non-performance by customers. The Company's abilit to collect on sales to its customers is dependent on the liquidity of its customer base. Volatility in credit markets may reduce the liquidity of the Company's customer base. To manage customer credit risk, the Company monitors historical collection statistics, current retaile credit information, the current economic environment, the aging of receivables, the evaluation of compliance with lending covenant and the offsetting amounts due to Retailers for stock, notes, interest and anticipated but unpaid patronage distributions. From certain customers, the Company also obtains collateral as considered necessary to reduce risk of loss. The Company does not believe the los of any single customer would have a material adverse effect on its results of operations.

Interest Rate Volatility The Company has exposure to fluctuations in interest rates on its floating rate debt. The Company believes that its exposure to interest rate fluctuations is not significant to its financial condition and results of operations.

Certain Other Risks

The Company relies extensively on information technology systems, some of which are managed or provided by third-part service providers, to analyze, process, store, manage and protect transactions and data. In managing the business, the Company als relies heavily on the integrity of, security of and consistent access to this data for information such as customer data, merchandis planning and inventory replenishment, supplier purchases, sales to Retailers, warehousing, distribution, inventory control, and variou corporate-level financial and other functions including communication with the Retailers. The Company's systems and the system managed by third parties are subject to damage or interruption from a number of causes, including power outages; computer antelecommunications failures; computer viruses; security breaches or cyber-attacks. Although the Company and the third-party service providers seek to maintain the Company's respective systems effectively and to successfully address the risk of compromise of the integrity, security and consistent operations of these systems, such efforts may not be successful. If the Company's or our service providers' systems are damaged, breached or cease to function properly, this could significantly disrupt the Company's operations and be costly, time consuming and resource-intensive to remedy.

The Company and its third-party system providers receive, collect and store personally identifiable, confidential, proprietary and other information from our vendors, customers and employees so that they may, among other things, purchase products or service and enroll in promotional programs. Any breach of the Company's or its third-party system providers' security could result in significant costs to protect any customers, vendors, employees, and other parties whose personal data is compromised. Loss of customer, supplier employee or other business information could disrupt operations, damage the Company's reputation and expose the Company to claim from customers, vendors, financial institutions, regulators, payment card associations, employees and others, any of which could have a material adverse effect on the Company, its financial condition and results of operations.

The Company imports inventory from manufacturers outside the United States, particularly in Asia. Importing involves risk including duties, tariffs, quotas on imported merchandise and potential disruptions resulting from geopolitical problems in countrie from which merchandise is imported. Our ability to manage the importing of goods from overseas, their production, timing of deliverie and U.S. Customs-related compliance is an important component of our merchandising strategy. Geopolitical issues and military conflicts outside of the United States may increase the likelihood of supply chain interruptions and have a negative impact on our important activities.

The Company continues to monitor these market forces and risks but cannot reasonably predict the long-term impacts of these evolving developments.

Disclosure Regarding Forward-Looking Statements

This document includes certain forward-looking statements about the expectations of the Company. Although the Company believes these statements are based on reasonable assumptions, actual results may vary materially from stated expectations. Such forward-looking statements may be identified by the use of forward-looking words or phrases such as "anticipate," "believe," "expect, "intend," "may," "planned," "potential," "should," "will," "would," "project," "estimate," "ultimate," or similar phrases. Actual result may differ materially from those indicated in the Company's forward-looking statements and undue reliance should not be placed or such statements.

Factors that could cause materially different results include, but are not limited to, weather conditions, natural disasters, civi unrest; pandemics; acts of terrorism; fair value accounting adjustments; inventory valuation; health care costs; insurance costs of recoveries; legal costs; borrowing needs; interest rates; credit conditions; economic and market conditions; increases in fuel costs accidents, leaks, equipment failures, ransomware attacks, cybersecurity incidents, technology disruptions and other operating risks legislative actions; change in tax laws and tax rulings or audit results; asset sales; ineffective integration of acquired businesses significant unplanned capital needs; changes in accounting principles, interpretations, methods, judgments or estimates; performance of

major customers; reliance on third-party transporters, suppliers and contractors and interruption of product supply or increases in product costs; labor relations, increased labor costs and continued access to qualified labor.

Readers are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date of this report. The Company undertakes no obligation to publicly release any revision to these forward-looking statements to reflect events of circumstances after the date of this report.

FIVE YEAR SUMMARY OF EARNINGS AND DISTRIBUTIONS

	Years Ended									
		2024 2 Weeks)	10000	ember 30, 2023 2 Weeks)	107723	ember 31, 2022 Weeks)		nuary 1, 2022 Weeks)		2021 Weeks)
Revenues	S	9,491.4	S	9,131.1	S	9,169.8	S	8,594.2	\$	7,762.7
Cost of revenues		7,946.3		7,653.2		7,808.4		7,299.8		6,549.2
Gross profit		1,545.1		1,477.9		1,361.4		1,294.4		1,213.5
Total operating and other expenses, net		1,230.1		1,151.8		1,019.7		963.7		895.9
Net income attributable to Ace Hardware Corporation	S	315.0	s	326.1	s	341.7	S	330.7	\$	317.6
Distribution of net income:										
Patronage distributions to third party Retailers	s	335.6	s	335.7	S	323.6	s	294.1	s	275.4
Net (loss) income of non-patronage activities		(20.6)		(9.6)		18.1		36.6		42.2
Net income attributable to Ace Hardware Corporation	s	315.0	S	326.1	S	341.7	s	330.7	s	317.6

MANAGEMENT'S RESPONSIBILITY FOR FINANCIAL STATEMENTS

The consolidated financial statements presented in this Annual Report have been prepared with integrity and objectivity and are the responsibility of the management of Ace Hardware Corporation. These consolidated financial statements have been prepared in conformity with U.S. generally accepted accounting principles and properly reflect certain estimates and judgments based upon the best available information.

The Company maintains a system of internal accounting controls, which is supported by an internal audit program and is designed to provide reasonable assurance, at an appropriate cost, that the Company's assets are safeguarded and transactions are properly recorded. This system is continually reviewed and modified in response to changing business conditions and operations and as a result of recommendations by the internal and external auditors. In addition, the Company has distributed to employees its policies for conducting business affairs in a lawful and ethical manner.

The consolidated financial statements of the Company have been audited by Ernst & Young LLP, independent accountants. Thei accompanying report is based upon audits conducted in accordance with auditing standards generally accepted in the United States o America.

The Audit and Finance Committee of the Board of Directors meets periodically with the independent auditors and with the Company' internal auditors, both privately and with management present, to review accounting, auditing, internal control and financial reporting matters. The Audit and Finance Committee recommends to the full Board of Directors the selection of the independent auditors are regularly reviews the internal accounting controls, the activities of the outside auditors and internal auditors and the financial condition of the Company. Both the Company's independent auditors and the internal auditors have free access to the Audit and Finance Committee.

February 11, 2025

/s/ John Venhuizen
John Venhuizen
President and Chief Executive Officer

/s/ William M. Guzik
William M. Guzik
Executive Vice President,
Chief Financial Officer and Chief Risk Officer

/s/ Steven G. Locanto
Steven G. Locanto
Vice President, Controller

GUARANTEE OF ACE HARDWARE CORPORATION

Guaranty of Performance

For value received, Ace Hardware Corporation, a Delaware corporation (the "Guarantor"), located at 2915 Jorie Boulevard, Oak Brook, Illinois 60523-2100, absolutely and unconditionally guarantees to assume the duties and obligations of Ace Handyman Franchising, Inc., a Colorado corporation located at 390 Union Boulevard, Suite 520, Lakewood, Colorado 80228 (the "Franchisor"), under its franchise registration in each state where the franchise is registered, and under its Franchise Agreement identified in its 2025 Franchise Disclosure Document, as it may be amended, and as that Franchise Agreement may be entered into with franchisees and amended, modified or extended from time to time. This guarantee continues until all such obligations of the Franchisor under its franchise registrations and the Franchise Agreement are satisfied or until the liability of Franchisor to its franchisees under the Franchise Agreement has been completely discharged, whichever first occurs. The Guarantor is not discharged from liability if a claim by a franchisee against the Franchisor remains outstanding. Notice of acceptance is waived. The Guarantor does not waive receipt of notice of default on the part of the Franchisor. This guarantee is binding on the Guarantor and its successors and assigns.

The Guaranter signs this guarantee through its duly authorized officer at Oak Brook, Illinois on this ^{25th}day of March, 2025.

ACE HARDWARE CORPORATION

By: William Guzik

By: 9ABEE655BDEE424...

Name: William M. Guzik

Title: Executive Vice President, Chief Financial
Officer and Chief Risk Officer

ATTACHMENT K

LIST OF STATE FRANCHISE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

LIST OF STATE FRANCHISE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

STATE	STATE ADMINISTRATOR	AGENT FOR
CALIFORNIA	Department of Financial Protection and	SERVICE OF PROCESS Commissioner
CALIFORNIA	Innovation 320 West 4th Street, Suite 750 Los Angeles, California 90013-2344 (213) 576-7500	Department of Financial Protection and Innovation 320 West 4th Street, Suite 750 Los Angeles, California 90013-2344 (213) 576-7500
	One Sansome Street, Suite 600 San Francisco, California 94104-4448 (415) 972-8565	(866) 275-2677 (toll free)
	2101 Arena Boulevard Sacramento, California 95834 (916) 445-7205	
	(866) 275-2677 (toll free) www.dfpi,ca.gov ask.dfpi@dfpi.ca.gov	
FLORIDA	Florida Department of Agriculture and Consumer Services Division of Consumer Services Attn: Finance & Accounting 407 South Calhoun Street Tallahassee, Florida 32399-0800 (850) 410-3800	None
HAWAII	Commissioner of Securities Department of Commerce & Consumer Affairs Business Registration Division Securities Compliance Branch 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722	Hawaii Commissioner of Securities Same Address
ILLINOIS	Franchise Bureau Office of the Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465	Illinois Attorney General Same Address
INDIANA	Indiana Secretary of State Division of Securities 302 West Washington Room E-111 Indianapolis, Indiana 46204 (317) 232-6681	Indiana Secretary of State 201 State House 200 West Washington Street Indianapolis, Indiana 46204 (317) 232-6531
IOWA	Iowa Secretary of State 321 E. 12 th Street Des Moines, Iowa 50319 (515) 281-5204	Same

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
MARYLAND	Office of Attorney General Maryland Division of Securities 200 St. Paul Place Baltimore, Maryland 21202-2020 (410) 576-6360	Maryland Securities Commissioner Same Address
MICHIGAN	Michigan Attorney General Franchise Section - Consumer Protection Division G. Mennen Williams Building, 1st Floor 525 W. Ottawa Street Lansing, Michigan 48933 P.O. Box 30213 Lansing, Michigan 48909 (517) 373-7117	Michigan Department of Commerce Same Address
MINNESOTA	Minnesota Department of Commerce 85 7 th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1600	Minnesota Commissioner of Commerce Same Address
NEBRASKA	Department of Banking and Finance 1526 K Street, Suite 300 Lincoln, Nebraska 68508-2732 P.O. Box 95006 Lincoln, Nebraska 68509-5006 (402) 471-3445	None
NEW YORK	NYS Department of Law Investor Protection Bureau 28 Liberty Street, 21 st Floor New York, New York 10005 (212) 416-8222	Secretary of State 99 Washington Avenue Albany, New York 12231
NORTH DAKOTA	North Dakota Securities Department 600 East Boulevard Avenue, Fourteenth Floor, Dept 414 Bismarck, North Dakota 58505-0510 (701) 328-4712	North Dakota Securities Commissioner Same
OREGON	Department of Consumer and Business Services Division of Finance and Corporate Securities Labor and Industries Building 350 Winter Street NE, Room 410 Salem, Oregon 97301-3881 (503) 378-4140	Director of Oregon Department of Consumer and Business Services Division of Finance and Corporate Securities Labor and Industries Building 350 Winter Street NE, Room 410 Salem, Oregon 97301-3881 (503) 378-4140
RHODE ISLAND	State of Rhode Island and Providence Plantations Department of Business Regulation Securities Division 1511 Pontiac Avenue John O. Pastore Complex - Building 69-1 Cranston, Rhode Island 02920 (401) 462-9500	Director of Rhode Island Department of Business Regulation Same Address

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
SOUTH DAKOTA	South Dakota Division of Insurance	Director of South Dakota
	Securities Regulation	Division of Insurance
	124 S Euclid, Suite 104	Same Address
	Pierre, South Dakota 57501	
	(605) 773-3563	
TEXAS	Secretary of State	None
	Statutory Documents Section	
	James E. Rudder Building	
	1019 Brazos Street	
	Austin, Texas 78701	
	P.O. Box 13550	
	Austin, Texas 78711	
	(512) 463-5705	
UTAH	Utah Department of Commerce	Same
	Consumer Protection Division	
	160 East 300 South, 2 nd Floor	
	Salt Lake City, UT 84114	
	801-530-6601	
VIRGINIA	State Corporation Commission	Clerk of the State Corporation
	Division of Securities and Retail Franchising	Commission
	1300 E. Main Street, 9th Floor	State Corporation Commission
	Richmond, Virginia 23219	1300 E. Main Street, 1st Floor
	(804) 371-9051	Richmond, Virginia 23219
		(804) 371-9733
WASHINGTON	Department of Financial Institutions	Director of Financial Institutions
	Securities Division	150 Israel Road SW
	P.O. Box 41200	Tumwater, Washington 98501
	Olympia, Washington 98504-1200	(360) 902-8760
	(360) 902-8760	
WISCONSIN	Department of Financial Institutions	Administrator, Division of Securities
	Division of Securities	Same Address
	4822 Madison Yards Way, North Tower	
	Madison, Wisconsin 53705	
	P.O. Box 1768	
	Madison, Wisconsin 53701-1768	
	(608) 266-8557	

If a state is not listed, we have not appointed an agent for service of process in that state in connection with the requirements of franchise laws. There may be states in addition to those listed above in which we have appointed an agent for service of process.

There may also be additional agents appointed in some of the states listed.

ATTACHMENT L STATE SPECIFIC ADDENDUM TO DISCLOSURE DOCUMENT

STATE LAW ADDENDA TO THE ACE HANDYMAN SERVICES DISCLOSURE DOCUMENT

The following modifications are to the Ace Handyman Franchising, Inc. Franchise Disclosure Document for the states noted below.

CALIFORNIA

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.

OUR WEBSITES (www.AceHandymanServices.com, www.AceHandymanFranchising.com) HAVE NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THESE WEBSITES MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT www.dfpi.ca.gov.

1. The following is added to the end of the section titled "Regulations" in Item 1:

Businesses and individuals who construct or alter any building or other structure in California must be licensed by the California Contractors State License Board (the "CSLB") if the total cost, including labor and materials, of one or more contracts on the project is \$500 or more. Contractors, including subcontractors, specialty contractors, and persons engaged in the business of home improvement, with the exception of joint ventures and projects involving federal funding, must be licensed by the CSLB before submitting bids. Licenses may be issued by the CSLB to individuals, partnerships, corporations, joint ventures, or limited liability companies. It is your responsibility to become knowledgeable of, and comply with, the licensing requirements of the CSLB, to the extent that they are applicable to your AHS Business.

2. The following is added to the end of Item 11:

The Antitrust Law Section of the Office of the California Attorney General views maximum price agreements as per se violations of the Cartwright Act. As long as this represents the law of the State of California, we will not interpret the Franchise Agreement as permitting or requiring maximum price limits.

3. The following is added to the end of Item 12:

We reserve the right to establish alternative channels of distribution within your Territory.

4. The following paragraphs are added to the end of Item 17:

Neither we nor any person listed in Item 2 of this Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C. § 78a et seq., suspending or expelling these persons from membership in this association or exchange.

Section 31125 of the Franchise Investment Law requires us to give you a disclosure document approved by the Commissioner of the Department of Financial Protection and Innovation before we ask you to consider a material modification of your Franchise Agreement.

You may be required to sign a general release of claims if you renew or transfer your franchise. California Corporations Code § 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code §§ 31000 through 31516). Business and Professions Code § 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code §§ 20000 through 20043).

The California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, nonrenewal, or transfer of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control.

The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law. (11 U.S.C. § 101, et seq.)

The Franchise Agreement contains a covenant not to compete, which extends beyond the termination of the Franchise. This provision may not be enforceable under California law.

The Franchise Agreement requires binding arbitration. The arbitration will occur in Denver, Colorado, with the costs being awarded to the prevailing party. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and Federal laws (such as Business and Professions Code 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provision of the Franchise Agreement restricting venue to a forum outside the State of California.

The Franchise Agreement requires application of the laws of the State of Colorado. This provision may not be enforceable under California law.

5. The following paragraph is added to the end of Item 19:

Except as specifically listed in Item 19, the earnings claims figures do not reflect the costs of sales, operating expenses, or other costs or expenses that must be deducted from the Gross Revenue or gross sales figures to obtain your net income or profit. You should conduct an independent investigation of the costs and expenses you will incur in operating your franchise business. Franchisees or former franchisees, listed in the Disclosure Document, may be one source of this information.

- 6. Initial Franchise Fees paid by franchisees are typically applied towards our general operating expenses, which include but are not limited to those expenses we incur in fulfilling our initial obligations to franchisees.
- 7. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

HAWAII

THESE FRANCHISES HAVE BEEN FILED UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING. IF YOU LEARN THAT ANYTHING IN THIS DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE STATE AUTHORITY LISTED IN ATTACHMENT K.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE, OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE DISCLOSURE DOCUMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS DISCLOSURE DOCUMENT CONTAINS A SUMMARY OF ONLY CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

Registered agent in the state authorized to receive service of process:

Commissioner of Securities
Department of Commerce & Consumer Affairs
Business Registration Division
Securities Compliance Branch
335 Merchant Street, Room 203
Honolulu, Hawaii 96813
(303) 586-2722

1. The following paragraph is added to Item 17:

Section 482E-6(3) of the Hawaii Revised Statutes provides that upon termination or refusal to renew the Franchise, we are obligated to compensate you for the fair market value, at the time of the termination or expiration of the Franchise, of your inventory, supplies, equipment and furnishings purchased from us or a supplier designated by us; provided that personalized materials which have no value to us need not be compensated for. If we refuse to renew a Franchise for the purpose of converting your business to one owned and operated by us, we, in addition to the remedies provided above, shall compensate you for the loss of goodwill. We may deduct from such compensation reasonable costs incurred in removing, transporting and disposing of your inventory, supplies, equipment and furnishings pursuant to this requirement, and may offset from such compensation any monies due us.

- 2. The following list reflects the status of the franchise registration of the Franchisor in the states which require registration:
- A. The states in which this proposed registration is effective: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington and Wisconsin.
 - B. The states in which this proposed registration is or will be shortly on file: None.
- C. The states, if any, which have refused, by order or otherwise, to register these franchises: None.
- D. The states, if any, which have revoked or suspended the right to offer these franchises: None.
- E. The states, if any, in which the proposed registration of these franchises has been withdrawn by the Franchisor: None.
- 3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

ILLINOIS

- 1. Illinois law governs the Franchise Agreement.
- 2. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
- 3. Your rights upon termination and non-renewal are set forth in Articles 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

INDIANA

The following modifications are made to the Disclosure Document only to the extent required by the Indiana Franchises Act, IND. CODE § 23-2-2.5, and the Indiana Deceptive Franchise Practices Act, IND. CODE § 23-2-2.7:

1. The following statement is added to Item 12:

Indiana law prohibits us from establishing a Franchisor-owned outlet engaged in a substantially identical business within your Territory that competes unfairly with you within a reasonable area.

- 2. The Summary column of Items 17.r and 17.w in the Chart in Item 17 are deleted and replaced by the following:
 - 17.r: No involvement in a competitive business for 2 years within your Territory.

17.w: Except to the extent governed by federal law, disputes related to a violation of the Indiana Franchises Act or the Indiana Deceptive Franchise Practices Act shall be governed by those laws, and all other matters regarding the Franchise Agreement shall be governed by Federal and Colorado law. The Colorado Consumer Protection Act does not apply.

The changes to Item 17.w have been included in this Disclosure Document as a condition to registration. We and you do not agree with the above language and believe that each of the provisions of the Franchise Agreement, including all choice-of-law and venue provisions, are fully enforceable. We and you intend to fully enforce all of the provisions of the Franchise Agreement and all other documents signed by us, including but not limited to, all choice-of-law, venue, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.

MARYLAND

1. The following information is added to the Summary column of Items 17.c and 17.m:

The release will not apply to any liability that may arise under the Maryland Franchise Registration and Disclosure Law.

2. The Summary column of Item 17.v is deleted and the following is inserted in its place:

Colorado (except for claims arising under the Maryland Franchise Registration and Disclosure Law). Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within three years after the grant of the franchise. If the Franchise Agreement permits you to file a lawsuit, rather than requiring arbitration, you may file that suit in Maryland.

3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the

franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

MINNESOTA

1. The following statements are added to the Cover Page:

THIS FRANCHISE HAS BEEN REGISTERED UNDER THE MINNESOTA FRANCHISE ACT. REGISTRATION DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE COMMISSIONER OF COMMERCE OF MINNESOTA OR A FINDING BY THE COMMISSIONER THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE MINNESOTA FRANCHISE ACT MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WHICH IS SUBJECT TO REGISTRATION WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, AT LEAST 7 DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST 7 DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION, BY THE FRANCHISEE, WHICHEVER OCCURS FIRST, A COPY OF THIS PUBLIC OFFERING STATEMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE FRANCHISE. THIS PUBLIC OFFERING STATEMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR AN UNDERSTANDING OF ALL RIGHTS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

2. The following language is added at the end of Item 17:

We will comply with Minnesota Statutes, Section 80C.14, Subd. 3-5, which require (except in certain specified cases) (1) that a franchisee be given 90 days notice of termination (with 60 days to cure) and 180 days notice for nonrenewal of the franchise agreement, and (2) that consent to the transfer of the franchise will not be unreasonably withheld.

Minnesota Statutes, Section 80C.21 and Minnesota Rule 2860.4400(J) prohibit us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties, or judgment notes. In addition, nothing in the Disclosure Document or agreement can abrogate or reduce (1) any of your rights as provided for in Minnesota Statutes, Chapter 80C, or (2) your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction. (The foregoing language has been included in this Disclosure Document as a condition of registration. We and you do not agree with the above language and believe that each of the provisions of the Franchise Agreement are fully enforceable. We and you intend to fully enforce all of the provisions of the Franchisee Agreement and all other documents signed by us, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.)

As we deem necessary, we will protect your right to use our Marks. The Minnesota Department of Commerce requires Franchisors to indemnify franchisees against liability to third parties resulting from claims by third parties that the franchisees' use of our Marks infringes upon the trademark rights of the third party.

Minnesota Statutes, Section 80C.17, Subd. 5 provides that any claims and actions based on a violation of Chapter 80C of the Minnesota statutes or any rule or order thereunder shall be commenced within three years from the occurrence of the facts giving rise to such claim or action.

You cannot consent to us obtaining injunctive relief. We may seek injunctive relief. See Minnesota Rule 2860.4400(J). Also, a court will determine if a bond is required.

Minnesota Statutes, Section 181.991 prohibits a franchisor from restricting, restraining, or prohibiting in any way a franchisee from soliciting or hiring an employee of the franchisor or an employee of a franchisee of the same franchisor. Any such restrictions in the Franchise Agreement are deemed deleted.

3. The following language is added at the end of Items 17.c and 17.m:

(Any release executed in connection herewith shall not apply to any claims that may arise under the Minnesota Franchise Act. Minnesota Rule 2860.4400(D) prohibits a franchisor from requiring a franchisee to assent to a general release from liability imposed by the Minnesota Franchise Act, provided, it does not bar the voluntary settlement of disputes.)

4. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

NEW YORK

1. The following information is added to the Cover Page of the Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN <u>ATTACHMENT K</u> OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A

PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. Special Risks to Consider About This Franchise:

WE HAVE THE RIGHT TO INSPECT AND AUDIT RECORDS OF YOUR AFFILIATES AND FAMILY MEMBERS INCLUDING PARENTS, SIBLINGS, AND ADULT CHILDREN, THAT WE DETERMINE IN OUR SOLE DISCRETION MAY BE RELEVANT IN DETERMINING THE BUSINESS RESULTS OF YOUR FRANCHISED BUSINESS.

3. The following paragraphs are added at the end of Item 3:

With the exception of what is stated above, the following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.
- 4. The following paragraph is added at the end of Item 4:

Neither the franchisor, its affiliate, its predecessor, officers, or general partner during the 10-year period immediately before the date of the offering circular: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the bankruptcy code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that

obtained a discharge of its debts under the U.S. Bankruptcy Code during or within 1 year after that officer or general partner of the franchisor held this position in the company or partnership.

5. The following statement is added at the end of Item 5:

The Initial Franchise Fee constitutes part of our general operating funds and will be used as such in our discretion.

6. The following statement is added to the end of the "Summary" sections of Item 17.c, titled "Requirements for you to renew or extend," and Item 17.m, titled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued hereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687(4) and 687(5) be satisfied.

7. The following language replaces the "Summary" section of Item 17.d, titled "Termination by franchisee":

You may terminate the Franchise Agreement upon any grounds available by law.

8. The following statement is added to the end of the "Summary" section of Item 17.j, titled "Assignment of contract by franchisor":

However, no assignment will be made except to an assignee who, in our good faith judgment, is willing and able to assume our obligations under the Franchise Agreement.

9. The following statement is added to the end of the "Summary" sections of Item 17.v, titled "Choice of forum," and Item 17.w, titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon either us or you by Article 33 of the General Business Law of the State of New York. (The foregoing language has been included in this Disclosure Document as a condition of registration. We and you do not agree with the above language and believe that each of the provisions of the Franchise Agreement, including all choice of law provisions, are fully enforceable. We and you intend to fully enforce all of the provisions of the Franchisee Agreement and all other documents signed by us, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.)

- 10. Franchise Questionnaires and Acknowledgements -- No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 11. Receipts -- Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 et seq.), which describes the time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchisee before a sale may be made. New York law

requires a franchisor to provide the Franchise Disclosure Document at the earlier of the first personal meeting, 10 business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

WE REPRESENT THAT THIS DISCLOSURE DOCUMENT DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF A MATERIAL FACT.

NORTH DAKOTA

- 1. The North Dakota Securities Commissioner has held the following to be unfair, unjust or inequitable to North Dakota franchisees:
 - A. Restrictive Covenants: Franchise Disclosure Documents which disclose the existence of covenants restricting competition contrary to Section 9-08-06, N.D.C.C., without further disclosing that such covenants will be subject to this statute.
 - B. Situs of Arbitration Proceedings: Franchise agreements providing that the parties must agree to the arbitration of disputes at a location that is remote from the site of the franchisee's business.
 - C. Restrictions on Forum: Requiring North Dakota franchisees to consent to the jurisdiction of courts outside of North Dakota.
 - D. Liquidated Damages and Termination Penalties: Requiring North Dakota franchisees to consent to liquidated damages or termination penalties.
 - E. Applicable Laws: Franchise agreements which specify that they are to be governed by the laws of a state other than North Dakota.
 - F. Waiver of Trial by Jury: Requiring North Dakota franchisees to consent to the waiver of a trial by jury.
 - G. Waiver of Exemplary & Punitive Damages: Requiring North Dakota franchisees to consent to a waiver of exemplary and punitive damages.
 - H. General Release: Franchise agreements that require the franchisee to sign a general release upon renewal of the franchise agreement.
 - I. Limitation of Claims: Franchise agreements that require the franchisee to consent to a limitation of claims. The statute of limitations under North Dakota law applies.

RHODE ISLAND

1. The following paragraph is added at the end of Item 17:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application

of the laws of another state is void with respect to a claim otherwise enforceable under this Act." The preceding language has been included in this Disclosure Document as a condition to registration. Franchisor and Franchisee do not agree with the above language and believe that each of the provisions of the Franchise Agreement, including all choice of law provisions, are fully enforceable. Franchisor and Franchisee intend to fully enforce all of the provisions of the Franchise Agreement and all other documents signed by them, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.

VIRGINIA

1. The following statement is added to the Summary column of Item 17.h:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in the Franchise Agreement does not constitute "reasonable cause," as that term is defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

WISCONSIN

REGISTRATION OF THIS FRANCHISE IN THE STATE OF WISCONSIN DOES NOT MEAN THAT THE STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS DISCLOSURE DOCUMENT.

The conditions under which the Franchise Agreement can be terminated or not renewed may be affected by the Wisconsin Fair Dealership Law, Wisconsin Statutes 1981-82, Title XIV-A, Chapter 135.

ATTACHMENT M AHF DATA COLLECTION AGREEMENT

AHF DATA COLLECTION AGREEMENT

By execution of this Data Collection Agreement (the "Agreement"), Ace Handyman Franchising, Inc. ("AHF"), with offices located at 390 Union Boulevard, Suite 520, Lakewood, Colorado 80228, and the undersigned AHF Handyman Services Franchisee ("Franchisee", and together with AHS, the "Parties") agree as follows:

1. Services and Data. AHF agrees to collect from Franchisee, and Franchisee agrees to share with AHF, data related to Franchisee's handyman business operations, including, but not limited to, customer data (collectively, the "Data") which AHF may use for the betterment of the business of AHF, its parents, affiliates and the franchisees of AHF, its parents and affiliates, including, but not limited to, for purposes of advertising, analytics, compliance, and operations. These uses and the sharing of Data will be collectively referred to herein as the "Services". For the purposes of this Agreement, Franchisee's handyman business operations shall include any of the Franchisee's territories.

2. Data Collection:

- a) Franchisee agrees that AHF may collect Data as needed.
- b) The frequency and timing of Data collection is subject to change at any time by AHF without notice at its discretion.
- c) A third-party agent, service provider, or contractor may be used to collect and store Data from Franchisee.
- **3. Mutual Data Privacy.** To the extent the provision of Services hereunder involves the Processing of Personal Information (as such terms are defined below) on Franchisee's or AHF's behalf, the Parties have agreed to enter into and to be bound by this Agreement to ensure compliance with applicable privacy laws.

4. Definitions:

- a) "Personal Information" means information that identifies, relates to, describes, is capable of being associated with, or could reasonably be linked, directly or indirectly, with a particular consumer, household, or device. For the avoidance of doubt, Personal Information includes, but not limited to, name, mailing address, email address, phone number, and purchase history.
- b) "Process" or "Processing" means any operation or set of operations that are performed on Personal Information or on sets of Personal Information (including, but not limited to, retaining, using, disclosing, accessing, storing, or creating derivative works therefrom), regardless of whether by automated means.
- 5. Processing of Personal Information. The Parties shall only Process Personal Information for the purpose of providing the Services. The Parties shall not retain, use, or disclose Personal Information for any purpose (including, but not limited to, any commercial purpose) other than to perform the Services. If requested by a Party, the other Party agrees it will not sell, rent, release, disclose, disseminate, make available, transfer or otherwise communicate Personal Information related to a particular individual, household or device to any third party for monetary or other valuable consideration.
- 6. Consumer Requests. If a Party receives a request from an individual to take certain actions related to that individual's Personal Information collected by said Party, such as, but not limited to, delete, stop selling or return said individual's Personal Information (a "Consumer Rights Request"), the Party receiving said Consumer Rights Request agrees to respond to it in accordance with applicable law. If

a Party requests the other Party also comply with said Consumer Rights Request, the other Party agrees it will also respond to said request in accordance with applicable law. In addition, if a Party requests information from the other Party to help it comply with a Consumer Rights Request, such as, but not limited to, asking for the details relating to such Consumer Rights Request, the other Party will respond to such request, in writing, within five (5) calendar days of receipt of said request, which response may include, as applicable, a copy of said individual's Personal Information Processed by such Party. In response to a Consumer Rights Request to delete an individual's Personal Information, each Party agrees to securely and permanently erase and delete all copies of Personal Information Processed. If requested, written certification of compliance with this Section 6 shall be promptly returned to the requesting Party.

- 7. **Disclaimer.** NEITHER PARTY, NOR ANY THIRD PARTY SERVICE PROVIDER OR CONTRACTOR OF AHF, WILL BE LIABLE TO THE OTHER PARTY FOR ANY DIRECT, SPECIAL, EXEMPLARY, INDIRECT, INCIDENTAL OR CONSEQUENTIAL DAMAGES IN CONNECTION WITH THE PERFORMANCE OF ITS OBLIGATIONS ARISING FROM OR UNDER THIS AGREEMENT.
- **8. Amendment.** This Agreement may be amended by AHF upon ten (10) days prior written notice to Franchisee. This Agreement supersedes any prior written agreements between the parties relating to the Data Collection Agreement.
- **9. Compliance with Laws.** Each party shall comply with all statutes, laws and regulations applicable to the Processing of Personal Information hereunder.
- **10.** Certification. By executing this Agreement, Franchisee and AHF each certify it understands the restrictions on the Processing of Personal Information as set forth in this Agreement.

SIGNATURE PAGE TO FOLLOW

AHF: ACE HANDYMAN FRANCHISING, INC., a Colorado corporation	FRANCHISEE: IF AN INDIVIDUAL:
By:	Signature
	Print Name:
	IF A PARTNERSHIP, LIMITED LIABILITY COMPANY, CORPORATION, OR OTHER ENTITY:
	Print Company Name
	By:Signature Print Name and Title:
	By:
	Date:

ATTACHMENT N IGX PARTICIPATION AGREEMENT

ACE HANDYMAN FRANCHISING PARTICIPATION AGREEMENT – INGENIUX SAAS SERVICES

This Ace Handyman Franchising Participation Agreement – Ingeniux SaaS Services ("Participation Agreement") is made and entered into as of ______ ("Effective Date"), by and between Ace Handyman Franchising, Inc. ("AHF") and the undersigned independent franchisee and services provider (each a "Franchisee") that operates the service provider(s) business.

WHEREAS, AHF has designated Ingeniux Corporation ("IGX" or "Vendor") as the content management SaaS services provider for the AHF website. Franchisee may access and use the Vendor SaaS platform and obtain certain content management software services from Vendor as hosted through Vendor's platform (the "Vendor SaaS Services") for the purposes of providing content to be displayed on Franchisee's pages or other pages within AHF's website (collectively the "Program"), provided that Franchisee agrees to the following requirements for participation and executes this Participation Agreement; and,

WHEREAS, Franchisee wishes to participate in the Program; and,

NOW THERFORE, in consideration of the premises and other good and valuable consideration, the receipt of which is acknowledged, Franchisee and AHF hereby agree as follows:

- **1. The Program.** Upon execution of this Participation Agreement, Franchisee will receive access to Vendor SaaS Services.
- **Terms of Participation.** By participating in the Program, Franchisee agrees to adhere to Vendor's terms of use set forth in Schedule A, attached hereto and incorporated herein by reference, and the terms set forth in this Participation Agreement. Franchisee shall work cooperatively with AHF and Vendor in any investigation involving Franchisee's compliance thereof.

3. <u>Cost</u>.

- a. The fees payable by Franchisee for the use of the Vendor SaaS Services shall be included within the Software and Internet Fees, as updated from time to time, payable by Franchisee pursuant to the Ace Handyman Franchise Agreement.
- b. In the event that additional services are requested by Franchisee or otherwise become available, Franchisee agrees it is solely responsible for the fees associated with such additional services that it purchases directly from Vendor in excess of the base services and tools offered as part of the Program.
- **Term and Termination.** The term of this Participation Agreement shall commence as of the Effective Date and shall continue until such time that Franchisee's access to the Vendor SaaS Services has been terminated including, but not limited to, automatically upon:
 - a. termination or suspension of Franchisee's access to Vendor SaaS Services by Vendor or AHF as a result of the breach of this Participation Agreement as determined by AHF or Vendor.
 - b. termination for any reason of the Ace Handyman Franchise Agreement.
 - c. discontinuation of the Program by AHF in its sole discretion.
- **Release and Indemnity.** Franchisee acknowledges and agrees that Franchisee shall be responsible for any claims, liabilities, losses, damages and injuries that may arise out of its participation in the Program, including, without limitation, from its customers, employees (former or current), consultants or agents, and Franchisee agrees to indemnify, defend and hold AHF and their respective parents, affiliates, directors, officers, and employees harmless from and against any claims, liabilities, losses, damages and injuries arising out of (i) the Franchisee's participation in the Program, (ii) the Franchisee's access or use of Vendor SaaS Services, (iii) Franchisee's violation of this Participation Agreement including, but not limited to, Vendor's

terms of use set forth in Schedule A, and/or (iv) any alleged infringement of any patent, trademark, copyright, or other intellectual property right or third party right arising out of or in connection with (a) any specifications, documentation, trademarks, trade names, logos, ad imagery, display and video ad creative and copy, content, keywords, materials, or other content or creative assets provided by or on behalf of Franchisee to be published in the Vendor SaaS Services, or (b) modifications, enhancements or other alterations by or on behalf of Franchisee to its content or the Vendor SaaS Services; or (c) malware or disabling code provided by or on behalf of Franchisee in connection with its use of the Vendor SaaS Services. This obligation shall survive termination of this Participation Agreement.

- **Status of Participation Agreement with Respect to Other Documents.** Nothing in this Participation Agreement is intended to limit, reduce, terminate, amend, or modify any obligations Franchisee may have pursuant to any other agreement between Franchisee and AHF, including, without limitation, the Ace Handyman Franchise Agreement (collectively "Other Agreements"). In the event of any conflict between this Participation Agreement and any of the Other Agreements, such Other Agreements shall prevail. AHF and Franchisee acknowledge that IGX is a third-party beneficiary under Schedule A of this Agreement.
- **Governing Law.** This Participation Agreement shall be construed in accordance with the laws of the State of Colorado, regardless of any conflict of law rules. The parties hereby submit to the exclusive jurisdiction of the federal and state courts located in the State of Colorado.

IN WITNESS WHEREOF, the parties have executed this Participation Agreement by their duly authorized representatives.

PARTICIPATI	NG FRANCHISEE:
Legal Busine	ss Name
Ву:	
[Authori	zed Signature and Title]
Franchisee A Street: City: State: Zip Code: ACE HANDYI	MAN FRANCHISING, INC.
Ву:	
	e, President
[Sign	ature and Title of Authorized Officer]
Street:	390 Union Boulevard, Suite 520
City:	<u>Lakewood</u>
State:	<u>Colorado</u>
Zip Code:	80228

SCHEDULE A

Participation Terms of Use

1. <u>Use</u>.

- a. Franchisee will not use hosting services ("Hosting Services") of Ingeniux Corporation or its affiliates ("Vendor") in a manner that is prohibited by any law, or to facilitate the violation of any law. Transmission, storage, or presentation of any information, data or material in violation of any United States Federal, State or Local law is prohibited. This includes but is not limited to: uploading content that infringes or misappropriates a third party's intellectual property or violates applicable data privacy laws, material judged to be threatening or obscene, material that jeopardizes national security, or material protected by trade secret or other laws.
- b. Franchisee may not utilize the Hosting Services in connection with any disruptive or abusive activity, including but not limited to, denial of service attacks, degradation or impairment of the network servers, gain access to restricted or confidential information stored in the network, gain unauthorized access to network accounts, or otherwise utilize the hosting services to post any disruptive or harmful software code nor include any tracking pixel code snippets.
- 2. <u>Removal of Content; Suspension; Termination</u>. If Vendor has reasonable grounds to believe that Franchisee is utilizing the Hosting Services in violation of the foregoing terms, Vendor, Ace Handyman Franchising, Inc. ("AHF"), or AHF's affiliates may remove the infringing or unlawful content or material and suspend or terminate the Franchisee's access and use of the Hosting Services, and promptly provide written notice to AHF.
 - Franchisee acknowledges and agrees that AHF, its affiliates, and Ingeniux Corporation will not have an obligation to Franchisee to monitor content. To the extent AHF, its affiliates, or Vendor does monitor content, Franchisee acknowledges and agrees that AHF, its affiliates, and Vendor shall be the sole arbiters as to what constitutes a violation of this provision.
- 3. Indemnification. In addition to Franchisee's indemnification obligations set forth in Section 5 of the Agreement, and without limiting any of said obligations, Franchisee agrees to defend Vendor and AHF and its affiliates, including their respective officers, directors, agents, and employees ("Indemnified Parties"), through final judgment or settlement, against all third-party claims, actions, or suits against the Indemnified Parties ("Claims") that (a) are caused by any negligent act or omission or willful misconduct of Franchisee or violation of applicable laws by Franchisee; (b) arise from Franchisee's access or use of Vendor's hosting services; (c) arise from its postings, data or transmissions, including, but not limited to, any malware or disabling code provided by or on behalf of Franchisee in connection with its use of Vendor's hosting services and/or any breach or non-compliance of these Participation Terms by Franchisee; or (d) arise from any alleged infringement of any patent, trademark, copyright, or other intellectual property right or third party right arising out of or in connection with (i) any specifications, documentation, trademarks, trade names, logos, ad imagery, display and video ad creative and copy, content, keywords, materials, or other content or creative assets provided by or on behalf of Franchisee to be published in the Vendor's hosting services, or (ii) modifications, enhancements or other alterations by or on behalf of Franchisee to its content or Vendor's hosting services. Franchisee will indemnify and hold the Indemnified Parties harmless from and against all damages, expenses and costs, including reasonable attorneys' fees, incurred by the Indemnified Parties for such Claims or amounts agreed to by Franchisee in settlement of the Claims. Vendor and AHF's affiliates are intended third-party beneficiaries of this provision. This obligation shall survive termination of the Participation Agreement in which these Participation Terms are incorporated.

4

4. <u>LIMITATION OF LIABILITY</u>. EXCEPT AS OTHERWISE AGREED IN AHF'S AGREEMENT WITH VENDOR, VENDOR SHALL NOT BE LIABLE FOR SPECIAL, INCIDENAL OR CONSEQUENTIAL DAMAGES, EVEN IF

ADVISED OF THEIR POSSIBILITY.

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	March 27, 2024 PENDING
Hawaii	June 18, 2024 PENDING
Illinois	March 27, 2024 PENDING
Indiana	April 11, 2024PENDING
Maryland	April 1, 2024PENDING
Michigan	March 27, 2024 PENDING
Minnesota	April 17, 2024PENDING
New York	April 30, 2024PENDING
North Dakota	April 3, 2024PENDING
Rhode Island	April 1, 2024PENDING
South Dakota	March 28, 2024PENDING
Virginia	April 2, 2024PENDING
Washington	SEE WASHINGTON SPECIFIC FDD
Wisconsin	April 1, 2024 PENDING

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

RECEIPT

This Disclosure Document summarizes certain provisions of the Franchise Agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If Ace Handyman Franchising, Inc. ("AHF") offers you a franchise, it must provide this Disclosure Document to you 14 calendar days before you sign a binding agreement with, or make a payment to, the franchisor, or an affiliate, in connection with the proposed franchise sale.

New York requires that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration, whichever occurs first.

Michigan requires that we give you this Disclosure Document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

Iowa requires that we give you this Disclosure Document at the earlier of the first personal meeting or 14 calendar days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship.

If AHF does not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Attachment K.

AHF authorizes the parties identified on <u>Attachment K</u> to receive service of process for AHF in the particular state.

The following Franchise Sellers were involved in the offering of this franchise:

The following employee(s) of AHF, having a principal business address and telephone number the same as AHF: Andy Bell, Chris Bue, Colette Bell, LaShelle Taylor, Nicole Carson, Maria Hargett, David Ohl, Fatima Agin, Kent Miller, and Jessica Weeks

vinici, and Jessica weeks	
The following independent sales agent (AHF requests that prospective franchisee fill in the information, if kn	iown):
, having a principal business address at:	
, telephone number:	
ssuance Date: March 27 27 2024 2025	

I received a Disclosure Document dated Mareh 2727, 2024 2025, that included the following Attachments: Franchise Agreement with Exhibits (Attachment A); Nondisclosure and Noncompetition Agreement (Attachment B); Statement of Prospective Franchisee (Attachment C); Multi-Territory Addendum to Franchise Agreements (Attachment D); Form of Successor Franchise Rider to Franchise Agreement (Attachment E); Current Form of General Release (Attachment F); Operations Manual Table of Contents (Attachment G); List of Franchisees (Attachment H); List of Franchisees Who Have Left The System (Attachment I); Financial Statements (Attachment J); List of State Franchise Administrators and Agents for Service of Process (Attachment K); State Addendum to Disclosure Document (Attachment L); AHF Data Collection Agreement (Attachment M); and IGX Participation Agreement (Attachment N).

PROSPECTIVE FRANCHISEE

Signature:	
Print Name:	
Date:	

RECEIPT

This Disclosure Document summarizes certain provisions of the Franchise Agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If Ace Handyman Franchising, Inc. ("AHF") offers you a franchise, it must provide this Disclosure Document to you 14 calendar days before you sign a binding agreement with, or make a payment to, the franchisor, or an affiliate, in connection with the proposed franchise sale.

New York requires that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration, whichever occurs first.

Michigan requires that we give you this Disclosure Document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

Iowa requires that we give you this Disclosure Document at the earlier of the first personal meeting or 14 calendar days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship.

If AHF does not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Attachment K.

AHF authorizes the parties identified on Attachment K to receive service of process for AHF in the particular state.

The following Franchise Sellers were involved in the offering of this franchise:

The following employee(s) of AHF, having a principal business address and telephone number the same as AHF: Andy Bell, Chris Bue, Colette Bell, LaShelle Taylor, Nicole Carson, Maria Hargett, David Ohl, Fatima Agin, Kent Miller, and Jessica Weeks,

The following independent sales agent (AHF requests that prospective franchisee fill in the information, if kno	wn): _
, having a principal business address at:	
, telephone number:	
ssuance Date: March 27 27, 2024. 2025	

I received a Disclosure Document dated Mareh 2727, 2024 2025, that included the following Attachments: Franchise Agreement with Exhibits (Attachment A); Nondisclosure and Noncompetition Agreement (Attachment B); Statement of Prospective Franchisee (Attachment C); Multi-Territory Addendum to Franchise Agreements (Attachment D); Form of Successor Franchise Rider to Franchise Agreement (Attachment E); Current Form of General Release (Attachment F); Operations Manual Table of Contents (Attachment G); List of Franchisees (Attachment H); List of Franchisees Who Have Left The System (Attachment I); Financial Statements (Attachment J); List of State Franchise Administrators and Agents for Service of Process (Attachment K); State Addendum to Disclosure Document (Attachment L); AHF Data Collection Agreement (Attachment M); and IGX Participation Agreement (Attachment N).

PROSPECTIVE FRANCHISEE

Signature:			
Print Name:			
Date:			

IMPORTANT: PLEASE IMMEDIATELY SIGN, SCAN, AND E-MAIL THIS PAGE TO CBELL@ACEHANDYMANCORP.COM, THEN PROMPTLY RETURN THIS PAGE BY MAIL OR COURIER TO ACE HANDYMAN FRANCHISING, INC., 390 UNION BOULEVARD, SUITE 520, LAKEWOOD, COLORADO 80228.